WILMS TUMOR

Cover image: Disruption of the differentiation of renal progenitor cells may result in Wilms tumor formation. See page 157, chapter 10 for details. Copyright: Carraro DM, Ramalho RF and Maschietto M.

WILMS TUMOR

Edited by

Marry M. van den Heuvel-Eibrink

Princess Maxima Center for Pediatric Oncology/Hematology, Lundlaan 6, 3584 EA Utrecht, The Netherlands



Wilms Tumor

ISBN: 978-0-9944381-1-9 DOI: http://dx.doi.org/10.15586/codon.wt.2016

Edited by

Marry M. van den Heuvel-Eibrink, MD, PhD Princess Maxima Center for Pediatric Oncology/Hematology, Lundlaan 6, 3584 EA Utrecht, The Netherlands

Published by

Codon Publications Brisbane, QLD 4122, Australia

Copyright© 2016 Codon Publications

Published under Creative Commons Attribution 4.0 International (CC BY 4.0). Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.

Notice to the user

The views and opinions expressed in this book are believed to be accurate at the time of publication. The publisher, editors or authors cannot be held responsible or liable for any errors, omissions or consequences arising from the use of the information contained in this book. The publisher makes no warranty, implicit or explicit, with respect to the contents of this book, or its use.

First Published in March 2016 Printed in Australia

A free online version is available at http://codonpublications.com

Contents

	Foreword Bengt Sandstedt	ix
	Preface Marry M. van den Heuvel-Eibrink	xiii
	Contributors	xvii
Section I	Epidemiology, Diagnostics, and Treatment	1
Chapter 1	Wilms' Tumour – Histology and Differential Diagnosis Sergey D. Popov, Neil J. Sebire, Gordan M. Vujanic	3
Chapter 2	The Clinical Relevance of Age at Presentation in Nephroblastoma Saskia L. Gooskens,Heidi Segers, Kathy Pritchard-Jones, Norbert Graf, Jeffrey S. Dome, Marry M. van den Heuvel-Eibrink	23
Chapter 3	Histopathological and Molecular Characteristics of Wilms Tumor Gulden Diniz	31
Chapter 4	Wilms Tumor and Its Management in a Surgical Aspect Basak Erginel	43
Chapter 5	Management of Bilateral Wilms Tumours Alastair J. W. Millar, Sharon Cox, Alan Davidson	61
Chapter 6	Extrarenal Wilms' Tumor: Challenges in Diagnosis, Embryology, Treatment and Prognosis Reza Shojaeian, Mehran Hiradfar, Parisa Saeedi Sharifabad, Nona Zabolinejad	77

Chapter 7	Neoadjuvant Transcatheter Arterial Chemoembolization and Systemic Chemotherapy for the Treatment of Wilms Tumor Min-Ju Li, Da-Xing Tang, Shan Xu, Yong Huang, De-Hua Wu, Jin-Hu Wang, Can Lai, Hong-Feng Tang, Qiang Shu	95
Chapter 8	Dendritic Cell-Based Cancer Immunotherapy Targeting Wilms' Tumor 1 for Pediatric Cancer Shigetaka Shimodaira, Koichi Hirabayashi, Ryu Yanagisawa, Yumiko Higuchi, Kenji Sano, Tomonobu Koizumi	113
Chapter 9	Chronic Kidney Disease in Wilms Tumour Survivors – What Do We Know Today? Joanna Stefanowicz, Mateusz Kosiak, Wojciech Kosiak, Beata S. Lipska-Ziętkiewicz	131
Section II	Biology	147
Chapter 10	Gene Expression in Wilms Tumor: Disturbance of the Wnt Signaling Pathway and MicroRNA Biogenesis Dirce Maria Carraro, Rodrigo F. Ramalho, Mariana Maschietto	149
Chapter 11	Transcriptional Regulation of the Human Thromboxane A₂ Receptor Gene by Wilms' Tumour (WT)1 B. Therese Kinsella	163
Chapter 12	The Inflammatory Microenvironment in Wilms Tumors Paramahamsa Maturu	189

Section III	WT1 Gene Aberrations in Other Malignancies	209
Chapter 13	WT1 in Cardiac Development and Disease Sjoerd N. Duim, Marie-José Goumans, Boudewijn P. T. Kruithof	211
Chapter 14	Functional Role of WT1 in Prostate Cancer Gail C. Fraizer, Kurtis Eisermann, Sony Pandey, Adina Brett-Morris, Anton Bazarov, Sarah Nock, Nirmala Ghimirey, Steven J. Kuerbitz	235
Chapter 15	Functional Roles of Wilms' Tumor 1 (WT1) in Malignant Brain Tumors Noriyuki Kijima, Naoya Hashimoto, Yasuyoshi Chiba, Yasunori Fujimoto, Haruo Sugiyama, Toshiki Yoshimine	261
Chapter 16	Wilms' Tumor Gene (WT1) Expression and Minimal Residual Disease in Acute Myeloid Leukemia Giovanni Rossi, Maria Marta Minervini, Angelo Michele Carella, Lorella Melillo, Nicola Cascavilla	273
	Index	287

DOI: http://dx.doi.org/10.15586/codon.wt.2016

Foreword

I have been lucky and privileged to work with the Société Internationale d'Oncologie Pédiatrique (SIOP) panel of pathologists for 40 years. During this time, I have seen steady and remarkable progress and success in the treatment of Wilms' tumour, with an increase in survival rates from around 50 to more than 90%.

Max Wilms, a German pathologist and surgeon, hardly realized that his thesis on "Mischgeschwulste der Niere" ("Mixed tumours of the kidney"; 1899) would link his name to the most common renal tumour in children and also turn out to be an example of successful multimodal treatment. He collected nine, mostly large, tumours from children aged 11 weeks to 11 years, which were described as round cells or myosarcomas with a content of epithelial structures in which he saw the confusing similarity to the embryonic kidney. He suspected that the component he termed "round cell sarcoma" represented tumour stem cells with the potential for differentiating into mesenchyme and epithelium and proposed its origin from the "kidney blastema" and extensively discussed oncogenesis compared with renal embryology.

In the first half of the 20th century, surgical excision was the only treatment often with a fatal outcome partly due to large tumour size. An early attempt (1916) to treat an inoperable tumour had shown that X-rays could shrink a tumour and it gave initial success, but the method was not commonly used. Slowly, surgery improved and saved some children with small tumours. A general breakthrough in treatment came with advanced surgery together with irradiation and chemotherapy as reported by Sidney Farber and his group (1956), resulting in a 2-year survival rate of 81%. The next step came with the creation and contribution of two major groups, which gave an enormous impact on treatment success. The National Wilms' Tumor Study (NWTS) began in 1969, and through national and international collaboration, it collected a large number of patients and ran several clinical and randomized trials aiming to optimize treatment for various risk groups and possibly also to identify genetic risk factors. This has led to using loss of heterozygosity of 1p and 16q to stratify patients in the current Children's Oncology Group Wilms' Tumor risk stratification protocol. A cornerstone right from the beginning of NWTS clinical trials was the work of the iconic pathologist Bruce Beckwith, who firmly related histopathology to prognosis and identified tumours with favourable or unfavourable morphology. This classification is still valid for tumours without upfront treatment as NWTS never adopted this mode until recently in some clinical settings. Among his enormous contributions, the documentation and clinical importance of nephrogenic rests must also be mentioned.

In Europe, a small group of dedicated French doctors started a paediatric oncology club in 1961, which in 1969 transformed to SIOP. At first a bilingual society, but with the intention

Foreword

of becoming international, it soon had members from all over Europe, and today, with members from all over the world, it warrants the English name International Society of Pediatric Oncology ("SIOP" still used reflecting the origins). It must also be mentioned that SIOP is dovetailed with United Kingdom Children's Cancer Study Group (UKCCSG) and the German organization, The Society for Paediatric Oncology and Haematology (GPOH). They share committee and panel members, as well as general rules for treatment and results, with a hub for statistics in Amsterdam although having different national offices.

From the first study in 1971, preoperative treatment was given to children >6 months to reduce operative rupture by shrinking and encapsulating the tumour and thereby lowering the stage level. Initially, radiation was used, but in a subsequent study, it was shown that preoperative two-drug chemotherapy was as efficient as radiotherapy. This became the standard in the SIOP protocol for the treatment of renal tumours in children. An added bonus was that the responsiveness to chemotherapy was revealed by reduced tumour volume, as well as the extent of regression seen at the pathological examination. Regressive changes were a challenge for us in the pathology panel, which I joined in 1973. The dilemma was how to assign risk group or grade tumours based on the amount of regression due to chemotherapy and to relate it to the different viable components. The main issue during the first studies was to register the amount of all these elements, which later led to the three-tier risk classification in SIOP 9301 trial and study. This was updated to "the revised SIOP working classification" used in the latest study (SIOP 2001) with the important change of placing the blastemic subtype in the high-risk group. Staging was also necessary to adapt to regression, a deviation from pure anatomical grounds. Compared with the straightforward grading and staging of nonpretreated tumours, there are quantitative histological threshold values, which sometimes are difficult to interpret and make high demands on local pathologists and also make access to reference pathology important. The SIOP risk classification, however, has shown to be of significant value for distinguishing between low-, intermediate-, and high-risk tumours. The guiding light for all these trials and studies was not only to titrate the optimal amount and type of chemotherapy and irradiation but also to lower the intensity or exclude components when possible in defined risk groups to reduce toxicity but retaining cure.

Over time there has been an increasing demand to find biomarkers for those tumours that are resistant to chemotherapy, markers which are not obvious with conventional histopathology. After recognition of the mutation in the *WT1* gene led to intensive molecular research, this field has expanded at a pace which is beyond keeping up with for an old histopathologist without at least one foot in molecular research. This new constellation of clinically active doctors will be evident in the present issue. It is noteworthy that this research now focuses on normal kidney embryology to relate it genetically to Wilms' tumour development, exactly what Max Wilms also was doing with the help of a light microscope more than a hundred years ago.

Foreword

In this book, you will find selected topics that cover the most recent developments spiced with some new findings in pathology, clinical management, and biological research in Wilms' tumour based on what has been achieved by a long row of hardworking devoted individuals in the large international collaborative groups. I have been fortunate to work with, and meet, most of these extraordinary persons. Some sadly passed away, some retired but most still active and found among the authors here. Enjoy their important effort!

Bengt Sandstedt, MD, PhD Childhood Cancer Research Unit, Karolinska Institutet Astrid Lindgren Children's Hospital SE-17176 Stockholm, Sweden March 2016 DOI: http://dx.doi.org/10.15586/codon.wt.2016.fr

Preface

Wilms' tumor (WT), also called nephroblastoma, is a rare kidney cancer that is usually diagnosed in children under the age of 6. WT arises from nephrogenic rests, which are undifferentiated embryonic tissues retained after birth. At molecular level, in a proportion of patients, WT has been shown to be the result of aberrations in *WT1* gene, located on chromosome 11p13. In addition to being a risk factor for WT, germ line *WT1* aberrations can cause renal and extrarenal developmental abnormalities and predispose to other malignancies. In the past two decades, there has been a considerable improvement in our understanding of WT and *WT1*. This book brings together recently uncovered basic and clinical aspects of the burgeoning WT and *WT1* gene aberrations in other malignancies.

Section I provides a comprehensive guide to the epidemiology, diagnosis, management, and treatment of WT. Chapter 1 describes the morphology and differential diagnosis of WT. It presents a clear view of the common histological components of WT. While stage and histological subtypes are well-known prognostic factors for WT, age at diagnosis is also an independent risk factor for recurrence. Chapter 2 elegantly summarizes the clinical relevance of age at presentation in WT management. Chapter 3 provides a comprehensive review of the histopathology, genetics, and molecular biology of WT. Also, this chapter discusses how these changes influence the prognosis and differential diagnosis. The clinical features and surgical management of WT is discussed in chapter 4. Especially, this chapter emphasizes the necessity of a multidisciplinary approach for the effective surgical management of WT.

Bilateral WT represents 4–7% of all WT, typically presenting at a younger age than unilateral WT. The major challenge in the treatment of bilateral WT is the preservation of renal function. Chapter 5 gives an overview of the current status of management of bilateral WT. Extrarenal WT is a rare entity, which usually occurs in the retroperitoneum or inguinal region. Chapter 6 presents a comprehensive review of the challenges in diagnosis, histopathology, staging, treatment, and prognosis of extrarenal WT. In chapter 7, the authors share their experience on the use of preoperative transcatheter arterial chemoembolization combined with systemic chemotherapy for the management of unilateral advanced WT. Treatment of advanced cancers that have metastasized to distant parts, irrespective of the cancer type, continues to be a challenge. Dendritic cell-based immunotherapy has been presented as a viable treatment option in many cancers. In chapter 8, the authors present autologous dendritic cell vaccines for the treatment of WT

Preface

as a possible future option. Chapter 9 addresses the problem of chronic kidney disease and renal function in WT survivors.

Section II covers the biological aspects of WT and *WT1* under three headings. WT displays morphological and molecular characteristics that resemble early stages of kidney development. Therefore, a study of molecular pathways relevant to normal kidney development may provide insights into the events that drive WT. Based on this rationale, chapter 10 gives an overview of the link between Wnt signaling, microRNA biogenesis, and β -catenin in regulating kidney differentiation. Chapter 11 focuses on the transcriptional regulation of the human thromboxane A2 receptor gene by WT1. The prostanoid thromboxane A2 is implicated in neoplastic diseases. In humans, TXA2 signals through the T-prostanoid (TP) α and TP β isoforms of the TP receptor, two structurally related receptors transcriptionally regulated by distinct promoters, Prm1 and Prm3, respectively, within the TP gene (*TBXA2R*). A particular focus is placed on the role of WT1 in the regulation of TP α expression through Prm1 in megakaryoblastic and endothelial cells of vascular origin and in prostate and breast carcinoma cells. Chapter 12 gives a comprehensive review of the inflammatory microenvironment of human WT with a comprehensive picture of various immune cells and inflammatory markers.

Section III focuses on the role of WT1 in cardiac development, prostate cancer, glioblastoma, and minimal residual disease. WT1 has been identified as a crucial player in cardiac development. Absence of WT1 leads to major cardiac malformations, including incomplete formation of coronary vasculature, resulting in embryonic lethality. Chapter 13 describes the diverse and unique roles of WT1 during heart development and disease. WT1 is expressed in prostate cancer (PC) epithelial cells and regulates PC critical genes. WT1 promotes metastatic disease by enhancing motility of PC cells with low-migratory and metastatic potential. While the mechanisms are multifactorial, chapter 14 focuses on how WT1 interacts with vascular endothelial growth factor (VEGF) and androgen receptor to promote prostate cancer progression and metastasis. While WT1 is widely considered as a tumor suppressor, it can also act as an oncogene in some cancers. For example, WT1 is overexpressed in most glioblastoma. Chapter 15 describes the functional role of WT1 in glioblastoma and how it regulates proliferation and apoptosis of glioblastoma cells. Finally, chapter 16 focuses on the role of WT1 in minimal residual disease in acute myeloid leukemia (AML). WT1 is overexpressed at mRNA level in 80-90% of AML cases, and there are reports of poor outcome for patients having WT1 levels above reference thresholds at specific time points. This chapter gives a comprehensive review of the role of WT1 in AML, molecular markers to stratify high-risk AML patients, and interventional therapy based on WT1 expression.

Preface

The intended audience of this book is students, basic scientists, and clinicians who are interested in the basic and/or clinical aspects of WT and *WT1*. It is our wish that this book would serve as an authoritative source for readers who want a comprehensive understanding of the development, progression, management, and treatment of WT.

Marry M. van den Heuvel-Eibrink, MD, PhD Professor of Oncology/Hematology Princess Máxima Center for Pediatric Oncology, Utrecht, The Netherlands March 2016 DOI: http://dx.doi.org/10.15586/codon.wt.2016.pr

Adina Brett-Morris, PhD

Case Western Reserve University, Cleveland, OH, USA

Alan Davidson, MBChB, FCPaed, Cert Med Onc (Paeds), MPhil

Department of Haematology and Oncology, University of Cape Town and Red Cross War Memorial Children's Hospital, Cape Town, South Africa

Alastair J. W. Millar, FRCS, FRACS (Paed Surg), FCS(SA), DCH

Department of Paediatric Surgery, University of Cape Town and Red Cross War Memorial Children's Hospital, Cape Town, South Africa

Angelo Michele Carella, MD

Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, San Giovanni Rotondo, Italy

Anton Bazarov, BS, MD

Kent State University, Kent, OH, USA

B. Therese Kinsella, PhD

UCD School of Biomolecular and Biomedical Sciences, UCD Conway Institute of Biomolecular and Biomedical Research, University College Dublin, Belfield, Dublin 4, Ireland

Basak Erginel, MD

Department of Pediatric Surgery, Istanbul University Istanbul Medical Faculty, Istanbul, Turkey

Beata S. Lipska-Ziętkiewicz, MD, PhD

Department of Biology and Genetics, Medical University of Gdansk, Gdansk, Poland

Boudewijn P. T. Kruithof, PhD

Department of Molecular Cell Biology, Leiden University Medical Center, Leiden, The Netherlands

Can Lai, MD

Department of Radiology, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Da-Xing Tang, MD

Division of Urology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

De-Hua Wu

Division of Urology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Dirce Maria Carraro, PhD

Laboratory of Genomics and Molecular Biology, CIPE, AC Camargo Cancer Center, Sao Paulo, SP, Brazil

Gail C. Fraizer, MPH, PhD Kent State University, Kent, OH, USA

Giovanni Rossi, MD, PhD

Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, San Giovanni Rotondo, Italy

Gordan M. Vujanic, MD, PhD, FRCPath

Department of Cellular Pathology, University Hospital of Wales, Cardiff, UK

Gulden Diniz, MD, PhD Izmir Tepecik Education and Research Hospital, Izmir, Turkey

Haruo Sugiyama, MD, PhD

Department of Functional Diagnostic Science, Osaka University Graduate School of Medicine, Osaka, Japan

Heidi Segers, MD, PhD Department of Pediatric Oncology, University Hospital Leuven, Leuven, Belgium

Hong-Feng Tang, MD Department of Pathology, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Jeffrey S. Dome, MD, PhD

Division of Oncology, Children's National Health System, Washington, District of Columbia, USA

Jin-Hu Wang, MD

Division of Surgical Oncology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Joanna Stefanowicz, MD, PhD

Department of Paediatrics, Haematology and Oncology, Medical University of Gdansk, Gdansk, Poland

Kathy Pritchard-Jones, MD, PhD

Department of Pediatric Hematology and Oncology, Institute of Child Health, University College, London, United Kingdom

Kenji Sano, MD, PhD

Department of Laboratory Medicine, Shinshu University Hospital, Matsumoto, Japan

Koichi Hirabayashi, MD, PhD

Center for Advanced Cell Therapy, Department of Pediatrics, Shinshu University Hospital, Matsumoto, Japan

Kurtis Eisermann, PhD

Kent State University, Kent, OH, USA, University of Pittsburgh, Pittsburgh, PA, USA

Lorella Melillo, MD

Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, San Giovanni Rotondo, Italy

Maria Marta Minervini, MD

Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, San Giovanni Rotondo, Italy

Mariana Maschietto, PhD

Brazilian Biosciences National Laboratory (LNBio), Center for Research in Energy and Materials (CNPEM), Campinas, SP, Brazil

Marie-José Goumans, PhD

Department of Molecular Cell Biology, Leiden University Medical Center, Leiden, The Netherlands

Marry M. van den Heuvel-Eibrink, MD, PhD

Department of Pediatric Oncology, Princess Máxima Center for Paediatric Oncology, Utrecht, The Netherlands

Mateusz Kosiak, MD, PhD

Department of Family Medicine, Medical University of Gdansk, Gdansk, Poland, Medical Centre, Puck, Poland

Mehran Hiradfar, MD

Department of Pediatric Surgery, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran

Min-Ju Li, MD

Division of Surgical Oncology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Naoya Hashimoto, MD, PhD

Department of Neurosurgery, Kyoto Prefectural University of Medicine, Kyoto, Japan

Neil J. Sebire, MD, FRCPath

Department of Histopathology, Great Ormond Street Hospital, London, UK

Nicola Cascavilla, MD

Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, San Giovanni Rotondo, Italy

Nirmala Ghimirey

Kent State University, Kent, OH, USA

Nona Zabolinejad, MD

Department of Pathology, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran

Norbert Graf, MD, PhD

Department of Pediatric Hematology/Oncology, University Hospital for Children, Homburg, Germany

Noriyuki Kijima, MD, PhD

Department of Neurosurgery, Osaka National Hospital, Osaka, Japan

Paramahamsa Maturu, PhD

Department of Genetics, The University of Texas MD Anderson Cancer Center, 1515 Holcombe Blvd, Unit 1010, Houston, TX 77030, USA, Department of Pediatrics, Section of Neonatology, Texas Children's Hospital, Baylor College of Medicine, 1102 Bates Avenue, Houston, TX 77030, USA

Parisa Saeedi Sharifabad, MD

Department of Pediatric Urology, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran

Qiang Shu, MD

Division of Surgical Oncology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Reza Shojaeian, MD

Department of Pediatric Surgery, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran

Rodrigo F. Ramalho, PhD

Laboratory of Genomics and Molecular Biology, CIPE, AC Camargo Cancer Center, Sao Paulo, SP, Brazil

Ryu Yanagisawa, MD, PhD

Center for Advanced Cell Therapy, Department of Pediatrics, Shinshu University Hospital, Matsumoto, Japan

Sarah Nock, BS

Kent State University, Kent, OH, USA

Saskia L. Gooskens, MD

Department of Pediatric Oncology, Princess Máxima Center for Paediatric Oncology, Utrecht, The Netherlands, Department of Pediatric Hematology and Oncology, Erasmus MC – Sophia Children's Hospital, Rotterdam, The Netherlands

Sergey D. Popov, MD, PhD

Department of Cellular Pathology, University Hospital of Wales, Cardiff, UK

Shan Xu, MD

Division of Urology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Sharon Cox, MbChB, FCS(SA), Cert Paed Surg (SA)

Department of Paediatric Surgery, University of Cape Town and Red Cross War Memorial Children's Hospital, Cape Town, South Africa

Shigetaka Shimodaira, MD, PhD

Center for Advanced Cell Therapy, Shinshu University Hospital, Matsumoto, Japan

Sjoerd N. Duim, MSc

Department of Molecular Cell Biology, Leiden University Medical Center, Leiden, The Netherlands

Sony Pandey, BS Kent State University, Kent, OH, USA

Steven J. Kuerbitz, MD Akron Children's Hospital, Akron, OH

Tomonobu Koizumi, MD, PhD

Shinshu Cancer Center, Shinshu University Hospital, Matsumoto, Japan

Toshiki Yoshimine, MD, PhD

Department of Neurosurgery, Osaka University Graduate School of Medicine, Osaka, Japan

Wojciech Kosiak, MD, PhD

Department of Paediatrics, Haematology and Oncology, Medical University of Gdansk, Gdansk, Poland

Yasunori Fujimoto, MD, PhD

Department of Neurosurgery, Osaka University Graduate School of Medicine, Osaka, Japan

Yasuyoshi Chiba, MD, PhD

Department of Neurosurgery, Osaka Medical Center and Research Institute for Maternal and Child Health, Osaka, Japan

Yong Huang

Division of Urology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Yumiko Higuchi, PhD

Center for Advanced Cell Therapy, Shinshu University Hospital, Matsumoto, Japan

DOI: http://dx.doi.org/10.15586/codon.wt.2016.cont

Section I

Epidemiology, Diagnostics, and Treatment

Chapter 1

Wilms' Tumour – Histology and Differential Diagnosis

Sergey D. Popov,¹ Neil J. Sebire,² Gordan M. Vujanic¹

¹Department of Cellular Pathology, University Hospital of Wales, Cardiff, UK; ²Department of Histopathology, Great Ormond Street Hospital, London, UK

Author for correspondence: Sergey D. Popov, MD, PhD, Department of Cellular Pathology, University Hospital of Wales, Cardiff CF14 4XW, UK. Email: sergey.popov@icr.ac.uk; sergey.popov@wales.nhs.uk

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch1

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Wilms' tumour (WT) is the most common paediatric renal tumour, which can present as a single nodule, as multifocal unilateral lesions or as bilateral tumours. Typically, WT comprises three histological components namely blastemal, epithelial and stromal. The proportion and the degree of maturation of these components vary significantly, making the histological appearance of each tumour unique. Classical triphasic WT rarely presents diagnostic difficulty for pathologists, but when only one component is present, especially in a small biopsy specimen, the differential diagnosis may

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Popov et al.

include renal cell carcinoma, metanephric adenoma and hyperplastic nephrogenic rest for epithelial elements and clear cell sarcoma of the kidney, mesoblastic nephroma and synovial sarcoma for stromal elements. Pure blastemal-type WT may be difficult to distinguish from other embryonal 'small round blue cell tumours', including neuroblastoma, primitive neuroectodermal tumour/Ewing sarcoma, desmoplastic small round cell tumour and lymphoma. All the three components, though usually blastema, can become anaplastic, leading to the diagnosis of either focal or diffuse anaplasia. WT with diffuse anaplasia and WT with blastemal predominance (after preoperative chemotherapy) are regarded as high-risk tumours and require more aggressive treatment. Careful assessment of the tumour and the normal kidney is critical for accurate subtyping and staging of WT, which is the basis for post-operative treatment. In addition, the identification and correct interpretation of nephrogenic rests may affect prognosis and management. Histological distinction between WT and nephrogenic rest is not always possible based on morphology alone, and implementation of new molecular genetic tools may aid in this regard. Other molecular genetic signatures of WT, such as P53 mutation and MYCN dysregulation, may provide future additional prognostic and therapeutic information.

Key words: Nephrogenic rest; Pathology; Wilms' tumour

Introduction

Renal tumours comprise 7–8% of all paediatric tumours in children under 15 years of age, and among those, Wilms' tumour (WT) or nephroblastoma is the most common neoplasm (1). The frequency of renal malignancies in childhood is listed in Table 1.

Tumour	Relative frequency (%)
Wilms' tumour	85
Mesoblastic nephroma	2-3
Clear cell sarcoma	3
Rhabdoid tumour	2
Renal cell carcinoma	5
Others	3

Table 1. Primary renal tumours in childhood

Wilms' tumour – pathology

There are several reasons why making correct diagnosis of WT may be challenging for general or paediatric pathologists:

- Rarity of the paediatric renal tumours results in lack of experience with these entities for most of the pathologists (2)
- 2. Presence of several subtypes of WT (morphological heterogeneity)
- 3. Morphological appearances may vary dramatically from case to case
- 4. Histological patterns of certain WT subtypes may appear initially similar to those of other rare paediatric renal tumours
- 5. Lack of sharp differential criteria distinguishing WT from nephrogenic rests (NRs), especially in limited biopsy material
- 6. Assessment of the tumour and determination of the local pathology stage are multistep and time-consuming processes

Preoperative chemotherapy may create additional difficulty in precise tumour assessment because the criteria for tumour subtyping and risk-group stratification are different for treated and untreated cases (3–5).

Gross appearance

Macroscopically, WTs are usually large masses disconfigurating the renal contours, which can vary in size significantly. Multicentric tumours occur in 5%, and they are usually associated with NRs (6). Precaution should be taken for the cutting procedure because the cut surface of the tumour may expand from the surrounding pseudocapsule, making the microscopic assessment of tumour margins more difficult. Macroscopic appearance of the cut surface is heterogeneous in many cases, with areas of viable tumour, haemorrhage and necrosis, especially in pre-treated specimens. Viable tumour is usually solid, pale grey to slightly pink or yellow-grey with soft consistency. Some tumours are markedly cystic, and careful search for the presence of solid foci is required. To avoid artificial contamination by the tumour cells, it is important to sample the hilar margins, including vessels, if possible before the tumour is incised.

Histological features

Classical histological features of WT include a triphasic pattern of epithelial, stromal and blastemal components (Figure 1). The proportions of these components and their lines and degree of differentiation vary significantly, resulting in countless tumour appearances. Biphasic and monophasic variants are not uncommon. Preoperative chemotherapy, given to children treated according to the International Society of Paediatric Oncology (SIOP) protocol, may affect the original histology dramatically by reducing or enhancing certain elements or by inducing maturation (7, 8).

Blastema represents the least differentiated, and presumed most malignant, component and consists of small round blue cells with overlapping nuclei and brisk mitotic activity. Several histological Popov et al.



Figure 1. Wilms' tumour: mixed pattern with blastema, stroma and single epithelial structures.

patterns of blastema, including diffuse, serpentine, nodular and basaloid, have been described. The serpentine pattern of growth is characterised by broad bands of undifferentiated cells surrounded by fibromyxoid stroma. In the basaloid variant, nests or cords of blastema have a distinctive peripheral palisading of elongated cells with epithelial differentiation. All four above-mentioned patterns may be found in the same tumour and have no prognostic significance; however, their recognition in the histological slides can be helpful in differential diagnosis with other 'small round blue cell tumours' when the tumour is composed of the blastemal component only. It is worth noting that although WTs are mostly well circumscribed and surrounded by a pseudocapsule, which is used as one of the differential diagnostic criteria, blastemal-type WTs, usually with diffuse growth pattern, can show marked infiltrative growth with no pseudocapsule between the tumour and adjacent tissues. Primitive tubular epithelial structures sometimes present in the centre of blastemal nodules may morphologically mimic neuroblastoma-like areas with pseudorosettes. Vague epithelioid or spindle cell appearances are other possible histological features of blastema depending on the extent and pattern of early differentiation. There are no strict criteria to discriminate blastema from early epithelial differentiation (Figure 2) or stromal lineage, with almost all literature describing WT subtypes being based on subjective morphological criteria.

The epithelial component may demonstrate the whole spectrum of differentiation from early stages of tubular formation with primitive epithelial rosette-like structures to somewhat differentiating tubules or glomeruli-like structures, reflecting different stages of nephrogenesis. Squamous epithelial islands and mucinous epithelium are examples of heterologous differentiation within the epithelial component of WT. Wilms' tumour - pathology



Figure 2. Blastemal-type Wilms' tumour with early epithelial differentiation.

The stromal component may include densely packed undifferentiated mesenchymal cells or loose cellular myxoid areas. The latter areas may be difficult to distinguish from nontumorous stroma associated with chemotherapy-induced change (CIC). Heterologous differentiation of neoplastic stroma in the form of well-differentiated smooth or skeletal muscle cells, fat tissue, cartilage, bone and even glial tissue is present in some cases, especially in tumours that have undergone preoperative chemotherapy (Figure 3).

CIC includes areas of necrosis, haemorrhage and fibrosis of varying degree and areas with foamy and/or haemosiderin-laden macrophages. Primitive, highly proliferative blastemal component more readily responds to chemotherapy, leaving homogeneous eosinophilic areas where 'shadows' of pre-existing cells and structures may be seen. Mature epithelial and stromal components are often less sensitive to chemotherapy, and such tumours may show no significant response to pre-operative therapy in terms of tumour-size shrinkage. It is worth emphasising that the criteria and terminology used by the SIOP and National Wilms' Tumor Study/Children's Oncology Group (NWTS/COG) differ, so direct comparison of certain subtypes is not feasible. Histological assessment of tumour responsiveness to chemotherapy is important for risk-group stratification by the SIOP. For instance, completely necrotic WT is regarded as a low-risk tumour and requires less post-operative therapy than WTs from other groups. Further, stromal- or epithelial-type WTs are terms used by the SIOP for pre-treated tumours, whereas the NWTS/COG uses terms such as stromal or epithelial

Popov et al.



Figure 3. Preoperatively treated Wilms' tumour with prominent skeletal muscle differentiation and cartilage.

predominant WT (Table 2). In non-treated cases, stromal or epithelial predominant tumours may contain up to one-third of the blastemal component, whereas in pre-treated cases, the finding of >10% of blastema would result in the tumour being sub-classified as mixed type (9).

Anaplastic Wilms' tumours account for 5–8% of all WTs, and the majority of patients with anaplastic WT (Figure 4) are older than those with non-anaplastic WT. The criteria necessary for the diagnosis of anaplasia are the presence of large, atypical multipolar mitotic figures and significantly enlarged and hyperchromatic nuclei (10). These tumours are generally aneuploid. Anaplasia may be focal or diffuse. Focal anaplasia means that there is a localised and definitely completely excised area with anaplastic features. All other cases where anaplasia is found should be regarded as diffuse anaplasia. Diffuse anaplasia is regarded as the only unfavourable histological feature in WTs undergoing primary nephrectomy. Anaplasia is responsible for adverse outcome, especially in the cases with advanced tumour stage; thus, its recognition is essential for the prognosis and treatment. Because anaplasia is regarded as a chemo-resistant cell clone, it may be easier to detect it in pre-treated cases due to loss of other chemo-sensitive elements. Anaplastic tumours often express p53 on immunohistochemical staining and bear mutants in the *TP53* gene (11–14). *TP53* mutation has been shown to compromise patients' survival, overall and event-free, and therefore has the potential as an adverse prognostic factor combined with anaplastic

Wilms' tumour - pathology

Pre-treated tumours*	Primary nephrectomy tumours	
Low risk	Low risk	
Mesoblastic nephroma	Mesoblastic nephroma	
Cystic partially differentiated nephroblastoma	Cystic partially differentiated nephroblastoma	
Completely necrotic nephroblastoma		
Intermediate risk	Intermediate risk	
Nephroblastoma – epithelial type	Non-anaplastic nephroblastoma and its variants	
Nephroblastoma – stromal type	Nephroblastoma – focal anaplasia type	
Nephroblastoma – mixed type		
Nephroblastoma – regressive type		
Nephroblastoma – focal anaplasia type		
High risk	High risk	
Nephroblastoma – blastemal type	Nephroblastoma – diffuse anaplasia type	
Nephroblastoma – diffuse anaplasia type	Clear cell sarcoma of the kidney	
Clear cell sarcoma of the kidney	Rhabdoid tumour of the kidney	
Rhabdoid tumour of the kidney		

Table 2. Current SIOP classification of paediatric renal tumours

*The criteria for subclassifying pre-treated WTs are as follows: completely necrotic type shows no viable tumour elements. If more than 66% (two-thirds) of the tumour is non-viable (i.e., shows chemotherapy-induced changes), it is regarded as regressive type, irrespective of the presence of remaining viable tumour components. If viable tumour comprises more than one-third of the tumour mass, subtyping depends on the percentage of viable components: in mixed type, none of the components comprise more than 66% of the tumour; in epithelial (or stromal) type, in addition to having more than 66% of the tumour being composed of epithelial (or stromal) elements, the finding of only up to 10% of blastema is allowed (if the finding is more, then the tumour is subclassified as mixed type).

morphological features (15). Dysregulation of *MYCN* gene in WTs with anaplastic histology has also been reported to be involved in the development of tumours with adverse outcome (16).

Handling of the nephrectomy specimen

Core biopsies are done in some cases, and their main purpose is to confirm whether a tumour is a WT, in order to give appropriate pre-operative chemotherapy. If biopsies contain enough

Popov et al.



Figure 4. Wilms' tumour with anaplasia.

tissue for diagnostic purpose, some material should be kept frozen for molecular biology studies.

Immediately after surgery, the tumour should be delivered to the pathology department for appropriate handling of the specimen. Careful assessment of the surface and of the margins of renal vessels and the ureter and the assessment of the renal capsule for breaches are critical points for adequate staging (17). The nephrectomy specimen should be inked after photography and measurement. After opening (bivalving), tumour and normal renal tissues are taken for biological studies. Additional parallel slices are usually needed for a large tumour, but they should not compromise staging assessment of the fixed neoplasm. Careful mapping of the specimen, photographs and precise block guides are crucial in the staging assessment. At least one whole longitudinal slice of the tumour is sampled, with additional blocks taken from grossly different areas. When multicentric tumour is present, each nodule is sampled for histology and molecular biology study. Interface between the tumour and normal kidney as well as blocks containing renal and tumour capsule are always taken for histological examination. Evaluation of the renal sinus involvement is very important for the staging purpose; hence, this part of the specimen is a subject of thorough investigation especially when the tumour compromises the normal sinus architecture (18). The residual kidney is also sampled for possible presence of NRs. The hilar fat and all lymph nodes are sampled in search for possible metastases.

Wilms' tumour – pathology

Current staging criteria for pre-treated and non-treated tumours are shown in Tables 3 and 4. The presence of tumour cells in the vessels within the tumour mass does not generally change the stage unless it is found in the vessels of the renal sinus. The finding of non-viable tumour and/or secondary inflammatory changes in the renal sinus or perirenal fat is not the criterion for stage II. However, the finding of non-viable tumour at the resection margins is currently regarded as a reason for stage III in the SIOP protocol (9). The presence of Tamm-Horsfall protein and mature tubules in lymph nodes is occasionally seen, but it should not be regarded as an evidence of metastatic disease (19). Recent studies have shown that there is considerable discrepancy in diagnosing and staging of these tumours between the institutional pathologists and the central pathology reviewers (around 20% of cases), so rapid central pathology review is being introduced and recommended in renal tumour trials (2, 20, 21).

Table 3. SIOP staging system

WT 2001 staging criteria for pre-operatively chemotherapy-treated tumour*

Stage I

- a. The tumour is limited to the kidney or surrounded with a fibrous pseudocapsule if outside the normal contours of the kidney. The renal capsule or pseudocapsule may be infiltrated by the tumour, but it does not reach the outer surface
- b. The tumour may be protruding ('bulging') into the pelvic system and 'dipping' into the ureter, but it is not infiltrating their walls
- c. The renal sinus (its vessels and soft tissues) is not involved
- d. Intrarenal vessels may be involved

Notes: Fine-needle aspiration or percutaneous core needle biopsy does not upstage the tumour, but the size of the needle gauge should be mentioned to the pathologist

The presence of necrotic tumour or chemotherapy-induced change in the renal sinus and/or within the perirenal fat should not be regarded as a reason for upstaging the tumour, provided it is completely excised and does not reach the resection margins

Stage II

- a. Viable tumour penetrates through the renal capsule and/or fibrous pseudocapsule into perirenal fat but is completely resected (resection margins 'clear')
- b. Viable tumour infiltrates the soft tissues and/or blood and/or lymphatic vessels of the renal sinus
- c. Viable tumour infiltrates the perirenal tissue, but it is completely resected
- d. Viable tumour infiltrates the renal pelvic or ureter's wall
- e. Viable tumour infiltrates adjacent organs or vena cava but is completely resected

Notes: Infiltration of the adrenal gland is not regarded as stage II if there is a (pseudo)capsule. Equally, tumour adherence to the liver is not regarded as stage II for which there should be a genuine infiltration of the liver parenchyma

(Continued)

Popov et al.

Table 3. (Continued)

WT 2001 staging criteria for pre-operatively chemotherapy-treated tumour*

Stage III

- a. Viable or non-viable tumour present at resection margins
- b. Any abdominal lymph nodes are involved
- c. Tumour rupture before or intraoperatively (irrespective of other criteria for staging)
- d. Tumour penetration through the peritoneal surface
- e. Tumour implants are found on the peritoneal surface
- f. Tumour thrombi present at resection margins of extra-renal vessels, transected or removed piecemeal by the surgeon
- g. The tumour has been surgically biopsied (wedge biopsy) prior to preoperative chemotherapy or surgery

Note: The presence of necrotic tumour or chemotherapy-induced changes in a lymph node or at the resection margins is regarded as a proof of previous tumour with microscopic residue, and therefore, the tumour is assigned stage III (because of the possibility that some viable tumour is left behind in the adjacent lymph node or beyond the resection margins)

Stage IV

a. Haematogenous metastases (lung, liver, bone, brain, etc.) or lymph node metastases outside the abdominopelvic region

Stage V

a. Bilateral renal tumours at diagnosis. Each side should be substaged according to the above criteria

*Data from reference (9) with additional notes for stage II.

Nephrogenic rests and nephroblastomatosis

NRs are abnormal areas of embryonic tissue persisting beyond 36 weeks of development. They are found in 30–44% of kidneys with WT. The term 'nephroblastomatosis' was introduced in 1961 by Hou and Holman (22) in their description of a lesion composed of immature renal tissue in the kidney of a premature infant. Later, the term was adopted by Beckwith et al. (23)and Beckwith (24) who developed the theory of WT origin from NR.

There are two main types of NR – perilobar (PLNR) and intralobar (ILNR). The former is located at the periphery of the renal lobules and the latter in the central part of the lobe. ILNR is believed to arise earlier in the development when compared with PLNR, which may explain the higher frequency of heterologous elements in ILNR, such as striated muscle, fat, cartilage and bone. Depending on the stage of their development, both ILNR and PLNR might present with different morphological patterns. Beckwith suggested several histological types, including incipient (in newborns and young infants) or dormant (in older infants or children), regressing or sclerotic, obsolescent, and hyperplastic NR.
Wilms' tumour - pathology

Tal	ble	4.	COG	staging	system
-----	-----	----	-----	---------	--------

WT staging criteria for non-treated tumours prior to operation*
 Stage I a. Tumour limited to the kidney and completely resected b. Renal capsule intact c. The tumour was not ruptured or biopsied prior to removal d. Renal vein contains no tumour (intrarenal vessel involvement may be present) e. No residual tumour apparent beyond the margins of excision
 Stage II a. Tumour extends beyond the kidney but is completely resected b. Regional extension of tumour (vascular invasion outside the renal parenchyma or within the renal sinus and/or capsular penetration with negative excision margin) c. Operative tumour spill confined to flank (no peritoneal contamination) d. Tumour biopsy (except fine-needle aspiration) prior to surgery
 Stage III a. Non-haematogenous metastases confined to the abdomen (e.g., tumour in regional lymph nodes), including tumour implants on or penetrating the peritoneum b. Gross or microscopic tumour remains post-operative (tumour at the margins of resection) c. Tumour spill before or during surgery not confined to flank d. Piecemeal excision of the tumour (removal in >1 piece)
Stage IV a. Presence of haematogenous metastases or metastases to distant lymph nodes
Stage V a. Bilateral renal involvement at the time of initial diagnosis
*Data from reference (2)

*Data from reference (3).

The presence of multifocal NRs is defined as nephroblastomatosis. In the condition called diffuse hyperplastic perilobar nephroblastomatosis, which is often bilateral, a large portion of the cortical renal parenchyma is replaced with a thick 'crust' composed of proliferating nephroblastic tissue. It is important to distinguish NR from WT because their clinical management differs significantly. The usual differential diagnostic guides emphasise the criteria such as a lack of fibrous pseudocapsule in NR, which is almost always present in WT cases (Figures 5 and 6). This observation provides a useful tool for pathologists dealing with untreated nephrectomy specimens. However, for patients treated according to the SIOP protocol receiving pre-operative chemotherapy, a fibrous capsule may be present even around the foci of NRs. Conversely, blastemal-type WT may show no separation from the renal parenchyma by the pseudocapsule. In addition, because their microscopic features may be very similar, distinguishing WT from NR in limited needle biopsy material is virtually impossible. In such cases, it has been suggested to use the term

Popov et al.

'nephroblastic process, consistent with either WT or NR' as optimal, with further radiologic-pathologic correlation being required (25). The main differential diagnostic criteria for NR and WT are summarised in Table 5, but one has to bear in mind that none of them is absolutely conclusive.

Another challenge for pathologists is to assess the local stage of the tumour in the presence of ILNR. Providing the frequent location of the ILNR next to the renal sinus or even in the sinus or in the calyceal wall can be misinterpreted as renal sinus invasion by the tumour, leading to upstaging and unnecessary more aggressive treatment.

There are no reliable immunohistochemical or molecular markers facilitating differential diagnosis of NR and WT. A recent study showed significant variability of methylation profiles in NRs and WTs and reported changes in the methylome to underlie NR formation and transformation to WTs in a subset of cases (26). These data have the potential for being implemented into the clinical differential diagnosis of these two lesions, but more extensive work is required.

Differential diagnosis

The diagnosis is usually straightforward in triphasic or even biphasic WTs, although their subclassification may be challenging (27). However, monophasic WTs may be very difficult



Figure 5. View of treated case of hyperplastic perilobar nephroblastomatosis. Partially developed fibrous capsule is seen.

Wilms' tumour - pathology

	Table 5.	Features	of NR	and	WT
--	----------	----------	-------	-----	----

WT	NR
Shape – spherical	Shape – oval
Fibrous capsule is present	No fibrous capsule*
Skeletal muscle differentiation is common	Skeletal muscle differentiation is uncommon
Usually solitary	Often multifocal

NR, nephrogenic rest; WT, Wilms' tumour.

*In untreated cases but in pre-treated cases, capsule may be present.

to separate from other renal tumours with similar histological features. Pure blastemal-type WTs have to be distinguished from other undifferentiated tumours, such as neuroblastoma, primitive neuroectodermal tumour/Ewing sarcoma of the kidney (28), desmoplastic small round cell tumour (29) and synovial sarcoma (30). It is particularly important to consider non-WTs in older patients (Table 5) and adults – WT in adults definitely exists, but many of the renal tumours that in the past were labelled as adult WTs proved to be some of the mentioned entities. In order to reach the correct diagnosis in such cases, it is critical to apply immunohistochemistry and molecular biology investigations looking for characteristic features. Although blastemal components may show focal CD99 positivity, it



Figure 6. Hyperplastic perilobar nephroblastomatosis – direct interface with normal renal parenchyma with no pseudocapsule.

Popov et al.

is usually not diffuse and membranous as in Ewing sarcoma of the kidney, where genetic studies also show characteristic translocations, with t(11;22)(q24;q12) being the most common (28). Desmoplastic small round cell tumour shares many immunohistochemical features with blastemal-type WT but is rare, and the diagnosis should only be made if genetic investigations demonstrate the EWS-WT1 t(11;22)(q13;q12) translocation(29). Neuroblastoma usually shows elevated levels of catecholamines, and on histological examination, its cells reveal non-overlapping nuclei and coarse 'salt and pepper' chromatin. Both tumours may be positive for neuron-specific enolase and CD56, but WT1 marker is negative in neuroblastoma and NB84a marker is negative in WT. In the past, in rare cases, a rhabdoid tumour could be mistaken for a WT, but now it is simple to distinguish between them based on immunohistochemistry, with the lack of nuclear INI1 expression in rhabdoid tumour (31). Pure epithelial-type WT may be difficult to distinguish from metanephric adenoma, renal cell carcinoma and hyperplastic PLNR. Highly differentiated epithelial-type WT may be composed of small, well-differentiated and closely packed tubules similar to metanephric adenoma, but the latter can be diagnosed by the lack of capsule between the tumour and renal parenchyma and the absence of mitotic activity. The combination of CK7-, AMACR-, WT1+ and CD57+ has been shown as an immunohistochemical pattern of metanephric adenoma (32). Renal cell carcinomas in children associated with translocations show distinctive histological features, but papillary renal cell carcinoma (as seen in adults) may be more challenging to diagnose. Immunohistochemistry demonstrating the expression of markers such as CK7 and CD10 (33, 34) and cytogenetic findings may be very helpful (35).

In the differential diagnosis of pure stromal-type WTs, a clear cell sarcoma of the kidney and mesoblastic nephroma should be considered. In WTs treated with preoperative chemotherapy, the stroma may show a striking clear cell sarcoma-like appearance, and extensive sampling may be required in order to find the foci with other WT components.

WT with prominent cystic appearance has to be differentiated from cystic nephroma (CN) and cystic partially differentiated nephroblastoma(CPDN). CN and CPDN share some clinical-pathological features and were regarded as related lesions. They occur in young children, are well demarcated from the kidney and are composed of cysts only, with no solid nodules. Histologically, the only difference between these lesions is the finding of blastema in the septa of CPDN, whereas CN contain no blastema (36). However, despite clear similarities, recent studies showed that these lesions are not related at all and that CN shows DICER1 mutations that are never found in CPDN or WT (37). Still, both CN and CPDN are adequately treated with resection alone, with an excellent prognosis and no chemotherapy required, whereas cystic WTs, which are usually stage I and therefore having a fairly good

Wilms' tumour - pathology

Age (years)	Most common	Possible	Rare
Birth	MN	WT	RTK
<1	WT, MN	RTK, CCSK	
1-5	WT	CCSK	MN (<3 years), RTK
5-10	WT	CCSK, RCC	
11-15	WT, RCC	PNET	

Table 6.	Age and	likely renal	tumour*
----------	---------	--------------	---------

*Data from reference (9).

CCSK, clear cell sarcoma of the kidney; MN, mesoblastic nephroma; PNET, primitive neuroectodermal tumour; RCC, renal cell carcinoma; RTK, rhabdoid tumour of the kidney; WT, Wilms' tumour.

prognosis, should be treated according to the current protocol for WT. Although CN is a benign neoplasm, its association with the malignant pleuropulmonary blastoma has been reported recently (38).

Awareness of the age distribution for paediatric kidney tumours (Table 6) might assist in their differential diagnosis (39–41).

Conclusion

Remarkable progress in classification, treatment and understanding of the pathology and molecular biology of WT has been made over recent decades. Because this tumour is rare, it still represents a diagnostic problem, and awareness of the potentially complex pathological features of this malignancy is required for the accurate diagnosis, subtyping and staging to allow appropriate treatment. Preoperative chemotherapy may affect histological and staging features, and diagnostic pathologists should be familiar with these when assessing such tumours. Adequate handling and sampling are essential prerequisites for correct diagnosis. Molecular biology markers are likely to play an even more important role in the tumour prognosis and differential diagnosis in future, but at present, pathological examination represents the gold standard for diagnosis, subtyping and prognosis.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgement

Not relevant.

Popov et al.

References

- Birch JM, Breslow N. Epidemiologic features of Wilms tumor. Hematol Oncol Clin North Am. 1995;9:1157–78.
- 2. Vujanic GM, Sandstedt B, Kelsey A, Sebire NJ. Central pathology review in multicenter trials and studies: lessons from the Nephroblastoma trials. Cancer. 2009;115:1977–83. http://dx.doi.org/10.1002/cncr.24214
- Qualman SJ, Bowen J, Amin MB, Srigley JR, Grundy RE, Perlman EJ. Protocol for the examination of specimens from patients with Wilms tumor (nephroblastoma) or other renal tumors or childhood. Arch Pathol Lab Med. 2003;127:1280-9. http://dx.doi.org/10.1043/1543-2165(2003)127<1280:PFTEOS>2.0.CO;2
- 4. Delemarre JF, Sandstedt B, Harms D, Boccon-Gibod L, Vujanić GM. The new SIOP (Stockholm) working classification of renal tumours of childhood. International Society of Paediatric Oncology. Med Pediatr Oncol. 1996;26:145-6. http://dx.doi.org/10.1002/(SICI)1096-911X(199602)26:2<145::AID-MPO15>3.0.CO;2-H
- Vujanić GM, Sandstedt B, Harms D, Kelsey A, Leuschner I, de Kraker J; SIOP Nephroblastoma Scientific Committee. Revised International Society of Paediatric Oncology (SIOP) working classification of renal tumors of childhood. Med Pediatr Oncol. 2002;38:79–82.
- Breslow NE, Beckwith JB, Perlman EJ, Reeve AE. Age distributions, birth weights, nephrogenic rests, and heterogeneity in the pathogenesis of Wilms tumor. Pediatr Blood Cancer. 2006;47:260–7.

http://dx.doi.org/10.1002/pbc.20891

- Sebire NJ, Vujanic GM. Paediatric renal tumours: recent developments, new entities and pathological features. Histopathology. 2009;54:516–28. http://dx.doi.org/10.1111/j.1365-2559.2008.03110.x
- Vujanic GM. Renal tumours of childhood: an overview. Diagnostic Histopathology. 2009;15:501–9.
- Vujanić GM, Sandstedt B. The pathology of Wilms' tumour (nephroblastoma): the International Society of Paediatric Oncology approach. J Clin Pathol. 2010;63:102–9. http://dx.doi.org/10.1136/jcp.2009.064600
- Dome JS, Cotton CA, Perlman EJ, Breslow NE, Kalapurakal JA, Ritchey ML, et al. Treatment of anaplastic histology Wilms' tumor: results from the fifth National Wilms' Tumor Study. J Clin Oncol. 2006;24:2352–8. http://dx.doi.org/10.1200/JCO.2005.04.7852
- Jadali F, Sayadpour D, Rakhshan M, Karimi A, Rouzrokh M, Shamsian BS, et al. Immunohistochemical detection of p53 protein expression as a prognostic factor in Wilms tumor. Iran J Kidney Dis. 2011;5:149–53.

- Ghanem MA, van Steenbrugge GJ, Nijman RJ, van der Kwast TH. Prognostic markers in nephroblastoma (Wilms' tumor). Urology. 2005;65:1047–54. http://dx.doi.org/10.1016/j.urology.2004.12.005
- Sredni ST, de Camargo B, Lopes LF, Teixeira R, Simpson A. Immunohistochemical detection of p53 protein expression as a prognostic indicator in Wilms tumor. Med Pediatr Oncol. 2001;37:455–8.
- Lahoti C, Thorner P, Malkin D, Yeger H. Immunohistochemical detection of p53 in Wilms' tumors correlates with unfavorable outcome. Am J Pathol. 1996;148:1577–89.
- 15. Maschietto M, Williams RD, Chagtai T, Popov SD, Sebire NJ, Vujanic G, et al. TP53 mutational status is a potential marker for risk stratification in Wilms tumour with diffuse anaplasia. PLoS One. 2014;9:e109924. http://dx.doi.org/10.1371/journal.pone.0109924
- Williams RD, Chagtai T, Alcaide-German M, Apps J, Wegert J, Popov S, et al. Multiple mechanisms of MYCN dysregulation in Wilms tumour. Oncotarget. 2015;6:7232–43. http://dx.doi.org/10.18632/oncotarget.3377
- Zuppan CW. Handling and evaluation of pediatric renal tumors. Am J Clin Pathol. 1998;109 (Suppl 1):S31–7.
- Perlman EJ. Pediatric renal tumors: practical updates for the pathologist. Pediatr Dev Pathol. 2005;8:320–38.

http://dx.doi.org/10.1007/s10024-005-1156-7

- 19. Weeks DA, Beckwith JB, Mierau GW. Benign nodal lesions mimicking metastases from pediatric renal neoplasms: a report of the National Wilms' Tumor Study Pathology Center. Hum Pathol. 1990;21:1239–44.
- 20. Teot LA, Sposto R, Khayat A, Qualman S, Reaman G, Parham D. The problems and promise of central pathology review: development of a standardized procedure for the Children's Oncology Group. Pediatr Dev Pathol. 2007;10:199–207. http://dx.doi.org/10.2350/06-06-0121.1
- 21. de Kraker J, Delemarre JF, Lilien MR, Tournade MF. Misstaging in nephroblastoma. Causes and consequences. A report of the Sixth Nephroblastoma Trial and Study of the International Society of Paediatric Oncology. Eur J Pediatr Surg. 1999;9:153–7. http://dx.doi.org/10.1055/s-2008-1072232
- 22. Hou LT, Holman RL. Bilateral nephroblastomatosis in a premature infant. J Pathol Bacteriol. 1961;82:249–55.
- 23. Beckwith JB, Kiviat NB, Bonadio JF. Nephrogenic rests, nephroblastomatosis, and the pathogenesis of Wilms' tumor. Pediatr Pathol. 1990;10:1–36.
- Beckwith JB. Precursor lesions of Wilms tumor: clinical and biological implications. Med Pediatr Oncol. 1993;21:158–68.

Popov et al.

- 25. Perlman EJ, Faria P, Soares A, Hoffer F, Sredni S, Ritchey M, et al.; National Wilms Tumor Study Group. Hyperplastic perilobar nephroblastomatosis: long-term survival of 52 patients. Pediatr Blood Cancer. 2006;46:203–21. http://dx.doi.org/10.1002/pbc.20386
- 26. Charlton J, Williams RD, Sebire NJ, Popov S, Vujanic G, Chagtai T, et al. Comparative methylome analysis identifies new tumour subtypes and biomarkers for transformation of nephrogenic rests into Wilms tumour. Genome Med. 2015;7(1):11. http://dx.doi.org/10.1186/s13073-015-0136-4
- Vujanić GM, Sandstedt B, Kelsey A, Sebire NJ. Central pathology review in multicenter trials and studies: lessons from the nephroblastoma trials. Cancer. 2009;115:1977–83.

http://dx.doi.org/10.1002/cncr.24214

- 28. Parham DM, Roloson GJ, Feely M, Green DM, Bridge JA, Beckwith JB. Primary malignant neuroepithelial tumors of the kidney: a clinicopathologic analysis of 146 adult and pediatric cases from the National Wilms' Tumor Study Group Pathology Center. Am J Surg Pathol. 2001;25:133–46.
- Wang LL, Perlman EJ, Vujanic GM, Zuppan C, Brundler MA, Cheung CR, et al. Desmoplastic small round cell tumor of the kidney in childhood. Am J Surg Pathol. 2007;31:576–84.

http://dx.doi.org/10.1097/01.pas.0000213432.14740.14

- Argani P, Faria PA, Epstein JI, Reuter VE, Perlman EJ, Beckwith JB, et al. Primary renal synovial sarcoma: molecular and morphologic delineation of an entity previously included among embryonal sarcomas of the kidney. Am J Surg Pathol. 2000;24:1087–96.
- 31. Hoot AC, Russo P, Judkins AR, Perlman EJ, Biegel JA. Immunohistochemical analysis of hSNF5/INI1 distinguishes renal and extra-renal malignant rhabdoid tumors from other pediatric soft tissue tumors. Am J Surg Pathol. 2004;28:1485–91.
- 32. Kinney SN, Eble JN, Hes O, Williamson SR, Grignon DJ, Wang M, et al. Metanephric adenoma: the utility of immunohistochemical and cytogenetic analyses in differential diagnosis, including solid variant papillary renal cell carcinoma and epithelial-predominant nephroblastoma. Mod Pathol. 2015;28:1236–48. http://dx.doi.org/10.1038/modpathol.2015.81
- Muir TE, Cheville JC, Lager DJ. Metanephric adenoma, nephrogenic rests, and Wilms' tumor: a histologic and immunophenotypic comparison. Am J Surg Pathol. 2001;25:1290–6.
- Truong LD, Shen SS. Immunohistochemical diagnosis of renal neoplasms. Arch Pathol Lab Med. 2011;135:92–109. http://dx.doi.org/10.1043/2010-0478-RAR.1.

- Soller MJ, Kullendorff CM, Békássy AN, Alumets J, Mertens F. Cytogenetic findings in pediatric renal cell carcinoma. Cancer Genet Cytogenet. 2007;173:75–80. http://dx.doi.org/10.1016/j.cancergencyto.2006.09.018.
- Joshi VV, Beckwith JB. Multilocular cyst of the kidney (cystic nephroma) and cystic, partially differentiated nephroblastoma. Terminology and criteria for diagnosis. Cancer. 1989;64:466–79.
- 37. Doros LA, Rossi CT, Yang J, Field A, Williams GM, Messinger Y, et al. DICER1 mutations in childhood cystic nephroma and its relationship to DICER1-renal sarcoma. Mod Pathol. 2014;27:1267–80.
 http://lib.ic./lib.ic.

http://dx.doi.org/10.1038/modpathol.2013.242

38. Boman F, Hill DA, Williams GM, Chauvenet A, Fournet JC, Soglio DB, et al. Familial association of pleuropulmonary blastoma with cystic nephroma and other renal tumors: a report from the International Pleuropulmonary Blastoma Registry. J Pediatr. 2006;149:850–54.

http://dx.doi.org/10.1016/j.jpeds.2006.08.068

 Vujanic GM, Charles AK. Renal tumours of childhood: an update. Pathology. 2008;40:217–27.

http://dx.doi.org/10.1080/00313020701813776

- 40. Popov SD, Sebire NJ, Pritchard-Jones K, Vujanić GM. Renal tumors in children aged 10–16 years: a report from the United Kingdom Children's Cancer and Leukaemia Group Pediatr Dev Pathol. 2011;14:189–93. http://dx.doi.org/10.2350/10-06-0855-OA.1
- Grabowski J, Silberstein J, Saltzstein SL, Saenz N. Renal tumors in the second decade of life: results from the California Cancer Registry. J Pediatr Surg. 2009;44:1148–51. http://dx.doi.org/10.1016/j.jpedsurg.2009.02.019

Chapter 2

The Clinical Relevance of Age at Presentation in Nephroblastoma

Saskia L. Gooskens,^{1,2} Heidi Segers,³ Kathy Pritchard-Jones,⁴ Norbert Graf,⁵ Jeffrey S. Dome,⁶ Marry M. van den Heuvel-Eibrink¹

 ¹Department of Pediatric Oncology, Princess Máxima Center for Paediatric Oncology, Utrecht, The Netherlands; ²Department of Pediatric Hematology and Oncology, Erasmus MC – Sophia Children's Hospital, Rotterdam, The Netherlands;
 ³Department of Pediatric Oncology, University Hospital Leuven, Leuven, Belgium;
 ⁴Department of Pediatric Hematology and Oncology, Institute of Child Health, University College, London, United Kingdom; ⁵Department of Pediatric Hematology/ Oncology, University Hospital for Children, Homburg, Germany; ⁶Division of Oncology, Children's National Health System, Washington, District of Columbia, USA

Author for correspondence: Marry M. van den Heuvel-Eibrink, MD, PhD, Associate Professor of Pediatric Oncology/Hematology, Princess Máxima Center for Paediatric Oncology, Lundlaan 3, 3584 EA Utrecht, The Netherlands. Email: m.m.vandenheuvel-eibrink@prinsesmaximacentrum.nl

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch2

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

The most important prognostic factors for Wilms tumor (WT) patients seem to be stage, histological subtype, and 1p/16q loss of heterozygosity (LOH) in chemotherapy-naive WTs.

In: *Wilms Tumor*. Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Gooskens et al.

Over the last decade, age at diagnosis also was suggested to be an important risk factor for WT recurrence in Children's Oncology Group (COG), United Kingdom (UK), and International Society of Pediatric Oncology (SIOP) studies. Several studies have analyzed age as a prognostic factor; these studies revealed age <2 years as a favorable prognostic factor, while age >4 years has been described as an adverse prognostic factor. In adults (>18 years of age), WT represents less than 1% of all diagnosed renal tumors; therefore, diagnosis of WT in adults is often unexpected and poorly recognized, thereby inducing treatment delay with subsequent adverse outcome. One explanation for the higher risk of recurrence with increasing patient age is the higher frequency of anaplasia at higher age. Other suggested reasons are delay in diagnosis, advanced tumor stage at presentation, and intrinsically different biological behaviors. Whether age is really an *independent* risk factor, and whether age is a stronger prognostic factor than stage, histology, and LOH 1p/16q, needs to be further explored. This may provide some insight into whether older patients need to be treated more intensively, as is already advised for adult WT patients.

Key words: Age; Prognostic factor; Wilms tumor

Introduction

Wilms tumor (WT) is the most common type of childhood renal cancer. It affects approximately one child per 10,000 worldwide before the age of 15 years (1). The median age at diagnosis of WT is approximately 3.5 years (1). The two treatment approaches (European and North America) available for children with WT result in comparable overall survival rates, currently reaching 90%. The International Society of Pediatric Oncology (SIOP) in Europe advocates chemotherapy before nephrectomy, whereas the Children's Oncology Group (COG) in North America recommends immediate surgery (2).

The most important prognostic factors for WT patients seem to be stage, histological anaplastic subtype and blastemal subtype (the latter in chemotherapy-pretreated nephroblastoma cases only), and 1p/16q loss of heterozygosity (LOH) in chemotherapy-naive WTs (3–5). Tumor stage is the original prognostic factor for WT, while tumor histology is perhaps the most powerful prognostic factor for WT; (diffuse) anaplasia is associated with adverse outcome in both the COG and SIOP histologic classification systems, while the adverse prognostic effect of residual blastemal cells after pre-operative chemotherapy is only recognized in the SIOP classification system (4, 5). LOH of 1p/16q is found in around 5% of favorable-histology WTs, and it has been demonstrated to be significantly correlated with less favorable outcome (3).

Over the last decade, age at diagnosis also was suggested to be an important risk factor for WT recurrence in COG, UK (United Kingdom), and SIOP studies (6–9).

Relevance of age in nephroblastoma

The clinical relevance of age

General

Cooperative studies have shown that increasing age is associated with an increased risk of recurrence of nonmetastatic WT (6, 7, 9–11). This is only partly explained by the fact that the occurrence of anaplasia increases with age (12); even in patients with favorable histology, older age seems to be associated with less favorable outcome. It still needs to be determined what the exact age threshold is at which outcome starts to deteriorate.

Infants

The "chemotherapy before surgery strategy" has been under debate internationally for years. SIOP protocols recommend to treat patients >6 months with preoperative chemotherapy; this has the clear evidence-based benefit of downstaging tumors, thereby sparing survivors the late effects of doxorubicin or radiotherapy (14). However, in young infants, the so-called non-WTs tend to occur up to a substantial proportion in the younger age group (13). This initiated a study on all renal tumors in infants (under the age of 7 months at presentation) on a global level, based on data in 750 children, treated in UK, COG, and SIOP protocols, showing that above 2 months of age at presentation, WT is the most common tumor type, while congenital mesoblastic nephroma occurred more often than WT under the age of 3 months at presentation (Figure 1) (13). In addition, the biologically more aggressive malignant rhabdoid tumor of the kidney has a high propensity in this young age group. This has forced



Figure 1. Distribution of renal tumors in children aged 7 months or less (13). CMN: congenital mesoblastic nephroma; CCSK: clear cell sarcoma of the kidney; MRTK: malignant rhabdoid tumor of the kidney.

Gooskens et al.

an international recommendation to, even in the SIOP community, immediately perform surgery instead of pre-operative chemotherapy in these young children with renal tumors.

Survival rates for WT patients below the age of 7 months are very good (5-year overall survival 93.4%). In addition, the incidence of metastatic WT in this age group is very low (<1%) (13).

Children >6 months of age

Several studies have analyzed age as a prognostic factor (Table 1). These studies revealed age <2 years as a favorable prognostic factor, while age >4 years has been described as an adverse prognostic factor. One study specifically addressed the adverse outcome in teenagers (10–16 years of age) (15).

Currently, age is already incorporated into the risk stratification of COG studies (AREN0532); it is predicted that children under the age of 2 years with small tumors (<550 g) and stage I favorable-histology WTs can benefit from surgical treatment only (nephrectomy alone without adjuvant chemotherapy). As stage and histology are considered to be stronger prognostic markers, age is not used for risk stratification in the SIOP trials.

Adults

In adults (>18 years of age), WT represents less than 1% of all diagnosed renal tumors (17–22). The most common type of adult renal cancer is renal cell carcinoma (approximately 85%); therefore, diagnosis of WT in adults is often unexpected and poorly recognized, thereby inducing treatment delay with subsequent adverse outcome (23). This treatment delay, rather than more aggressive biology seems to determine the worse outcome in adults with WT as compared to in children (17–22, 24, 25). More recent data indicate the potential for improvement in adults when pediatric treatment approaches, including multimodality chemo- and radiotherapy adapted from the pediatric treatment protocols, are used (18, 19, 21, 22, 24, 25).

Multiple factors, including the unfamiliarity of adult oncologists with WT, lack of standardized treatment, delay in initiating the appropriate therapy and also a possible more biologically aggressive tumor type, may contribute to poor outcome (22).

This prompted several representatives of the renal tumor committees of the COG and the SIOP to develop, together with adult urologists, medical oncologists, and radiotherapists, a consensus "best practice" guideline for the management of WT in adults (26). The aim of this international consensus recommendation is to further improve outcome by shortening adjuvant treatment delay and by using standardized treatment (26).

Age in correlation with other prognostic factors

One explanation for the higher risk of recurrence with increasing patient age is the higher frequency of anaplasia at higher age. Anaplasia is only very rarely seen in WT diagnosed

Relevance of age in nephroblastoma

Report	Study	Result	Multivariate analysis	Outcome measures
Green et al. (JCO 1993) (6)	NWTS1, NWTS2, NWTS3	Age <2 y as favorable prognostic factor in small (<550 g) stage I favorable-histology tumors	Not performed	4 y EFS
Pession et al. (EJC 2008) (16)	AIEOP 1989–1998	Age ≤2 y as favorable prognostic factor	Age <i>not</i> an indepen- dent prognostic factor	OS
Pritchard-Jones et al. (JCO 2003) (7)	UKW2, UKW3	Age >4 y as adverse prognostic factor in stage I favorable- histology tumors	Age an independent prognostic factor	4 y EFS and OS
Irtan et al. (EJC 2015) (10)	UKW3	Age >4 y as adverse prognostic factor	Age an independent prognostic factor	EFS
Shamberger et al. (Ann Surg 1999) (9)	NWTS4	Age >4 y as adverse prognostic factor	Age <i>not</i> an indepen- dent prognostic factor	EFS
Reinhard et al. (Oncol Rep 2008) (8)	SIOP- GPOH 1989–2003	Age >4 y as adverse prognostic factor	Age <i>not</i> an indepen- dent prognostic factor	OS
Breslow et al. (Cancer 1991) (11)	NWTS3	Age 0-23 m, 5.4% relapse; age 24-47 m, 9.5% relapse; age 48+ m, 16.3% relapse	Not performed	EFS
Popov et al. (Ped and Dev Pathology 2011) (15)	UKW3, SIOP 2001	Age 10–16 y as adverse prognostic factor	Not performed	5 y OS

Table 1. Age as a prognostic factor

during the first year of life and is also rare in the second year of life (12). Nevertheless, even in the group of patients with favorable histology, older age seems to be correlated with a higher risk of relapse and death, although prognostic factors such as stage or histology seem to be more powerful (7). Other suggested reasons for the adverse survival rates in older children are delay in diagnosis, advanced tumor stage at presentation, and intrinsically different biological behaviors (7).

NWTS: National Wilms Tumor Study; AIEOP: Associazione Italiana Ematologia ed Oncologia Pediatrica; UKW: United Kingdom Wilms Tumour Study; SIOP: International Society of Pediatric Oncology; GPOH: Gesellschaft fur Padiatrische Onkologie und Hamatologie; y: year; m: month; EFS: event-free survival; OS: overall survival.

Gooskens et al.

While age has been described as an independent risk factor in two (UK) studies, it did not remain significant after multivariate analysis in other studies (Table 1). It is important to stress that studies reported are heterogeneous with respect to design, outcome measures, and treatment regimens. Whether age is really an *independent* risk factor, and whether age is a stronger prognostic factor than stage, histology, and LOH 1p/16q, needs to be further explored. This may provide some insight into whether older patients need to be treated more intensively, as is already advised for adult WT patients.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

This work was supported by the Pediatric Oncology Center Society for Research (KOCR), Rotterdam, The Netherlands and by the DaDa foundation, Nieuwerkerk aan de Ijssel, The Netherlands.

References

- Breslow N, Olshan A, Beckwith JB, Green DM. Epidemiology of Wilms tumor. Med Pediatr Oncol. 1993;21(3):172–81.
- D'Angio GJ. Pre- or postoperative therapy for Wilms' tumor? J Clin Oncol. 2008;26(25):4055–7.

http://dx.doi.org/10.1200/JCO.2008.16.5316

3. Grundy PE, Breslow NE, Li S, Perlman E, Beckwith JB, Ritchey ML, et al. Loss of heterozygosity for chromosomes 1p and 16q is an adverse prognostic factor in favorable-histology Wilms tumor: a report from the National Wilms Tumor Study Group. J Clin Oncol. 2005;23(29):7312–21.

http://dx.doi.org/10.1200/JCO.2005.01.2799

 van den Heuvel-Eibrink MM, van Tinteren H, Bergeron C, Coulomb-L'Hermine A, de Camargo B, Leuschner I, et al. Outcome of localised blastemal-type Wilms tumour patients treated according to intensified treatment in the SIOP WT 2001 protocol, a report of the SIOP Renal Tumour Study Group (SIOP-RTSG). Eur J Cancer. 2015;51(4):498–506.

http://dx.doi.org/10.1016/j.ejca.2014.12.011

5. Beckwith JB, Palmer NF. Histopathology and prognosis of Wilms tumors: results from the First National Wilms' Tumor Study. Cancer. 1978;41(5):1937–48.

Relevance of age in nephroblastoma

- Green DM, Breslow NE, Beckwith JB, Takashima J, Kelalis P, D'Angio GJ. Treatment outcomes in patients less than 2 years of age with small, stage I, favorablehistology Wilms' tumors: a report from the National Wilms' Tumor Study. J Clin Oncol. 1993;11(1):91–5.
- 7. Pritchard-Jones K, Kelsey A, Vujanic G, Imeson J, Hutton C, Mitchell C. Older age is an adverse prognostic factor in stage I, favorable histology Wilms' tumor treated with vincristine monochemotherapy: a study by the United Kingdom Children's Cancer Study Group, Wilm's Tumor Working Group. J Clin Oncol. 2003;21(17):3269–75. http://dx.doi.org/10.1200/JCO.2003.01.062
- Reinhard H, Schmidt A, Furtwangler R, Leuschner I, Rube C, Von Schweinitz D, et al. Outcome of relapses of nephroblastoma in patients registered in the SIOP/ GPOH trials and studies. Oncol Rep. 2008;20(2):463–7.
- 9. Shamberger RC, Guthrie KA, Ritchey ML, Haase GM, Takashima J, Beckwith JB, et al. Surgery-related factors and local recurrence of Wilms tumor in National Wilms Tumor Study 4. Ann Surg. 1999;229(2):292–7.
- Irtan S, Jitlal M, Bate J, Powis M, Vujanic G, Kelsey A, et al. Risk factors for local recurrence in Wilms tumour and the potential influence of biopsy – the United Kingdom experience. Eur J Cancer. 2015;51(2):225–32. http://dx.doi.org/10.1016/j.ejca.2014.10.026
- Breslow N, Sharples K, Beckwith JB, Takashima J, Kelalis PP, Green DM, et al. Prognostic factors in nonmetastatic, favorable histology Wilms' tumor. Results of the Third National Wilms' Tumor Study. Cancer. 1991;68(11):2345–53.
- 12. Bonadio JF, Storer B, Norkool P, Farewell VT, Beckwith JB, D'Angio GJ. Anaplastic Wilms' tumor: clinical and pathologic studies. J Clin Oncol. 1985;3(4):513–20.
- 13. van den Heuvel-Eibrink MM, Grundy P, Graf N, Pritchard-Jones K, Bergeron C, Patte C, et al. Characteristics and survival of 750 children diagnosed with a renal tumor in the first seven months of life: a collaborative study by the SIOP/GPOH/SFOP, NWTSG, and UKCCSG Wilms tumor study groups. Pediatr Blood Cancer. 2008;50(6):1130–4. http://dx.doi.org/10.1002/pbc.21389
- 14. Mitchell C, Pritchard-Jones K, Shannon R, Hutton C, Stevens S, Machin D, et al. Immediate nephrectomy versus preoperative chemotherapy in the management of non-metastatic Wilms' tumour: results of a randomised trial (UKW3) by the UK Children's Cancer Study Group. Eur J Cancer. 2006;42(15):2554–62. http://dx.doi.org/10.1016/j.ejca.2006.05.026
- Popov SD, Sebire NJ, Pritchard-Jones K, Vujanic GM. Renal tumors in children aged 10–16 years: a report from the United Kingdom Children's Cancer and Leukaemia Group. Pediatr Dev Pathol. 2011;14(3):189–93. http://dx.doi.org/10.2350/10-06-0855-OA.1

Gooskens et al.

 Pession A, Dama E, Rondelli R, Magnani C, De Rosa M, Locatelli F, et al. Survival of children with cancer in Italy, 1989–98. A report from the hospital based registry of the Italian Association of Paediatric Haematology and Oncology (AIEOP). Eur J Cancer. 2008;44(9):1282–9.

http://dx.doi.org/10.1016/j.ejca.2008.03.020

- 17. Izawa JI, Al-Omar M, Winquist E, Stitt L, Rodrigues G, Steele S, et al. Prognostic variables in adult Wilms tumour. Can J Surg. 2008;51(4):252–6.
- Kalapurakal JA, Nan B, Norkool P, Coppes M, Perlman E, Beckwith B, et al. Treatment outcomes in adults with favorable histologic type Wilms tumor-an update from the National Wilms Tumor Study Group. Int J Radiat Oncol Biol Phys. 2004;60(5):1379–84.

http://dx.doi.org/10.1016/j.ijrobp.2004.05.057

- 19. Kattan J, Tournade MF, Culine S, Terrier-Lacombe MJ, Droz JP. Adult Wilms' tumour: review of 22 cases. Eur J Cancer. 1994;30A(12):1778–82.
- Mitry E, Ciccolallo L, Coleman MP, Gatta G, Pritchard-Jones K. Incidence of and survival from Wilms' tumour in adults in Europe: data from the EUROCARE study. Eur J Cancer. 2006;42(14):2363–8. http://dx.doi.org/10.1016/j.ejca.2006.04.009

Reinhard H. Aliani S. Rucha C. Stockla M. Louschn

- 21. Reinhard H, Aliani S, Ruebe C, Stockle M, Leuschner I, Graf N. Wilms' tumor in adults: results of the Society of Pediatric Oncology (SIOP) 93-01/Society for Pediatric Oncology and Hematology (GPOH) Study. J Clin Oncol. 2004;22(22):4500–6. http://dx.doi.org/10.1200/JCO.2004.12.099
- Terenziani M, Spreafico F, Collini P, Piva L, Perotti D, Podda M, et al. Adult Wilms' tumor: a monoinstitutional experience and a review of the literature. Cancer. 2004;101(2):289–93.

http://dx.doi.org/10.1002/cncr.20387

23. McLaughlin JK, Lipworth L, Tarone RE. Epidemiologic aspects of renal cell carcinoma. Semi Oncol. 2006;33(5):527–33.
 http://dlank.ic./10.1052/jimes.com/2006.00.010

http://dx.doi.org/10.1053/j.seminoncol.2006.06.010

- 24. Arrigo S, Beckwith JB, Sharples K, D'Angio G, Haase G. Better survival after combined modality care for adults with Wilms' tumor. A report from the National Wilms' Tumor Study. Cancer. 1990;66(5):827–30.
- 25. Byrd RL, Evans AE, D'Angio GJ. Adult Wilms tumor: effect of combined therapy on survival. J Urol. 1982;127(4):648–51.
- 26. Segers H, van den Heuvel-Eibrink MM, Pritchard-Jones K, Coppes MJ, Aitchison M, Bergeron C, et al. Management of adults with Wilms' tumor: recommendations based on international consensus. Expert Rev Anticancer Ther. 2011;11(7):1105–13. http://dx.doi.org/10.1586/era.11.76

Chapter 3

Histopathological and Molecular Characteristics of Wilms Tumor

Gulden Diniz

Izmir Tepecik Education and Research Hospital, Izmir, Turkey

Author for correspondence: Gulden Diniz, MD, PhD, Associate Prof., Pathologist, Microbiologist, Basic Oncologist, Izmir Tepecik Education and Research Hospital, Izmir, Turkey. Email: agdiniz@gmail.com

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch3

Copyright: The Author.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the author and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Diagnosis of malignant renal tumors does not mostly create difficulties. Although micrometastases may be encountered during postmortem examination, kidney is not a preferred organ for clinically detected metastases of malignant tumors. Therefore, almost all renal tumors in adults and children are primary tumors. When primary renal tumors are encountered, most of the cases pose a diagnostic simplicity. Indeed, diagnosis of malignant kidney tumors in children is Wilms tumor (WT) in 80–90% of the cases, while it is renal cell carcinoma in adults. In fact, a typical WT contains tissue components in three different morphologies. These are mesenchymal component resembling primitive fetal mesenchyme, epithelial component that reminds us fetal renal tubules and glomeruli, and blastomatous component consisting of clusters of blast

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia cells that contributed to the coinage of the term "nephroblastoma." However, not all WTs are triphasic, and different tissue components in very restricted areas may be overlooked. Besides, immunohistochemical staining methods helpful in the differential diagnosis of other tumors are not much useful in WT. Embryonic development of kidney is a complex process in which different transcription factors, proto-oncogenes, and various types of growth factors are effective. WT can be considered a failure of this transition. A number of genes are involved in nephrogenesis, as well as in Wilms tumorigenesis. Recently, some of these genes are believed to be regulated by *HACE1*, *glypican 3* (GPC3), and six *WT genes*. The incidence of WT is 1:10,000 worldwide. Currently, high cure rates can be achieved, and multimodality treatment has resulted in a significant improvement in outcomes. In this chapter, histopathological features of WT, genetic and molecular modifications related to WT, the effects of these genetic abnormalities on prognosis, and clues for differential diagnosis were evaluated.

Key words: Anaplastic; Blastemal type; Differential diagnosis; Favorable histology

Introduction

Diagnosis of malignant renal tumors does not mostly create difficulties. Although micrometastases may be encountered during autopsy, kidney is not a preferred organ for clinically detected metastases of malignant tumors. Therefore, almost all renal tumors in adults and children are primary tumors (1–3). When primary renal tumors are encountered, most of the cases pose a diagnostic simplicity. Indeed, diagnosis of malignant kidney tumors in children is Wilms tumor (WT) in 80–90% of the cases, while it is renal cell carcinoma in most adults. In fact, a typical WT contains tissue components in three different morphologies. These components are mesenchymal component resembling primitive fetal mesenchyme, epithelial component that reminds us fetal renal tubules and glomeruli, and blastomatous component consisting of clusters of blast cells that contributed to the coinage of the term "nephroblastoma." However, not all WTs are triphasic, and different tissue components in very restricted areas may be overlooked. Immunohistochemical staining methods helpful in the differential diagnosis of other tumors are not much use in WT, such as clear cell sarcoma or even renal cell carcinoma subtypes or other even more rare renal tumors (1, 3, 4).

Embryonic development of kidney is a complex process in which different transcription factors, proto-oncogenes, and various types of growth factors are effective. WT can be considered a failure of this transition. A number of genes are involved in nephrogenesis, as well as in Wilms tumorigenesis. Recently, some of these genes are believed to be regulated by *HACE1*, *glypican 3* (GPC3), and six *WT genes*. In addition, several studies have demonstrated that Cav-1 interacts with multiple members of the EGF-R/RAS/ERK and PI3/AKT pathways to modify signaling activity (5, 6).

Diniz

Characteristics of Wilms Tumor

The incidence of WT, the most common primary malignant renal tumor of childhood, is 1:10,000 worldwide. Currently, high cure rates can be achieved, and multimodality treatment has resulted in a significant improvement in outcomes. Recent studies have revealed that several genetic abnormalities are associated with a worse prognosis in WT, even in those with localized stage and favorable histology (7, 8). In this chapter, histopathological features of WT, genetic and molecular modifications related to WT, the effects of these genetic abnormalities on prognosis, and clues for differential diagnosis were analyzed.

Pathogenesis of WT

Kidney development is a complex process, consisting of two distinct embryological origins, the nephrogenic (mesenchymal) and the ductogenic (ureteric) (9). Both development pathways are regulated by transcription factors, proto-oncogenes, polypeptide growth factors that act as signaling molecules, and their receptors (10, 11). WT is the direct result of maldevelopment of the embryonic kidney and has led to many fundamental insights such as the link between normal development and tumorigenesis. Understanding the normal kidney development has helped in our understanding and treatment of WT. The metanephric kidney develops from the intermediate mesoderm, and this structure gives rise to three cell types that will form the kidney. In conclusion, this structure consists of the epithelial nephric or Wolffian duct, Six2-positive mesenchymal cells that will form the nephrons, and Foxd1positive cells that will give rise to the stromal cells (6). WT can be considered a failure of this transition. It arises from pluripotent renal precursors that undergo excessive proliferation resulting in undifferentiated stromal components, blastemal cells similar to the condensing mesenchyme, and primitive epithelial structures resembling comma and S-shaped bodies and glomeruli. The presence of associated nephrogenic rests consisting of foci of persistent embryonic remnant tissues that failed to mature to normal renal parenchyma further points toward impaired differentiation in early renal development (6, 9, 11-14). WT was one of the three types of cancer in which Knudson and Strong (15) based his two-hit model for tumor suppressor genes, and the loss of WT1 in a subset of WT cases remains an archetypal example of a classic tumor suppressor gene, as originally proposed (6). Since then, many variations in classifications and the genetics and mechanics of tumor suppressor genes have been found (16), and the biological basis of the multiple tumors that arise in genetically predisposed individuals may clearly involve genes other than WT1. A number of genes involved in nephrogenesis, especially in the mesenchymal to epithelial transition, have also been implicated in Wilms tumorigenesis (9, 17).

Common genetic abnormalities in WT

WT, or nephroblastoma which is currently the preferred term, is the most common pediatric renal cancer (6). The biology of WT illustrates some important aspects of childhood

Diniz

neoplasms such as the relationship between malformation and neoplasia, the histological similarities between the organogenesis and oncogenesis, and the two-hit theory of recessive tumor suppressor genes (7). WT morphologically resemble embryonic kidneys with a disrupted architecture, associated with undifferentiated metanephric precursors (6–8). Previous studies demonstrated that the risk of WT is increased in at least three groups of congenital malformations associated with distinct chromosomal loci. Although WT arising in these malformations accounts for no more than 10% of cases, these syndromic tumors have provided important insight into the biology in this neoplasm (7).

The first disorder that is associated with WT is WAGR syndrome, characterized by WT, aniridia, genital anomalies, and mental retardation. Lifetime risk of developing WT in these patients is approximately 33%. Patients with WAGR syndrome carry germ line deletion of chromosome 11p13, and the first identified WT-associate gene, WT1, is located on this chromosome. WT1 deletion in WAGR syndrome represents a "first hit"; the development of WT in these individuals frequently correlates with the occurrence of the mutation in the second WT1 allele as the "second hit" (6-9). The second disorder, Denys-Drash syndrome (DDS), is characterized by gonadal dysgenesis and early-onset nephropathy based on glomerulosclerosis leading to renal failure. Lifetime risk of WT in patients with DDS is approximately 90%. These patients demonstrated germ line point mutations in the zinc finger region of the WT1 protein that affects its DNA-binding properties (7). However, bi-allelic inactivation of WT1 must be required for the development of the WT phenotype in DDS (13-17). The third disorder, Beckwith-Wiedemann syndrome (BWS), is characterized by organomegaly, macroglossia, hemihypertrophy, and omphalocele. BWS has served as a model for tumorigenesis by genomic imprinting. Genetic locus of BWS or WT2 gene is on the 11p15.5. Unlike WAGR syndrome or DDS, the genetic basis for BWS is considerably more heterogeneous, in that no single genetic region is involved in all cases. Recent genetic studies have also elucidated the role of beta-catenin in WT. Beta-catenin belongs to the WNT (wingless) signaling pathway. Gain-of-function mutations have been demonstrated in 10% of sporadic WT. Similarly, mutations and deletions of WT1 gene are less common in sporadic WT cases (7, 8, 17, 18).

Histopathological features of WT

WT recapitulates normal nephrogenic differentiation, but while normal developing nephrons are beautifully structured, nephrogenic structures in WTs are disorganized (6). Most WTs show triphasic patterns such as blastemal, epithelial, and stromal (Figure 1). Clinical investigations reveal that the outcome of children with WT is dependent on histology. The cure rate in these cases is close to 90% (12). Favorable histology is characterized by the presence of all three histological elements and the absence of diffuse anaplasia (12, 14, 19–22). In

Characteristics of Wilms Tumor

WT cases that had been pretreated with chemotherapy before surgery, the blastemal type also has been well recognized now as an adverse prognostic subtype (23, 24). However, the histological features are not sufficient to predict the prognosis of WT, and some chromosomal mutations may play a role as adverse biological markers, even in those with localized (stage I and II) favorable histology WT (25–33).



Figure 1. (A) Gross pathology of WT, (B) entrapped two normal glomeruli in a WT, (C, D) typical triphasic WTs, (E) differences after therapy, (F) blastemal and stromal areas in a WT, and (G, H) anaplastic WT.

Diniz

The three histological components of WT have different proliferation potentials and different responses to therapy. In most reports, the lowest proliferation index was determined in the stromal component, and this component generally was not affected by chemotherapy. There were different results for the highest Ki-67 index in the literature. For example, the blastemal component had the highest proliferative activity in three studies, and the authors demonstrated that the surviving blastemal component after chemotherapy was a highly significant indicator of metastases and adverse outcome in WT (14, 30, 31). However, in two other studies, the highest Ki-67 index was determined in the epithelial component (22, 32). A fundamental difference in the behavior of normal versus tumor cells in culture is that normal cells divide for a limited number of times and exhibit cellular senescence, whereas tumor cells usually have the limitless proliferative capacity (14). The most prominent hypothesis is that the maintenance of telomere stability is required for the long-term proliferation of tumors. The tumor cells may escape from cellular senescence and become immortal by telomere maintenance. The simplest way of this maintenance is the activation of telomerase. Telomerase activity has been found in almost all tumors but not in adjacent normal cells (34-37). This activity was mainly evaluated with molecular studies, but it was also determined that the immunohistochemical staining pattern of TERT was correlated with telomerase activity (14, 34–38). The high telomerase reverse transcriptase (hTERT) staining was restricted to the nucleus in both normal telomerase-positive cells and cancer cells. The immunolocalization of hTERT in specimens of adult cancers revealed that the levels of telomerase activity mainly depended on the number of tumor cells with telomerase activity (14). Telomere maintenance is evident in virtually all types of malignant cells where either a telomerase-dependent or alternative lengthening of telomeres (ALT) mechanism exists. For this reason, effective strategies targeting telomere maintenance in cancer cells require telomerase inhibitors or ALT inhibitors (14, 34–38). The importance of telomerase activity is novel and potentially relevant in WT biology and progression because WT1 has been identified as a repressor of telomerase protein catalytic subunit promoter (36). In addition, functions of TERT other than telomere lengthening such as oncogenic transcriptional activation were reported (14). Although several genes such as HACE1, GPC3, and six WTs have been reported to involve in the pathogenesis of WT, they are not associated with specific histological features of WT (39-45).

Clues for differential diagnosis of WT

If a WT shows triphasic patterns, the diagnostic procedure is often not difficult. Wherein the case is of a monophasic pattern, differential diagnosis may be tiresome. In this condition, the main differential diagnosis of WT includes the so-called non-WT renal tumors,

Characteristics of Wilms Tumor

that is, clear cell sarcomas of kidney, congenital mesoblastic nephroma, renal malignant rhabdoid tumors, neuroblastoma, and primitive neuroectodermal tumors (PNETs). In a pure epithelial tumor, metanephric adenoma should be considered for differential diagnosis. Especially, positivity of WT1 in metanephric adenoma creates the diagnostic difficulty in this tumor (1, 8, 17). Pure stromal WT is also rare, and in those cases, differential diagnosis includes the congenital mesoblastic nephroma. The age of cases is helpful for differential diagnosis, as most cases with mesoblastic nephroma occur in children younger than 6 months. In addition, WT with purely blastemal appearance after chemotherapy can be too hard to differentiate from neuroblastomas and PNETs (23, 24). Immunohistochemical stains provide limited benefit in the differential diagnosis of WT subtypes. Immunohistochemical profile of the various components of WT mirrors that of their counterparts in the developing kidney. For example, the blastematous elements show focal positivity for vimentin, the epithelial elements react for keratin and epithelial membrane antigen (EMA), and the mesenchymal elements show a heterogeneous reactivity according to the morphological appearances. Immunoreactivity for WT1 antigen is determined in the 90% of WTs, and it is the most useful marker for differential diagnosis. By contrary, positive immunoreactivity for TTF-1 is determined in 17% of WTs, and it represents a potential source of misdiagnosis (6, 7, 17). However, IHC can be very helpful in the conformation of non-WT subtypes.

Conclusion

In conclusion, WT that demonstrates monophasic appearance can be too hard to discriminate from other primary renal tumors, such as neuroblastoma, clear cell sarcoma, rhabdoid tumor, mesoblastic nephroma, or even sarcomatoid-type renal cell carcinoma (1, 3, 4). Apart from histology, genetic risk factors may aid in stratifying patients for future treatment.

Conflict of Interests

The author declares no potential conflicts of interest with respect to research, authorship and/or publication of this article.

References

- Diniz G, Ortac R, Aktas S. Panoramic view of malignant renal tumors in Childhood. Turk J Pathology. 2001;17(1-2) 54-56 (Turkish).
- Peterson RO. Kidney. In: Urologic Pathology. 2nd ed. Philadelphia: Lippincott Company, 1992;50–150.

- 3. Breslow NE, Beckwith JB. Epidemiological features of Wilms' tumor: results of the National Wilms' Tumor Study. J Natl Cancer Inst. 1982;68(3):429–36.
- 4. Beckwith JB. Histopathological aspects of renal tumors in children. Prog Clin Biol Res. 1982;100:1–14.
- Chu A, Heck JE, Ribeiro KB, Brennan P, Boffetta P, Buffler P, et al. Wilms' tumour: a systematic review of risk factors and meta-analysis. Paediatr Perinat Epidemiol. 2010;24(5):449–69.

http://dx.doi.org/10.1111/j.1365-3016.2010.01133.x

- Hohenstein P, Pritchard-Jones K, Charlton J. The yin and yang of kidney development and Wilms' tumors. Genes Dev. 2015 Mar 1;29(5):467–82. http://dx.doi.org/10.1101/gad.256396.114
- Kumar V, Abbas AK, Aster JC. Robbins and Cotran's Pathologic Basis of Disease. Philadelphia: Elsevier, 2015;473–81.
- Ordonez NG, Rosai J. Urinary tract. In: Rosai and Ackerman's Surgical Pathology. 10th ed. Philadelphia: Elsevier, 2011;1120–286.
- 9. Horster MF, Braun GS, Huber SM. Embryonic renal epithelia: induction, nephrogenesis, and cell differentiation Physiol Rev. 1999;79:1157–91.
- 10. Quaggin SE, Kreidberg JA. Development of the renal glomerulus: good neighbors and good fences. Development. 2008;135(4):609–20.
- 11. Fischer EG, Carney JA, Anderson SR, Klatt EC, Lager DJ. An immunophenotypic comparison of metanephric metaplasia of Bowman capsular epithelium with metanephric adenoma, Wilms tumor, and renal development: a case report and review of the literature. Am J Clin Pathol. 2004;121(6):850–6.
- 12. Diniz G, Aktas S, Cubuk C, Ortac R, Vergin C, Olgun N. Tissue expression of MLH1, PMS2, MSH2, and MSH6 proteins and prognostic value of microsatellite instability in Wilms tumor: experience of 45 cases. Pediatr Hematol Oncol. 2013;30(4):273–84.
- 13. Costantini F, Kopan R. Patterning a complex organ: branching morphogenesis and nephron segmentation in kidney development. Dev Cell. 2010;18:698–712.
- Diniz G, Aktas S, Turedi A, Temir G, Ortac R, Vergin C. Telomerase reverse transcriptase catalytic subunit expression and proliferation index in Wilms tumor. Tumour Biol. 2011;32(4):761–7.

http://dx.doi.org/10.1007/s13277-011-0178-1

- 15. Knudson AG Jr, Strong LC. Mutation and cancer: a model for Wilms' tumor of the kidney. J Natl Cancer Inst. 1972;48:313–24.
- 16. Berger AH, Knudson AG, Pandolfi PP. A continuum model for tumour suppression. Nature. 2011;476:163–9.
- 17. Zhou M, Netto GJ, Epstein JI. Uropathology: High Yield Pathology. Elsevier, 2012;266–338.

Diniz

Characteristics of Wilms Tumor

- Ramburan A, Hadley GP, Govender D. Expression of E-cadherin, cadherin-11, alpha-, beta- and gamma-catenins in nephroblastomas: relationship with clinicopathological parameters, prognostic factors and outcome. Pathology. 2006;38(1):39–44.
- Husain AN, Pysher TJ, Dehner LP: The kidney and lower urinary tract. In: Stocker JT, Dehner LP (Eds). Pediatric Pathology. Philadelphia: Lippincott Williams & Wilkins Publications, 2001;834–903.
- 20. Perlman EJ. Renal tumors. In: Gilbert-Barness E (Ed). Potter's Pathology of the Fetus, Infant and Child. Philadelphia: Mosby Elsevier, 2007;1347–75.
- Argani P, Beckwith JB. Renal neoplasms of childhood. In: Sternberg's Diagnostic Surgical Pathology. Philadelphia: Lippincott Williams and Wilkins, 2004;p2001–33.
- 22. Metzger ML, Dome JS. Current therapy for Wilms' tumor. Oncologist. 2005;10(10):815–26.
- 23. Wegert J, Ishaque N, Vardapour R, Geörg C, Gu Z, Bieg M, et al. Mutations in the SIX1/2 pathway and the DROSHA/DGCR8 miRNA microprocessor complex underlie high-risk blastemal type Wilms tumors. Cancer Cell. 2015;27(2):298–311. http://dx.doi.org/10.1016/j.ccell.2015.01.002
- 24. van den Heuvel-Eibrink MM, van Tinteren H, Bergeron C, Coulomb-L'Hermine A, de Camargo B, Leuschner I, et al. Outcome of localised blastemal-type Wilms tumour patients treated according to intensified treatment in the SIOP WT 2001 protocol, a report of the SIOP Renal Tumour Study Group (SIOP-RTSG). Eur J Cancer. 2015;51(4):498–506.

http://dx.doi.org/10.1016/j.ejca.2014.12.011

- 25. Jurić I, Pogorelić Z, Kuzmić-Prusac I, Biočić M, Jakovljević G, Stepan J, et al. Expression and prognostic value of the Ki-67 in Wilms' tumor: experience with 48 cases. Pediatr Surg Int. 2010;26:487–93.
- Scott RH, Walker L, Olsen ØE, Levitt G, Kenney I, Maher E, et al. Surveillance for Wilms tumour in at-risk children: pragmatic recommendations for best practice. Arch Dis Child. 2006;91(12):995–9.
- Terenziani M, Spreafico F, Collini P, Piva L, Perotti D, Podda M, et al. Adult Wilms' tumor: a monoinstitutional experience and a review of the literature. Cancer. 2004;101(2):289–93.
- Scott RH, Stiller CA, Walker L, Rahman N. Syndromes and constitutional chromosomal abnormalities associated with Wilms tumour. J Med Genet. 2006;43(9):705–15.
- 29. Hohenstein P, Hastie ND. The many facets of the Wilms' tumour gene, WT1. Hum Mol Genet. 2006;15 Spec No 2:R196–201.
- 30. Maw MA, Grundy PE, Millow LJ, Eccles MR, Dunn RS, Smith PJ, et al. A third Wilms' tumor locus on chromosome 16q. Cancer Res. 1992;52(11):3094–8.

- 31. Delahunt B, Farrant GJ, Bethwaite PB, Nacey JN, Lewis ME. Assessment of proliferative activity in Wilms' tumour. Anal Cell Pathol. 1994;7(2):127–38.
- Dome JS, Bockhold CA, Li SM, Baker SD, Green DM, Perlman EJ, et al. High telomerase RNA expression level is an adverse prognostic factor for favorable-histology Wilms' tumor. J Clin Oncol. 2005;23(36):9138–45.
- 33. Berrebi D, Leclerc J, Schleiermacher G, Zaccaria I, Boccon-Gibod L, Fabre M, et al. High cyclin E staining index in blastemal, stromal or epithelial cells is correlated with tumor. aggressiveness in patients with nephroblastoma. PLoS One. 2008;3(5):e2216.
- 34. Khine MM, Aung W, Sibbons PD, Howard CV, Clapham E, McGill F, et al. Analysis of relative proliferation rates of Wilms' tumor components using proliferating cell nuclear antigen and MIB-1 (Ki-67 equivalent antigen) immunostaining and assessment of mitotic index. Lab Invest. 1994;70(1):125–9.
- Green DM, D'Anjio GJ, Beckwith JB. Wilms' tumor. In: Pizzo PA, Fobcek DG (eds). Principles and Practice of Pediatric Oncology. 2nd ed. Philadelphia: JB Lippincott Company, 1997;733–59.
- Oh S, Song Y, Yim J, Kim TK. The Wilms' tumor 1 tumor suppressor gene represses transcription of the human telomerase reverse transcriptase gene. J Biol Chem. 1999;274(52):37473–8.
- Yashima K, Maitra A, Timmons CF, Rogers BB, Pinar H, Shay JW, et al. Expression of the RNA component of telomerase in Wilms tumor and nephrogenic rest recapitulates renal embryogenesis. Hum Pathol. 1998;29(5):536–42.
- 38. Dome JS, Chung S, Bergemann T, Umbricht CB, Saji M, Carey LA, et al. High telomerase reverse transcriptase (hTERT) messenger RNA level correlates with tumor recurrence in patients with favorable histology Wilms' tumor. Cancer Res. 1999;59(17):4301–7.
- 39. Rahman N, Arbour L, Tonin P, Renshaw J, Pelletier J, Baruchel S, et al. Evidence for a familial Wilms' tumour gene (FWT1) on chromosome 17q12-q21. Nat Genet. 1996;13(4):461–3.
- Miozzo M, Perotti D, Minoletti F, Mondini P, Pilotti S, Luksch R, et al. Mapping of a putative tumor suppressor locus to proximal 7p in Wilms tumors. Genomics. 1996;37(3):310–5.
- 41. White GR, Kelsey AM, Varley JM, Birch JM. Somatic glypican 3 (GPC3) mutations in Wilms' tumour. Br J Cancer. 2002;86(12):1920–2.
- 42. Rivera MN, Kim WJ, Wells J, Driscoll DR, Brannigan BW, Han M, et al. An X chromosome gene, WTX, is commonly inactivated in Wilms tumor. Science. 2007;315(5812):642–5.
- Slade I, Stephens P, Douglas J, Barker K, Stebbings L, Abbaszadeh F, et al. Constitutional translocation breakpoint mapping by genome-wide paired-end sequencing identifies HACE1 as a putative Wilms tumour susceptibility gene. J Med Genet. 2010;47(5):342–7.

Diniz

Characteristics of Wilms Tumor

- 44. Ramachandran C, Melnick SJ, Escalon E, Jhabvala P, Khatib Z, Alamo A, et al. Expression of apoptosis, cell proliferation, and drug resistance genes in pediatric Wilms' tumors. Anticancer Res. 2000;20(5C):3759–65.
- 45. Wittmann S, Wunder C, Zirn B, Furtwängler R, Wegert J, Graf N, et al. New prognostic markers revealed by evaluation of genes correlated with clinical parameters in Wilms tumors. Genes Chromosomes Cancer. 2008;47(5):386–95.

Chapter 4

Wilms Tumor and Its Management in a Surgical Aspect

Basak Erginel

Department of Pediatric Surgery, Istanbul University Istanbul Medical Faculty, Istanbul, Turkey

Author for correspondence: Basak Erginel, MD, Department of Pediatric Surgery, Istanbul University Istanbul Medical Faculty, Millet Caddesi, Capa, 34093 Fatih, Istanbul, Turkey. Email: basakerginel@hotmail.com

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch4

Copyright: The Author.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt

(remix, transform, and build upon the material for any purpose, even commercially), as long as the author and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Nephroblastoma [Wilms tumor (WT)] is a rare, but the most common, primary renal tumor in children. WT is usually diagnosed between the ages of 1 and 5, with the most common diagnosis at the age of 3. While imaging (ultrasound, computed tomography, and magnetic resonance) can accurately predict up to 95% of WTs, they cannot predict the histologic subtypes and require tissue examination. Surgery is one of the cornerstones of WT treatment. Other aspects of management include chemotherapy and radiation therapy. The Societe Internationale D'oncologie Pediatrique (SIOP) advocates primary chemotherapy in patients less than 6 months of age, whereas the

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Erginel

Children's Oncology Group (COG) recommends primary surgery in all cases except those considered not resectable by the surgeon. In this chapter, the surgical therapy of WT is reviewed.

Key words: Nephroblastoma; Pediatric oncology; Surgery for Wilms tumor

Introduction

Wilms tumor (WT), also called nephroblastoma, was first described by Thomas F. Rance in 1814 (1, 2). In 1899, Carl Max Wilhelm Wilms, a German surgeon and pathologist, gave a detailed histological description and since then the tumor bears his name (1, 3). The incidence of WT is 1:10000 in children under 15 (4). WT is the most common renal malignancy in children, and it represents 6% of all childhood cancers. It is also the second most common intra-abdominal cancer, and it is an embryonal malignancy of the kidney (5). About 75% of children are diagnosed before the age of 5, and the median age is 3.5 years (6).

WT is primarily a sporadic disease, but family history exists in 1–2% of cases (7). There are a number of syndromes associated with WT, including WAGR (WT, aniridia, genitourinary anomalies, and mental retardation), Denys-Drash syndrome (progressive renal disease, male pseudohermaphroditism, and WT), and Beckwith-Wiedemann syndrome (8). While WAGR syndrome is associated with large deletion on the WT1 chromosome located on 11p13, Denys-Drash syndrome has point mutations on WT1 (9, 10). Beckwith-Wiedemann syndrome is a syndrome characterized by macrosomia, macroglossia, omphalocele, and growth retardation, and hemihypertrophy is caused by a mutation located on 11p15, also known as the WT2 gene locus (11). Normal kidneys complete differentiation at the end of the 36th week of gestation; however, in about 1-2% of newborns, nephrogenic blastemal cells, also called nephrogenic rests, persist. WT is thought to arise from these nephrogenic rests (12). These nephrogenic rests have been detected in almost 35% of unilateral and 100% of bilateral WT patients (13). The prognosis of WT is closely related to its histology. An unfavorable prognosis occurs when the tumor consists of anaplastic cells; the prognosis is favorable otherwise. This anaplastic type represents 11.5% of all WTs and is responsible for 52% of the mortality rate (14). However, for pretreated patients, there exist three subgroups: low-, intermediate-, or high-risk tumor, according to the Stockholm working classification of renal tumors used by the Societe Internationale D'oncologie Pediatrique (SIOP) (15). WT is associated with mutations in various tumor suppressor genes (16). In the anaplastic type of WTs, p53 mutation is frequent (17).

Wilms tumor - surgical treatment

Clinical presentation

WT is usually unilateral, and the mean age of presentation is 3.3 years. In 4 to 7% of cases, they are synchronous bilateral and appear at a younger age (mean age of 2.6 years) (18). The most common clinical presentation of WT is an abdominal mass. Over 90% of children are referred with an abdominal mass. Other common symptoms are abdominal pain, macroscopic hematuria, fever, and hypertension (19). Abdominal pain can be a symptom of a rupture or intratumoral hemorrhage. Macroscopic hematuria may occur when the tumor has extended to the collecting system (20). Nonspecific symptoms, such as microscopic hematuria, urinary disturbances, malaise, weight loss, and anemia, may be present at initial presentation. If spermatic veins are occluded, varicocele can occur (3) and a thorough examination of the abdomen is necessary. Varicocele is always an alarming symptom, but even more if it occurs on the right side [symptom of inferior vena cava (IVC) thrombosis] or occlusion of the right spermatic vein by the lower part of the tumor (right spermatic vein goes directly to IVC). During the physical examination, a firm, nontender mass in the abdomen is usually identified. Due to elevated renin levels, a follow-up of the blood pressure is very important in WT patients. The most common intravascular tumor extension sites are the renal vein, IVC, and atrium (21). Although the lung is the most common metastatic site, respiratory symptoms are not common. In summary, during examination, the associated anomalies should be considered, and the patient should be examined for aniridia, hemihypertrophy, and genitourinary anomalies.

Diagnosis

Ultrasonography (US) is the primary diagnostic tool for children suspected of having WT. US allows for measuring the tumor size, identifying its origin, establishing the relationship between vena cava and aorta, as well as possible IVC or renal vein. The second important diagnostic tool is computerized tomography (CT). CT is recommended for WT along with US (Figure 1). During interpretation of the CT, and also the US, the contralateral kidney and the liver should always be carefully examined. Magnetic resonance imaging (MRI) is also helpful in detecting the vascular involvement. MRI, although requires longer general anesthesia in the preschool age, offers the most accurate imaging of the kidney. It serves best when nephron-sparing resection is considered or in case of the need to distinguish between WT and nephroblastomatosis. A plain chest X-ray is a routine procedure for the evaluation of pulmonary metastases. A routine pulmonary CT is still controversial (22); however, many physicians prefer the thoracic CT because of its high sensitivity. The definite diagnoses of WT and its subtypes are made by histological evaluation. The preoperative laboratory tests that should be performed are the total blood count, renal and liver function tests, calcium level examination, and urinary examination. In rhabdoid types, the serum calcium increases (23).



Figure 1. A CT scan of a left Wilms tumor.

In 8% of children with WTs, von Willebrand disease is a comorbidity (24). The routinely evaluated tumor markers include neuron-specific enolase, lactate dehydrogenase, alpha-fetoprotein, β -human chorionic gonadotropin, and ferritin. High-risk patients for the growth of WTs, such as patients with Beckwith-Wiedemann syndrome, should be routinely examined with an US (25). In addition, a 24-hour urine catecholamine test is essential to avoid misdiagnosing with neuroblastoma.

Staging

There are currently two major staging systems, the National Wilms Tumor Study (NWTS) and the SIOP, as summarized in Tables 1 and 2, respectively (26).

Surgical management

Surgery is the cornerstone for the treatment of WT. The Children's Oncology Group (COG) from North America, a group that conducted the NWTS trials, recommends surgery before chemotherapy, whereas SIOP in Europe suggests preoperative chemotherapy (26). As the SIOP group, the National Wilms Tumor Study Group (NWTSG) has concerns about performing a biopsy first because of the risk of tumor upstaging (27). The SIOP recommends preoperative chemotherapy to decrease the risk of intraoperative rupture, downstage the tumor, and to reduce the need for irradiation. The advantage of preoperative chemotherapy is the identification of chemoresistant high-risk blastemal predominant subtype that benefits from treatment intensification.

Erginel

Stages	Description
Stage 1	(a) Tumor is limited to the kidney and completely excised
	(b) The tumor is not ruptured before or during removal
	(c) The vessels of the renal sinus are not involved beyond 2 mm
	(d) There is no residual tumor apparent beyond the margins of excision
Stage 2	(a) Tumor extends beyond the kidney but is completely excised
	(b) No residual tumor is apparent at or beyond the margins of excision
	(c) Tumor thrombus in vessels outside the kidney is stage II if the thrombus is removed en bloc with the tumor
Stage 3	Residual nonhematogenous tumor is present and confined to abdomen
	(a) Lymph nodes in the renal hilum, the periaortic chains, or beyond are found to contain a tumor
	(b) Diffuse peritoneal contamination by the tumor
	(c) Implants are found on the peritoneal surfaces
	(d) Tumor extends beyond the surgical margins either microscopically or grossly
	(e) Tumor is not completely resectable because of local infiltration into vital struc- tures
Stage 4	Hematogenous metastasis or lymph node metastasis
Stage 5	Bilateral renal involvement

Table 1. The NWTS staging system*

*This system relies on surgical and pathological evaluation of a tumor in patients not submitted to the preoperative chemotherapy (4).

The management of WT requires a multidisciplinary approach with a pediatric radiologist, an oncologist, a surgeon, and a radiotherapist. First, the patient must be carefully evaluated using appropriate imaging techniques in the preoperative period to identify the origin of the tumor, its position in relation to the adjacent tissue, and vascular involvement. The bilaterality of the tumor, and presence of nephrogenic rests, must be evaluated. If the imaging is not suggestive of any bilateral lesion, there is no need to explore the contralateral kidney at surgery (28, 29). Among the results of WTSG-4, WT is detected in 7% of the patients whose preoperative abdominal CT of contralateral kidney is normal (30). An abdominal Doppler US is advised to check for possible thrombus in the renal vein and the IVC. In tumors with a renal vein and caval extension, it is advised to delay surgery and to start with chemotherapy (31). If the thrombus extends to the thoracic vena cava, an echocardiography should be performed. In case of the thrombus extending to the heart, the risk of pulmonary thrombosis produced by fragmented floating atrial/ventricular thrombus versus benefits of chemotherapy-induced regression must be carefully balanced. Some of these patients are cardiosurgical emergencies

Erginel

Stages	Description		
Stage 1	(a) Tumor is limited to kidney and is completely resected (resection margins "clear")		
	(b) The tumor may be protruding into the pelvic system and "dipping" into the ureter (but it is not infiltrating their walls)		
	(c) The vessels of the renal sinus are not involved		
	(d) Intrarenal vessel involvement may be present		
Stage 2	 (a) The tumor extends beyond kidney or penetrates through the renal capsule and/or fibrous pseudocapsule into perirenal fat but is completely resected (resection margins "clear") 		
	(b) The tumor infiltrates the renal sinus and/or invades blood and lymphatic vessels outside the renal parenchyma but is completely resected		
	(c) The tumor infiltrates adjacent organs or vena cava but is completely resected		
Stage 3	(a) Incomplete excision of the tumor, which extends beyond the resection margins		
	(b) Any abdominal lymph nodes are involved		
	(c) Tumor rupture before or intraoperatively (regardless of other criteria for staging)		
	(d) The tumor has penetrated through the peritoneal surface		
	(e) Tumor thrombi present at resection margins of vessels or ureter, transected or removed piecemeal by surgeon		
	(f) The tumor has been surgically biopsied (wedge biopsy) prior to preoperative		
	chemotherapy or surgery		
Stage 4	Hematogenous metastases (lung, liver, bone, brain, etc.) or lymph node metasta- ses outside the abdominopelvic region		
Stage 5	Bilateral renal tumors at diagnosis		

Table 2. The SIOP staging system*

*This system relies on findings at postchemotherapy tumor nephrectomy and the microscopical examination of the whole sample.

and therefore an oncology-oriented pediatric surgeon and a cardiac surgeon are necessary, and the equipment must include the cardiopulmonary bypass device.

For unilateral WTs, a transperitoneal radical nephrectomy is the standard operation. Nephron-sparing surgery (NSS) is advocated only in selected cases of patients with solitary kidney or bilateral WTs (20).
Wilms tumor - surgical treatment

Before the incision, rolled sterile pads should be placed under the patient in supine position. A nasogastric tube, a transient catheter, an arterial line, and a Foley catheter are placed in our center (Figure 2). A large transverse supraumbilical incision should be made, permitting the exploration of the entire abdomen and the contralateral kidney, if necessary (Figure 3). In patients with an intracardiac extension, where an upfront excision is not encouraged, a chevron incision or a combined transverse abdominal and sternal incision can be performed. Because the rupture of the tumor capsule and the spillage of the tumor increase with the tumor stage, and decrease survival, it is important to perform a suitably long incision to remove the tumor from the abdomen safely. At the beginning of the operation, the abdomen should be explored for any intra-abdominal implants. If deemed necessary based on preoperative imaging, the opening of Gerota fascia and careful exploration of the whole contra lateral kidney is recommended. Any suspicious areas should be biopsied. The peritoneum covering the tumor is opened as laterally as possible for easy closure of the peritoneum after the tumor resection (Figures 4 and 5). Most frequently, the best access to the right-sided WT is Kocher' maneuver, which offers a good exposition of the right renal vessels. For left-sided WTs, laterocolonal access is usually sufficient (Figure 6). The renal vein and the IVC should be palpated first to exclude thrombus. The nephrectomy should begin by ligating the renal artery to avoid thrombus embolism. Afterward, the renal vein should be ligated immediately (Figure 7). The ureter should be ligated as low as possible. Titanium clips are mostly used for the marking of the tumor area for possible further radiotherapy. An en bloc resection of the tumor without any tumor spillage is the most important aspect. Lymph node sampling is another important goal in WT surgery. Lymph node samples should be collected from the renal hilum, iliac, paracaval, or para-aortic areas for accurate staging. Spillage of the tumor or inadequate lymph node dissection results



Figure 2. The position of the patient.



Figure 3. A large transverse supraumbilical incision, permitting the exploration of the entire abdomen and the contralateral kidney.

in incomplete chemotherapy and thus decreased survival rates. NWTS suggests seven regional nodes to get reliable information on lymph node involvement (32).

There is an emerging consensus on performing routine adrenalectomy during unilateral nephrectomy for WTs. One recent study proved that an adrenalectomy does not affect the 5-year survival rate, but it does increase the intraoperative spillage rates (33). Most surgeons leave the adrenal gland intact if it is not infiltrated by the tumor (34). In the study of van Waas et al. (35), it is stated that the removal of one adrenal gland does not result in clinical adrenal insufficiency.

Performing NSS by partial nephrectomy or enucleation in unilateral cases is also a matter of debate (36). We recommend this approach only in synchronous or metachronous bilateral cases or in solitary kidneys. Only less than 5% of all unilateral WTs are eligible for NSS because most of the tumors are locally advanced at the time of diagnosis (37). The surgical criteria for a partial resection are as follows: tumor is located in one pole and infiltrates approximately less than 1/3 of the kidney; no invasion of the renal vein; and the surgeon's experience in pediatric oncology (38). The SIOP WT 2001 trial reported 91 children (3%) with excellent survival rates in which NSS was performed (39). Minimally invasive nephrectomy can offer the same outcome as the classical laparotomic approach (40, 41).

For bilateral cases, the current approach is preoperative chemotherapy followed by bilateral NSS (42). Timing of surgery is important. Both COG/NWTSG and SIOP/RTSG recommend surgery after 9–12 weeks of chemotherapy. Where possible, both sides can be operated in the same session. In difficult cases, the easier side can be operated first. The more difficult

Erginel

Wilms tumor - surgical treatment



Figure 4. The peritoneal reflection covering a right Wilms tumor. P: peritoneum.

side can be operated after one or two courses of chemotherapy and stabilization of the renal function. The major goal of bilateral WT surgery is to achieve a cure by removing all tumoral renal tissue while preserving the maximum functioning kidney. Some authors suggest *in situ* topical cooling or perfusion with preservation solutions for meticulous dissection without the loss of renal function from ischemia. The ultrasonic scalpel is a useful tool in such resections (43). Millar et al. (44) reported 19 children with bilateral WT cases in which they achieved good results with appropriate chemotherapy and conservative NSS. They suggested a revision surgery if needed. Perioperative US is also a useful tool in NSS of bilateral cases for detecting the margins of normal renal tissue and the tumoral tissue.

Intracardiac extension of WT has been an important surgical challenge. First, an accurate preoperative radiological evaluation of the tumor and thrombus is necessary, and second, a multidisciplinary treatment plan by the cardiovascular surgeon and the pediatric surgeon is important. In general, upfront treatment with chemotherapy is advised. Then, surgical treatment can be considered as reported by SIOP group, which shows favorable results in patients who underwent surgery, including a cardiopulmonary bypass and hypothermia (45). Some surgeons have used this reported technique in pediatric WT cases (46–48).

Common surgical complications in primary nephrectomy patients with WT are ruptures, intestinal obstruction, bleeding, and surgical site infections. NWTSG conducted an



Figure 5. The lateral peritoneal reflection opened from an avascular area on the lateral side. WT: Wilms tumor, P: peritoneum, C: colon.

analysis on 3335 children who received primary nephrectomy for WT. They observed surgical complications in 12.7% of the patients. These complications were intestinal obstruction (5.1%), excessive bleeding (1.9%), surgical site infection (1.9%), and major vascular injuries (1.5%). Also, the risk of complications increased when the tumor size had exceeded 10 cm (49). The risk of tumor rupture was 15% in NWTSG for primary surgery and 3% for chemotherapy-pretreated cases in SIOP. This was replicated by a randomized study in UK (50). Godzinski (51) reported that not only the intraoperative tumor rupture but also other surgery-related complications became rare after the pretreatment. The rate of these complications did not exceed 8% in the pretreated patients. In selected cases, preoperative biopsy was considered. Tru-cut biopsy instead of a needle biopsy may be used to avoid upstaging of the tumor. Preoperative chemotherapy is advised in most bilateral cases, inoperable tumors, and tumors with an intracaval/cardiac extension (43).

Recurrent disease

About 10–15% of WT results in recurrence (52). Spillage occurs in almost every 10 unilateral nephrectomies and is correlated with right-side and larger tumors (53). The recent analysis of COG demonstrated that a relapse in the flank or abdominal site occurred in only 7.4% and 9.5%, respectively, of stage II WT patients with spill, whereas it occurred in 2.5% and 3%, respectively, for those without spill (54). According to SIOP trials, even though

Erginel

Wilms tumor - surgical treatment



Figure 6. Image showing access to the right-sided WT in which the colon is reflected medially. C: Colon, I: ileum, GB: gall bladder, RK: right kidney, LK: left kidney, RRV: right renal vein, LRV: left renal vein, VCI: vena cava inferior, A: aorta.

the rupture frequency was much less in prechemotherapy group, event-free follow-up or overall survival rate was reported similarly (55). Drugs such as platinum compounds, ifos-famide, cyclophosphamide, etoposide (ICE), and their combinations are used in relapsed WT. Postrelapse survival rates of 50–60% have been reported with ICE chemotherapy (56). Survival rates also depend on the initial stage, initial treatment, metastatic burden, and the relapse-free interval (57). A complete resection of the recurrent lesion(s) has also been shown to be a favorable prognostic factor.

Conclusion

Whether chemotherapy is given preoperatively or not, surgery comprises the main part of the WT surgery. WT patients need a multimodal, multidisciplinary treatment with a close follow-up.



Figure 7. The ligation of the renal vein.

Conflict of Interest

The author declares no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

I convey my sincere gratitude to Prof. Dr. J. Godzinski for the revision of the chapter.

References

- 1. Varan A. Wilms' tumor in children: an overview. Nephron Clin Pract. 2008;108:c83–90.
- 2. Rance TF. Case of fungus haematodes of the kidneys. Med Phys J. 1814;32:19–25.
- 3. Wilms M. Die Mischgeschwülste der Niere. Leipzig: A Georgi, 1899;1-90.
- Ehrlich PF, Shamberger RC. Wilms' tumor. In: Coran A, Adzic NS, Krummel TM, Laberge JM, Shamberger RC, Caldamone AA. Pediatric Surgery. Vol. 1. 7th ed. Philadelphia, PA: Elsevier Saunders, 2012;423–40.
- Adam MP, Fechner PY, Ramsdell LA, Badaru A, Grady RE, Pagon RA, et al. Ambiguous genitalia: what prenatal genetic testing is practical? Am J Med Genet A. 2012;158:1337–43.
 - http://dx.doi.org/10.1002/ajmg.a.35338
- Breslow NE, Beckwith JB, Perlman EJ, Reeve AE. Age distributions, birth weights, nephrogenic rests, and heterogeneity in the pathogenesis of Wilms tumor. Pediatr Blood Cancer. 2006;47(3):260–7. http://dx.doi.org/10.1002/pbc.20891

Erginel

 Perlman EJ, Grundy PE, Anderson JR, Jennings LJ, Green DM, Dome JS, et al. WT1 mutation and 11P15 loss of heterozygosity predict relapse in very low-risk Wilms' tumors treated with surgery alone: a children's oncology group study. J Clin Oncol. 2011;29(6):698–703.

http://dx.doi.org/10.1200/JCO.2010.31.5192

 Scott RH, Stiller CA, Walker L, Rahman N. Syndromes and constitutional chromosomal abnormalities associated with Wilm's tumour. J Med Genet. 2006 Sep;43(9):705–15.

http://dx.doi.org/10.1136/jmg.2006.041723

- 9. Pelletier J, Bruening W, Kashtan CE, Mauer SM, Manivel JC, Striegel JE, et al. Germline mutations in the Wilms' tumor suppressor gene are associated with abnormal urogenital development in Denys-Drash syndrome. Cell. 1991;67(2):437–47.
- Pelletier J, Bruening W, Li FP, Haber DA, Glaser T, Housman DE. WT1 mutations contribute to abnormal genital system development and hereditary Wilms' tumour. Nature. 1991 Oct 3;353(6343):431–4. http://dx.doi.org/10.1038/353431a0
- 11. Koufos A, Grundy P, Morgan K, Aleck KA, Hadro T, Lampkin BC, et al. Familial Wiedemann-Beckwith syndrome and a second Wilms' tumor locus both map to 11p15.5. Am J Hum Genet. 1989 May;44(5):711–9.
- 12. Hammer E, Ernst FD, Thiele A, Karanam NK, Kujath C, Evert M, et al. Kidney protein profiling of Wilms' tumor patients by analysis of formalin-fixed paraffin-embedded tissue samples. Clin Chim Acta. 2014 Jun 10;433:235–41. http://dx.doi.org/10.1016/j.cca.2014.03.020
- Beckwith JB. Precursor lesions of Wilms' tumor: clinical and biological implications. Med Pediatr Oncol. 1993;21(3):158–68.
- 14. Beckwith JB, Palmer NF. Histopathology and prognosis of Wilms' tumors: results from the First National Wilms' Tumor Study. Cancer. 1978 May;41(5):1937–48.
- Vujanić GM, Sandstedt B. The pathology of Wilms' tumour (nephroblastoma): the International Society of Paediatric Oncology approach. J Clin Pathol. 2010;63:102–9.

http://dx.doi.org/10.1136/jcp.2009.064600

- Coppes MJ, Haber DA, Grundy PE. Genetic events in the development of Wilms' tumor. N Engl J Med. 1994 Sep 1;331(9):586–90. http://dx.doi.org/10.1056/NEJM199409013310906
- 17. Bardeesy N, Falkoff D, Petruzzi MJ, Nowak N, Zabel B, Adam M, et al. Anaplastic Wilms' tumour, a subtype displaying poor prognosis, harbours p53 gene mutations. Nat Genet. 1994 May;7(1):91–7. http://dx.doi.org/10.1038/ng0594-91

Erginel

- Neville HL, Ritchey ML. Wilms' tumor. Overview of National Wilms' Tumor Study Group results. Urol Clin North Am. 2000 Aug;27(3):435–42.
- Ganguly A, Gribble J, Tune B, Kempson RL, Luetscher JA. Renin-secreting Wilms' tumor with severe hypertension. Report of a case and brief review of renin-secreting tumors. Ann Intern Med. 1973 Dec;79(6):835–7.
- 20. Ko EY, Ritchey ML. Current management of Wilms' tumor in children. J Pediatr Urol. 2009 Feb;5(1):56–65.

http://dx.doi.org/10.1016/j.jpurol.2008.08.007

- 21. Aspiazu D, Fernandez-Pineda I, Cabello R, Ramirez G, Alvarez-Madrid A, De Agustin JC. Surgical management of Wilms' tumor with intravascular extension: a single-institution experience. Pediatr Hematol Oncol. 2012 Feb;29(1):50–4. http://dx.doi.org/10.3109/08880018.2011.642941
- 22. Owens CM, Veys PA, Pritchard J, Levitt G, Imeson J, Dicks-Mireaux C. Role of chest computed tomography at diagnosis in the management of Wilms' tumor: a study by the United Kingdom Children's Cancer Study Group. J Clin Oncol. 2002 Jun 15;20(12):2768–73.
- Green DM, D'Angio GJ, Beckwith JB, Breslow NE, Grundy PE, Ritchey ML, et al. Wilms tumor. CA Cancer J Clin. 1996 Jan-Feb;46(1):46–63. Review.
- Coppes MJ, Zandvoort SW, Sparling CR, Poon AO, Weitzman S, Blanchette VS. Acquired von Willebrand disease in Wilms' tumor patients. J Clin Oncol. 1992 Mar;10(3):422–7.
- Choyke PL, Siegel MJ, Craft AW, Green DM, DeBaun MR. Screening for Wilms' tumor in children with Beckwith-Wiedemann syndrome or idiopathic hemihypertrophy. Med Pediatr Oncol. 1999 Mar;32(3):196–200.
- 26. Metzger ML, Dome JS. Current therapy for Wilms' tumor. Oncologist. 2005 Nov-Dec;10(10):815–26.

http://dx.doi.org/10.1634/theoncologist.10-10-815

- Vujanić GM, Sandstedt B, Harms D, Kelsey A, Leuschner I, de Kraker J; SIOP Nephroblastoma Scientific Committee. Revised International Society of Paediatric Oncology (SIOP) working classification of renal tumors of childhood. Med Pediatr Oncol. 2002 Feb;38(2):79–82.
- Green DM. The evolution of treatment for Wilms' tumor. J Pediatr Surg. 2013 Jan;48(1):14–9.

http://dx.doi.org/10.1016/j.jpedsurg.2012.10.012

29. Powis M, Messahel B, Hobson R, Gornall P, Walker J, Pritchard-Jones K. Surgical complications after immediate nephrectomy versus preoperative chemotherapy in non-metastatic Wilms' tumour: findings from the 1991–2001 United Kingdom Children's Cancer Study Group UKW3 Trial. J Pediatr Surg. 2013 Nov;48(11):2181–6. http://dx.doi.org/10.1016/j.jpedsurg.2013.07.001

- 30. Ritchey ML, Green DM, Breslow NB, Moksness J, Norkool P. Accuracy of current imaging modalities in the diagnosis of synchronous bilateral Wilms' tumor. A report from the National Wilms' Tumor Study Group. Cancer. 1995 Jan 15;75(2):600–4.
- 31. Nakayama DK, Norkool P, deLorimier AA, O'Neill JA Jr, D'Angio GJ. Intracardiac extension of Wilms' tumor. A report of the National Wilms' Tumor Study. Ann Surg. 1986;204(6):693–7.
- 32. Kieran K, Anderson JR, Dome JS, Ehrlich PF, Ritchey ML, Shamberger RC, et al. Lymph node involvement in Wilms tumor: results from National Wilms Tumor Studies 4 and 5. J Pediatr Surg. 2012;47:700–6.

http://dx.doi.org/10.1016/j.jpedsurg.2011.08.017

- 33. Kieran K, Anderson JR, Dome JS, Ehrlich PF, Ritchey ML, Shamberger RC, et al. Is adrenalectomy necessary during unilateral nephrectomy for Wilms' tumor? A report from the Children's Oncology Group. J Pediatr Surg. 2013 Jul;48(7):1598-603. http://dx.doi.org/10.1016/j.jpedsurg.2013.04.019
- 34. Yao W, Li K, Xiao X, Gao J, Dong K, Xiao X, et al. Outcomes of Wilms' tumor in eastern China: 10 years of experience at a single center. J Invest Surg. 2012 Jun;25(3):181–5. http://dx.doi.org/10.3109/08941939.2011.615893
- 35. van Waas M, Neggers SJ, van Eck JP, van Noesel MM, van der Lely AJ, de Jong FH, et al. Adrenal function in adult long-term survivors of nephroblastoma and neuroblastoma. Eur J Cancer. 2012;48(8):1159–66. http://dx.doi.org/10.1016/j.ejca.2012.02.046.
- Ritchey ML. Nephron sparing surgery for Wilms tumor: where is the future? J Urol.
 2011 Oct;186(4):1179–80.

http://dx.doi.org/10.1016/j.juro.2011.07.048

- Wilimas JA, Magill L, Parham DM, Kumar M, Douglass EC. The potential for renal salvage in nonmetastatic unilateral Wilms' tumor. Am J Pediatr Hematol Oncol. 1991;13(3):342–4.
- 38. van Dalen EC, de Lijster MS, Leijssen LG, Michiels EM, Kremer LC, Caron HN, et al. Minimally invasive surgery versus open surgery for the treatment of solid abdominal and thoracic neoplasms in children. Cochrane Database Syst Rev. 2015 Jan 5;1:CD008403. http://dx.doi.org/10.1002/14651858.CD008403.pub3
- 39. Wilde JC, Aronson DC, Sznajder B, Van Tinteren H, Powis M, Okoye B, et al. Nephron sparing surgery (NSS) for unilateral Wilms tumor (UWT): the SIOP 2001 experience. Pediatr Blood Cancer. 2014;61:2175–9. http://dx.doi.org/10.1002/pbc.25185
- 40. Godzinski J, Graf N, Audry G. Current concepts in surgery for Wilms tumor--the risk and function-adapted strategy. Eur J Pediatr Surg. 2014 Dec;24(6):457–60. http://dx.doi.org/10.1055/s-0034-1396425

Erginel

- 41. Warmann SW, Godzinski J, van Tinteren H, Heij H, Powis M, Sandstedt B, et al. Minimally invasive nephrectomy for Wilms' tumors in children: data from SIOP 2001. Surgical Panel of the SIOP Renal Tumor Strategy Group. J Pediatr Surg. 2014; 49(11):1544–8. http://dx.doi.org/10.1016/j.jpedsurg.2014.06.005
- 42. User IR, Ekinci S, Kale G, Akyüz C, Büyükpamukçu M, Karnak I, et al. Management of bilateral Wilms' tumor over three decades: the perspective of a single center. J Pediatr Urol. 2015; 11(3):118.e1–6.

http://dx.doi.org/10.1016/j.jpurol.2014.11.012

- Ritchey ML. The role of preoperative chemotherapy for Wilms' tumor: the NWTSG perspective. National Wilms' Tumor Study Group. Semin Urol Oncol. 1999 Feb;17(1):21–7.
- Millar AJ, Davidson A, Rode H, Numanoglu A, Hartley PS, Daubenton JD, et al.
 Bilateral Wilms' tumors: a single-center experience with 19 cases. J Pediatr Surg. 2005;40(8):1289–94.

http://dx.doi.org/10.1016/j.jpedsurg.2005.05.013

45. Szavay P, Luithle T, Semler O, Graf N, Fuchs J. Surgery of cavoatrial tumor thrombus in nephroblastoma: a report of the SIOP/GPOH study. Pediatr Blood Cancer. 2004;43:40–5.

http://dx.doi.org/10.1002/pbc.20056

 Pannek J, Goepel M, Kremens B, Otto T, Sadony V. Surgical management of Wilms' tumor with intracardiac neoplastic extension. Thorac Cardiovasc Surg. 1994;42(2):108–11.

http://dx.doi.org/10.1055/s-2007-1016467

- 47. Hunt TM, Firmin RK, Johnstone MJ. Management of a patient with Wilms' tumour extending into the right heart chambers: a case report and a review of other published reports. Br Heart J. 1988;60(2):165–8.
- 48. Rioja Zuazu J, Rodríguez-Rubio Cortadillas F, Zudaire Bergera JJ, Saiz Sansi A, Rosell Costa D, Robles García JE, et al. Extracorporeal circulation and hypothermy surgery in tumors with vena cava extension: 20 years experience at the University Clinic of Navarra. Actas Urol Esp. 2008;32(4):396–405.
- 49. Ritchey ML, Shamberger RC, Haase G, Horwitz J, Bergemann T, Breslow NE. Surgical complications after primary nephrectomy for Wilms' tumor: report from the National Wilms' Tumor Study Group. J Am Coll Surg. 2001 Jan;192(1):63–8.
- 50. Mitchell C, Pritchard-Jones K, Shannon R, Hutton C, Stevens S, Machin D, et al. Immediate nephrectomy versus preoperative chemotherapy in the management of non-metastatic Wilms' tumour: results of a randomised trial (UKW3) by the UK Children's Cancer Study Group. United Kingdom Cancer Study Group. Eur J Cancer. 2006 Oct;42(15):2554–62.

http://dx.doi.org/10.1016/j.ejca.2006.05.026

- 51. Godzinski J. The current status of treatment of Wilms' tumor as per the SIOP trials.
 J Indian Assoc Pediatr Surg. 2015 Jan;20(1):16–20.
 http://dx.doi.org/10.4103/0971-9261.145439
- 52. Shamberger RC, Guthrie KA, Ritchey ML, Haase GM, Takashima J, Beckwith JB, et al. Surgery-related factors and local recurrence of Wilms' tumor in National Wilms Tumor Study 4. Ann Surg. 1999 Feb;229(2):292–7.
- 53. Gow KW, Barnhart DC, Hamilton TE, Kandel JJ, Chen MK, Ferrer FA, et al. Primary nephrectomy and intraoperative tumor spill: report from the Children's Oncology Group (COG) renal tumors committee. J Pediatr Surg. 2013 Jan;48(1):34–8. http://dx.doi.org/10.1016/j.jpedsurg.2012.10.015.
- 54. Green DM, Breslow NE, D'Angio GJ, Malogolowkin MH, Ritchey ML, Evans AE, et al. Outcome of patients with Stage II/favorable histology Wilms' tumor with and without local tumor spill: a report from the National Wilms' Tumor Study Group. Pediatr Blood Cancer. 2014 Jan;61(1):134–9.

http://dx.doi.org/10.1002/pbc.24658. Epub 2013 Aug 23.

- 55. Lemerle J, Voute PA, Tournade MF, Delemarre JF, Jereb B, Ahstrom L, et al. Preoperative versus postoperative radiotherapy, single versus multiple courses of actinomycin D, in the treatment of Wilms' tumor. Preliminary results of a controlled clinical trial conducted by the International Society of Paediatric Oncology (S.I.O.P). Cancer. 1976;38:647–654.
- 56. Dome JS, Liu T, Krasin M, Lott L, Shearer P, Daw NC, et al. Improved survival for patients with recurrent Wilms' tumor: the experience at St. Jude Children's Research Hospital. J Pediatr Hematol Oncol. 2002;24:192–8.
- 57. Grundy P, Breslow N, Green DM, Sharples K, Evans A, D'Angio GJ. Prognostic factors for children with recurrent Wilms' tumor: results from the Second and Third National Wilms' Tumor Study. J Clin Oncol. 1989;7:638–47.

Chapter 5

Management of Bilateral Wilms Tumours

Alastair J. W. Millar,¹ Sharon Cox,¹ Alan Davidson²

¹Department of Paediatric Surgery, University of Cape Town and Red Cross War Memorial Children's Hospital, Cape Town, South Africa; ²Department of Haematology and Oncology, University of Cape Town and Red Cross War Memorial Children's Hospital, Cape Town, South Africa

Author for correspondence: Alastair J. W. Millar, FRCS, FRACS (Paed Surg), FCS(SA), DCH, Emeritus Professor of Paediatric Surgery, Red Cross War Memorial Children's Hospital, Rondebosch 7700, Cape Town, South Africa. Email: alastair.millar@uct.ac.za

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch5

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Synchronous bilateral Wilms tumours (BWTs) represent 4% to 7% of all Wilms tumours (WTs) and present at a younger age than unilateral WTs do. At least 10% of synchronous BWTs have unfavourable histology, and up to 22% are associated with genitourinary abnormalities, aniridia, WAGR (WT, aniridia, genitourinary anomalies, and retardation) syndrome, Denys–Drash syndrome, hemihypertrophy or one of the other overgrowth syndromes. The long-term disease-free survival rate of patients with unilateral WT is now approaching 90% and is around 70% for those with metastatic disease. For both synchronous and metachronous WTs, the prognosis is less favourable, with reported cure rates

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

approaching 80% in the best centres but are often considerably less in resource-poor settings. Also, there is the potential for a reduced quality of life due to renal insufficiency and the possible need for renal transplantation. Thus, the major clinical challenge in BWTs is the preservation of functioning renal tissue, while achieving cure with the minimum of therapy-related morbidity. Although mortality is generally associated with progressive disease of anaplastic tumours, the emphasis of management has been increasingly placed on nephron-sparing surgical approaches in an attempt to reduce ultimate renal insufficiency. Chemotherapy followed by nephron-sparing surgery has been able, in most cases, to eradicate the tumour while preserving renal function. Radiotherapy has largely been avoided because of fear of long-term radiation injury to the residual functioning renal mass. Patient selection, appropriate pre- and post-operative chemotherapy, and skilful surgical techniques all contribute to excellent outcomes, where these are achievable.

Key words: Bilateral Wilms tumour; Nephroblastomatosis; Nephron-sparing surgery

Introduction

Nephroblastoma named after Max Wilms (1867–1918), a German surgeon, who published a monograph on 'Mixed tumours of the kidney' in 1899 while in Leipzig, working under the famous surgeon Professor Friedrich Trendelenburg, has become synonymous with the name "Wilms" (1). The tumour is an embryonal tumour derived from the metanephros containing epithelial, blastemal and stromal tissues. It is the third most frequently seen paediatric malignancy, is the most common renal tumour of childhood and is particularly seen in the under 5-year age group. It is associated with a number of 'overgrowth' syndromes mostly related to chromosome 11, as well as other genitourinary abnormalities.

Synchronous bilateral Wilms tumours (BWTs) make up 4% to 7% of all Wilms tumours (WTs) and present at a younger age than unilateral WTs (mean age, 2.6 vs ~3.3 years) (2, 3) do. Current results after the treatment of unilateral WT are excellent, even in some resource-limited settings, and protocols of management have been extensively interrogated on both sides of the Atlantic, in Children's Oncology Group (COG) and Société Internationale d'Oncologie Pediatrique (SIOP) trials. Both investigative groups have equivalent final outcomes with the 5-year overall survival in the region of 90% for favourable-histology unilateral WT, although there are differences between the groups in the approach to management, that is, in the use and timing of chemotherapy, surgery and radiotherapy. This has been emulated in some middle-income settings (4), but the outcome in patients with bilateral tumours is not as favourable, with the overall survival at 4 years varying between 81% for favourable histology and 55% for anaplastic histology in the National Wilms Tumor Study-5 trial (5, 6) and the overall 5-year survival being 85% in the SIOP-9 trial (7).

Predisposing factors

Nephrogenic rests are areas of embryonal metanephric tissue still present after the 36th week of life. Nephroblastomatosis refers to multiple or diffuse rests. While only a few will undergo clonal transformation into WTs, they are considered precursor lesions. Actively proliferating rests, termed hyperplastic rests, can cause the greatest diagnostic challenge and can appear radiologically and histologically similar to a WT (8). Rests, present on histological examination of resected tumours, increase the risk of a metachronous tumour developing in the other kidney, especially in the under 1-year age group. Diligence in follow-up and monitoring is thus required if nephroblastomatosis is present in a resected tumour. In a selected group of infants where unilateral tumours are well circumscribed and where nephroblastomatosis is identified pre- or intra-operatively, nephron-sparing surgery (NSS) may have a place. The NWTS group has recommended that this approach be assessed in a formal study. However, exclusion criteria for NSS such as those used in the SIOP 2001 protocol for unilateral WTs also need to be considered for BWT in order to stay oncologically safe (9). These are preoperative tumour rupture, abdominal lymph node metastases, tumour in the renal vein, multifocal tumour and infiltration into the renal pelvis (10). Nephrogenic rests may be seen in up to 90% of synchronous BWTs and 94% of metachronous BWTs (78% in our series) (11); about 70% of children with synchronous BWT in the NWTS series had multiple nephrogenic rests or nephroblastomatosis (12-14).

The prevalence of bilateral disease is higher among children with genetic predisposition syndromes such as WAGR (WT, aniridia, genitourinary anomalies and retardation), Beckwith–Wiedemann syndrome, hemihypertrophy and one of the overgrowth syndromes. These patients contribute to 22% of BWT series (15, 16).

Investigation

After investigation of history, blood pressure measurement, and physical examination looking specifically for signs of associated syndromes, all patients require routine testing of complete blood count, liver functions and renal functions and also urinalysis. Some centres advise coagulation studies because WT patients can develop acquired von Willebrand disease (17, 18).

Abdominal imaging includes a computed tomography (CT) scan with contrast or magnetic resonance imaging (MRI) (Figure 1a and b). Ultrasound scanning of the abdomen is often performed after clinical examination, which reveals a mass and guides more definitive imaging. In addition, ultrasound with Doppler may be the preferred method to assess intravascular extension of the tumour. MRI is of major value in the identification of nephrogenic rests. Nowadays metastatic workup includes a CT scan of the chest.



Figure 1. Axial (a) and sagittal (b) MRI scans of a 2-year-old child with bilateral Wilms tumours after 12 weeks of chemotherapy. Reprinted with permission from Red Cross War Memorial Children's Hospital Radiology Department.

There is only scarce evidence that an 18F-fluorodeoxyglucose (FDG)-positron emission tomography (PET)-CT may be helpful in patients with bilateral disease (19). WT is 18F-FDG avid; 18F-FDG-PET-CT imaging adds clinically applicable information to conventional imaging, and avid areas correspond to histologically active disease, but currently there is no evidence that it will discriminate WT from nephroblastomatosis.

Management

The aim of BWT management is to achieve cure with a minimum treatment-related morbidity, with the challenge in BWT being the preservation of adequate functioning renal tissue (20, 21). Renal tissue preservation may, however, be particularly difficult in cases with a delay in presentation and advanced local disease, in cases in which the chemotherapy response is poor, or in cases of a metachronous presentation where a contralateral nephrectomy has already been performed. Management may also be complicated by the presence of multiple nephrogenic rests or areas of nephroblastomatosis, as these are difficult to clinically distinguish from WTs (8, 22).

Management should commence with early clinical diagnosis, and for those presenting with ostensibly unilateral disease, a very close look at the contralateral kidney with biopsy of any suspicious areas is essential (23). Preoperative imaging is increasingly accurate (24) but should not be relied on due to some inconsistencies in imaging characteristics (8) and should not replace careful visualization of the contralateral kidney (25).

Preoperative chemotherapy in synchronous tumours is now the standard of care on both sides of the Atlantic. Previous protocols allowed for primary resection of tumours followed by chemotherapy. While the overall survival in a study by the United Kingdom Children's Cancer Study Group was similar in those undergoing primary surgery followed by chemotherapy and those treated with neoadjuvant chemotherapy followed by surgery, the incidence of renal failure was 20% in the initial surgery group versus 6% in the initial chemotherapy group (26).

With synchronous BWT, the traditional approach has been bilateral renal biopsies and staging of each kidney, followed by neoadjuvant chemotherapy and then renal salvage procedures (partial nephrectomy or tumourectomy) (14, 22). However, data from NWTS-4 and NWTS-5 have shown conclusively that a significant number of patients have unfavourable histology, which is revealed at the time of definitive surgery following chemotherapy but missed during initial biopsy.

Because anaplasia is difficult to detect (27) and bilateral childhood renal tumours are most likely to be WTs, initial biopsy is now controversial in patients with BWT. The current COG study of BWT and of patients with unilateral WT predisposed to the development of bilateral tumours attempts to avoid upfront biopsy but mandates biopsy after 6 weeks of three-drug chemotherapy (AREN0534, National Cancer Institute) (19, 28). The COG studies have shown a higher risk of local recurrence in patients who had tumour spillage or rupture, irrespective of the cause or extent of the soiling (with tumour biopsy being included as a cause of spillage). Thus, in the COG studies, all patients with tumour spillage, including biopsy, are considered stage III. When performing a biopsy, it is important to biopsy all lesions in both kidneys as discordant pathology occurs in up to 20% of cases with favourable histology on one side and anaplastic histology on the other (27).

In suspected BWT, where doubt exists due to atypical imaging or with a patient older than 10 years, an upfront biopsy may be indicated to rule out a non-Wilms tumour (AREN0534) even though it may not rule out anaplasia (28).

While most deaths occur because of progressive disease especially in the case of anaplastic tumours and usually in the first 2 years after diagnosis, emphasis has been increasingly placed on nephron-sparing surgical approaches to avoid subsequent renal insufficiency (15, 16, 29, 30). Nephron sparing is contraindicated in the face of diffuse anaplasia (3) and in cases of the Denys–Drash syndrome (31).

With metachronous presentation, the contralateral kidney has been removed and only the affected side remains. In this scenario, an attempt at NSS is indicated. Alternatively, should excision be performed, the patient may be free of disease but anephric, requiring renal transplantation. This has lifelong consequences of immune suppression and drug toxicity.

Chemotherapy

The current COG BWT study protocol is a response-based protocol. Neoadjuvant chemotherapy consists of three drugs (vincristine, dactinomycin and doxorubicin) given for 6 weeks, with a provision for a further 6 weeks if NSS is not feasible. Surgery is mandated at week 12. This protocol does not require pretreatment biopsies because bilateral tumours are very rarely clear cell sarcoma or rhabdoid tumours of the kidney. They are invariably WT, and therefore, biopsy does not change the therapy. In addition, anaplasia is difficult to diagnose, and biopsy may upstage the tumour and increase the chance of local recurrence.

SIOP (32) uses vincristine and dactinomycin for preoperative chemotherapy for 8 weeks, adding doxorubicin after 4 weeks only if there is a poor response. Despite adequate neo-adjuvant chemotherapy, there is a subgroup of patients with BWT with progressive or non-responsive disease. This non-responsiveness may be due to anaplasia, and thus insensitivity to administered therapy, necrosis and rhabdomyomatous or mature stromal differentiation (33, 34). While the second group of patients do not respond radiologically to therapy, they have improved outcomes when compared with those with anaplasia. It is thus crucial to establish the exact histology, and therefore, all such patients are best served by surgery.

Post-operative chemotherapy is based on the histology of the surgical specimen and the stage and is chosen according to the kidney with the highest risk. Cyclophosphamide, etoposide and carboplatin are added for unfavourable histology.

Surgery

The surgery for BWTs needs a multidisciplinary approach for diagnosis and treatment, and BWT patients benefit from surgery in the national centres of excellence with the most experience in BWT.

The decision to operate on both kidneys at the outset or to do so sequentially depends on the size and site of the tumours and their response to therapy. In every case, assessment of differential renal function by way of radionucleotide scan is mandatory prior to surgery. In cases with small peripheral tumours, both sides can be approached at the same laparotomy. When the tumours are very large and/or centrally placed, we prefer to operate on the difficult side first so that we are aware of how much renal function remains before operating on the relatively easier side.

Traditionally, these tumours are resected via a transabdominal approach. More recently, a retroperitoneal approach has been advocated, citing earlier post-operative recovery and slightly less associated surgical morbidity with equivalent outcomes when compared to the usual transabdominal approach, but this approach is yet to be widely adopted (35).

Several innovations have facilitated the NSS. In situ topical cooling with vascular pedicle cross-clamping or ex situ perfusion with preservation solution and 'bench surgery', first described by Lilly and Starzl in 1975 (36), allows careful and extensive dissection and reconstruction in a bloodless field without the loss of renal function from ischaemia (37). Using in situ topical cooling results in a relatively short cross-clamp time and topical cooling with ice is sufficient to preserve the renal function in most cases. Bilateral extensive NSS with our technique of ice-dam cooling (Figure 2) without the need for bench surgery does allow for better preservation of renal function and avoids significant acute tubular necrosis in most cases. However, some surgeons have preferred not to use any form of cooling (38).

It is important to avoid traction or torsion of the renal artery to prevent spasm, intimal damage and vessel occlusion. Some surgeons have specifically avoided using a clamp that may injure the vessels and have used finger occlusion instead (38). Either way, prior to commencing the resection, the kidney should be fully mobilized on its vascular and ureteric pedicle. The resection line is then marked on the tumour mass, identifying the normal renal tissue which is to be preserved, prior to any cross-clamp in an attempt to minimize the ischaemic time. Considerable mobilization of the kidney is possible as the renal vessels usually display an increased diameter, due to the increased needs of the growing tumour prior to chemotherapy, and this allows for making surgical access relatively easy. At this point, the renal hilar and peri-aortic nodes are sampled to rule out lymphatic spread. The renal vein and inferior vena cava should be palpated for evidence of tumour extension.

Gerota's fascia is opened, and the perirenal fat is dissected off the renal surface, excluding the fat attached to the tumour mass. If the tumour is peripherally situated or is well localized to an upper or lower pole, a wedge or guillotine tumourectomy or partial nephrectomy is performed. For large tumours, the outline of the tumour to be resected is scored with diathermy on the renal mass prior to proceeding with the surgery (Figure 3). The capsule is peeled back to expose the adjacent renal parenchyma. Using either fine bipolar cautery or the ultrasonic scalpel in our centre to facilitate dissection, as both cut well in a wet environment, we resect the tumour while the residual part of the kidney to be preserved remains in ice (Figure 4). Fuchs has described three zones of technical repair, with the cortex lending itself to cautery haemostasis, the medulla to suture/ligation haemostasis and the pelvicalyceal system to fine absorbable suture repair (39). After the pelvi-calyceal repair, a double I stent can be placed to ensure better urine drainage and thus reduce the chance of a urine leak. The Cavitron ultrasonic surgical aspirator (Cavitron Surgical Systems, Stanford, CA) has also been used effectively by Ritchey and Coppes (37). Where possible, the residual renal parenchyma is 'folded' on itself with suture reconstitution of the renal capsule to achieve a near-normal post-operative appearance (Figure 5).



Figure 2. (a) Ice-bath kidney cooling in a 2-year-old child with bilateral nephroblastomatosis and an upper-pole Wilms tumour in the kidney. The score line indicating the line of dissection is clearly shown. (b) The resected specimen (cut surface shown by black arrow) shows the anaplastic Wilms tumour (white arrow) surrounded by nephrogenic rests. Reprinted with permission from Red Cross War Memorial Children's Hospital Surgery and Pathology Department.



Figure 3. The left kidney of the child depicted in Figure 1. The kidney is cooled in an ice bath after dissection of the kidney to isolate the pedicle, and the outline of tumour to be resected is scored with diathermy on the renal mass. Reprinted with permission from Red Cross War Memorial Children's Hospital Surgery Department.

In exceptional circumstances, bench surgery with autotransplantation can be performed, as demonstrated in one of our cases (37, 40). Vessel re-anastomosis can usually be performed in the orthotopic position as the renal artery usually has an enlarged diameter; alternatively, the reconstructed residual kidney can be transplanted onto the iliac vessels. The disadvantage, how-ever, is that it is very difficult to visually discriminate between tumour, nephroblastomatosis and normal renal tissue while performing ex vivo perfusion. In this situation, multiple frozen-section biopsies may be required. These are generally not satisfactory as these can also be difficult for the histopathologist to interpret. At the end of the procedure, a drain is inserted and a Foley catheter should remain in the bladder for a few days. The double J stent should be removed via cystoscopy in 4 to 6 weeks. If the surgical specimen reveals diffuse anaplasia and there is incomplete resection, additional surgery is indicated to ensure complete resection of the tumour (28). Successful bench surgery and subsequent contralateral tumourectomy were performed in one patient in our series. The patient then developed recurrence of tumour in that kidney, but he remains a long-term disease-free survivor with good renal function after a second resection.

Enucleation of the tumour by blunt dissection should only be considered for patients with favourable-histology WT. If anaplasia is present, enucleation is contraindicated as clear margins are mandated for anaplastic tumours (28).



Figure 4. Dissection of the 'fillet' of normal renal tissue in the same child. Reprinted with permission from Red Cross War Memorial Children's Hospital Surgery Department.

Tumours not responding to chemotherapy and still not amenable to NSS may require nephrectomy. In this instance, the dissection plane is outside the Gerota's fascia. Unless the mass is in the upper pole or the adrenal is abutting the mass, the adrenal gland should be left in situ.

Resection of local recurrence in the residual kidney is possible, and it has shown some promising results (41). In our series, apart from the patient indicated above, there have been four other occasions where we have resected recurrences with good long-term results, two of which had tumour extending into the pelvi-calyceal system.

Radiotherapy

In BWT, radiotherapy use has decreased over the years; 57% of patients on NWTS-2 and NWTS-3 (12) received renal or renal-bed irradiation, while 42 (21.4%) of the 196 renal units registered on the renal salvage procedure arm of NWTS-4 were treated with radiotherapy (14). Radiation therapy is considered for abdominal stage III tumours or for stage II cases of anaplasia with involved margins at tumour resection. Its use could come at the cost of reduced renal function, particularly in young patients with the added toxicity of anthracy-cline chemotherapy (42). Fortunately, it has been shown that anthracyclines can be omitted



Figure 5. The residual kidney folded over to close the renal capsule with preserved renal artery and vein (arrow). Reprinted with permission from Red Cross War Memorial Children's Hospital Surgery Department.

in Intermediate risk stage III cases in SIOP; however, this needs to be determined in stage V cases (43). Other than cardiomyopathy and second malignant neoplasms, renal failure is the most common source of morbidity in BWT.

Prognosis

The variables identified for poor outcome have been unfavourable histology, advanced (local) stage and age over 3 years at diagnosis. Six of the seven patients in our series who presented with metastases died, although none had anaplastic histopathology (44). About 1% of unilateral WT will subsequently develop a contralateral tumour, with the risk for this tumour development being higher in children younger than 12 months who have nephrogenic rests at diagnosis (10).

Renal failure is an obvious concern in patients with BWT, the aetiology being multifactorial, and it can be due to the compounded effects of the inherent renal disease related to the patient's genetics, nephrotoxic chemotherapy, effects of radiotherapy and the loss of renal parenchyma. Risk factors for renal failure include Denys–Drash and other congenital syndromes, metachronous tumour and progressive disease in BWT with the need for bilateral nephrectomies and radiation nephritis, while the greatest risk factor is that of BWT (40, 45).

The risk of renal failure increases with the loss of more than 50% of renal mass. With the changing nature of treatment and the increasing efforts to preserve renal parenchyma, the rate of renal failure has decreased from 16.4% in NWTS-1 and NWTS-2 (1969–1979) to 9.9% in NWTS-3 (1979–1986) and 3.8% in NWTS-4 (1986–1998) (46). However, more recent long-term studies reflect a higher incidence of renal failure. The SIOP 9301 study reported a 14% rate of end-stage renal disease in 49 children treated from 1993 to 2001 (16). Dekkers et al. (47) showed that tumour nephrectomy, as well as radiotherapy, carries a higher risk of impaired renal function and hypertension. Nonetheless, the absence of significant renal impairment among most survivors is a proof of the success of NSS following initial chemotherapy. However, all BWT patients are eligible to be surveilled and to have systematic follow-up of blood pressure, urine protein and renal function.

In the few patients requiring bilateral nephrectomy, such as those with unresectable tumours or Denys–Drash syndrome, renal transplantation is usually performed between 1 and 2 years without the evidence of malignancy (34). The general consensus is to wait till at least 2 years of disease-free survival for a cadaver donor and 1 year for a living, related donor (34).

Conclusion

With appropriate patient selection and both pre- and post-operative chemotherapy and skilful surgical techniques, an excellent outcome can be achieved in most cases.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

References

- Raffensperger J. Max Wilms and his tumor. J Pediatr Surg. 2015;50(2):356–9. http://dx.doi.org/10.1016/j.jpedsurg.2014.10.054
- De Backer A, Lamote J, Keuppens F, Willems G, Otten J. Bilateral Wilms' tumor: in situ cooling of the kidney facilitates curative excision of tumors, with preservation of renal function. J Pediatr Surg. 1995;30(9):1338–40.
- 3. Neville HL, Ritchey ML. Wilms' tumor. Overview of National Wilms' Tumor Study Group results. Urol Clin North Am. 2000;27(3):435–42.
- Davidson A, Hartley P, Desai F, Daubenton J, Rode H, Millar A. Wilms tumour experience in a South African centre. Pediatr Blood Cancer. 2006;46(4):465–71. http://dx.doi.org/10.1002/pbc.20388

- Ehrlich PF. Bilateral Wilms' tumor: the need to improve outcomes. Expert Rev Anticancer Ther. 2009;9(7):963–73. http://dx.doi.org/10.1586/era.09.50
- Dome JS, Cotton CA, Perlman EJ, Breslow NE, Kalapurakal JA, Ritchey ML, et al. Treatment of anaplastic histology Wilms' tumor: results from the fifth National Wilms' Tumor Study. J Clin Oncol. 2006;24(15):2352–8. http://dx.doi.org/10.1200/JCO.2005.04.7852
- Weirich A, Ludwig R, Graf N, Abel U, Leuschner I, Vujanic GM, et al. Survival in nephroblastoma treated according to the trial and study SIOP-9/GPOH with respect to relapse and morbidity. Ann Oncol. 2004;15(5):808–20.
- Cox SG, Kilborn T, Pillay K, Davidson A, Millar AJ. Magnetic resonance imaging versus histopathology in Wilms tumor and nephroblastomatosis: 3 examples of noncorrelation. J Pediatr Hematol Oncol. 2014;36(2):e81–4. http://dx.doi.org/10.1097/MPH.0b013e318290c60d
- Wilde JC, Aronson DC, Sznajder B, Van Tinteren H, Powis M, Okoye B, et al. Nephron sparing surgery (NSS) for unilateral Wilms tumor (UWT): the SIOP 2001 experience. Pediatr Blood Cancer. 2014;61(12):2175–9. http://dx.doi.org/10.1002/pbc.25185
- 10. Desai D, Nicholls G, Duffy PG. Bench surgery with autotransplantation for bilateral synchronous Wilms' tumor: a report of three cases. J Pediatr Surg. 1999;34(4):632–4.
- Horwitz JR, Ritchey ML, Moksness J, Breslow NE, Smith GR, Thomas PR, et al. Renal salvage procedures in patients with synchronous bilateral Wilms' tumors: a report from the National Wilms' Tumor Study Group. J Pediatr Surg. 1996;31(8):1020–5.
- 12. Blute ML, Kelalis PP, Offord KP, Breslow N, Beckwith JB, D'Angio GJ. Bilateral Wilms tumor. J Urol. 1987;138(4 Pt 2):968–73.
- 13. Cooper CS, Jaffe WI, Huff DS, Canning DA, Zderic SA, Meadows AT, et al. The role of renal salvage procedures for bilateral Wilms tumor: a 15-year review. J Urol. 2000;163(1):265–8.
- Coppes MJ, Arnold M, Beckwith JB, Ritchey ML, D'Angio GJ, Green DM, et al. Factors affecting the risk of contralateral Wilms tumor development: a report from the National Wilms Tumor Study Group. Cancer. 1999;85(7):1616–25.
- Indolfi P, Jenkner A, Terenziani M, Crocoli A, Serra A, Collini P, et al. Synchronous bilateral Wilms tumor: a report from the Associazione Italiana Ematologia Oncologia Pediatrica (AIEOP). Cancer. 2013;119(8):1586–92. http://dx.doi.org/10.1002/cncr.27897
- 16. Sudour H, Audry G, Schleimacher G, Patte C, Dussart S, Bergeron C. Bilateral Wilms tumors (WT) treated with the SIOP 93 protocol in France: epidemiological survey and patient outcome. Pediatr Blood Cancer. 2012;59(1):57–61. http://dx.doi.org/10.1002/cncr.27897

- Coppes MJ, Zandvoort SW, Sparling CR, Poon AO, Weitzman S, Blanchette VS. Acquired von Willebrand disease in Wilms' tumor patients. J Clin Oncol. 1992;10(3):422–7.
- Jonge Poerink-Stockschlader AB, Dekker I, Risseeuw-Appel IM, Hahlen K. Acquired Von Willebrand disease in children with a Wilms' tumor. Med Pediatr Oncol. 1996;26(4):238–43.

```
http://dx.doi.org/10.1002/(SICI)1096-911X(199604)26:4<238::AID-MPO3>3.0.CO;2-K
```

- Wilms Tumor and Other Childhood Kidney Tumors. Treatment for Health Professionals (PDQ[®]). National Cancer Institute; 2015.
- 20. Kieran K, Ehrlich PF. Current surgical standards of care in Wilms tumor. Urol Oncol. 2016;34(1):13–23.
 - http://dx.doi.org/10.1016/j.urolonc.2015.05.029
- 21. Vanden Berg RN, Bierman EN, Noord MV, Rice HE, Routh JC. Nephron-sparing surgery for Wilms tumor: a systematic review. Urol Oncol. 2016;34(1):24–32. http://dx.doi.org/10.1016/j.urolonc.2015.07.003
- 22. Anderson J, Slater O, McHugh K, Duffy P, Pritchard J. Response without shrinkage in bilateral Wilms tumor: significance of rhabdomyomatous histology. J Pediatr Hematol Oncol. 2002;24(1):31–4.
- 23. Kessler O, Franco I, Jayabose S, Reda E, Levitt S, Brock W. Is contralateral exploration of the kidney necessary in patients with Wilms tumor? J Urol. 1996;156 (2 Pt 2):693–5.
- 24. Merchant SA, Badhe PB. Nephroblastomatosis pathologic and imaging characteristics. J Postgrad Med. 1995;41(3):72–80.
- Montgomery BT, Kelalis PP, Blute ML, Bergstralh EJ, Beckwith JB, Norkool P, et al. Extended followup of bilateral Wilms tumor: results of the National Wilms Tumor Study. J Urol. 1991;146(2 Pt 2):514–8.
- 26. Kumar R, Fitzgerald R, Breatnach F. Conservative surgical management of bilateral Wilms tumor: results of the United Kingdom Children's Cancer Study Group. J Urol. 1998;160(4):1450–3.
- Hamilton TE, Green DM, Perlman EJ, Argani P, Grundy P, Ritchey ML, et al. Bilateral Wilms' tumor with anaplasia: lessons from the National Wilms' Tumor Study. J Pediatr Surg. 2006;41(10):1641–4. http://dx.doi.org/10.1016/j.jpedsurg.2006.05.053
- 28. Children's Oncology Group. Treatment for Patients with Bilateral, Multicentric, or
- Bilaterally-Predisposed Unilateral Wilms Tumor. AREN0534. 2012.
- Millar AJ, Davidson A, Rode H, Numanoglu A, Hartley PS, Daubenton JD, et al. Bilateral Wilms' tumors: a single-center experience with 19 cases. J Pediatr Surg. 2005;40(8):1289–94.

http://dx.doi.org/10.1016/j.jpedsurg.2005.05.013

- 30. Millar AJ, Davidson A, Rode H, Numanoglu A, Hartley PS, Desai F. Nephron-sparing surgery for bilateral Wilms' tumours: a single-centre experience with 23 cases. Afr J Paediatr Surg. 2011;8(1):49-56. http://dx.doi.org/10.4103/0189-6725.78669
- 31. Paya K, Horcher E, Lawrenz K, Rebhandl W, Zoubek A. Bilateral Wilms' tumor surgical aspects. Eur J Pediatr Surg. 2001;11(2):99-104. http://dx.doi.org/10.1055/s-2001-13787
- 32. Pritchard-Jones K. International Society of Paediatric Oncology Nephroblastoma (Wilms Tumour) Clinical Trial and Study: SIOP WT 2001. 2001.
- Shamberger RC, Haase GM, Argani P, Perlman EJ, Cotton CA, Takashima J, et al. 33. Bilateral Wilms' tumors with progressive or nonresponsive disease. J Pediatr Surg. 2006;41(4):652-7; discussion 652. http://dx.doi.org/10.1016/j.jpedsurg.2005.12.004
- 34. Rudin C, Pritchard J, Fernando ON, Duffy PG, Trompeter RS. Renal transplantation in the management of bilateral Wilms' tumour (BWT) and of Denys-Drash syndrome (DDS). Nephrol Dial Transplant. 1998;13(6):1506-10.
- 35. Lim II, Honeyman JN, Fialkowski EA, Murphy JM, Price AP, Abramson SJ, et al. Experience with retroperitoneal partial nephrectomy in bilateral Wilms tumor. Eur J Pediatr Surg. 2015;25(1):113-7.

http://dx.doi.org/10.1055/s-0034-1387944

- 36. Lilly JR, Phister RR, Putnam CW, Kosloske AM, Starzl TE. Bench surgery and renal autotransplantation in the pediatric patient. jpediatr surg. 1975;10:623-30.
- 37. Ritchey ML, Coppes MJ. The management of synchronous bilateral Wilms tumor. Hematol Oncol Clin North Am. 1995;9(6):1303-15.
- 38. Kieran K, Davidoff AM. Nephron-sparing surgery for bilateral Wilms tumor. Pediatr Surg Int. 2015;31(3):229-36. http://dx.doi.org/10.1007/s00383-015-3668-1
- 39. Fuchs J, Szavay P, Seitz G, Hundgretinger R, Schafer JF, Warmann S. Nephron sparing surgery for synchronous bilateral nephroblastoma involving the renal hilus. J Urol. 2011;186940:1430-6, doi:10.1016/j.juro.2
- 40. Ritchey ML, Green DM, Thomas PR, Smith GR, Haase G, Shochat S, et al. Renal failure in Wilms' tumor patients: a report from the National Wilms' Tumor Study Group. Med Pediatr Oncol. 1996;26(2):75-80.
 - http://dx.doi.org/10.1002/(SICI)1096-911X(199602)26:2<75::AID-MPO1>3.0.CO;2-R
- 41. Kieran K, Williams MA, McGregor LM, Dome JS, Krasin MJ, Davidoff AM. Repeat nephron-sparing surgery for children with bilateral Wilms tumor. J Pediatr Surg. 2014;49(1):149-53.

http://dx.doi.org/10.1016/j.jpedsurg.2013.09.048

- Smith GR, Thomas PR, Ritchey M, Norkool P. Long-term renal function in patients with irradiated bilateral Wilms tumor. National Wilms' Tumor Study Group. Am J Clin Oncol. 1998;21(1):58–63.
- 43. Pritchard-Jones K, Bergeron C, de Camargo B, van den Heuvel-Eibrink MM, Acha T, Godzinski J, et al. Omission of doxorubicin from the treatment of stage II-III, intermediate-risk Wilms' tumour (SIOP WT 2001): an open-label, non-inferiority, randomised controlled trial. Lancet. 2015;386(9999):1156–64. http://dx.doi.org/10.1016/S0140-6736(14)62395-3
- 44. Beckwith JB. Precursor lesions of Wilms tumor: clinical and biological implications. Med Pediatr Oncol. 1993;21(3):158–68.
- 45. Breslow NE, Collins AJ, Ritchey ML, Grigoriev YA, Peterson SM, Green DM. End stage renal disease in patients with Wilms tumor: results from the National Wilms Tumor Study Group and the United States Renal Data System. J Urol. 2005;174(5):1972–5.

http://dx.doi.org/10.1097/01.ju.0000176800.00994.3a

- 46. Shearer P, Parham DM, Fontanesi J, Kumar M, Lobe TE, Fairclough D, et al. Bilateral Wilms tumor. Review of outcome, associated abnormalities, and late effects in 36 pediatric patients treated at a single institution. Cancer. 1993;72(4):1422–6.
- Dekkers IA, Blijdorp K, Cransberg K, Pluijm SM, Pieters R, Neggers SJ, et al. Longterm nephrotoxicity in adult survivors of childhood cancer. Clin J Am Soc Nephrol. 2013;8(6):922–9.

http://dx.doi.org/10.2215/CJN.09980912

Chapter 6

Extrarenal Wilms' Tumor: Challenges in Diagnosis, Embryology, Treatment and Prognosis

Reza Shojaeian,¹ Mehran Hiradfar,¹ Parisa Saeedi Sharifabad,² Nona Zabolinejad³

¹Department of Pediatric Surgery, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran; ²Department of Pediatric Urology, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran; ³Department of Pathology, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran

Author for correspondence: Reza Shojaeian, MD, Assistant Professor of Pediatric Surgery, Head of Pediatric Surgery Department, Sarvar Children's Hospital, Mashhad University of Medical Sciences, Mashhad, Iran. Email: Drshojaeian@ymail.com

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch6

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Wilms' tumor is one of the most common childhood solid malignancies, which classically arises from primitive metanephric cells, but exceptionally it may arise in places other than kidneys. Extrarenal Wilms' tumor is a rare but challenging entity, considering its diagnosis,

In: *Wilms Tumor*. Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Shojaeian et al.

histopathology, staging, treatment, and prognosis. Diagnosis of extrarenal Wilms' tumor is always postsurgical, which may jeopardize treatment planning and consulting with parents in the first step. The histopathology of Wilms' tumor is very confusing. While most authors believe that it arises from primitive ectopic nephrogenic rests, teratoid Wilms' tumor leads to the debate whether this tumor is neoplastic or embryonic. Staging of extrarenal Wilms' tumor is also a challenge when we consider the National Wilms' Tumor Study (NWTS) recommendations; all these tumors should be considered as stage II or higher as they are beyond the renal capsule. This will mandate chemotherapy for all patients while most of the reported cases have a favorable histology, and long-term tumor-free survival has been reported even with exclusive surgery in some case reports. Although treatment strategies for extrarenal Wilms' tumor are the same as those for renal Wilms' tumor, different locations and neighboring organs may invoke special considerations and scenarios while planning for surgery and adjuvant therapies. Consulting with the parents is also a problem, considering the rarity of the disease and limited publications. In this chapter, we discuss all these topics in detail after a systematic review of extrarenal Wilms' tumor cases to date in order to provide a clear perspective for confronting this rare disease.

Key words: Extrarenal; Nephroblastoma; Pediatrics; Wilms' tumor

Introduction

Nephroblastoma or Wilms' tumor is one of the most common childhood malignancies, which accounts for almost 95% of renal malignancies in pediatrics. Extrarenal nephroblastoma is a rare entity, which was first described by Moyson et al. (1) in 1961. The estimated rate of occurrence of nephroblastoma outside the kidneys is almost 0.5 to 1% of all cases of Wilms' tumor. Extrarenal Wilms' tumor (ERWT) occurs mostly in childhood; however, it is also rarely reported in adults (2). Apart from the primary ERWT, nephroblastoma may be observed outside the kidneys in two other situations: metastatic disease and nephroblastoma arising in a teratoma; therefore, in the case of ERWT, it is mandatory to evaluate the kidneys for primary tumor preoperatively and search the whole specimen for any teratoid element postoperatively (3).

We reviewed all the reported childhood ERWTs (under 14 years), excluding those arising from teratomas (teratoid Wilms' tumor). The results are summarized in Table 1. Among 80 reported ERWT cases, more than 60% were younger than 4 years and a female predominance was observed while the female-to-male ratio was 3:2 (4).

The association of ERWT with a horseshoe kidney has been reported previously, and almost 7% of the reported ERWTs were found to be associated with the horseshoe kidney. Dysraphism is the second commonly found abnormality among ERWT patients (5).

Extrarenal Wilms' tumor: challenges in diagnosis and treatment

Author	Year	Gender	Age	ERWT location	Stage	Treatment	Follow-up
1. Moyson et al. (1)	1961	F	3	Mediastinal	II	Surgery + chemo	NA
2. Bhajkar et al. (6)	1964	М	2	Retroperitoneal	II	Surgery + chemo	NA
3. Edelstein et al. (7)	1965	М	3	Retroperitoneal	II	Surgery + chemo + RAD	2
4. Wu and Garcia (8)	1971	F	7	Pelvic	IV	Surgery + chemo + RAD	1
5. Thompson et al. (9)	1973	F	4	Inguinal	III	Surgery + chemo + RAD	2
6. Thompson et al. (9)	1973	М	3	Inguinal	IV	Surgery + chemo + RAD	0.5
7. Akhtar et al. (10)	1977	М	0.2	Inguinal	II	Surgery + chemo	1.5
8. Gaikwad et al. (11)	1977	М	0.2	Retroperitoneal	II	Surgery	0.5
9. Madanat et al. (12)	1978	F	9	Mediastinal	II	Surgery + chemo + RAD	3
10. Madanat et al. (12)	1978	М	0.3	Inguinal	NA	Surgery + chemo	2
11. McCauley et al. (13)	1979	F	4	Retroperitoneal	NA	Surgery + chemo + RAD	4
12. Aterman et al. (14)	1979	F	5	Retroperitoneal	NA	Surgery + chemo + RAD	0.7
13. Orlowski et al. (15)	1980	М	3.5	Paratesticular	II	Surgery	11
14. Fried et al. (16)	1980	М	3	Retroperitoneal	NA	Surgery + chemo	NA
15. Fernandes et al. (17)	1980	М	6	Retroperitoneal	III	Surgery + chemo + RAD	6
16. Johnson et al. (18)	1980	F	1	Retroperitoneal	NA	Surgery + chemo	1
17. Taylor et al. (19)	1980	М	0.5	Inguinal	NA	Surgery + chemo + RAD	0.5
18. Ho et al. (20)	1981	М	1.2	Paratesticular	Ι	Surgery	1
19. Bittencourt et al. (21)	1981	F	14	Female genital organs	III	Surgery + chemo + RAD	5.5

Table 1. Review of the reported childhood ERWTs in literature

Shojaeian et al.

Author	Year	Gender	Age	ERWT location	Stage	Treatment	Follow-up
20. Tamaro et al. (22)	1982	F	4	Retroperitoneal	NA	Surgery + chemo + RAD	1
21. Fernandes et al. (17)	1982	F	2	Retroperitoneal	II	Surgery + chemo	5
22. Adam et al. (23)	1983	М	10	Retroperitoneal	NA	NA	NA
23. Meng and Jagadeesan (24)	1983	М	3	Retroperitoneal	I	Surgery	1
24. Lüchtrath et al. (25)	1984	F	1.2	Inguinal	NA	Surgery + chemo	1.3
25. Bell et al. (26)	1985	F	13	Female genital organs	Ι	Surgery	9.5
26. Naito et al. (27)	1985	F	3	Retroperitoneal	NA	Surgery + chemo	2.3
27. Fernandes et al. (17)	1988	F	2	Retroperitoneal	II	Surgery + chemo	1
28. Lai et al. (28)	1888	F	3	Inguinal	II	Surgery + chemo	1.5
29. Narasim- harao et al. (29)	1989	F	NA	Retroperitoneal	NA	NA	NA
30. Broecker et al. (30)	1989	F	0.9	Pelvic	II	Surgery + chemo	1
31. Fernandes et al. (17)	1989	М	6	Retroperitoneal	II	Surgery + chemo + RAD	7
32. Broecker et al. (30)	1989	F	2	Retroperitoneal	II	Surgery + chemo + RAD	7
33. Wakely et al. (31)	1989	F	4	Female genital organs	II	Surgery + chemo + RAD	6
34. Broecker et al. (30)	1989	F	2	Retroperitoneal	IV	Surgery + chemo	1.3
35. Wakely et al. (31)	1989	F	1.5	Retroperitoneal	II	Surgery + chemo + RAD	6
36. Strand et al. (32)	1990	М	12	Inguinal	III	Surgery + chemo	NA
37. Simha and Doctor (33)	1991	F	3	Inguinal	III	Surgery + chemo + RAD	NA

Table 1. Continued

Extrarenal Wilms' tumor: challenges in diagnosis and treatment

Author	Year	Gender	Age	ERWT location	Stage	Treatment	Follow-up
38. Andrews et al. (34)	1992	F	NA	Lumbosacral	II	Surgery + chemo	1.4
39. Andrews et al. (34)	1992	М	NA	Retroperitoneal	II	Surgery + chemo	0.7
40. Andrews et al. (34)	1992	F	NA	Lumbosacral	II	Surgery + chemo	6.5
41. Andrews et al. (34)	1992	М	NA	Retroperitoneal	IV	Surgery + chemo + RAD	2
42. Andrews et al. (34)	1992	F	NA	Lumbosacral	III	Surgery + chemo + RAD	4
43. Andrews et al. (34)	1992	F	NA	Retroperitoneal	Ι	Surgery + chemo	3
44. Andrews et al. (34)	1992	М	NA	Retroperitoneal	II	Surgery + chemo	2
45. Andrews et al. (34)	1992	F	NA	Pelvic	II	Surgery + chemo	0.7
46. Suzuki et al. (35)	1993	М	2	Retroperitoneal	II	Surgery + chemo	NA
47. Rasheed et al. (36)	1993	М	3	Retroperitoneal	III	Surgery + chemo + RAD	7
48. Rasheed et al. (36)	1993	F	4	Retroperitoneal	III	Surgery + chemo + RAD	2
49. Fahner et al. (37)	1995	F	2.5	Lumbosacral	II	Surgery + chemo	1
50. Arkovitz et al. (38)	1996	М	3.5	Inguinal	III	Surgery + chemo + RAD	NA
51. López Cubillana et al. (39)	1997	F	2	Retroperitoneal	II	Surgery + chemo	3
52. Kapur et al. (40)	1998	F		Retroperitoneal	NA	Surgery + chemo	3
53. Kapur et al. (40)	1998	F	NA	Retroperitoneal	NA	Surgery + chemo	0.7
54. Benatar et al. (41)	1998	F	11	Female genital organs	II	Surgery + chemo	NA
55. Massarelli et al. (42)	1999	F	2	Female genital organs	III	Surgery + RAD	2.5

Table 1. Continued

Shojaeian et al.

Author	Year	Gender	Age	ERWT location	Stage	Treatment	Follow-up
56. Iraniha et al. (43)	1999	F	12	Female genital organs	II	Surgery + chemo	1
57. Govender et al. (44)	2000	М	4	Lumbosacral	III	Surgery + RAD	NA
58. Babin et al. (45)	2000	F	13	Female genital organs	III	Surgery + chemo + RAD	5
59. Arda et al. (46)	2001	F	5	Lumbosacral	III	Surgery + chemo + RAD	NA
60. Oner et al. (47)	2002	F	3.5	Female genital organs	II	Surgery + chemo	7
61. Yunus et al. (48)	2003	NA	NA	Lumbosacral	NA	NA	NA
62. Cojean et al. (49)	2003	М	0.2	Retroperitoneal	III	Surgery + chemo	NA
63. Ngan et al. (50)	2009	F	6	Retroperitoneal	II	Surgery	1
64. Cooke et al. (51)	2009	М	1.2	Inguinal	II	Surgery	2
65. Jeong et al. (52)	2011	М	9	Inguinal	III	Surgery + chemo + RAD	NA
66. Teerthanath (53)	2011	F	6	Retroperitoneal	II	Surgery + chemo	4
67. Chowhan (54)	2012	М	1.3	Retroperitoneal	II	Surgery + chemo	NA
68. Armanda et al. (2)	2012	F	0.1	Lumbosacral	Ι	Surgery + chemo	2
69. Yamamoto et al. (55)	2012	М	NA	Paratesticular	NA	NA	NA
70. Li et al. (56)	2012	F	2	Pelvic	III	Surgery + chemo + RAD	3
71. Marwah et al. (57)	2012	F	1.2	Retroperitoneal	II	Surgery + chemo	NA
72. Hiradfar et al. (58)	2012	F	9	Inguinal	II	Surgery + chemo	3
73. Baskaran (59)	2013	М	3	Retroperitoneal	II	Surgery	NA
74. Rojas et al. (60)	2013	М	2	Lumbosacral	II	Surgery + chemo	NA

Table 1. Continued

Extrarenal Wilms' tumor: challenges in diagnosis and treatment

Author	Year	Gender	Age	ERWT location	Stage	Treatment	Follow-up
75. Morandi et al. (61)	2013	М	3	Paratesticular	Ι	Surgery	2
76. Goel et al. (62)	2014	NA	NA	Retroperitoneal	NA	Surgery	NA
77. Wu et al. (63)	2014	М	0.8	Retroperitoneal	II	Surgery	0.5
78. Wu et al. (63)	2014	М	1.5	Inguinal	II	Surgery	0.5
79. Al-Nsoor et al. (64)	2014	F	1.7	Retroperitoneal	II	Surgery	NA
80. Kumar et al. (5)	2015	F	7	Retroperitoneal	NA	Surgery	0.8

Table 1. Continued

Age and follow-up time are in years.

NA, data not available; RAD, radiotherapy.

Embryogenesis

Embryonic nephrogenic tissue normally differentiates into metanephric blastema, which is considered as the precursor of nephrons and mesenchymal stroma. The pronephros is developed during the third gestational week in the cervical region, extending to the more caudal parts, the cloaca. The upper part of pronephros regresses, and in the caudal part, it gives rise to the mesonephric duct that persists in two lateral foci. While the mesonephric duct extends to the cloaca, it enters to the metanephric blastema that surrounds the ureteric bud to develop the kidneys.Therefore, the progenitors of renal tissue goes through a journey from the cranial to the caudal part of the embryo, and during the longitudinal growth of the fetus, the intrapelvic kidneys ascend up to their expected level in the flank.

It is believed that the Wilms' tumor is the result of a developmental abnormality in the metanephric blastema. Persistent metanephric tissue after the 36th week of gestation could be the precursor of nephroblastoma. Nephrogenic rests may be observed anywhere in the craniocaudal migration line of primitive mesonephros and metanephros cells (Figure 1).

The pathogenesis of ERWT remains elusive, and several theories are discussed about it. Some authors have suggested that ERWT may arise from teratomas although later classifications divided teratoid Wilms' tumor and a true ERWT into two different entities while a true primary ERWT lacks any teratogenic element.

The most widely accepted hypothesis for the pathogenesis of ERWT suggests that the ectopic nephrogenic rest develops into a nephroblastoma. It is well known that the persistent

Shojaeian et al.

intrarenal fetal nephrogenic blastemal tissue may undergo oncogenic mutation and develop nephroblastoma. Several reports pointed to the observation of ectopic nephrogenic rests, especially in inguinal or retroperitoneal and lumbosacral regions (63).

The hypothesis of ectopic blastematous cells explains the development of ERWT in the craniocaudal migration pathway of primitive metanephros cells. The observation of WT1 gene in 25% of ERWT supports the oncogenic mutation of nephrogenic rests causing ERWT (65). The question is how we can explain the relatively high prevalence of ERWT in the inguinal region. Some genital structures such as Gartner's duct, seminal vesicles, vas deferens, and epididymis are differentiated mesonephric ducts that could explain the presence of ectopic nephrogenic rests as a precursor of ERWT in the inguinal region. Currently, several evidences support the hypothesis of ectopic nephrogenic blastemal cells causing the ERWT, which help us in better understanding of Wilms' tumor pathogenesis. Early diagnosis of nephrogenic rests outside the kidneys as a precancerous tissue will mandate close observation and prompt intervention while facing any evidence of atypia or malignancy (66).

Pathology

A classic microscopic feature of Wilms' tumor consists of the triphasic pattern that includes mesenchymal, epithelial, and blastemal elements. Histologic diagnosis of ERWT is supported by the observation of classic triphasic histology in the absence of any teratoid component. The presence of heterotopic teratomatous elements in more than 50% of total microscopic field suggests the diagnosis of teratoid Wilms' tumor, which is a quite different entity in the pathogenesis and embryology with germ cell origin. While confronting an extrarenal nephroblastoma, whole specimen should be examined in multiple cuts to exclude the diagnosis of teratoid Wilms' tumor that accounts for half of the reported extrarenal nephroblastic malignancies (67).

As discussed previously, ectopic nephrogenic rest is believed to be the precursor of Wilms' tumor, and histologic discrimination between these two is always a challenge (68). The presence of disordered structures, atypical mitosis, and marked pleomorphisms indicate the presence of Wilms' tumor in contrast with proliferative nephrogenic rests without atypia (63).

Considering the histologic findings, ERWTs could be classified as favorable or unfavorable. The review of reported cases shows predominant favorable histology among ERWTs. Beckwith and Palmer (69) proposed the criteria for ERWT pathological diagnosis that include the documented classic triphasic Wilms' pattern outside kidneys in the absence of teratoid or anaplastic elements while both kidneys are tumor free in imaging (Figure 2).
Extrarenal Wilms' tumor: challenges in diagnosis and treatment



Figure 1. Schematic sagittal view of a 4-week-old embryo. Mesonephros consists of mesonephric duct and nephrogenic cord (Drawing by R. Shojaeian originally).

Clinical presentation and diagnosis

Clinical presentation of ERWTs depends on the location and stage of the tumor. As the symptoms are nonspecific and mostly due to the mass effect of tumor, diagnosis of ERWT often becomes apparent postoperatively. Common ERWT sites include retroperitoneum, inguinal area, lumbosacral and pelvic, female genital organs (uterus, cervix, vagina, and ovaries), mediastinum and chest wall, and spermatic cord and paratesticular region. Like classic Wilms' tumor, ERWT commonly manifests with asymptomatic mass or nonspecific symptoms, such as abdominal pain and discomfort, weight loss, and urologic or gynecologic symptoms. Symptoms related to the tumor site and size, such as inguinal mass that resembles hernia or lymphadenitis, vaginal bleeding or discharge, hematuria or dysuria, dyspnea, spinal cord compression signs, and even paraplegia (61, 3) may be observed.

Relation between ERWT and horseshoe kidney is suggested as we observed the coexistence of horseshoe kidney and ERWT in almost 13% of all previously reported cases. This may be explained by the higher chance of ectopic primitive renal tissue among patients with abnormal migration of nephrogenic cells, so the diagnosis of ERWT must be kept in mind while confronting with an abdominal mass in a patient with horseshoe kidney (59).

Ultrasound study is usually the first paraclinical step in the evaluation of an abdominal or pelvic mass. Further imaging such as computed tomography (CT) or magnetic resonance study may be needed in some cases. However, ERWT does not have a pathognomonic radiologic image, and the diagnosis of ERWT is almost always postoperative. After pathologic

Shojaeian et al.



Figure 2. Microscopic evaluation of the specimen revealed composition of sheets, which were randomly arranged and tightly packed. Small blue cells were arranged in serpiginous aggregates (blastemal component), sharply circumscribed by focal spindling and intervening collagenous bundles apart from the surrounding stromal elements. There are also a few small tubules lined by primitive cuboidal cells and a small area of nephrogenic rest at the periphery. Pathological features suggested extrarenal Wilms' tumor (from Authors' personal archive).

confirmation of ERWT, both kidneys should be evaluated precisely with multislice spiral CT images to exclude any intrarenal tumor. CT scan with intravenous and oral contrast is often indicated and helpful to evaluate the tumor location and its resectability. Magnetic resonance imaging may also be helpful in paraspinal and thoracic tumors, especially with cord compression symptoms (64) (Figure 3). Recently, due to probable hazardous effects of CT scan especially in pediatric imaging, it is preferred to use MRI for evaluation of tumors in pediatric.

Tumor markers such as AFP (alpha-fetoprotein) and β HCG (beta-human chorionic gonadotropin) are useful to discriminate ERWT from other pediatric neoplasms such as the highly frequent germ cell tumors in childhood. Although a systemic search should be done for the tumor spread, ERWTs rarely metastasize. The most common sites of metastasis include lungs and liver. Three percent of the reported ERWT cases were metastatic based on our review.

Treatment

It is widely believed that National Wilms' Tumor Study (NWTS) system could be applied for ERWTs staging, while stage I definition should be modified as a localized tumor that could be completely excised with microscopic clear margins, no residue, and no tumor rupture during surgery; otherwise, there would not be any stage I ERWT as all tumors are located

Extrarenal Wilms' tumor: challenges in diagnosis and treatment



Figure 3. Left, Clinical manifestation of an inguinal ERWT. Right, Gross macroscopic view of inguinal ERWT (from Authors' personal archive).

beyond the borders of the kidneys. Although ERWT is uncommon, it should be kept in mind as a differential diagnosis of retroperitoneal or inguinal masses in childhood. Surgical approach depends on the location and surrounding structures. Rare cases of retroperitoneal or thoracic ERWT with intraspinal component and neurologic symptoms should be managed by multidisciplinary approach promptly to prevent irreversible neurologic sequels (64).

The role of intraoperative frozen section in an unidentified childhood mass or ERWTs has not been discussed clearly before and is not considered as a part of surgical principle, while total excision is the mainstay of treatment in most pediatric solid tumors when applicable (50).

Surgical excision remains the key step in the treatment of ERWT, especially when performed radically (70). Regional lymph node sampling is a part of the surgical principle as that for classic renal Wilms' tumor. Careful inspection of solid organs such as kidneys or liver and also peritoneum for tumor implants is recommended in abdominal ERWTs. Adjuvant chemotherapy is recommended for all ERWT cases postoperatively in spite of favorable histopathology in most of them. However, there are a few cases of successful treatment of stage I ERWTs with pure surgery.

Chemotherapy regimen is determined by histology and stage of the tumor, considering the NWTS protocols that consist of the administration of vincristine, actinomycin D, and doxorubicin. Regarding the current guidelines of NWTS, completely resected ERWTs with no evidence of tumor at or beyond the margins are considered as stage II and treated with vincristine and actinomycin D, while addition of doxorubicin will have benefits in stage III ERWTs.

Shojaeian et al.

Most ERWTs have favorable histology, but local recurrence is observed in about 11% of the reported cases, which is comparable with 15% predicted recurrence rate in classic renal Wilms' tumor with favorable histology. We found that 70% of ERWTs were in stage II and 23% in stage III, while distant metastasis was reported in 6% of patients. Two-year event-free survival of the reported ERWT cases was almost 85% and mortality rate was 5%, which are comparable with renal Wilms' tumor (60). Radiotherapy is reserved for unresectable tumors or for those with gross residue, recurrence, or metastasis (67). Bilateral ERWT has not been reported to date.

Conclusion

ERWT is considered a rare childhood malignancy with atypical presentations. The pathogenesis of ERWT becomes clearer by the popular theory, which suggests the heterotopic metanephric blastema as the precursor of ERWT while the diagnosis, staging, and treatment remain challenging. NWTS protocols are applied for ERWTs due to the rarity of the disease and lack of systematic data. We reviewed 87 reported childhood ERWT cases and observed favorable histology in most cases, which made the prognosis good and comparable to that of classic Wilms' tumor with the same stage and histology.

Conflict of Interests

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

The authors would like to thank Miss Maryam Hiradfar and Mr. Mohankumar for their help in grammatically editing the manuscript.

References

- Moyson F, Maurus-Desmarez R, Gompel C. [Mediastinal Wilms' tumor?]. Acta Chir Belg. 1961;Suppl 2:118–28.
- Armanda V, Culić S, Pogorelić Z, Kuljiš D, Budimir D, Kuzmić-Prusac I. Rare localization of extrarenal nephroblastoma in 1-month-old female infant. J Pediatr Urol. 2012 Aug;8(4):e43-5.
 - http://dx.doi.org/10.1016/j.jpurol.2012.03.005
- Song JS, Kim IK, Kim YM, Khang SK, Kim KR, Lee Y. Extrarenal teratoid Wilms' tumor: two cases in unusual locations, one associated with elevated serum AFP. Pathol Int. 2010 Jan;60(1):35–41. http://dx.doi.org/10.1111/j.1440-1827.2009.02468.x

Extrarenal Wilms' tumor: challenges in diagnosis and treatment

- 4. Thakkar NC, Sarin YK. Extra-renal Wilms' tumor: a rare diagnosis. APSP J Case Rep. 2015 May 1;6(2):17.
- 5. Kumar S, Narayanankutty Sunilkumar M, Surendran D. Extrarenal nephroblastoma in a 7 year old child: a rare case report with review of literature. Int J Contemp Pediatr. 2015 May;2(2):155–8.
- 6. Bhajkar AB, Joseph M, Bhat HS. Unattached nephroblastoma. Br J Urol. 1964 Jun;36:187-90.
- Edelstein G, Webb RS, Romsdahl MM, Arboit JM. Extrarenal Wilms' tumor. Am J Surg. 1965 Apr;109:509–12.
- Wu JP, Garcia J. Supernumerary kidney with Wilms' tumor. Wis Med J. 1971 Oct;70(10):211-6.
- 9. Thompson MR, Emmanuel IG, Campbell MS, Zachary RB. Extrarenal Wilm's tumors. J Pediatr Surg. 1973 Feb;8(1):37–41.
- 10. Akhtar M, Kott E, Brooks B. Extrarenal Wilms' tumor: report of a case and review of the literature. Cancer. 1977 Dec;40(6):3087–91.
- 11. Gaikwad KD, Patil PS, Bhavthankar AG, Dave VB, Chavan SB. Extra renal Wilms' tumour. Indian Pediatr. 1977 Aug;14(8):657–9.
- 12. Madanat F, Osborne B, Cangir A, Sutow WW. Extrarenal Wilms tumor. J Pediatr. 1978 Sep;93(3):439-43.
- McCauley RG, Safaii H, Crowley CA, Pinn VW. Extrarenal Wilms' tumor. Am J Dis Child. 1979 Nov;133(11):1174–7.
- 14. Aterman K, Grantmyre E, Gillis DA. Extrarenal Wilms' tumour: a review and case report. Invest Cell Pathol. 1979 Oct-Dec;2(4):309–18.
- 15. Orlowski JP, Levin HS, Dyment PG. Intrascrotal Wilms' tumor developing in a heterotopic renal anlage of probable mesonephric origin. J Pediatr Surg. 1980 Oct;15(5):679–82.
- 16. Fried AM, Hatfield DR, Ellis GT, Fitzgerald KW. Extrarenal Wilms' tumor: sonographic appearance. J Clin Ultrasound. 1980 Aug;8(4):360–2.
- Fernandes ET, Kumar M, Douglass EC, Wilimas J, Parham DM, Rao BN. Extrarenal Wilms' tumor. J Pediatr Surg. 1989 May;24(5):483–5.
- Johnson F, Luttenton C, Limbert D. Extrarenal and urothelial Wilms tumor. Urology. 1980 Apr;15(4):370–3.
- 19. Taylor WF, Myers M, Taylor WR. Extrarenal Wilms' tumour in an infant exposed to intrauterine phenytoin. Lancet. 1980 Aug 30;2(8192):481–2.
- Ho J, Ma L, Wong KC. An extrarenal Wilms' tumour arising from an undescended testis. Pathology. 1981 Jul;13(3):619–24.
- 21. Bittencourt AL, Britto JF, Fonseca LE Jr. Wilms' tumor of the uterus: the first report of the literature. Cancer. 1981 May 15;47(10):2496–9.

Shojaeian et al.

- 22. Tamaro P, Fonda E, Valente M, Zanazzo GA. [New case of extrarenal Wilms' tumor]. Pediatr Med Chir. 1982 Jan-Apr;4(1-2):143–7.
- 23. Adam YG, Rosen A, Oland J, Wallach N, Reif R. Extrarenal Wilms tumor. J Surg Oncol. 1983 Jan;22(1):56–8.
- 24. Meng LL, Jagadeesan K. Extrarenal Wilms' tumour. Med J Malaysia. 1983 Jun;38(2):134-6.
- 25. Lüchtrath H, de Leon F, Giesen H, Gök Y. Inguinal nephroblastoma. Virchows Arch A Pathol Anat Histopathol. 1984;405(1):113–8.
- Bell DA, Shimm DS, Gang DL. Wilms' tumor of the endocervix. Arch Pathol Lab Med. 1985 Apr;109(4):371–3.
- Naito K, Yokoyama O, Yamaguchi K, Hisazumi H, Ohzeki S, Kajiwara S, et al. [Extrarenal nephroblastoma: report of a case and review of the literature]. Hinyokika Kiyo. 1985 Oct;31(10):1773–80.
- Lai HS, Hung WT, How SW. Extrarenal Wilms' tumor--a case report. J Pediatr Surg. 1988 May;23(5):454–6.
- 29. Narasimharao KL, Marwaha RK, Kaushik S, Bharati B, Katariya S, Mitra SK, et al. Extrarenal Wilms' tumor. J Pediatr Surg. 1989 Feb;24(2):212–4.
- 30. Broecker BH, Caldamone AA, McWilliams NB, Maurer H, Salzberg A. Primary extrarenal Wilms' tumor in children. J Pediatr Surg. 1989 Dec;24(12):1283–8.
- 31. Wakely PE Jr, Sprague RI, Kornstein MJ. Extrarenal Wilms' tumor: an analysis of four cases. Hum Pathol. 1989 Jul;20(7):691–5.
- 32. Strand WR, Chou P, Pero JE, Kaplan WE. Extrarenal Wilms tumor occurring in the inguinal canal. J Urol. 1990 Apr;143(4):783–5.
- Simha MR, Doctor VM. Extrarenal Wilm's tumour. A case report and review of literature. Indian J Cancer. 1991 Mar;28(1):16–21.
- Andrews PE, Kelalis PP, Haase GM. Extrarenal Wilms' tumor: results of the National Wilms' Tumor Study. J Pediatr Surg. 1992 Sep;27(9):1181–4.
- 35. Suzuki K, Miyake H, Tashiro M, Mori H, Fukushige T, Tanimura R, et al. Extrarenal Wilms' tumour. Pediatr Radiol. 1993;23(2):149–50.
- Rasheed K, O'Meara A, Kelleher J, Breatnach F, Fitzgerald RJ. Extrarenal Wilms' tumor. Eur J Pediatr Surg. 1993 Apr;3(2):121–3. http://dx.doi.org/10.1055/s-2008-1063527
- Fahner JB, Switzer R, Freyer DR, Mann JD, Mann RJ. Extrarenal Wilms' tumor. Unusual presentation in the lumbosacral region. Am J Pediatr Hematol Oncol. 1993 Feb;15(1):117–9.
- Arkovitz MS, Ginsburg HB, Eidelman J, Greco MA, Rauson A. Primary extrarenal Wilms' tumor in the inguinal canal: case report and review of the literature. J Pediatr Surg. 1996 Jul;31(7):957–9.

Extrarenal Wilms' tumor: challenges in diagnosis and treatment

- 39. López Cubillana P, Nortes Cano L, Martínez Barba E. [Extrarenal Wilms' tumor: diagnosis and treatment review regarding a case report]. Arch Esp Urol. 1997 Nov;50(9):999-1001. Spanish
- 40. Kapur VK, Sakalkale RP, Samuel KV, Meisheri IV, Bhagwat AD, Ramprasad A, et al. Association of extrarenal Wilms' tumor with a horseshoe kidney. J Pediatr Surg. 1998 Jun;33(6):935-7.
- Benatar B, Wright C, Freinkel AL, Cooper K. Primary extrarenal Wilms' tumor of 41. the uterus presenting as a cervical polyp. Int J Gynecol Pathol. 1998 Jul;17(3):277-80.
- 42. Massarelli G, Bosincu L, Costanzi G, Onida GA. Uterine Wilms' tumor. Int J Gynecol Pathol. 1999 Oct;18(4):402-3.
- 43. Iraniha S, Shen V, Kruppe CN, Downey EC. Uterine cervical extrarenal Wilms tumor managed without hysterectomy. J Pediatr Hematol Oncol. 1999 Nov-Dec;21(6):548-50.
- 44. Govender D, Hadley GP, Nadvi SS, Donnellan RB. Primary lumbosacral Wilms tumour associated with occult spinal dysraphism. Virchows Arch. 2000 May;436(5):502-5.
- Babin EA, Davis JR, Hatch KD, Hallum AV 3rd. Wilms' tumor of the cervix: a case 45. report and review of the literature. Gynecol Oncol. 2000 Jan;76(1):107-11. http://dx.doi.org/10.1006/gyno.1999.5625
- 46. Arda IS, Tüzün M, Demirhan B, Sevmis S, Hicsönmez A. Lumbosacral extrarenal Wilms' tumour: a case report and literature review. Eur J Pediatr. 2001 Oct;160(10):617-9.
- 47. Oner UU, Tokar B, Acikalin MF, Ilhan H, Tel N. Wilms' tumor of the ovary: a case report. J Pediatr Surg. 2002 Jan;37(1):127-9.
- 48. Yunus M, Hashmi R, Hasan SH, Brohi HM. Extrarenal Wilms' tumor. J Pak Med Assoc. 2003 Sep;53(9):436-9
- 49. Cojean N, Entz-Werle N, Eyer D, Becmeur F, Kehrli P, Marcellin L, et al. [Dumbbell nephroblastoma: an uncommon cause of spinal cord compression]. Arch Pediatr. 2003 Dec;10(12):1075-8.
- Ngan KW, Shaari S, Subramaniam T. Juxtarenal/pararenal Wilms' tumour in a six-50. year-old Malay girl. Singapore Med J. 2009 Sep;50(9):e329-31.
- 51. Cooke A, Deshpande AV, La Hei ER, Kellie S, Arbuckle S, Cummins G. Ectopic nephrogenic rests in children: the clinicosurgical implications. J Pediatr Surg. 2009 Dec;44(12):e13-6.

http://dx.doi.org/10.1016/j.jpedsurg.2009.09.015

52. Jeong YJ, Sohn MH, Lim ST, Kim DW, Jeong HJ, Yim CY. F-18 FDG PET/CT imaging of metastatic extrarenal Wilms tumor arising in the inguinal canal. Clin Nucl Med. 2011 Jun;36(6):475-8. http://dx.doi.org/10.1097/RLU.0b013e31820ade92

Shojaeian et al.

- 53. Teerthanath S. Extrarenal nephroblastoma. J Lab Physicians. 2011 Jan;3(1):59–60. http://dx.doi.org/10.4103/0974-2727.78572
- 54. Chowhan AK. Editorial Comment to paratesticular extrarenal Wilms' tumor. Int J Urol. 2012 May;19(5):491–2. http://dx.doi.org/10.1111/j.1442-2042.2012.03007.x
- 55. Yamamoto T, Nishizawa S, Ogiso Y. Paratesticular extrarenal Wilms' tumor. Int J Urol. 2012 May;19(5):490–1.

http://dx.doi.org/10.1111/j.1442-2042.2012.02971.x

- 56. Li K, Xiao X, Gao J, Yao W, Chen H, Zhang B. Pelvic Wilms tumor in a child with an absent right kidney and spinal malformations. J Pediatr Surg. 2012 Oct;47(10):e11-4. http://dx.doi.org/10.1016/j.jpedsurg.2012.05.025
- 57. Marwah N, Rattan KN, Rana P, Goyal V, Sen R. Extrarenal Wilms' tumor of the ovary: a case report and short review of the literature. J Gynecol Surg. 2012 August; 28(4): 306–8.
- 58. Hiradfar M, Shojaeian R, Zabolinejad N, Saeedi P, Joodi M, Khazaie Z, et al. Extrarenal Wilms' tumour presenting as an inguinal mass. Arch Dis Child. 2012 Dec;97(12):1077.

http://dx.doi.org/10.1136/archdischild-2012-302605

59. Baskaran D. Extrarenal teratoid Wilms' tumor in association with horseshoe kidney. Indian J Surg. 2013 Apr;75(2):128–32.

http://dx.doi.org/10.1136/archdischild-2012-302605

 Rojas Y, Slater BJ, Braverman RM, Eldin KW, Thompson PA, Wesson DE, et al. Extrarenal Wilms tumor: a case report and review of the literature. J Pediatr Surg. 2013 Jun;48(6):E33-5.

http://dx.doi.org/10.1016/j.jpedsurg.2013.04.021

- 61. Morandi A, Fagnani AM, Runza L, Farris G, Zanini A, Parolini F, et al. Extrarenal testicular Wilms' tumor in a 3-year-old child. Pediatr Surg Int. 2013 Sep;29(9):961–4. http://dx.doi.org/10.1007/s00383-013-3338-0
- 62. Goel V, Verma AK, Batra V, Puri SK. 'Primary extrarenal Wilms' tumour': rare presentation of a common paediatric tumour. BMJ Case Rep. 2014 Jun 6;2014. pii: bcr2013202172.
- 63. Wu Y, Zhu X, Wang X, Wang H, Cao X, Wang J. Extrarenal nephroblastomatosis in children: a report of two cases. BMC Pediatr. 2014 Oct 6;14:255. http://dx.doi.org/10.1186/1471-2431-14-255
- 64. Al-Nsoor N, Al-Emam O, Khader M. "Extrarenal Wilms' tumor with intraspinal extension: a case report." JRMS. 2014:21.3:71–4.

- 65. Roberts DJ, Haber D, Sklar J, Crum CP. Extrarenal Wilms' tumors. A study of their relationship with classical renal Wilms' tumor using expression of WT1 as a molecular marker. Lab Invest. 1993 May;68(5):528–36.
- Coli A, Angrisani B, Chiarello G, Massimi L, Novello M, Lauriola L. Ectopic immature renal tissue: clues for diagnosis and management. Int J Clin Exp Pathol. 2012;5(9):977–81.
- Sastri J, Dedhia R, Laskar S, Shet T, Kurkure P, Muckaden M. Extra-renal Wilms' tumour – is it different? Pediatr Nephrol. 2006 Apr;21(4):591–6. http://dx.doi.org/10.1007/s00467-005-0002-5
- 68. Beckwith JB. Nephrogenic rests and the pathogenesis of Wilms tumor: developmental and clinical considerations. Am J Med Genet. 1998 Oct 2;79(4):268-73.
- 69. Beckwith JB, Palmer NF. Histopathology and prognosis of Wilms tumors: results from the First National Wilms' Tumor Study. Cancer. 1978 May;41(5):1937–48.
- 70. Apoznański W, Sawicz-Birkowska K, Palczewski M, Szydełko T. Extrarenal nephroblastoma. Cent European J Urol. 2015;68(2):153–6. http://dx.doi.org/10.5173/ceju.2015.571

Chapter 7

Neoadjuvant Transcatheter Arterial Chemoembolization and Systemic Chemotherapy for the Treatment of Wilms Tumor

Min-Ju Li,¹ Da-Xing Tang,² Shan Xu,² Yong Huang,² De-Hua Wu,² Jin-Hu Wang,¹ Can Lai,³ Hong-Feng Tang,⁴ Qiang Shu¹

¹Division of Surgical Oncology, Department of Pediatric Surgery, ²Division of Urology, Department of Pediatric Surgery, ³Department of Radiology, and ⁴Department of Pathology, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China

Author for correspondence: Min-Ju Li, MD, Professor, Division of Surgical Oncology, Department of Pediatric Surgery, Children's Hospital, Zhejiang University School of Medicine, Hangzhou, China. Email: minjuli@qq.com

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch7

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

From 2003 to 2013, 55 patients (median age 3.3 years; 29 males, 26 females) with unresectable, metastatic, or diffuse anaplastic histology (AH) Wilms tumor were treated with neoadjuvant transcatheter arterial chemoembolization (TACE) and systemic chemotherapy. Characteristics

In: *Wilms Tumor.* Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

of patients were maximal tumor diameter greater than 10 cm, involvement of periaortic lymph nodes, tumor thrombus in inferior vena cava/right atrium, distal metastasis, or diffuse AH. The chemoembolic emulsion for TACE consisted of pirarubicin, vindesine, and iodized oil. For the tumor with distal metastasis or diffuse AH, cisplatin was added in the chemoembolic emulsion. Intravenous chemotherapy with vindesine and actinomycin D was administered 2-3 weeks after TACE. For the patients with distal metastasis or diffuse AH Wilms tumor, intravenous chemotherapy consisted of ifosfamide and etoposide. Nephrectomy was performed 2-3 weeks after preoperative combination therapy. Surgical stage was assigned according to local operative findings in terms of the National Wilms Tumor Study (NWTS) Group combined with the pretherapeutic imaging to define metastatic disease. Postoperative treatment was based on tumor histology and surgical stage. All patients were followed up for 17-141 months (median: 82 months). No cardiotoxicity, renal insufficiency, and hepatic dysfunction after neoadjuvant TACE and systemic chemotherapy were found. Oral mucositis developed in 5 patients, grade I-II marrow suppression developed in 12 patients, and 19 patients became moderately febrile. In terms of response evaluation criteria in solid tumors, partial response (PR) in 34 (61.8%), stable disease (SD) in 19 (34.5%), and progressive disease (PD) in 2 (3.6%) patients were observed. Four of five patients had complete regression of inferior vena cava tumor thrombus. Atrial tumor thrombus retreated to inferior vena cava in one of two patients. Distant metastasis disappeared in four of six cases. Fifty patients (90.1%) underwent complete tumor resection. Tumor spillage occurred in 3 patients (5.5%). Two patients (3.6%) had microscopic residual disease. Surgical stages were stage II in 25, stage III in 24, and stage IV in 6 patients. On pathologic examination, tumor necrosis was >90% in 14 (25.5%), 50%-90% in 23 (41.8%), and <50% in 18 cases (32.7%). The 5-year event-free survival was 92.7%, and the overall survival was 94.5%. These preliminary results suggest that the use of neoadjuvant TACE and systemic chemotherapy may provide a promising choice in the treatment of unresectable, metastatic, or diffuse AH Wilms tumor in children. Further investigations are necessary.

Key words: Neoadjuvant therapy; Systemic chemotherapy; Transcatheter arterial chemoembolization; Wilms tumor

Introduction

Wilms tumor accounts for about 6% of all malignant tumors in children, and it is the most common malignant renal tumor of childhood. The outcome for patients with Wilms tumor has improved remarkably during the past decades owing to the use of adjuvant chemotherapy and neoadjuvant chemotherapy (1). However, the treatment of children with unresectable, metastatic, or diffuse anaplastic histology (AH) Wilms tumor remains a challenge (2–4). The unresectable criteria most commonly utilized are the tumor diameter greater than or equal to 10 cm, involvement of adjacent vital structures, and intracaval/atrial tumor extension. These

factors significantly increase the risk of surgical morbidity (4). Novel treatment strategies are needed to maximize survival and minimize long-term morbidity for these patients.

Preoperative embolization of the renal artery as a coadjuvant treatment in high-risk renal neoplasia has benefits for the subsequent nephrectomy (5–20). In an attempt to improve the outcome of patients with unresectable, metastatic, or diffuse AH Wilms tumor, we have performed preoperative transcatheter arterial chemoembolization (TACE) since 1995 (21–24). In previous studies, we found preoperative TACE combined with systemic chemotherapy could induce more massive necrosis of the tumor, further improving the complete resection rate of the tumor (25). In this phase II study, we examined the efficacy and safety of combined-modality neoadjuvant therapy using TACE and systemic chemotherapy as a first-line treatment for unresectable, metastatic, or diffuse AH Wilms tumor.

Patients and methods

From January 2003 to December 2013, 55 patients with unilateral unresectable, metastatic, or diffuse AH Wilms tumor were treated using preoperative TACE combined with systemic chemotherapy at our hospital.

All patients underwent abdominal computed tomography (CT), magnetic resonance imaging, ultrasound scan, and chest CT examination at admission. When metastasis in liver or lung was visible on CT, the patient was classified as stage IV disease. A core-needle biopsy for histologic diagnosis was performed before the treatment. Histology results were classified as unfavorable if diffuse anaplastic (AH) features were present and favorable (FH) if absent. This study was approved by the institutional ethics committee, and informed consent was obtained from the children's parents before enrollment. Characteristics of the patients were as follows: maximal tumor diameter greater than 10 cm, involvement of periaortic lymph nodes, tumor thrombus in inferior vena cava (IVC)/right atrium, distal metastasis based on the imaging studies, or tumor with diffuse AH according to the biopsy report (Table 1). Eligible patients were between 5 months and 11 years of age (median: 3.3 years), 29 boys and 26 girls. The right kidney was treated in 33 patients and the left in 22 patients. Patients with bilateral renal tumors, congenital mesoblastic nephroma, clear cell sarcoma of kidney, rhabdoid tumor of kidney, and renal cell carcinoma were excluded from this study.

The preoperative treatment consisted of alternating TACE and intravenous chemotherapy. Patients underwent TACE under intravenous and caudal epidural anesthesia. The femoral artery was catheterized using the Seldinger technique. A 5-F Pigtail catheter (Cook Vascular Incorporated, Pennsylvania) was introduced into the abdominal aorta to perform aortography and to define the tumor blood supply for the purpose of planning the chemoembolization (Figure 1A and B). The selective renal arterial catheterization and angiography of the

 Table 1. Demographics and tumor characteristics of 55 patients with unresectable, metastatic, or diffuse AH Wilms tumor

Median age, y (range)	3.3 (0.5-11)
Sex, M:F ratio	29:26
Tumor side, right/left	33/22
Tumor characteristics on admission	
Greatest dimension >10 cm	29 (52.7%)
Involvement of periaortic lymph nodes	9 (16.4%)
Tumor thrombus in inferior vena cava/right atrium	7 (12.7%)
Distal metastasis	6 (10.9%)
Diffuse anaplastic histology	4 (7.3%)

involved kidney were performed using a 4-F or 5-F Cobra catheter (Cook Vascular Incorporated) (Figure 1C). The chemoembolic emulsion consisted of pirarubicin (Main Luck Pharmaceuticals Inc, Shenzhen, China) 40 mg/m²; vindesine (Minsheng Pharmaceuticals Inc, Hangzhou, China) 3 mg/m²; and iodized oil (Lipiodol; Guerbet, Aulnay-sous-Bois, France) 0.5 mL per tumor maximal diameter (in centimeters). The drugs were mixed, diluted in 120 mL of normal saline, and infused into the renal artery over 60 minutes. For the tumor with distal metastasis or diffuse AH, cisplatin 80 mg/m² was added in the chemoembolic emulsion. The catheter was removed after treatment. Intravenous hydration and alkalization were administered before, during, and after TACE. Postprocedure nausea and vomiting were treated with antiemetics.

Intravenous chemotherapy was administered 2–3 weeks after TACE. It consisted of vindesine 3 mg/m² (maximum 4 mg) weekly and actinomycin D 3 days × 15 μ g/kg/d (maximum 500 μ g) weeks 1 and 3. For the patients with distal metastasis or diffuse AH Wilms tumor, the intravenous chemotherapy consisted of ifosfamide 1,200 mg/m² and etoposide 100 mg/m² on days 2–4. For the patients younger than 1 year or with a body weight of <12 kg, the dosages of drugs for TACE and intravenous chemotherapy were reduced to two thirds.

Pirarubicin is a new anthracycline antibiotic with an antitumor efficacy similar to that of doxorubicin but less cardiotoxic because of its different pharmacodynamic properties (26). To minimize the side effects of anthracycline, we used pirarubicin instead of doxorubicin hydrochloride in this study. Vindesine is an analogue of the vinca alkaloids. Its spectrum of antitumor activity is similar to that of vincristine, but with milder neurotoxicity (27).



Figure 1. Preoperative TACE in a 2-year-old boy with left renal Wilms tumor. (A) Computed tomography (CT) finding left retroperitoneal huge tumor, invasion to the right side. (B) Aortography showing a large hypervascular lesion in the left kidney with abundant neovascularity. (C) Selective renal arterial catheterization and the chemoembolic emulsion infusing into the left renal artery. (D) Tumor volume is significantly reduced after alternating TACE and intravenous chemotherapy. Iodized oil deposits still visible within the tumor in the repeated CT scan before operation.

Nephrectomy was performed 2–3 weeks after preoperative TACE combined with systemic chemotherapy. Four patients underwent repeated preoperative combined treatment due to pulmonary metastasis. Tumor volumes were measured on ultrasound scans using the ellipsoid formula (length × thickness × depth × 0.523). The measurement was performed before preoperative therapy and repeated before surgery. Tumor volume reduction in comparison with the initial volume was calculated. Tumor response to treatment was defined according to the new response evaluation criteria in solid tumors [Response Evaluation Criteria In Solid Tumors (RECIST) 1.1] (28). Toxicity was scored according to the Children's Cancer Group toxicity grading system (29). After tumor resection, surgical specimens were microscopically examined for features of tumor, such as the surgical margin and necrosis, which was defined as complete if no viable cells were found in the tumor and nodules. Histopathologic

classification and surgical stage were assigned according to the National Wilms Tumor Study (NWTS) Group (30). Postoperative treatment using systemic chemotherapy and radiotherapy was based on tumor histology and its surgical stage. The treatment protocol was worked out according to the NWTS protocol modified by the Beijing Children's Hospital (31). Fifteen patients received radiotherapy after operation due to stage III or diffuse AH Wilms tumor. Postoperative follow-up was performed at the first month and then every 3–6 months after surgery, including physical, imaging (abdominal sonography and CT, chest x-ray, and electrocardiogram), and laboratory screening (blood and urine analysis, and liver and renal function tests). All patients were followed up until December 31, 2014.

Statistical methods

Standard methods were used for the analysis of censored and noncensored data. Event-free survival (EFS) time was defined as the time from the date of diagnosis to the first occurrence of progression, relapse, or death. Overall survival (OS) time was measured from the date of diagnosis to death or the patients being still alive on December 31, 2014. The Kaplan–Meier method was used to calculate EFS and OS rates, and the rates are presented as the rate \pm SE. All analyses were carried out using the SPSS 16.0 statistical software system.

Response to treatment

No patients experienced death while preoperative treatment. No preoperative tumor rupture, drug-induced cardiotoxicity, renal insufficiency, or hepatic dysfunction were found in all of the patients. Oral mucositis developed in 5 (9.1%) patients, grade I–II marrow suppression developed in 12 (21.8%) patients, and 19 (34.5%) patients became moderately febrile after chemoembolization; this was controlled with symptomatic treatment (Table 2). Tumor volumes were significantly reduced after preoperative TACE and systemic chemotherapy. Iodized oil deposits were still visible within the tumor in the repeated CT scan before operation (Figure 1D). Color Doppler ultrasonography showed the abundant blood flow decreased significantly in the tumor after preoperative treatment (Figure 2A and B). Tumor volume was 488 mL (median) at diagnosis and 198 mL (median) before operation. In terms of new response evaluation criteria in solid tumors (RECIST 1.1) (28), partial response (PR) was observed in 34 (61.8%), stable disease (SD) in 19 (34.5%), and progressive disease (PD) in 2 (3.6%) patients after preoperative therapy (Table 2). Five patients had tumor invading the IVC. Four of them had complete regression of the IVC tumor thrombus before operation. One patient underwent vena caval thrombectomy during nephrectomy. Two cases additionally had extensive venothrombotic invasion to the right atrium at admission. The atrial tumor thrombus retreated to IVC in one patient who underwent radical nephrectomy and vena caval thrombectomy. The atrial tumor thrombus reduced obviously but not completely disappeared in another patient who underwent thrombectomy under cardiopulmonary bypass with deep

Table 2. Complications, tumor response, and outcome of 55 patients treated with preoperative TACE and systemic chemotherapy

Complications during preoperative treatment	
Death while treatment	0 (0.0%)
Cardiotoxicity	0 (0.0%)
Renal insufficiency	0 (0.0%)
Hepatic dysfunction	0 (0.0%)
Oral mucositis	5 (9.1%)
Grade I-II marrow suppression	12 (21.8%)
Moderate febrile	19 (34.5%)
Tumor volume, mL	
On admission	488 (292-804)
Before operation	198 (126–324)
Tumor response	
PR	34 (61.8%)
SD	19 (34.5%)
PD	2 (3.6%)
Inferior vena and atrial tumor thrombus disappeared	4/7 (57.1%)
Distal metastasis disappeared	4/6 (66.7%)
Complete tumor resection	50/55 (90.9%)
Rapture during operation	3/55 (5.5%)
Microscopic residual	2/55 (3.6%)
Tumor necrosis	
>90% tumor necrosis	14/55 (25.5%)
50–90% tumor necrosis	23/55 (41.8%)
<50% tumor necrosis	18/55 (32.7%)
Postoperative stage	
Ι	0 (0.0%)
Ш	25 (45.5%)
III	24 (43.6%)
IV	6 (10.9%)

(Continued)

Histology	
FH	51 (92.7%)
AH	4 (7.3%)
Outcome	
EFS	92.7% (95% CI: 85.8-99.6%)
OS	94.5% (95% CI: 88.5-100%)

Table 2. Continued

PR, partial response; SD, stable disease; PD, progressive disease; FH, favorable histology; AH, anaplastic histology; EFS, event-free survival; OS, overall survival.

hypothermia and circulatory arrest during nephrectomy. Histopathologic examination of the resected atrial thrombus showed epithelial and mesenchymal components. Six patients had distant metastasis on admission, including one case of liver metastasis and five cases of pulmonary metastasis. Distal metastasis disappeared in front of the renal tumor resection in four cases (Figure 3A–C) and during postoperative chemotherapy in two patients. Fifty patients (90.1%) underwent complete tumor resection after preoperative therapy. Tumor spillage occurred in 3 patients (5.5%). Two patients (3.6%) had microscopic residual disease. The overall distribution of patients in the series according to surgical staging was stage II in 25 (45.5%), stage III in 24 (43.6%), and stage IV in 6 (10.9%) patients (Table 2).

Histopathologic findings

Postoperative histopathology revealed FH Wilms tumor in 51 patients (92.7%) and diffuse AH in 4 patients (7.3%) (Table 2). Pathologic examination of the specimen found massive necrosis in the tumor and increased thickness of the fibrous envelope around the tumor (Figure 4A and B). Tumor necrosis was >90% in 14 (25.5%), 50%–90% in 23 (41.8%), and <50% in 18 cases (32.7%) (Table 2). Necrosis was visible not only in the main tumor but also in the metastases of periaortic lymph nodes (25). Iodized oil deposition in the para-aortic lymph nodes was observed in six cases. Postoperative histologic examination of these marked lymph nodes confirmed lymph node metastases with necrosis. This finding implies that the chemo-embolization agent flowed directly into the para-aortic lymph node metastases.

Outcomes

All patients were followed up until December 31, 2014. The median length of follow-up was 82 months (range: 17–141 months). Thirty-one patients had been followed up for more than 5 years. The 5-year EFS was 92.7% [95% confidence interval (CI): 85.8%–99.6%] and OS was 94.5% (95% CI: 88.5%–100%) (Table 2) (Figure 5A and B). Four patients relapsed.



Figure 2. Color Doppler ultrasonography appearances of the left renal Wilms tumor in a 3-yearold boy. (A) Diffusely increased blood flow in the tumor before treatment. (B) Blood flow deceased significantly within the tumor after preoperative TACE and systemic chemotherapy.

One of them relapsed with pulmonary metastases 8 months after operation and was cured by chemotherapy. Three of them died. The first patient was a 6.5-year-old boy with a stage III FH Wilms tumor. He had a relapse involving the proximal tibia 1 year after operation that was unresponsive to treatment, and he died 20 months after presentation. The second patient, a 7-year-old boy with a stage II FH tumor, died 18 months after operation from liver metastatic disease. The third was an 8-year-old girl with a surgical stage III AH tumor and IVC thrombus invasion to the right atrium at admission. The tumor thrombus retreated to suprahepatic IVC after preoperative therapy, and nephrectomy with thrombus removal was performed. She relapsed with pulmonary and liver metastases 6 months after operation and subsequently died. Late effects of therapy were evaluated in the survivors. One patient had scoliosis caused by radiotherapy after operation. No case(s) of anthracycline cardiotoxicity, liver disease, hypertension, and renal dysfunction were documented in survivors. Puberty and growth disturbances were not observed in all patients. No new late cancers were detected on follow-up.

Discussion

Although the FH Wilms tumors showed excellent outcome, the survival rate for unresectable, metastatic, or diffuse AH Wilms tumor cases remains to be improved. Patients with unresectable or metastatic Wilms tumor fare worse than patients with localized and resectable tumors (32). The unresectable criteria commonly utilized are huge size of the tumor, involvement of adjacent vital structures, and intracaval/atrial tumor extension. These factors significantly increase the risk of surgical morbidity, principally hemorrhage, and tumor spillage during initial nephrectomy (4). Larger tumors are at higher risk of intraoperative



Figure 3. A 7-year-old boy had left Wilms tumor with pulmonary metastases. (A) Enhanced CT of the abdomen showing a huge mass in the left renal fossa. (B) Chest CT examination demonstrated right lung metastasis and left pleural effusion. (C) Pulmonary metastases disappeared after preoperative TACE and systemic chemotherapy.

tumor spillage (33). Primary nephrectomy for Wilms tumor diameter greater than or equal to 10 cm was also associated with an increased risk of surgical complications (4). The patients with stage III disease, diffuse AH, and tumor spillage during surgery also observed the relative risks of local recurrence and poor survival (34). Patients with diffuse anaplastic Wilms tumor, particularly stages III and IV, continue to have poor outcomes and may benefit from new treatment strategies (3, 30, 35).

The SIOP studies largely focus on the issue of preoperative chemotherapy to facilitate surgery of a shrunken tumor and to treat metastasis as early as possible. The duration of conventional preoperative chemotherapy is 4 or 8 weeks (2, 36–38). Preoperative chemotherapy is also used for the treatment of "inoperable" or "unresectable" Wilms tumor by NWTS and



Figure 4. Pathologic examination of the surgical specimen. (A) Macroscopic examination of the specimen found massive necrosis and thickening of the tumor fibrous capsule. (B) Microscopically, extensive and homogenous necrosis was found in the tumor (hematoxylin and eosin stain, ×40).

United Kingdom Children's Cancer Study group in recent years (39, 40). However, some patients did not respond to conventional preoperative chemotherapy and died before the excision of the primary tumor. Ritchey et al. (4) reviewed 131 children in NWTS-3 who had received preoperative chemotherapy for unresectable tumors or were judged inoperable by imaging. Thirteen of them did not respond to chemotherapy, but the disease progressed. Eight children died before the removal of the primary tumor. Ora et al. (41) reported tumor progression during preoperative chemotherapy in 57 of 1,090 patients (5%) with localized Wilms tumors. Patients whose tumors do increase in size have poorer EFS and OS rates independently of stage distribution and histopathologic risk group.

Actinomycin D, vincristine, doxorubicin, ifosfamide, etoposide, and carboplatin are the commonly used drugs for more advanced and recalcitrant Wilms tumor (42). Actinomycin D and vincristine were used by the SIOP studies for patients with unilateral localized Wilms tumor and stage II or III with low-risk (LR) or intermediate-risk (IR) histology (43). Doxorubicin, cyclophosphamide, etoposide, and carboplatin are now the standard part of the treatment protocols for more advanced and recalcitrant cases (42, 44). NWTS-5 regimen for patients with stages II to IV diffuse AH Wilms tumor was treated with vincristine, doxorubicin, cyclophosphamide, and etoposide (3). Children's Cancer Group used ifosfamide, carboplatin, and etoposide in children with poor-risk relapsed Wilms tumor (45). Cisplatin is a well-known chemotherapeutic drug and is effective against various types of cancers. Combination therapies of cisplatin with other drugs have been highly considered to overcome drug resistance and to reduce toxicity (46). It is ideal to administer cisplatin intra-arterially because it has a very high affinity for tissue protein, which leads to the effective binding of cisplatin to the tumor tissue during its first pass (47, 48).

Almgard et al. (5) performed first embolization for the treatment of renal adenocarcinoma in 1973. Since then, renal artery embolization is increasingly being used for the treatment of advanced or unresectable renal tumors in adults. Clinical studies have shown that preoperative renal embolization significantly reduces blood loss during nephrectomy, especially in large hypervascular tumors (6–20). Renal artery embolization also has been used in the management of Wilms tumor in children (49–54). Although the value of preoperative embolization of Wilms tumor has been documented by many authors, opinions on its indication have differed, and its use in practice has remained relatively limited.

TACE has significantly contributed to the evolution of interventional radiology. TACE may effectively deliver highly concentrated doses of chemotherapy to the tumor bed. The merits of renal chemoembolization are based on the concept that the blood supply to tumor only comes from the renal artery. The anticancer drug and embolizing material are injected into the tumor-feeding artery, increasing the effect of the chemotherapy agents in the ischemic





Figure 5. Event-free survival and overall survival rate by Kaplan–Meier estimates. (A) EFS, 92.7% (95% CI: 85.8–99.6%). (B) OS 94.5% (95% CI: 88.5–100%). EFS, event-free survival; OS, overall survival. (All the images in this chapter have not published previously and do not violate the copyright of the original publisher.)

tissue and reducing the risk of bleeding during surgery. In patients with extended renal carcinoma, survival was significantly higher after chemoembolization than after standard embolization of the renal artery (55–58). Animal experiments revealed that renal arterial chemoembolization can maintain high local concentrations of the anticancer drug, while maintaining low blood levels of the anticancer drug (59, 60).

We performed preoperative TACE for the treatment of advanced Wilms tumors since 1995 (21-24). The benefits of TACE for the treatment of advanced Wilms tumor are based on the concept that the anticancer drugs were directly injected into the tumor-feeding artery, increasing the effect of the chemotherapy agent within the tumor, while avoiding concomitant systemic toxicity. In our previous studies, we found preoperative chemoembolization combined with systemic chemotherapy for the treatment of advanced Wilms tumor showed a higher response rate than TACE alone (25). We used neoadjuvant TACE and systemic chemotherapy as for the treatment of unresectable, metastatic, or diffuse AH Wilms tumor since 2003. Our regimen is a platinum-based combination chemotherapy. The scientific rationale for the use of combination chemotherapy is to overcome drug resistance to individual agents. In addition to providing a broader range of coverage against naturally resistant tumor cells, combined chemotherapy may also prevent or delay the development of acquired resistance in initially responsive tumors and provide additive or synergistic cytotoxic effects. This regimen is also a multimodal combination therapy consisting of localized arterial chemotherapy, arterial embolization, and intravenous chemotherapy. This combination can induce more massive necrosis of tumor, eliminate the distant metastases, improve the complete resection rate of tumor, and achieve excellent survival rate.

Conclusion

The preliminary results in this phase II study suggest that the use of neoadjuvant TACE and systemic chemotherapy is well tolerated and may provide a promising choice in the treatment of unresectable, metastatic, or diffuse anaplastic Wilms tumor in children.

There were limitations in this study. Because of the small number of cases in this group and the short observation period, long-term effects warrant further investigation.

Conflict of Interest

The authors declare that they have no conflicts of interest with respect to research, authorship and/or publication of this book chapter.

Acknowledgments

This study was approved by our institutional review board; the need for informed consent for publication of data was waived. The authors thank all the children and their parents for allowing us to publish the data collected during this research project. The authors also thank the staff of the Department of Pediatric Surgery, Radiology, and Pathology for their cooperation.

References

 D'Angio GJ. The National Wilms Tumor Study: a 40 year perspective. Lifetime Data Anal. 2007;13:463–70.

http://dx.doi.org/10.1007/s10985-007-9062-0

- Grundy RG, Hutton C, Middleton H, Imeson J, Pritchard J, Kelsey A, et al. Outcome of patients with stage III or inoperable WT treated on the second United Kingdom WT protocol (UKWT2); a United Kingdom Children's Cancer Study Group (UKCCSG) study. Pediatr Blood Cancer. 2004;42:311–9. http://dx.doi.org/10.1002/pbc.10477
- Dome JS, Cotton CA, Perlman EJ, Breslow NE, Kalapurakal JA, Ritchey ML, et al. Treatment of anaplastic histology Wilms' tumor: results from the fifth National Wilms' Tumor Study. J Clin Oncol. 2006;24:2352–8. http://dx.doi.org/10.1200/JCO.2005.04.7852
- 4. Ritchey ML, Shamberger RC, Haase G, Horwitz J, Bergemann T, Breslow NE. Surgical complications after primary nephrectomy for Wilms' tumor: report from the National Wilms' Tumor Study Group. J Am Coll Surg. 2001;192:63–8.
- 5. Almgard LE, Fernstrom I, Haverling M, Ljungqvist A. Treatment of renal adenocarcinoma by embolic occlusion of the renal circulation. Br J Urol. 1973;45:474–9.

- 6. Almgard LE, Slezak P. Treatment of renal adenocarcinoma by embolization: a follow-up of 38 cases. Eur Urol. 1977;3:279–81.
- Bakal CW, Cynamon J, Lakritz PS, Sprayregen S. Value of preoperative renal artery embolization in reducing blood transfusion requirements during nephrectomy for renal cell carcinoma. J Vasc Interv Radiol. 1993;4:727–31.
- 8. Saitoh H, Hayakawa K, Nishimura K, Kubo S, Hida S. Long-term results of ethanol embolization of renal cell carcinoma. Radiat Med. 1997;15:99–102.
- 9. Kalman D, Varenhorst E. The role of arterial embolization in renal cell carcinoma. Scand J Urol Nephrol. 1999;33:162–70.
- 10. Somani BK, Nabi G, Thorpe P, Hussey J, Mc Clinton S. Therapeutic transarterial embolisation in the management of benign and malignant renal conditions. Surgeon. 2006;4:348–52.
- 11. Jaganjac S, Sarajlić-Durović V, Duherić A, Herceglija E, Bulja D, Lincender L. Percutaneous transarterial kidney embolization. Med Arh. 2007;61:233–5.
- 12. Ginat DT, Saad WE, Turba UC. Transcatheter renal artery embolization: clinical applications and techniques. Tech Vasc Interv Radiol. 2009;12:224–39. http://dx.doi.org/10.1053/j.tvir.2009.097
- Subramanian VS, Stephenson AJ, Goldfarb DA, Fergany AF, Novick AC, Krishnamurthi V. Utility of preoperative renal artery embolization for management of renal tumors with inferior vena caval thrombi. Urology. 2009;74:154–9. http://dx.doi.org/10.1016/j.urology.2008.12.084
- Ginat DT, Saad WE, Turba UC. Transcatheter renal artery embolization for management of renal and adrenal tumors. Tech Vasc Interv Radiol. 2010;13:75–88. http://dx.doi.org/10.1053/j.tvir.2010.02.003
- Loffroy R, Rao P, Ota S, Geschwind JF. Renal artery embolisation prior to radical nephrectomy for renal cell carcinoma: when, how and why? Br J Radiol. 2010;83:630.

http://dx.doi.org/10.1259/bjr/34309294

- Rodríguez Carvajal R, Orgaz A, Leal JI, Peinado FJ, Vicente S, Gil J, et al. Renal embolization and nephrectomy in a single surgical act in high-risk renal tumor pathology. Ann Vasc Surg. 2011;25:222–8. http://dx.doi.org/10.1016/j.avsg.2010.03.037
- 17. Sauk S, Zuckerman DA. Renal artery embolization. Semin Intervent Radiol. 2011;28:396–406.

http://dx.doi.org/10.1055/s-0031-1296082

 Provenza G, Sparagna A, Cunsolo GV, Tierno SM, Centanini F, Bellotti C, et al. Renal artery embolization in a gross kidney neoplasm. Case report. G Chir. 2013;34:263–6.

 Zargar H, Addison B, McCall J, Bartlett A, Buckley B, Rice M. Renal artery embolization prior to nephrectomy for locally advanced renal cell carcinoma. ANZ J Surg. 2014;84:564–7.

http://dx.doi.org/10.1111/ans.12545

- 20. Muller A, Rouvière O. Renal artery embolization-indications, technical approaches and outcomes. Nat Rev Nephrol. 2015;11:288–301. http://dx.doi.org/10.1038/nrneph.2014.231
- 21. Li MJ, Tang DX, Zhou YB, Tang HF. Preoperative interventional therapy for children with advanced Wilms' tumor. Chin J Pediatr Surg. 2001;22:10–13 (in Chinese).
- 22. Li MJ, Zhou YB, Shen LG. Prospective study of preoperative transcatheter arterial chemo-embolization for Wilms' tumor. Zhejiang Da Xue Xue Bao Yi Xue Ban. 2003;32:69–71 (in Chinese).
- 23. Li MJ, Huang Y, Tang DX, Zhou YB, Tang HF, Liang JF. Treatment of advanced Wilms' tumor. Zhonghua Zhong Liu Za Zhi. 2006;28:791–5 (in Chinese).
- 24. Liu WG, Gu WZ, Zhou YB, Tang HF, Li MJ, Ma WX. The prognostic relevance of preoperative transcatheter arterial chemoembolization (TACE) and PCNA/VEGF expression in patients with Wilms' tumour. Eur J Clin Invest. 2008;38:931–8. http://dx.doi.org/10.1111/j.1365-2362.2008.02043.x
- 25. Li MJ, Zhou YB, Huang Y, Tang DX, Xu S, Wu DH, et al. A retrospective study of the preoperative treatment of advanced Wilms tumor in children with chemotherapy versus transcatheter arterial chemoembolization alone or combined with short-term systemic chemotherapy. J Vasc Interv Radiol. 2011;22:279–86. http://dx.doi.org/10.1016/j.jvir.2010.11.025
- 26. Munck JN, Riggi M, Rougier P, Chabot GG, Ramirez LH, Zhao Z, et al. Pharmacokinetic and pharmacodynamic advantages of pirarubicin over adriamycin after intraarterial hepatic administration in the rabbit VX2 tumor model. Cancer Res. 1993;53:1550–4.
- 27. Bayssas M, Gouveia J, de Vassal F, Misset JL, Schwarzenberg L, Ribaud P, et al. Vindesine: a new vinca alkaloid. Recent Results Cancer Res. 1980;74:91–7.
- 28. Eisenhauera EA, Therasseb P, Bogaertsc J, Schwartzd LH, Sargente D, Fordf R, et al. New response evaluation criteria in solid tumours: revised RECIST guideline (version 1.1). Eur J Cancer. 2009;45:228–47.
- 29. Trotti A, Colevas A, Setser A, Rusch V, Jaques D, Budach V, et al. CTCAE v3.0: development of a comprehensive grading system for the adverse effects of cancer treatment. Semin Radiat Oncol. 2003;13:176–81.

http://dx.doi.org/10.1016/S1053-4296(03)00031-6

 D'Angio GJ, Breslow N, Beckwith JB, Evans A, Baum H, deLorimier A, et al. Treatment of Wilms' tumor. Results of the Third National Wilms' Tumor Study. Cancer. 1989;64:349–60.

- 31. Bai JW. Wilms' tumor. In: Zhang JZ (ed.). Modern Pediatric Oncologic Surgery. Beijing: Beijing Science Publishing House, 2003;245–54.
- Ritchey ML, Pringle KC, Breslow NE, Takashima J, Moksness J, Zuppan CW, et al. Management and outcome of inoperable Wilms' tumor. A report of National Wilms Tumor Study-3. Ann Surg. 1994;220:683–90.
- 33. Gow KW, Barnhart DC, Hamilton TE, Kandel JJ, Chen MK, Ferrer FA, et al. Primary nephrectomy and intraoperative tumor spill: report from the Children's Oncology Group (COG) renal tumors committee. J Pediatr Surg. 2013;48:34–8. http://dx.doi.org/10.1016/j.jpedsurg.2012.10.015
- 34. Shamberger RC, Guthrie KA, Ritchey ML, Haase GM, Takashima J, Beckwith JB, et al. Surgery-related factors and local recurrence of Wilms tumor in National Wilms Tumor Study 4. Ann Surg. 1999;229:292–7.
- 35. Green DM, Beckwith JB, Breslow NE, Faria P, Moksness J, Finklestein JZ, et al. Treatment of children with stages II to IV anaplastic Wilms' tumor: a report from the National Wilms' Tumor Study Group. J Clin Oncol. 1994;12:2126–31.
- Graf N, Tournade MF, de Kraker J. The role of preoperative chemotherapy in the management of Wilms' tumor. The SIOP studies. International Society of Pediatric Oncology. Urol Clin North Am. 2000;27:443–54.
- 37. Tournade MF, Com-Nougué C, de Kraker J, Ludwig R, Rey A, Burgers JM, et al. Optimal duration of preoperative therapy in unilateral and nonmetastatic Wilms' tumor in children older than 6 months: results of the Ninth International Society of Pediatric Oncology Wilms' Tumor Trial and Study. J Clin Oncol. 2001;19:488–500.
- 38. Reinhard H, Semler O, Burger D, Bode U, Flentje M, Göbel U, et al. Results of the SIOP 93-01/GPOH trial and study for the treatment of patients with unilateral nonmetastatic Wilms' Tumor. KlinPediatr. 2004;216:132-40. http://dx.doi.org/10.1055/s-2004-822625
- Ritchey ML. The role of preoperative chemotherapy for Wilms' tumor: the NWTSG perspective. National Wilms' Tumor Study Group. Semin Urol Oncol. 1999;17:21–7.
- 40. Mitchell C, Pritchard-Jones K, Shannon R, Hutton C, Stevens S, Machin D, et al. Immediate nephrectomy versus preoperative chemotherapy in the management of nonmetastatic Wilms' tumour: results of a randomised trial (UKW3) by the UK Children's Cancer Study Group. Eur J Cancer. 2006;42:2554–62. http://dx.doi.org/10.1016/j.ejca.2006.05.026
- Ora I, van Tinteren H, Bergeron C, de Kraker J. Progression of localized Wilms' tumour during preoperative chemotherapy is an independent prognostic factor: a report from the SIOP 93-01 nephroblastoma trial and study. Eur J Cancer. 2007;43:131–6.

http://dx.doi.org/10.1016/j.ejca.2006.08.033

- Gleason JM, Lorenzo AJ, Bowlin PR, Koyle MA. Innovations in the management of Wilms' tumor. Ther Adv Urol. 2014;6:165–76. http://dx.doi.org/10.1177/1756287214528023
- 43. Graf N, van Tinteren H, Bergeron C, Pein F, van den Heuvel-Eibrink MM, Sandstedt B, et al. Characteristics and outcome of stage II and III non-anaplastic Wilms' tumour treated according to the SIOP trial and study 93-01. Eur J Cancer. 2012;48:3240-8.

http://dx.doi.org/10.1016/j.ejca.2012.06.007

- Hamilton TE, Shamberger RC. Wilms tumor: recent advances in clinical care and biology. Semin Pediatr Surg. 2012; 21:15–20. http://dx.doi.org/10.1053/j.sempedsurg.2011.10.002
- 45. Abu-Ghosh AM, Krailo MD, Goldman SC, Slack RS, Davenport V, Morris E, et al. Ifosfamide, carboplatin and etoposide in children with poor-risk relapsed Wilms' tumor: a Children's Cancer Group report. Ann Oncol. 2002;13:460–9.
- 46. Dasari S, Tchounwou PB. Cisplatin in cancer therapy: molecular mechanisms of action. Eur J Pharmacol. 2014;740:364–78. http://dx.doi.org/10.1016/j.ejphar.2014.07.025
- 47. Chen HS, Gross JF. Intra-arterial infusion of anticancer drugs: theoretic aspects of drug delivery and review of responses. Cancer Treat Rep. 1980;64:31–40.
- Collins JM. Pharmacologic rationale for regional drug delivery. J Clin Oncol. 1984;2:498–504.
- 49. Harrison MR, de Lorimier AA, Boswell WO. Preoperative angiographic embolization for large hemorrhagic Wilms' tumor. J Pediatr Surg. 1978;13:757–8.
- 50. Danis RK, Wolverson MK, Graviss ER, O'Connor DM, Joyce PF, Cradock TV. Preoperative embolization of Wilms' tumors. Am J Dis Child. 1979;133:503–6.
- 51. Gapchenko AS, Kononenko NG, Iugrinov OG, Galakhin KA, Siniuta BF, Shevchenko NV. Chemoembolization of blood vessels in the complex treatment of children with nephroblastoma. Klin Khir. 1992;5:18–21.
- 52. Zupancic B, Bradic I, Batinica S, Radanović B, Simunić S, Zupancić V, et al. Our 10-year experience with embolized Wilms' tumor. Eur J Pediatr Surg. 1995;5:88–91.
- Chitnis M, Chowdhary SK, Lazarus C. Preoperative angioembolisation for lifethreatening haemorrhage from Wilms' tumour: a case report. Pediatr Surg Int. 2004;20:290–1.

http://dx.doi.org/10.1007/s00383-003-1128-9

54. Smith NP, Jesudason EC, McDowell HP, Rowlands P, Ashworth M, Losty PD. Emergent embolisation to control severe haematuria in Wilms' tumour. Pediatr Surg Int. 2005;21:313–5.

http://dx.doi.org/10.1007/s00383-005-1402-0

- 55. Kato T, Nemoto R, Mori H, Takahashi M, Tamakawa Y. Transcatheter arterial chemoembolization of renal cell carcinoma with microencapsulated mitomycin C. J Urol. 1981;125:19–24.
- 56. Kato T, Sato K, Abe R, Moriyama M. The role of embolization/chemoembolization in the treatment of renal cell carcinoma. Prog Clin Biol Res. 1989;303:697–705.
- 57. Granov AM, Gorelov AI, Gershanovich ML, Karelin MI, Vorob'ev AV, Filov VA, et al. Results of endovascular interventions (embolization and chemoembolization) in the treatment of operable and extensive kidney cancer. Vopr Onkol. 1998;44:711–4.
- 58. Kónya A, Choi BG, Van Pelt CS, Wright KC. Transcatheter arterial embolization of renal VX-2 carcinoma: ethiodol-ethanol capillary embolization combined with carboplatin. Korean J Radiol. 2007;8:136–47.
- 59. Fujiwara K, Hayakawa K, Nagata Y, Hiraoka M, Nakamura T, Shimizu Y, et al. Experimental embolization of rabbit renal arteries to compare the effects of poly L-lactic acid microspheres with and without epirubicin release against intraarterial injection of epirubicin. Cardio Vasc Intervent Radiol. 2000;23:218–23.
- Kurzidem M, Seidensticker P, Rassweiler J. Renal chemoembolization with mitomycin c/Ethibloc: pharmacokinetics and efficacy in an animal model. J Endourol. 2002;16:515–8.

http://dx.doi.org/10.1089/089277902760367485

Chapter 8

Dendritic Cell-Based Cancer Immunotherapy Targeting Wilms' Tumor 1 for Pediatric Cancer

Shigetaka Shimodaira,¹ Koichi Hirabayashi,^{1,2} Ryu Yanagisawa,^{1,2} Yumiko Higuchi,¹ Kenji Sano,³ Tomonobu Koizumi⁴

¹Center for Advanced Cell Therapy, ²Department of Pediatrics, ³Department of Laboratory Medicine, and ⁴Shinshu Cancer Center, Shinshu University Hospital, Matsumoto, Japan

Author for correspondence: Shigetaka Shimodaira, MD, PhD, Professor (Director) of Advanced Center for Cell Therapy, Shinshu University Hospital, 3-1-1 Asahi, Matsumoto, Nagano 390-8621, Japan. Email: shimodai@shinshu-u.ac.jp

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch8

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

The treatment of advanced pediatric cancers that have metastasized to distant organs remains difficult. Investigations evaluating the potential treatment of these cancers using therapeutic vaccination with an active dendritic cell (DC)-based immunotherapy are also being conducted. This method induces an efficient immune response by the acquired immune system against tumor-associated antigens. Cancer vaccination therapies have been prepared using autologous monocyte-derived mature DCs exposed to granulocyte-macrophage colony-stimulating factor

In: *Wilms Tumor*. Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Shimodaira et al.

and interleukin-4, which are the molecules principally attributed to the presence of tumorassociated antigens. Wilms' tumor 1 (WT1), an attractive target antigen that has been widely detected in cancers including sarcoma and leukemia, has been shown to be the most potent tumor-associated antigen. DC-based immunotherapy targeting WT1 may have a potentially strong therapeutic activity against cancers. DC vaccines primed with human leukocyte antigen (HLA) class I-/II-restricted WT1 peptides (WT1-DC) are a feasible option. A 6-year-old girl with neuroblastoma and a 14-year-old girl with WT received autologous DC vaccination pulsed with a modified WT1 peptide compatible with HLA-A*24:02. The patients received 20 and 25 vaccines, respectively, and experienced no adverse effects aside from a grade 2 skin reaction at the injection site and a fever with tolerable elevation. WT1tetramer analysis after vaccination detected WT1-specific immune responses. This treatment strategy may be safe, tolerable, and even feasible for all patients who are refractory to treatment and for pediatric patients who have relapsed with neoplasms.

Key words: Cancer vaccination; Dendritic cells; Pediatric neoplasm; Tetramer analysis

Introduction

Despite significant advances in cancer therapeutics, including the introduction of immune checkpoint inhibitors (1–6), it remains extremely difficult to treat advanced cancers affecting multiple organs and involving distant metastases. Ralph Steinman, the Nobel Prize-winning scientist who discovered dendritic cells (DCs) in 1973 (7), experimentally immunized himself with DC vaccination therapy against his pancreatic cancer and survived for 4.5 years. The manufacturing technology used in the production of antigen-presenting cell (APC)-based immunotherapies involving active DCs, the immune system's most potent APCs, is currently under development as a means of therapeutic vaccination against cancer (8). DC-based immunotherapy not only appears to be associated with few adverse reactions but also has limited clinical effectiveness when assessed using conventional evaluation methods such as response evaluation criteria in solid tumors (9, 10). Due to a slow clinical response, a low response rate, and few differences in patient median survival time (MST), long-term cancer immunity results in a delayed separation of treated and untreated patient survival curves, with an eventual treatment advantage in prolonged overall survival (OS) (11, 12).

An *ex vivo* technique is being developed for DC-based cancer vaccination to promote strong induction of T cells against tumor antigens. Oil adjuvants for peptide vaccines act by locally accelerating the activation of lymphocytes (13). However, DCs have the potential antigen bio-activity and may be used as a suitable adjuvant (14–16). Human leukocyte antigen (HLA) molecules harbor cancer antigen peptides that promote DCs binding with receptors on CD8⁺ killer and CD4⁺ helper T cells, leading to an immune response against cancers (Figure 1). In contrast,

DC-based cancer immunotherapy for pediatric cancer



Figure 1. Dendritic cells and other immune cells in the cancer environment. Human leukocyte antigen (HLA) molecules harbor cancer antigen peptides, which induce DC binding with receptors on CD8⁺ killer and CD4⁺ helper T cells, leading to anticancer immune responses. In contrast, immune suppressor cells, such as regulatory T cells, tolerogenic DCs, and myeloid-derived suppressor cells, suppress autoreactive and cancer-derived mechanisms. (Original figure by Shimodaira S.)

immune suppressor cells, such as regulatory T cells, tolerogenic DCs, and myeloid-derived suppressor cells, suppress autoreactive and cancer-derived mechanisms (17–21). Immune suppressive factors are also stimulated by the presence of cancer cells. These factors are shown in Figure 1 and include transforming growth factor- β , interleukin (IL)-10, vascular endothelial growth factor, prostaglandin E2 (PGE2), and programmed death-ligand 1 (PD-L1) (22). The efficacy of DC vaccination can likely be attributed to the inhibition of these immune suppressors.

DCs are generated from peripheral monocytes following exposure to granulocyte-macrophage colony-stimulating factor (GM-CSF) and IL-4. DCs expressing tumor-specific antigens have been used in active cancer immunotherapies (23, 24). The most common approach to DC vaccination is the preparation of autologous, mature, monocyte-derived DCs *ex vivo* with consequent, homogeneous, and functional DC generation. Cancer vaccination therapies are principally attributed to the presence of tumor-associated antigens using peptide, protein, tumor lysate, and RNA (25–29). Sipuleucel-T (Provenge[®]) is a US Food and Drug Administration-approved autologous DC-based immunotherapy for men with metastatic hormonerefractory prostate cancer, which provides a new treatment option for patients with this type

Shimodaira et al.

of cancer. Sipuleucel-T is manufactured by exposing an individual patient's affected blood cells to a recombinant fusion protein composed of a prostatic acid phosphatase fused to GM-CSF, enhancing immune cell activity against this type of cancer. The patient's own DC product is administered intravenously as part of a three-dose schedule, with approximately 2-week intervals between each dose. This regimen yields a survival benefit of 4.1 months in patients with hormone-resistant prostate cancer (30). According to the requirement for antigens, such as Wilms' tumor 1 (WT1), mucin 1, cell surface associated, human epidermal growth factor receptor 2, carcinoembryonic antigen, survivin, and prostate-specific antigen, WT1 was identified as the most potent cancer-associated antigen. WT1 has confirmed immunological and clinical effectiveness with respect to therapeutic functions, immunogenicity, specificity, and oncogenicity (31). HLA-restricted WT1 peptides were identified as being compatible with HLA-A*02:01- or HLA-A*02:06-restricted (126-134: RMFPNAPYL) and class II compatible with HLA-DRB1*04:05 (332-347: KRYFKLSHLQMHSRKH). The WT1 peptide was restricted to HLA-A*24:02 and modified WT1235-243 peptide (CYTWNQMNL). Methionine (M), the second amino acid, was replaced with tyrosine (Y), which can induce cytotoxic T cells (CTLs) to be more effective than the wild-type peptide (32-35). The percentage results for HLA genotyping were as follows: genotypes of HLA-A*24:02 (60%), A*02:01 (20%), and A*02:06 (15%) and HLA class II genotypes of HLA-DRB1*04:05, DRB1*08:03, DRB1*15:01, DRB1*15:02, DPB1*05:01, or DPB1*09:01 (90%). Phase I clinical trials have been conducted with this regimen for various types of solid tumors and hematological malignancies (36-38). DC vaccines primed with HLA class I-/II-restricted WT1 peptides (WT1-DC) have been determined to be safe and feasible, with few adverse reactions reported by patients with advanced cancers, including lung, breast, stomach, biliary tract, pancreas, ovary, and even high-grade glioma (39-47). Clinical studies have indicated that the efficacy of DC vaccination may be enhanced by the off-target effects of chemotherapeutic drugs (39-44, 48) and chemoradiotherapy (47, 49, 50), suggesting a survival benefit in some patients. Different combinations with adjuvant chemotherapy and/or radiotherapy have been investigated, along with the periods required for adaptation. The development of combination therapy regimens, which could potentially include immune checkpoint inhibitors, should improve the outcomes of personalized therapy for patients with cancer (51). However, DC vaccination has been only rarely utilized to treat pediatric patients. There are a few reports describing its use in acute leukemia after allogeneic hematopoietic stem transplantation (52, 53). This article focuses on a pilot study evaluating autologous DC vaccination targeting WT1 in pediatric patients with neuroblastoma or WT.

Manufacture of a DC vaccine

Mature DCs (mDCs) were generated under Good Gene, Cell and Tissue Manufacturing Practice, conditions according to the "The Act on the Safety of Regenerative Medicine"

introduced in Japan on November 25, 2014 (54). Mononuclear cell-rich fractions (165 ml) were isolated from 4,000 ml of the patient's blood through apheresis using a COM.TEC® cell separator (Fresenius Kabi Japan K.K., Tokyo, Japan). Immature DCs were generated by culturing adherent cells in AIM-V® medium (Gibco, Gaithersburg, MD) containing GM-CSF (50 ng/ml; Gentaur, Brussels, Belgium) and IL-4 (50 ng/ml; R&D Systems Inc., Minneapolis, MN) in a CO₂ incubator equipped with a Cell Processing Isolator (H_2O_2 -sterilizing system, Panasonic Corporation, Osaka, Japan) at the Shinshu University Hospital Cell Processing Center. After 5 days of culture, immature DCs were differentiated into mDCs by stimulation with OK-432 (10 µg/ml of streptococcal preparation; Chugai Pharmaceutical Co., Ltd, Tokyo, Japan) and PGE2 (50 ng/ml; Daiichi Fine Chemical Co. Ltd., Toyama, Japan) for 24 h (55). The resulting mDCs were cryopreserved at -152°C or in the gas layer within a liquid nitrogen tank until the day of administration. Cell culture supernatants were collected for sterility testing at the time of mDC freezing. For each vaccination, an aliquot of frozen mDCs was thawed immediately prior to clinical use and primed with 100 µg/ml of good manufacturing practice-grade WT1 peptide (NeoMPS Inc., San Diego, CA) containing 1-2 KE of OK-432. WT1 peptides contained HLA-A*02:01- or A*02:06-restricted peptides (126-134: RMFPNAPYL), HLA-A*24:02-restricted modified WT1 peptides (CYTWNQML, residue 235-243), and/or class II peptide (332-347: KRYFKLSHLQMHSRKH) compatible with DRB1*04:05, DRB1*08:03, DRB1*15:01, DRB1*15:02, DPB1*05:01, or DPB1*09:01 (35, 43). One course of seven biweekly sessions was performed with $1-3 \times 10^7$ DCs with 1-2 KE of OK-432 intradermally injected at bilateral axillar and inguinal areas per session. For pediatric cases, the dose of adjuvant OK-432 was modified as 0.25-1.0 KE, and intradermal injection sits were selected at two points in either bilateral axillar or inguinal areas per session.

DC vaccine release criteria

The antigenic profiles of mDCs were determined using flow cytometry. mDCs were defined as CD11c⁺, CD14⁻, HLA⁻DR⁺, HLA⁻ABC⁺, CD80⁺, CD83⁺, CD86⁺, CD40⁺, and CCR7⁺ cells (55). The criteria for DC vaccine administration were as follows: purity defined as >90% proportion of CD11c⁺ CD14⁻ CD86⁺ HLA⁻DR⁺ >90% cells, >80% viability, mDC phenotype, negative for bacterial and fungal infection after 14 days, presence of endotoxin \leq 0.05 EU/ml, and negative for mycoplasma (55).

DC vaccine study

Application and conditions for DC vaccine therapy

- 1. Adjuvant therapy after surgical resection or high risk of disease relapse
- 2. De novo cancer at an advanced stage or recurrent cancer after standard therapies

Shimodaira et al.

Indication for DC vaccine therapy and eligibility

- 1. Performance status: 0/1
- No organ function abnormalities, no infectious diseases, no blood abnormalities, no bleeding tendency
- 3. Neither cardiovascular diseases nor respiratory disorders that would prevent blood apheresis
- 4. Tolerable to chemotherapy and radiotherapy as standard cancer treatments
- 5. Within 6 months of cancer diagnosis or recurrence, with cancer sensitivity to chemotherapy

Exclusion criteria

- 1. Requiring platelet or red blood cell transfusion or albumin infusion
- 2. Disseminated intravascular coagulation syndrome and deep vein thrombosis
- 3. An infectious disease such as viral hepatitis (following the standard of the Japanese Red Cross Blood Center)
- 4. Allergy to penicillin or OK-432
- 5. Steroid hormone therapy continuously administered for diseases other than the prevention of temporal chemotherapeutic drug allergy
- 6. Difficulty in arm vessel blood access for apheresis
- 7. A presumed length of survival period that would prevent seven sessions of one course at the outpatient clinic
- 8. No informed consent due to cancer
- 9. Inability to understand the risk and benefit of the DC vaccine therapy
- 10. Opposition to DC vaccine therapy
- 11. Pregnant or nursing women
- 12. Physician judgment that a patient is inappropriate for treatment

Evaluation of safety and effectiveness

- In terms of safety evaluation, we evaluated (i) any allergic reaction after the intradermal injection of the DC vaccine (presence of reduced blood pressure, tachycardia, breathing difficulties, or rash) and (ii) local reactions, fever onset, nausea, vomiting, diarrhea, loss of appetite, ulcer of the mucosa, central nervous system damage, anemia, reduced white blood cells, reduced platelets, abnormal kidney function, and abnormal liver function either during or after the completion of treatment.
- 2. We assessed the cancerous lesions during the treatment course using various imaging techniques, such as computerized tomography, magnetic resonance imaging, and positron emission tomography, approximately 4 weeks after the completion of DC

vaccination. The DC vaccination study was conducted at Shinshu University Hospital and was approved by the Ethics Committee of Shinshu University School of Medicine (Approval Number 1199, December 2, 2008; Approval Number 2704, April 8, 2014).

Case report

Case 1: Neuroblastoma

A 6-year-old girl presented with adrenal gland neuroblastoma in December 2008 at the age of 4. Bone metastasis and bone marrow involvement were detected, resulting in a diagnosis of stage IV disease according to International Neuroblastoma Staging System (56). The patient underwent systemic chemotherapy according to the protocol of the Japanese Neuroblastoma Study Group, followed by surgical resection of the primary adrenal gland neuroblastoma. After intensive chemotherapy was administered in combination with thiotepa and melphalan, the patient subsequently underwent autologous hematopoietic stem cell transplantation (HSCT). The patient received 20 Gy of radiation therapy to the primary right adrenal gland area after recovery from myeloablation and achieved complete disease remission in December 2009. However, at the age of 6, she developed bone marrow relapse in June 2010 and was admitted for DC vaccination in combination with etoposide chemotherapy. Her HLA genotype was confirmed as HLA-A*24:02 compatible with modified WT1-235 peptide. One course (seven sessions, once every 3 weeks) of DC vaccination containing modified WT1-235 peptide (a total of 7.22×10^7 DCs; mean, 1.03×10^7 DCs per session) was administered from March to July 2011. DC vaccine-related toxicities were tolerable and included grade 2 skin reactions and pain at the injection sites along with grade 1 low-grade fever within 48 h of treatment. There were no ≥grade 3 adverse effects due to DC vaccination based on Common Terminology Criteria for Adverse Events (CTCAE), version 4.0 (http://evs.nci.nih.gov/ftp1/ CTCAE/CTCAE_4.03_2010-06-14_QuickReference_5x7.pdf). The tumor markers (neuronspecific enolase, urinary vanillylmandelic acid, and homovanillic acid) were normalized in August 2011, and the magnetic resonance imaging indicated the lesion significantly reduced.

However, the increase of the recurrent tumor with multiple metastasis of bone marrow was detected by MRI and metaiodobenzylguanidine scintigraphy in November 2011. Temozolomide was added, and the DC vaccination was also continued after one course for an additional 15 sessions until October 2012. The patient died due to disease progression in August 2013. Progression-free survival and OS from diagnosis were 5 months after DC vaccination and 4 years and 8 months, respectively.

Case 2: Wilms' tumor

A 14-year-old girl presented with WT derived from her left kidney in January 2002 at the age of 4. Tumor cells were involved from inferior vena cava to the right atrium, with metastases to the

Shimodaira et al.

liver, lung, and iliac bone. She was diagnosed with stage IV disease according to the National Wilms' Tumor Study criteria, and pathological findings as favorable histology. She underwent systemic chemotherapy with the protocol of SIOP 93-01, followed by surgical resection of the primary left renal WT. After intensive chemotherapy according to the JWiTs DD-4A protocol of the Japan WT Study group was performed, the patient subsequently underwent radiation therapy targeting the primary left renal area and achieved complete disease remission in December 2002. However, she developed inferior vena cava relapse in November 2006. Although chemotherapy was started, effectiveness was few, and thereafter, localized radiotherapy was performed. She was admitted for DC vaccination at the age of 14. The HLA genotype was confirmed as HLA-A*24:02, which was compatible with the modified WT1-235 peptide. Two courses (seven sessions, once every 3 weeks) of DC vaccination containing modified WT1-235 peptide together with tumor lysate (a total of 31.46×10^7 DCs; mean, 2.25×10^7 DCs per session) were administered from November 2011 to August 2012. During DC vaccination, residual tumor cells extending from the inferior vena cava to the right atrium were surgically resected in March 2012. Despite the surgery, new tumor lesions were detected at the hepatic portal area in March 2013. DC vaccination at 1- to 3-month intervals was continued for a total of 11 additional sessions by November 2014. The patient died due to disease progression in May 2015. DC vaccination-related toxicities were tolerable and included grade 2 skin reactions and pain at the injection sites, along with grade 2 low-grade fever within 48 h of treatment. There were no \geq grade 3 adverse effects due to DC vaccination based on CTCAE ver.4.0. Disease-free survival during DC vaccination was achieved for 12 months, and OS since the time of initial diagnosis was 13 years and 4 months.

Immune monitoring with tetramer analysis

Freshly isolated peripheral blood mononuclear cells were stained with phycoerythrin (PE)conjugated human immunodeficiency virus/HLA-A*24:02 tetramer as a negative control or with PE-conjugated WT1-modified peptide/HLA-A*24:02 tetramer (MBL; Medical & Biological Laboratories Co., Ltd., Nagoya, Japan). Other stains included allophycocyanin-conjugated anti-CD3 mAb and fluorescein isothiocyanate-conjugated anti-CD8 mAb prior to the analysis by flow cytometry (BD FACSCaliburTM and BD FACSCantoTM II) in Figure 2A. The presence of WT1 antigen-specific CTLs (WT1-CTLs) was defined according to the following criteria: (i) greater than 0.02% WT1-positive cells of all CD8+ T cells analyzing 50,000–10,000 lymphocytes with no evidence of false-positive cells and (ii) WT1-positive population clustered and not diffused as described (57). WT1-CTLs were determined either by WT1-peptide/HLA-A*24:02 tetramer analysis or by interferon (IFN)- γ -producing clones used in enzyme-linked immunosorbent spot (ELISPOT) assays after DC vaccination as a proof-of-concept analysis. Before DC vaccination in both cases, WT1-CTLs were detectable at levels above 0.02% as previously defined (57). After one course of DC vaccination, the immune monitoring assay demonstrated that WT1-CTLs consisted of 0.05% and 2.05% of the CD8+ T-cell population in cases 1 and 2, respectively (Figure 2B
DC-based cancer immunotherapy for pediatric cancer



Figure 2. WT1 tetramer assay conducted during the course of DC vaccination. (A) PE-conjugated, WT1-modified peptide/HLA-A*24:02 tetramer was used to detect WT1-specific cytotoxic T cells. Before DC vaccination, WT1-CTLs were at detectable levels in both cases at a concentration of more than 0.02%. After one course of DC vaccination, the immune monitoring assay demonstrated that WT1-CTLs comprise 0.05% of CD8⁺ T cells in case 1 (B) and 2.05% of CD8⁺ T cells in case 2 (C). WT1-CTLs concentrations gradually increased after one course of DC vaccination. (Original figure by Shimodaira S.)

and C). WT1-specific T cells were markedly increased after one course during additional vaccination, contributing to the antitumor immune responses noted in our cases (Figure 2B and C).

DC vaccination technology for pediatric patients

Our preliminary study on pediatric patients has several limitations, such as the small sample size and a heterogeneous group of patients. However, DC vaccination targeting WT1 during a standard

Shimodaira et al.

therapy course may be both feasible and well tolerated for treating advanced neuroblastoma and WT. The findings also indicated that DC vaccination targeting WT1 generated immunogenicity. WT1-specific CTLs were detected at several levels at the time of initial vaccination, whereas they were distinctly increased after one course of additional vaccination and contributed to induction of the antitumor immune response in our cases. Spontaneous WT1-specific T-cell responses have been reported in acute myeloid leukemia patients (58). Therefore, one possible explanation for the immune response is that the WT1-specific T cells might have been spontaneously induced in these patients. WT1-specific T-cell responses were interestingly maximal at the last session of the course under their disease progression. It is possible that the response of WT1-specific CTLs might be merely boosted by tumor cell growth, although they were no longer able to control disease progression as described with a case of allogeneic DC vaccination targeting WT1 (53).

Case 1 with stage IV neuroblastoma, who relapsed 6 months, had the highest risk of death based on the time to first relapse (59). Our patient survived for 38 months after relapse under disease control with DC vaccination and low-dose etoposide, suggesting a survival benefit together with the maintenance of quality of life. WT1-DC vaccination would be helpful when selecting an optimal therapy for poor survival after neuroblastoma relapse. The WT in case 2 was classified as stage III very high risk for subsequent relapse among children with relapsed WTs (60). Despite the high-dose therapy, MST for very high-risk patients is less than 2 years. It is evident that an effect of the combined modality therapy including DC vaccination achieved the more than 8-year survival after the recurrence in this case, although the contribution of the WT1-DC vaccine to the patient's prolonged survival was unclear. As the number of WT1-CTLs was positively related to the WT1-specific IFN-y production according to ELISPOT assays (57), the efficacy of DC vaccination would be presumed to be dependent on the number of WT-CTLs. However, the WT1-CTL response to neuroblastoma cells might be limited due to a lack of and downregulation of HLA-class I antigens in neuroblastoma and other renal cell cancers (61-63). Despite an increase in HLA-class I expression on neuroblastoma cells following exposure to IFN- $_{\rm V}$ (61), there is a concern regarding the attenuation of WT1 antigen in tumor cells during the course of WT1-DC vaccination.

A breakthrough in DC-based vaccine technology is required to achieve further improvement in its cancer treatment efficacy. An allogeneic DC vaccination targeting WT1 may be another potential strategy for patients with relapsed leukemia after HSCT. This strategy may be safe, tolerable, and even feasible for pediatric donors and patients with relapsed leukemia after HSCT as described (52, 53). A 15-year-old girl with acute lymphoblastic leukemia received allogeneic DC vaccination pulsed with WT1 peptide after her third HSCT. The vaccines were generated from her third HSCT donor, the patient's younger 12-year-old sister, who matched with HLA-A*24:02. The patient received 14 vaccine doses with no occurrence of graft-versus-host disease and no systemic adverse effects apart from a grade 2 local skin reaction at the injection site.

DC-based cancer immunotherapy for pediatric cancer

WT1-specific immune responses were detected postvaccination by both WT1 tetramer analysis and ELISPOT assays. The patient experienced 44 months of remission after the third HSCT with DC vaccinations, whereas she had been in remission for less than 14 months between her second and third HSCT. This finding suggests that WT1-specific DC vaccination contributed to the extended period of remission following the patient's third HSCT (53). One potential approach to overcome the phenomenon of tumor cells escaping immune detection is the generation of IFN-DCs from monocytes using GM-CSF and IFN-α. Mature forms of IFN-DCs would induce CTLs together with their strong adaptive antitumor effects, with natural killer cell activity independent of HLA-class I antigen expression (64). Another approach is the administration of granulocyte colony-stimulating factor (G-CSF), resulting in the upregulation of monocyte adhesion molecules. An evaluation of the hypothesis that acceleration of acquired cancer immunity using a G-CSF-primed WT1-DC vaccine is related to the type of cancer is ongoing.

The efficacy of DC vaccination may be enhanced by the off-target effects of chemotherapeutic drugs such as gemcitabine (2',2'-difluorodeoxycytidine, GEM) and a combination of tegafur, gimeracil, and oteracil (48). It has also been reported that WT1 antigen expression in pancreatic cancer cell lines is increased by GEM treatment (65). Initial radiotherapy with additional chemotherapeutic drugs acting through their off-target effects may have accelerated the development of acquired cancer immunity and induced antigen-specific CTLs in patients receiving WT1-targeted DC vaccinations (47). Therefore, DC vaccines in combination with chemotherapy and radiotherapy should promote treatment efficacy against advanced disease. It is necessary to determine the best combinations of the DC vaccine with chemotherapeutic drugs for treating WT and pediatric neoplasms. Immune checkpoint inhibitors are rapidly being developed as chemotherapeutic agents (66). Further studies are required to evaluate whether effector memory T-cell numbers prior to vaccination and the exhaustion of markers for PD1-positive CTLs after DC vaccination influence the efficacy of DC vaccination. Targeted clinical trials could reveal the effectiveness of DC vaccine in combination with immune checkpoint inhibitors as cancer treatments in the near future. Predictive biomarkers for use with DC vaccination targeting WT1 are highly relevant to the personalized cancer therapy.

Conclusion

Our preliminary study suggests that DC vaccination targeting WT1 administered during the course of standard cancer therapies may be both feasible for and well tolerated by patients with neuroblastoma and WT. The study findings indicate that induction of acquired immunity by targeting WT1 was detected by immune monitoring with tetramer analysis during the course of DC vaccination, confirming the positive results for this proof-of-concept investigation in pediatric patients. The results also suggest that WT1-DC vaccination may prolong the survival of pediatric patients with neoplasms. In contrast, it was not clearly

Shimodaira et al.

determined whether there was an improvement in patient prognosis following WT1-DC vaccination because both patients died due to disease progression. Therefore, the efficacy and safety of DC vaccination should be determined by phase I/II prospective trials enrolling larger numbers of patients with pediatric neoplasms.

Conflicts of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgments

This study was supported by the Grant of Funding Programs: A-STEP (Adaptable & Seamless Technology Transfer Program through Target-driven R&D) from Japan Society and Technology Agency (JST).

References

 Hodi FS, O'Day SJ, McDermott DF, Weber RW, Sosman JA, Haanen JB, et al. Improved survival with ipilimumab in patients with metastatic melanoma. N Engl J Med. 2010;363:711–23.

http://dx.doi.org/10.1056/NEJMoa1003466

 Topalian SL, Hodi FS, Brahmer JR, Gettinger SN, Smith DC, McDermott DF, et al. Safety, activity, and immune correlates of anti-PD-1 antibody in cancer. N Engl J Med. 2012;366:2443–54.

http://dx.doi.org/10.1056/NEJMoa1200690

3. Hamid O, Robert C, Daud A, Hodi FS, Hwu WJ, Kefford R, et al. Safety and tumor responses with lambrolizumab (anti-PD-1) in melanoma. N Engl J Med. 2013;369:134–44.

http://dx.doi.org/10.1056/NEJMoa1305133

 Brahmer JR, Tykodi SS, Chow LQ, Hwu WJ, Topalian SL, Hwu P, et al. Safety and activity of anti-PD-L1 antibody in patients with advanced cancer. N Engl J Med. 2012;366:2455–65.

```
http://dx.doi.org/10.1056/NEJMoa1200694
```

- Wolchok JD, Kluger H, Callahan MK, Postow MA, Rizvi NA, Lesokhin AM, et al. Nivolumab plus ipilimumab in advanced melanoma. N Engl J Med. 2013;369:122–33. http://dx.doi.org/10.1056/NEJMoa1302369
- Weber JS, Kudchadkar RR, Yu B, Gallenstein D, Horak CE, Inzunza HD, et al. Safety, efficacy, and biomarkers of nivolumab with vaccine in ipilimumab-refractory or -naive melanoma. J Clin Oncol. 2013;31:4311–8. http://dx.doi.org/10.1200/JCO.2013.51.4802

DC-based cancer immunotherapy for pediatric cancer

- 7. Steinman RM, Cohn ZA. Identification of a novel cell type in peripheral lymphoid organs of mice. I. Morphology, quantitation, tissue distribution. J Exp Med. 1973;137:1142–62.
- 8. Steinman RM. Decisions about dendritic cells: past, present, and future. Annu Rev Immunol. 2012;30:1–22.

http://dx.doi.org/10.1146/annurev-immunol-100311-102839

- Rosenberg SA, Yang JC, Restifo NP. Cancer immunotherapy: moving beyond current vaccines. Nat Med. 2004;10:909–15. http://dx.doi.org/10.1038/nm1100
- 10. Ridgway D. The first 1000 dendritic cell vaccines. Cancer Invest. 2003;21:873–886.
- Hoos A. Evolution of end points for cancer immunotherapy trials. Ann Oncol. 2012;23:47–52.

http://dx.doi.org/10.1093/annonc/mds263

12. Schlom J, Arlen PM, Gulley JL. Cancer vaccines: moving beyond current paradigms. Clin Cancer Res. 2007;13:3776–782.

http://dx.doi.org/10.1158/1078-0432.CCR-07-0588

- Hailemichael Y, Dai Z, Jaffarzad N, Ye Y, Medina MA, Huang XF, et al. Persistent antigen at vaccination sites induces tumor-specific CD8⁺ T cell sequestration, dysfunction and deletion. Nat Med. 2013;19:465–72. http://dx.doi.org/10.1038/nm.3105
- 14. Banchereau J, Steinman RM. Dendritic cells and the control of immunity. Nature. 1998;392:245–52.

http://dx.doi.org/10.1038/32588

15. Steinman RM, Banchereau J. Taking dendritic cells into medicine. Nature. 2007;449:419–26.

http://dx.doi.org/10.1038/nature06175

- Figdor CG, de Vries IJ, Lesterhuis WJ, Melief CJ. Dendritic cell immunotherapy: mapping the way. Nat Med. 2004;10:475–80. http://dx.doi.org/10.1038/nm1039
- Nishikawa H, Sakaguchi S. Regulatory T cells in tumor immunity. Int J Cancer. 2010;127:759–67.

http://dx.doi.org/10.1002/ijc.25429

- Whiteside TL. Induced regulatory T cells in inhibitory microenvironments created by cancer. Expert Opin Biol Ther. 2014;14:1411–25. http://dx.doi.org/10.1002/ijc.25429
- Nagaraj S, Gabrilovich DI. Tumor escape mechanism governed by myeloid-derived suppressor cells. Cancer Res. 2008;68:2561–3. http://dx.doi.org/10.1158/0008-5472.CAN-07-6229

Shimodaira et al.

- Ezzelarab M, Thomson AW. Tolerogenic dendritic cells and their role in transplantation. Semin Immunol. 2011;23:252–63. http://dx.doi.org/10.1016/j.smim.2011.06.007
- Svajger U, Rozman P. Tolerogenic dendritic cells: molecular and cellular mechanisms in transplantation. J Leukoc Biol. 2014;95:53–69. http://dx.doi.org/10.1189/jlb.0613336
- 22. Massagué J. TGFbeta in cancer. Cell. 2008;134:215–30. http://dx.doi.org/10.1016/j.cell.2008.07.001
- O'Neill DW, Adams S, Bhardwaj N. Manipulating dendritic cell biology for the active immunotherapy of cancer. Blood. 2004;104:2235–46. http://dx.doi.org/10.1182/blood-2003-12-4392
- Nestle FO, Farkas A, Conrad C. Dendritic-cell-based therapeutic vaccination against cancer. Curr Opin Immunol. 2005;17:163–9. http://dx.doi.org/10.1016/j.coi.2005.02.003
- Copier J, Ward S, Dalgleish A. Cell based cancer vaccines: regulatory and commercial development. Vaccine. 2007;25:B35–46. http://dx.doi.org/10.1016/j.vaccine.2007.06.041
- 26. Ramakrishnan R, Antonia S, Gabrilovich DI. Combined modality immunotherapy and chemotherapy: a new perspective. Cancer Immunol Immunother. 2008;57:1523–9. http://dx.doi.org/10.1007/s00262-008-0531-4
- 27. Baxevanis CN, Perez SA, Papamichail M. Combinatorial treatments including vaccines, chemotherapy and monoclonal antibodies for cancer therapy. Cancer Immunol Immunother. 2009;58:317–24.

http://dx.doi.org/10.1007/s00262-008-0576-4

Koido S, Homma S, Takahara A, Namiki Y, Tsukinaga S, Mitobe J, et al. Current immunotherapeutic approaches in pancreatic cancer. Clin Dev Immunol. 2011;2011:267539.

http://dx.doi.org/10.1155/2011/267539

- 29. Koido S, Ohkusa T, Homma S, Namiki Y, Takakura K, Saito K, et al. Immunotherapy for colorectal cancer. World J Gastroenterol. 2013;19:8531–42. http://dx.doi.org/10.3748/wjg.v19.i46.8531
- 30. Kantoff PW, Higano CS, Shore ND, Berger ER, Small EJ, Penson DF, et al. Sipuleucel-T immunotherapy for castration-resistant prostate cancer. N Engl J Med. 2010;363:411–22.

http://dx.doi.org/10.1056/NEJMoa1001294

31. Cheever MA, Allison JP, Ferris AS, Finn OJ, Hastings BM, Hecht TT, et al. The prioritization of cancer antigens: a national cancer institute pilot project for the acceleration of translational research. Clin Cancer Res. 2009;15:5323–37. http://dx.doi.org/10.1158/1078-0432.CCR-09-0737

DC-based cancer immunotherapy for pediatric cancer

- 32. Oka Y, Elisseeva OA, Tsuboi A, Ogawa H, Tamaki H, Li H, et al. Human cytotoxic T-lymphocyte responses specific for peptides of the wild-type Wilms' tumor gene (WT1) product. Immunogenetics. 2000;51:99–107.
- Ohminami H, Yasukawa M, Fujita S. HLA class I-restricted lysis of leukemia cells by a CD8(+) cytotoxic T-lymphocyte clone specific for WT1 peptide. Blood. 2000;95:286–293.
- 34. Tsuboi A, Oka Y, Udaka K, Murakami M, Masuda T, Nakano A, et al. Enhanced induction of human WT1-specific cytotoxic T lymphocytes with a 9-mer WT1 peptide modified at HLA-A*2402-binding residues. Cancer Immunol Immunother. 2002;51:614–20.

http://dx.doi.org/10.1007/s00262-002-0328-9

- 35. Fujiki F, Oka Y, Tsuboi A, Kawakami M, Kawakatsu M, Nakajima H, et al. Identification and characterization of a WT1 (Wilms Tumor Gene) protein-derived HLA-DRB1*0405-restricted 16-mer helper peptide that promotes the induction and activation of WT1-specific cytotoxic T lymphocytes. J Immunother. 2007;30:282–93. http://dx.doi.org/10.1097/01.cji.0000211337.91513.94
- 36. Oka Y, Tsuboi A, Taguchi T, Osaki T, Kyo T, Nakajima H, et al. Induction of WT1 (Wilms' tumor gene)-specific cytotoxic T lymphocytes by WT1 peptide vaccine and the resultant cancer regression. Proc Natl Acad Sci U S A. 2004;101:13885–90. http://dx.doi.org/10.1073/pnas.0405884101
- 37. Oka Y, Tsuboi A, Murakami M, Hirai M, Tominaga N, Nakajima H, et al. Wilms tumor gene peptide-based immunotherapy for patients with overt leukemia from myelodysplastic syndrome (MDS) or MDS with myelofibrosis. Int J Hematol. 2003;78:56–61.
- 38. Nishida S, Koido S, Takeda Y, Homma S, Komita H, Takahara A, et al. Wilms tumor gene (WT1) peptide-based cancer vaccine combined with gemcitabine for patients with advanced pancreatic cancer. J Immunother. 2014;37:105–14. http://dx.doi.org/10.1097/CJI.00000000000020
- 39. Kimura Y, Tsukada J, Tomoda T, Takahashi H, Imai K, Shimamura K, et al. Clinical and immunologic evaluation of dendritic cell-based immunotherapy in combination with gemcitabine and/or S-1 in patients with advanced pancreatic carcinoma. Pancreas. 2012;41:195–205.

http://dx.doi.org/10.1097/MPA.0b013e31822398c6

 Takahashi H, Okamoto M, Shimodaira S, Tsujitani S, Nagaya M, Ishidao T, et al. DC-vaccine study group at the Japan Society of Innovative Cell Therapy (J-SICT). Impact of dendritic cell vaccines pulsed with Wilms' tumor-1 peptide antigen on the survival of patients with advanced non-small cell lung cancers. Eur J Cancer. 2013;49:852–9.

http://dx.doi.org/10.1016/j.ejca.2012.11.005

Shimodaira et al.

- 41. Kobayashi M, Sakabe T, Abe H, Tanii M, Takahashi H, Chiba A, et al. Dendritic cellbased immunotherapy targeting synthesized peptides for advanced biliary tract cancer. J Gastrointest Surg. 2013;17:1609–17. http://dx.doi.org/10.1007/s11605-013-2286-2
- 42. Kobayashi M, Shimodaira S, Nagai K, Ogasawara M, Takahashi H, Abe H, et al. Prognostic factors related to add-on dendritic cell vaccines on patients with inoperable pancreatic cancer receiving chemotherapy: a multicenter analysis. Cancer Immunol Immunother. 2014;63:797–806.

http://dx.doi.org/10.1007/s00262-014-1554-7

 Koido S, Homma S, Okamoto M, Takakura K, Mori M, Yoshizaki S, et al. Treatment with chemotherapy and dendritic cells pulsed with multiple Wilms' tumor 1 (WT1)specific MHC class I/II-restricted epitopes for pancreatic cancer. Clin Cancer Res. 2014;20:4228–39.

http://dx.doi.org/10.1158/1078-0432.CCR-14-0314

- 44. Kobayashi M, Chiba A, Izawa H, Yanagida E, Okamoto M, Shimodaira S, et al. The DC-vaccine study group at the Japan Society of Innovative Cell Therapy (J-SICT). The feasibility and clinical effects of dendritic cell-based immunotherapy targeting synthesized peptides for recurrent ovarian cancer. J Ovarian Res. 2014;7:48. http://dx.doi.org/10.1186/1757-2215-7-48
- 45. Kobayashi M, Sakabe T, Chiba A, Nakajima A, Okamoto M, Shimodaira S, et al. Therapeutic effect of intratumoral injections of dendritic cells for locally recurrent gastric cancer: a case report. World J Surg Oncol. 2014;12:390. http://dx.doi.org/10.1186/1477-7819-12-390
- 46. Sakai K, Shimodaira S, Maejima S, Udagawa N, Sano K, Higuchi Y, et al. Dendritic cell-based immunotherapy targeting Wilms' Tumor 1 (WT1) in patients with relapsed malignant glioma. J Neurosurg. 2015;123:989–97. http://dx.doi.org/10.3171/2015.1.JNS141554
- 47. Shimodaira S, Kobayashi T, Hirabayashi K, Horiuchi K, Koya T, Mizuno Y, et al. Induction of antigen-specific cytotoxic T lymphocytes by chemoradiotherapy in patients receiving Wilms' tumor 1-targetted dendritic cell vaccinations for pancreatic cancer. OMICS J Radiol. 2015;4:196.
- http://dx.doi.org/10.4172/2167-7964.1000196
 48. Duffy AG, Greten TF. Immunological off-target effects of standard treatments in gastrointestinal cancers. Ann Oncol. 2014;25:24–32.

http://dx.doi.org/10.1093/annonc/mdt349

49. Shibamoto Y, Okamoto M, Kobayashi M, Ayakawa S, Iwata H, Sugie C, et al. Immune-maximizing (IMAX) therapy for cancer: combination of dendritic cell vaccine and intensity-modulated radiation. Mol Clin Oncol. 2013;1:649–54. http://dx.doi.org/10.3892/mco.2013.108

DC-based cancer immunotherapy for pediatric cancer

- 50. Soukup K, Wang X. Radiation meets immunotherapy-a perfect match in the era of combination therapy? Int J Radiat Biol. 2015;91:299–305. http://dx.doi.org/10.3109/09553002.2014.995383
- 51. Lutz ER, Wu AA, Bigelow E, Sharma R, Mo G, Soares K, et al. Immunotherapy converts nonimmunogenic pancreatic tumors into immunogenic foci of immune regulation. Cancer Immunol Res. 2014;2:616–31. http://dx.doi.org/10.1158/2326-6066.CIR-14-0027
- 52. de Haar C, Plantinga M, Blokland NJ, van Til NP, Flinsenberg TW, Van Tendeloo VF, et al. Generation of a cord blood-derived Wilms Tumor 1 dendritic cell vaccine for AML patients treated with allogeneic cord blood transplantation. Oncoimmunology. 2015;4:e1023973.

http://dx.doi.org/10.1080/2162402X.2015.1023973

- 53. Saito S, Yanagisawa R, Yoshikawa K, Higuchi Y, Koya T, Yoshizawa K, et al. Safety and tolerability of allogeneic dendritic cell vaccination with induction of WT1-specific T cells in a pediatric donor and pediatric patient with relapsed leukemia: a case report and review of the literature. Cytotherapy. 2015;17:330–5. http://dx.doi.org/10.1016/j.jcyt.2014.10.003
- 54. Shimodaira S, Hirabayashi K, Kobayashi T, Higuchi Y, Yokokawa K. Future Prospective of Cancer Vaccination Technology in Japan. Pharmaceut Reg Affairs. 2015;4:2.

http://dx.doi.org/10.4172/2167-7689.1000143

55. Shimodaira S, Higuchi Y, Koya T, Kobayashi T, Yanagisawa R, Hirabayashi K, et al. Smoking influences the yield of dendritic cells for cancer immunotherapy. Pharmaceut Reg Affairs. 2015;4:133.

http://dx.doi.org/10.4172/2167-7689.1000133

- 56. Brodeur GM, Pritchard J, Berthold F, Carlsen NL, Castel V, Castelberry RP, et al. Revisions of the international criteria for neuroblastoma diagnosis, staging, and response to treatment. J Clin Oncol. 1993;11:1466–77.
- 57. Higuchi Y, Koya T, Yuzawa M, Yamaoka N, Mizuno Y, Yoshizawa K, et al. Enzymelinked immunosorbent spot assay for the detection of Wilms' tumor1-specific T cells induced by dendritic cell vaccination. Biomedicines. 2015;3(4):304–15. http://dx.doi.org/10.3390/biomedicines3040304
- 58. Scheibenbogen C, Letsch A, Thiel E, Schmittel A, Mailaender V, Baerwolf S, et al. CD8 T-cell responses to Wilms tumor gene product WT1 and proteinase 3 in patients with acute myeloid leukemia. Blood. 2002;100:2132–7. http://dx.doi.org/10.1182/blood-2002-01-0163
- 59. London WB, Castel V, Monclair T, Ambros PF, Pearson AD, Cohn SL, et al. Clinical and biologic features predictive of survival after relapse of neuroblastoma: a

Shimodaira et al.

report from the International Neuroblastoma Risk Group project. J Clin Oncol. 2011;29:3286–92.

http://dx.doi.org/10.1200/JCO.2010.34.3392

- 60. Ha TC, Spreafico F, Graf N, Dallorso S, Dome JS, Malogolowkin M, et al. An international strategy to determine the role of high dose therapy in recurrent Wilms' tumour. Eur J Cancer. 2013;49:194–210. http://dx.doi.org/10.1016/j.ejca.2012.07.010
- 61. Lampson LA, George DL. Interferon-mediated induction of class I MHC products in human neuronal cell lines: analysis of HLA and beta 2-m RNA, and HLA-A and HLA-B proteins and polymorphic specificities. J Interferon Res. 1986;6:257–65.
- 62. Corrias MV, Occhino M, Croce M, De Ambrosis A, Pistillo MP, Bocca P, et al. Lack of HLA-class I antigens in human neuroblastoma cells: analysis of its relationship to TAP and tapasin expression. Tissue Antigens. 2001;57:110–17.
- 63. Yuan J, Occhino M, Croce M, De Ambrosis A, Pistillo MP, Bocca P, et al. Down-regulation of human leukocyte antigen class I (HLA-I) is associated with poor prognosis in patients with clear cell renal cell carcinoma. Acta Histochem. 2013;115:470–4. http://dx.doi.org/10.1016/j.acthis.2012.11.002
- 64. Chan CW, Crafton E, Fan HN, Flook J, Yoshimura K, Skarica M, et al. Interferonproducing killer dendritic cells provide a link between innate and adaptive immunity. Nat Med. 2006;12:207–13.

http://dx.doi.org/10.1038/nm1352

65. Takahara A, Koido S, Ito M, Nagasaki E, Sagawa Y, Iwamoto T, et al. Gemcitabine enhances Wilms' tumor gene WT1 expression and sensitizes human pancreatic cancer cells with WT1-specific T-cell-mediated antitumor immune response. Cancer Immunol Immunother. 2011;60:1289–97.

http://dx.doi.org/10.1007/s00262-011-1033-3

 Shih K, Arkenau HT, Infante JR. Clinical impact of checkpoint inhibitors as novel cancer therapies. Drugs. 2014;74:1993–2013. http://dx.doi.org/10.1007/s40265-014-0305-6

Chapter 9

Chronic Kidney Disease in Wilms Tumour Survivors – What Do We Know Today?

Joanna Stefanowicz,¹ Mateusz Kosiak,² Wojciech Kosiak,¹ Beata S. Lipska-Ziętkiewicz³

¹Department of Paediatrics, Haematology and Oncology, Medical University of Gdansk, Gdansk, Poland; ²Department of Family Medicine, Medical University of Gdansk, Gdansk, Poland, Medical Centre, Puck, Poland; ³Department of Biology and Genetics, Medical University of Gdansk, Gdansk, Poland

Author for correspondence: Joanna Stefanowicz, MD, PhD, Department of Paediatrics, Haematology and Oncology, Medical University of Gdansk,
7 Debinki Street, 80-952 Gdansk, Poland. Email: jstefanowicz@gumed.edu.pl

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch9

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Currently, the treatment of Wilms tumour (WT) is successful in approximately 90% of cases, and consists of chemotherapy, nephrectomy, and, in some cases, radiation therapy. All treatments have potential long-term influence on the function of solitary kidneys in WT survivors (WTS). Severe reduction in glomerular filtration rate occurs after nephrectomy. All patients who underwent surgical treatment for WT could be considered to have

In: *Wilms Tumor*. Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Stefanowicz et al.

a risk of chronic kidney disease (CKD) because they lack a kidney. End-stage renal disease is rare (1.8% of National Wilms Tumour Study patients). Recent studies have revealed that patients with CKD have a greater risk of cardiovascular events and death. Most of the WTS have lower stages or no CKD. Regular biochemical studies and ultrasound examination at follow-up visits should be considered as indispensible elements of long-term care in uninephrectomized WTS. The evaluation of a single kidney function should be frequent, consisting of the assessment of estimated glomerular filtration rate (eGFR), assessment of albumin urine excretion, urine sediment analysis to detect abnormalities, ultrasound examination and measurements of blood pressure. According to Kidney Disease Improving Global Outcomes (KDIGO) recommendation and suggestions, GFR should be assessed using GFR-estimating equations that include serum creatinine and cystatin C concentrations. Cystatin C can be a more sensitive marker of kidney filtration function than creatinine, especially in diseases characterized by a mild decrease in glomerular filtration. This will facilitate the detection of early kidney impairment and assessment of the progression of CKD in WTS.

Key words: Chronic kidney disease; Renal function; Solitary kidney; Wilms tumour survivors

Introduction

Currently, the treatment of Wilms tumour (WT) is successful in approximately 90% of cases after chemotherapy, nephrectomy and, in some cases, radiation therapy (1). The number of survivors who have completed this treatment is increasing. All treatments can have potential long-term influence on the renal function of WT survivors (WTS) (2–4). A wide range of defects in kidney structure and function, from end-stage renal disease (ESRD) to varying degrees of chronic kidney disease (CKD), have been reported. CKD is associated with an increased risk of cardiovascular events, hospitalization and higher mortality (4, 5). The incidence and causes of renal dysfunctions vary depending on distinct clinical situations: sporadic (nonsyndromic) unilateral WT (UWT), sporadic bilateral WT (BWT) and WT arising in patients with genetic predisposition syndromes. All WTS are also considered to be at increased risk of acute kidney injury. Even mild deficiencies in the renal function may be associated with an increased risk of hypertension and cardiovascular disease.

End-stage renal disease in WT

ESRD, simply defined as the need for kidney replacement therapy, is very rare in WTS. Although observed mostly among patients who present with or develop BWT, it is also more frequent among children with syndromic WT. The latter include patients with microdeletion 11p13 syndrome (i.e., WAGR syndrome, MIM#194072: WT, Aniridia, Genitourinary

Chronic kidney disease in WT survivors

malformation, mental Retardation) and Denys–Drash syndrome (DDS, MIM#194080) (7). The 20-year cumulative risk of ESRD among WAGR syndrome and DDS patients was 43.3% [95% confidence interval (CI), 20.8–59.5%] and 82.7% (95% CI, 60.5–92.4%), respectively (8–10).

WT1 gene expression has a crucial role in normal kidney development. The disruption of the activity of WT1 protein may lead to fewer functional nephrons at birth, and histological studies of patients with WAGR syndrome indicate a reduction in the size of glomeruli that is presumably related to the *WT1* deletion. Reductions in nephron and podocyte number and mass could increase susceptibility to renal failure, particularly in patients with unilateral or partial bilateral nephrectomy. Reduced expression of WT1 in adult podocytes may reduce the GFR and eventually lead to glomerular sclerosis (9). Recently, Lipska et al. (11) evaluated genotype-phenotype associations in *WT1* glomerulopathy. The authors reported that diffuse mesangial sclerosis is largely specific for *WT1* disease, but focal segmental glomerulosclerosis was equally prevalent in *WT1*-positive and *WT1*-negative steroid-resistant nephrotic syndrome patients. According to the National Wilms Tumour Study (NWTS), the cumulative incidence of ESRD due to chronic renal failure (CRF) 20 years after WT diagnosis was 0.7% (9).

The low incidence of ESRD in WTS is also confirmed by European studies (12, 13). For ESRD due to progressive BWT, it was 4.0% at 3 years post-WT diagnosis in patients with synchronous BWT and 19.3% in patients with metachronous BWT. Lange et al. (9) concluded that metachronous BWT is associated with high rates of ESRD due to surgery for progressive WT. Carriers of germline *WT1* mutation had markedly increased risk of ESRD due to CRF, despite a low risk in non-*WT1* syndromic patients overall. Ritchey et al. (8) reported an incidence of renal failure of 0.25% among patients with UWT, with a median follow-up of 6 years from diagnosis (range: 2 months–22 years).

Analysis of NWTS trials performed by Grigoriev et al. (14) showed that ESRD was diagnosed in 173 patients among 9,162 individuals with WT treated between October 1969 and April 2002. The most common causes of ESRD were progressive BWT (55); DDS (27); WAGR syndrome (10); radiation nephritis (12); focal segmental glomerulosclerosis (18); CKD, aetiology unknown (16) and hypertension (7). Fifty-five patients whose ERSD resulted from progressive BWT experienced high early mortality from WT that limited their opportunity for transplant (47% at 5 years) and survival (44% at 10 years) compared with population controls. The remaining 118 patients, many of whom had *WT1*-associated congenital anomalies, had transplant (77% at 5 years) and survival (73% at 10 years) outcomes no worse than those for population controls. The risk of ESRD due to progressive BWT was largely confined to the first 3 years following the onset of bilateral disease, whereas the incidence rates of ESRD due to CKD continued to increase for 20–25 years from WT diagnosis (14).

Stefanowicz et al.

It is known that patients with hereditary predisposition syndromes are at a higher risk of developing bilateral tumours (both synchronous and metachronous). Two main molecular subgroups are recognized: syndromes associated with germline *WT1* mutation and overgrowth syndromes associated with epigenetic alterations in chromosome 11p15 (15, 16). Germline *WT1* mutations are also associated with renal developmental abnormalities and are risk factors for renal dysfunction regardless of the occurrence of WT. The following syndromes are associated with *WT1* mutation: DDS, WAGR syndrome, Frasier syndrome (MIM#136680) and isolated WT (MIM#194070). The 20-year cumulative incidence of ESRD in patients with DDS and WAGR syndrome treated for WT can be as high as 80% and 90%, respectively (9). In patients with overgrowth syndromes, the most common being Beckwith-Wiedemann syndrome (BWS, MIM#130650), there does not seem to be a higher risk for renal dysfunction. Nevertheless, almost 20% of WT are bilateral in individuals with BWS (16).

Romao et al. (15) suggests that patients with hereditary predisposition syndromes who develop UWT should be treated with preoperative chemotherapy followed by nephron-sparing surgery (NSS), with the goal of preserving normal kidney function. The issue remains, however, strongly controversial. For instance, data presented by Lipska et al. (11) support pre-emptive/ elective bilateral nephrectomy in patients with exonic germline *WT1* (i.e., Denys–Drash type) mutations (14). Using the international PodoNet cohort, Lipska et al. described the genotypic and phenotypic spectrum of *WT1*-associated kidney disease in 61 patients, the largest cohort of *WT1* nephropathy analyzed to date. Eighty-two percent of DDS patients needed kidney replacement therapy within 10 years of diagnosis. Among patients with exonic mutations who initially presented proteinuria of various degrees, 67% eventually developed WT, including 23% BWT. A total of 27 patients (44%), including 4 with intronic (Frasier type) mutations, underwent bilateral nephrectomy. Half of them, all with exonic mutations, underwent the surgery before their fifth birthday. Nephrectomy was performed electively before transplantation (n=18) due to BWT (n=5) or suspicious sonographic findings (n = 4) (11).

The development of renal dysfunction in survivors of BWT is a well-known complication. The philosophy of initial treatment with neoadjuvant chemotherapy for BWT is to avoid renal failure by maximal preservation of renal parenchyma (17). Bishop et al. (17) first reported a significant difference in the incidence of renal failure in NWTS patients with BWT (9% synchronous, 18% metachronous) versus unilateral involvement (1%). The primary cause of renal failure was bilateral nephrectomy for persistent or recurrent tumour. Within the NWTS 4, 23 out of 188 (12%) patients with bilateral disease followed from 1986 to 1994 developed ESRD (18). According to Lange et al. (9) who studied ESRD in non-*WT1*-syndromic patients treated by NWTS, the incidence of ESRD increased dramatically 20 years after diagnosis, reaching 3.1% for BWT. Non-*WT1*-syndromic BWTs have six times the risk of ESRD compared with unilateral ones.

The function of solitary kidney in WTS

The function of solitary kidney in WTS has been analyzed in several studies. Some investigators consider post-nephrectomy renal dysfunction as clinically insignificant (19, 20). In contrast, other investigators consider this renal dysfunction to be a harbinger of longterm consequences (21, 22). Romao et al. (15) were the first to draw attention to the need to develop risk stratification for renal dysfunction for unilateral nonsyndromic WT patients. Early detection of patients at risk may help tailor treatment (e.g., nephrectomy or NSS) and create focused monitoring protocols. As molecular biomarkers for both biological aggressiveness and multifocality are being discovered and incorporated into clinical practice, targeted interventions may be devised to improve the balance between cure and long-term morbidity.

Recently, Interiano et al. (23) evaluated the prevalence of hypertension and impaired renal function in a group of 75 long-term survivors of non-syndromic UWT (median length of follow-up, 19.6 years; range: 10.0–32.8 years) who were treated without nephrotoxic chemotherapy or ionizing radiation. Renal function was assessed by urinalysis and eGFR. Sixteen patients (21.3%) only had eGFR<90 ml/min/1.73 m2, no patient had an eGFR <60 ml/min/1.73 m2 and five patients (6.7%) had hypertension. At the time of last follow-up, no patient developed ESRD. The authors concluded that patients with UWT who were treated with unilateral radical nephrectomy without nephrotoxic chemotherapy or ionizing radiation appear to be at low risk of developing significant long-term renal dysfunction, but monitoring and counselling are important for early detection of subtle abnormalities. This group of WTS might be at an increased risk of adverse cardiovascular sequelae.

Hypertension is one of the components of the metabolic syndrome, which is indicated to be an important risk factor for developing cardiovascular diseases and type II diabetes mellitus. Van Waas et al. (24) reported that long-term adult survivors of childhood cancer are at increased risk of developing components of the metabolic syndrome. Their analysis of 500 adult survivors of childhood cancer provides information on the occurrence of components of the metabolic syndrome in long-term survivors of 11 types of childhood cancer. Systolic blood pressure was increased after the treatment of all types of malignancies, except for Langerhans cell histiocytosis and include WTS. It is unknown what determines the elevated blood pressure in survivors. Dysfunction of the endothelium has been hypothesized to be the initial step in the development of cardiovascular diseases. Chemotherapy agents like anthracyclines are known to damage the vascular endothelium. Additionally, radiotherapy could have a damaging effect on the endothelium.

Chronic kidney disease in WTS

According to the definition of the National Kidney Foundation (NKF), CKD is defined as abnormalities of kidney structure or function present for >3 months with implications for

Stefanowicz et al.

health. CKD is either a kidney damage or an eGFR below 60 ml/min/1.73 m² for over 3 months (6). Kidney damage is defined as pathologic abnormalities or markers of damage, including abnormalities in blood or urine tests or imaging studies (6, 25,). Signs of kidney damage may indicate the risk of deterioration of kidney function in the future. Long-term monitoring of renal function in WTS will facilitate the identification of those with treatment-related impairment of function.WTS are at risk of deterioration of renal function and CKD because of the following: decreased number of nephrons – after nephrectomy, nephrotoxic side effects of chemotherapy (carboplatin, cisplatin, ifosfamide, cyclophosphamide) or radiation therapy – if solitary kidney was in the field of radiation (3, 4, 26, 27). According to Daw et al. (3), the most severe reduction in GFR, measured by 99Tc-DTPA (technetium-99m-dieth-ylenetriamine pentaacetic acid) clearance, occurs after nephrectomy. Decreasing the number of nephrons causes a compensatory increase in the filtration of the remaining nephrons to maintain excretory demands. Subsequent glomerular hyperfiltration in the remaining kidney leads to further deterioration of viable nephrons. Over time, the number of viable nephrons may become insufficient, resulting in a further reduction in kidney function (28-30).

According to the Brenner theory, the reduced filtration surface area of the kidney resulting from an acquired deficit of glomeruli impairs the normal adjustment of blood pressure by pressure natriuresis (31). Therefore, patients with a solitary kidney reveal an increased risk of albuminuria, hypertension and CKD.CKD can be diagnosed in all WTS subjected to nephrectomy. According to Interiano et al. (23), the current guidelines do not recognize solitary kidney or unilateral nephrectomy as a structural abnormality, but further studies are needed to determine whether a lack of one kidney is a marker for CKD development. In our opinion, from the viewpoint of renal function and long-term survival among uninephrectomized WT patients, the above statement is neither certain nor obvious. There are few long-term studies that evaluate the renal function in adult WTS , and we do not have sufficient scientific evidence that confirms the validity of this thesis (4, 32-34). Certainly, solitary kidney is a risk factor for the progression of CKD.

Little is known about the renal function in adult WTS who underwent nephrectomy a long time ago. Patients with WT are mostly very young children, and the prevalence of severe renal dysfunction owing to multifactorial causes is likely to increase with longer follow-up and survival. Kern et al. (35) assessed the renal function in a group of 55 patients with non-syndromic UWT and reported that increasing time between surgery and the last known GFR follow-up was associated with decreased GFR. They concluded that longer follow-up may reveal that a clinically significant decline in the renal function occurs in the years following nephrectomy. Because of the potential for long-term renal insufficiency in children who undergo unilateral nephrectomy, some groups have advocated for NSS in patients with UWT to preserve renal parenchyma and function (32, 33). The use of NSS was judged by

Chronic kidney disease in WT survivors

these investigators as essential to reduce post-nephrectomy renal morbidity. NSS is infrequently performed in patients with UWT because the majority of cooperative group protocols recommend a radical nephrectomy (36). Partial nephrectomy for patients with UWT is a more controversial topic, particularly in the case of excellent oncologic outcomes with unilateral radical nephrectomy. The risk of local recurrence, need for therapy intensification, and unclear long-term renal function benefits have been the basis for debate against NSS in the treatment of unilateral non-syndromic WT (15). NSS is necessary in patients with BWT for the preservation of nephrons and renal function (17, 18).

Cozzi et al. (33) also examined and compared the renal function of 20 WTS treated with NSS, 40 WTS aged 2-30 years treated with nephrectomy, and 18 WTS aged 33-51 years treated with nephrectomy. While only 8% of NSS and 42% of nephrectomized young WTS presented mildto-moderate renal function, this was 78% in the oldest nephrectomized WTS. The authors demonstrated a significant reduction in eGFR in the fifth decade of life in a group of WTS who underwent nephrectomy compared with patients during the third decade after surgery. Currently, the role of NSS in the treatment of unilateral nonsyndromic WT patients is discussed in the context of ensuring adequate local control and protection of renal function. So far, the standard of UWT treatment according to the Societe International de Oncologie Pediatrique (SIOP) and Children's Oncology Group protocols includes radical nephrectomy. Data from the SIOP 2001 study showed that NSS was only performed in 3% of patients with UWT. Wilde et al. (36) concluded that NSS as a new approach for UWT has now been shown to be safe in a small and highly selected group of patients, concordant with the intention of the SIOP 2001 protocol. The event-free and overall survival after NSS appeared to be as good as total nephrectomy with an equal local relapse rate as that of total nephrectomy. Despite excellent survival, the gain of nephrons needs to be weighed against the risk to induce stage III with intensified therapy. Studies on the renal function after NSS are based on relatively small patient groups. Larger prospective studies are needed to fully assess the gain of renal function and oncological outcome.

Previously, we have analyzed the prevalence of CKD in nephrectomized WTS in a group of 32 patients (children and adolescents). All participants had undergone unilateral nephrectomy and had been treated according to the chemotherapy protocols SIOP 9, SIOP 1992, SIOP 2001 between the years 1987 and 2008. Kidney damage was established by the assessment of GFR using 99 Tc-DTPA clearance, the Schwartz formula, the new Schwartz equation, Filler formula, serum cystatin C concentration, β 2-microglobulin and albumin urine excretion, urine sediment and ultrasound examination. Blood pressure was measured. The mean values of GFR assessed with different methods, 99Tc-DTPA clearance, eGFR Schwartz, eGFR new Schwartz equation for children with CKD and eGFR Filler, were all well above 60 ml/min/1.73 m². Increased excretion of albumin (ACR >30 mg/g creatinine) and B-2-microglobuin (BCR >0.04 g/mol) was observed in 22% and 13% of patients, respectively.

are succes at harmonic and the second branch	Recommendations of monitoring	No eGFR<60 renal and/or injury treat- ment	1/y 2-4/y	1/y 2-4/y	1/5y 2-4/y	ecessary	necessary	at least every 5 years	regular follow up should included BP monitoring, USG, ECHO, kidney function parameters	necessary
			BP	ACR***	SCr***/ eGFR	5				
	BCR**				1		0.57 (0.2-1.0)	1	ı.	
	Albuminuria or proteinuria					A2 (30-300 mg/L), 14%	-	0.19 (0.02-13.2)	Normal/0%	%0
	Hypertension	26%				2.9%	-	24 h ABPM***** normal	24 h ABPM 20% 24 h SBP higher com- par to con- trol group (p<0.05)	30%
	eGFR * [ml/ min/1.73 m²] mean±SD/ median (range)	83±14 <60 - 4.2%			midly decreased G2, (84.5, 63-89) 22.9% CKD 8.6%	† 15%	103.8 (89.2-166.7)	%0	119	
	Treatment	SIOP				SIOP		VCR, ACTD	NWTS	NWTS (ICE-9 patients)
	Stage	I-IV				I-IV	1	I, II	I-IV	VI-I
	Follow up duration [years] mean ±SD/median (range)	14.8±3.3				20±5.43	6.3 (median)	13.3 (10.2-19.3)	9.9 (2-21)	4.8 (media)
	Age at follow up mean±SD/ median (range)	17.8±3.6				25		18.3 (12.7-30.4)	1	'
	Number of patients	33				35	55	15	15	30
		Mavinkurve- Groothuis 2015 [44]				Schiavetti 2015 [43]	Kern 2014 [36]	Spreafico 2014 [45]	Elli 2013 [46]	Sanpakit 2013 [47]

 Table. 1
 Renal function in Wilms tumour survivors based on publish data in recent years

138

Stefanowicz et al.

necessary	necessary		eGFR, ACR, BCR, urine sediment analyses, USG	necessary	necessary, not specified	
1	,	,	12,5%	←	1	
25%	,	,	21.9%	18%	5%	
31.4%	,	,	6.3%	18%	%0	
60-90 51.8% 30-59 4,7% 15-29 0% <15 3.5%	76.1±16.3 <90 - 78%	95.1±18.5 <90 - 43%	94.28 (10.24)	86 (10-169) <90 - 54 % Renal trans- plantation 1 (with FSGS)	100 (61-150) 60-90 25% 30-59 0% 15-29 0% <15 0%	
SIOP	VCR, ACTD, ADR	VCR, ACTD, ADR	SIOP	ICE (ifos- famide, carbopla- tin, eto- poside), nephrec- tomy	VCR, ACTD, ADR	
I-IV	I-IV	I-IV	I-IV	II-IV	I-IV	
24.4 (12.2- 41.1)	38.44±4.9	11.38±7.8	7.75 (0.3-20.6)	11.2 (7.3-12.8)	8.8 (0.06-27.5)	
27.9 (17.9- 49.0)	42.7±5.7	15.8±8.0	12.2 (3.6-24.3)	18.6 (12-25.6)	14.31 (3.38- 29.75)	ton action 12 m
85	18	42	32	11	40	l alomonul
Dekkers 2013 [48]	Cozzi 2013 [34]		Stefanowicz 2011 [28]	Daw 2009 [3]	Bailey 2002 [19]	Portene and and a

GFR – estimated glomerular filtration r

**BCR - β2microglobulin/urine creatinine.
***ACR-albumin/urine creatinine.

****SCr - serum creatinine.

***** ABPM - ambulatory blood pressure monitoring.

Chronic kidney disease in WT survivors

Stefanowicz et al.

Arterial hypertension, based on the mean values of systolic and diastolic blood pressures from three independent measurements, over the 95th centile was observed in 6.25% individuals (27).

Ultrasound examination of WTS provides an opportunity to detect signs of kidney damage. According to the National Kidney Foundation Kidney Disease Outcomes Quality Initiative (NKF KDOQI), sonographic features associated with CKD include the following: nephrocalcinosis, stones, hydronephrosis, cysts, increased echogenicity of kidney, small 'hyperechoic kidney', large kidneys, size disparities and scars, and venous thrombosis or renal stenosis (25). Thus, ultrasound examination can be used to identify WTS with CKD and risk of deterioration of eGFR. In our previous study, sonographic signs of kidney damage, including hyperechoic rings around renal pyramids (37.5%), renal scars (9%), increased echogenicity of renal cortex (15.5%) and cysts (3%), were observed in 43% of WTS (27).

Hypertrophy of a solitary kidney [length or volume of kidney over 2SD (standard deviation of reference value)] in WTS was observed in 50-88% of individuals (9, 37, 38). Mean value of renal length was $128 \pm 14\%$ of the reference value. Mean value of renal volume was from 155±35% to 213% (39, 40). In one study, correlation between microalbuminuria and renal volume was observed (40). The parenchymal thickness/kidney length ratio correlated with the deterioration of renal function (cystatin C serum concentration) (38). Currently, in light of the definition of CKD for individuals with the risk of deterioration of kidney function, it is essential not only to assess GFR but also to establish the presence of structural and functional markers of kidney damage. A marker for CKD, which may be more sensitive to the detection of early renal impairment, is cystatin C. According to Kazama et al. (41), a cystatin C serum concentration greater than 0.98 mg/dl has a sensitivity of 88.5% and a specificity of 95.2% for detecting GFR below 80 ml/min/1.73 m2. Recently, Schiavetti et al. (42) evaluated the prevalence of and the possible risk factors for the renal impairment in 35 adult WTS by estimating GFR categories and CKD according to KDIGO guidelines from 2012 (6). Only eight (23%) survivors presented a mildly decreased eGFR, three survivors (9%) had CKD and one (3%) hypertension. Data on the renal function in WTS from recent reports are included in Table 1.

Conclusion

To conclude, in our opinion, the evaluation of a single kidney in WTS should be regular and consist of the assessment of eGFR using equations, albumin excretion, urine sediment analysis, ultrasound examination and blood pressure measurements. If we do not have progressive renal injury, this nephrological follow-up should be once per year. It is necessary to diagnose early renal impairment. According to KDIGO recommendations, CKD is defined as abnormalities of kidney structure or function, present for >3 months, with implications for health, and CKD is classified based on cause, GFR category and albuminuria category. The assessment of eGFR and albumin excretion is necessary to determine the risk of CKD. For GFR evaluation, KDIGO recommends using serum creatinine and a GFR estimating equation for initial assessment and suggests using additional tests such as cystatin C for confirmatory testing in specific circumstances when eGFR based on serum creatinine is less accurate. In children, GFR can be estimated using Schwartz formula [41.3 × (height/serum creatinine)], where height is expressed in meters and serum creatinine in mg/dl (6).

Furthermore, according to Romao et al. (15), the discussion about renal dysfunction in WTS will evolve and receive more attention. At present, survival in patients with WT is very good. The number of survivors increases from year to year. It is necessary to agree on the standardized follow-up protocols and tools to measure renal dysfunction over time. A complete assessment of renal function in WTS should be simple, easy, generally available and, according to NKF recommendations, include estimated GFR, urine test with albuminuria, ultrasound examination and measurements of blood pressure. This will facilitate the assessment and detection of the progression of CKD in WTS.

Conflicts of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

References

- Tournade MF, Com-Nougué C, de Kraker J, Ludwig R, Rey A, Burgaers MB, et al. Optimal duration of preoperative therapy in unilateral and nonmetastatic Wilms tumor in children older than 6 months: results of the ninth International Society of Pediatric Oncology Wilms Tumor Trial and Study. J Clin Oncol. 2001;19:488–500.
- 2. Barrera M, Roy PL, Stevens M. Long-term follow-up after unilateral nephrectomy and radiotherapy for Wilms' tumour. Pediatr Nephrol. 1989;3:430–43.
- Daw NC, Gregornik D, Rodman J, Marina N, Wu J, Kun EL, et al. Renal function after ifosfamide, carboplatin and etoposide (ICE) chemotherapy, nephrectomy and radiotherapy in children with Wilms tumor. Eur J Cancer. 2009;45:99–106. http://dx.doi.org/10.1016/j.ejca.2008.09.017
- Mulder RL, Knijnenburg SL, Geskus RB, van Dalen E, van der Pal HJH, Koning CCE, et al. Glomerular function time trends in long-term survivors of childhood cancer: a longitudinal study. Cancer Epidemiol Biomarkers Prev. 2013;22:1736–46. http://dx.doi.org/10.1158/1055-9965.EPI-13-0036
- Go AS, Chertow GM, Fan D, McCulloch CF, Hsu C. Chronic kidney disease and the risk of death, cardiovascular events, and hospitalization. N Engl J Med. 2004;551:1296–305. http://dx.doi.org/10.1056/NEJMoa041031

Stefanowicz et al.

- http://www.kdigo.org/clinical_practice_guidelines/pdf/CKD/KDIGO_2012_ CKD_GL.pdf. Accessed February 07, 2016.
- Fischbach BV, Trout KL, Lewis J, Luis CA, Sika M. WAGR syndrome: a clinical review of 54 cases. Pediatrics. 2005;116:984–8. http://dx.doi.org/10.1542/peds.2004-0467
- 8. Ritchey ML, Green DM, Thomas PR, Smith GR, Haase G, Shochat S, et al. Renal failure in Wilms' tumor patients: a report from the National Wilms Tumor Study Group. Med Pediatr Oncol. 1996;26:75–80.

http://dx.doi.org/10.1002/(SICI)1096-911X(199602)26:2<75::AID-MPO1>3.0.CO;2-R

 Lange J, Peterson SM, Takashima JR, Grigoriev Y, Ritchey ML, Shamberger RC, et al. Risk factors for end stage renal disease in non WT1-syndromic Wilms tumor. J Urol. 2011;186:378–86.

http://dx.doi.org/10.1016/j.juro.2011.03.110

 Breslow NE, Collins AJ, Ritchey ML, Grigoriev YA, Peterson SM, Green DM. End stage renal disease in patients with Wilms tumor: results from the National Wilms Tumor Study Group and the United States Renal Data System. J Urol. 2005;174:1972–75.

http://dx.doi.org/10.1097/01.ju.0000176800.00994.3a

 Lipska BS, Ranchin B, Iatropoulos P, Gellermann J, Melk A, Ozaltin F, al.; PodoNet Consortium. Genotype-phenotype associations in WT1 glomerulopathy. Kidney Int. 2014;85:1169–78.

http://dx.doi.org/10.1038/ki.2013.519

- Ardissino G, Dacco V, Testa S, Bonaudo R, Claris-Appiani A, Taioli E, et al. Epidemiology of chronic renal failure in children: data from the ItalKid project. Pediatrics. 2003;111:e382–7.
- Knijnenburg SL, Mulder RL, Schouten-Van Meeteren AY, Bökenkamp A, Blufpand H, van Dulmen-den Broeder E, et al. Early and late renal adverse effects after potentially nephrotoxic treatment for childhood cancer. Cochrane Database Syst Rev. 2013;8:10.
- Grigoriev Y, Lange J, Peterson S, Takashima JR, Ritchey ML, Ko D, et al. Treatments and outcomes for end-stage renal disease following Wilms tumor. Pediatr Nephrol. 2012;27:1325–33.

http://dx.doi.org/10.1007/s00467-012-2140-x

- Romao RLP, Lorenzo AJ. Renal function in patients with Wilms tumor. Urol Oncol. 2016;34:33–41. http://dx.doi.org/10.1016/j.urolonc.2015.07.002
- Dome JS, Huff V. Wilms tumor. In: Pagon RA, Bird TD, Dolan CR, et al. (eds). GeneReviews[™] (Internet). Seattle, WA: University of Washington, 1993. http://www.ncbi.nlm.nih.gov/books/NBK1294/

- Bishop HC, Teft M, Evans A, D'Angio GJ. Survival in bilateral Wilms' tumor Review of 30 National Wilms' Tumor Study Cases. J Pediatr Surg. 1977;12:631–63.
- Hamilton TE, Ritchey ML, Haase GM, Argani P, Peterson SM, Anderson JR, et al. The management of synchronous bilateral Wilms tumor: a report from the Wilms Tumor Study Group. Ann Surg. 2011;253:1004–10. http://dx.doi.org/10.1097/SLA.0b013e31821266a0
- Bailey S, Roberts A, Brock C, Price L, Craft AW, Kilkarni R, et al. Nephrotoxicity in survivors of Wilms' tumours in the North of England. Br J Cancer. 2002;87:1092–98. http://dx.doi.org/10.1038/sj.bjc.6600608
- 20. Schell M, Cochat P, Hadj-Aissa A, Bouffet E, Dubourg L, Brunat-Mentigny M. Renal function following unilateral nephrectomy for neuroblastoma and Wilms' tumor. Pediatr Nephrol. 1995;9:579–82.
- 21. Argueso LR, Ritchey ML, Boyle ET Jr, Milliner DS, Bergstralh EJ, Kramer SA. Prognosis of children with solitary kidney after unilateral nephrectomy. J Urol. 1992;148:747–51.
- 22. Levitt GA, Yeomans E, Dicks Mireaux C, Breatnach F, Kingston J, Pritchard J. Renal size and function after cure of Wilms' tumour. Br J Cancer. 1992;66:877–82.
- 23. Interiano RB, Santos ND, Huang S, Srivastava DK, Robinson LL, Hudson MM, et al. Renal function in survivors of nonsyndromic Wilms tumor treated with unilateral radical nephrectomy. Cancer. 2015;121:2449–56. http://dx.doi.org/10.1002/cncr.29373
- 24. van Waas M, Neggers SJCMM, Pieters R, van den Heuvel-Elbrink MM. Components of the metabolic syndrome in 500 adult long-term survivors of childhood cancer. Ann Oncol. 2010;21:1121–26.

http://dx.doi.org/10.1093/annonc/mdp414

- National Kidney Foundation KDOQI GUIDELINES. http://www.kidney.org/professionals/KDOQI/guidelines_ckd/toc.htm. Accessed November 20, 2011.
- 26. Green DM. Evaluation of renal function after successful treatment for unilateral, non-syndromic Wilms tumor. Pediatr Blood Cancer. 2013;60:1929–35. http://dx.doi.org/10.1002/pbc.24738
- Stefanowicz J, Kosiak M, Romanowicz G, Owczuk R, Adamkiewicz-Drożyńska E, Balcerska A. Glomerular filtration rate and prevalence of chronic kidney disease in Wilms' tumour survivors. Pediatr Nephrol. 2011;26:759–66. http://dx.doi.org/10.1007/s00467-011-1759-3
- 28. Thorner P, Arbus G, Celermajer D, Baumal R. Focal segmental glomerulosclerosis and progressive renal failure associated with a unilateral kidney. Pediatrics. 1984;73:806–10.

Stefanowicz et al.

- Fotino S. The solitary kidney: a model of chronic hyperfiltration in humans. Am J Kidney Dis. 1989;13:88–98.
- 30. Zucchelli P, Cagnoli L, Casanova S, Donini U, Pasquali S. Focal glomerulosclerosis in patients with unilateral nephrectomy. Kidney Int. 1983;24:649–55.
- 31. Brenner B, Mackenzie H. Nephrons mass as a risk factor for the progression of renal disease. Kidney Int. 1997;52:124–27.
- Cozzi DA, Ceccanti S, Frediani S, Schiavetti A, Cozzi F. Chronic kidney disease in children with unilateral renal tumor. J Urol. 2012;187:1800–05. http://dx.doi.org/10.1016/j.juro.2011.12.109
- 33. Cozzi DA, Ceccanti S, Frediani S, Mele E, Cozzi F. Renal function adaption up to fifth decade after treatment of children with unilateral renal tumor: a cross sectional and longitudinal study. Pediatr Blood Cancer. 2013;60:1534–8. http://dx.doi.org/10.1002/pbc.24545
- 34. Knijnenburg SL, Mulder RL, Schouten-Van Meeteren AY, Bökenkamp A, Blufpand H, van Dulmen-den Broeder E, et al. Early and late renal adverse effects after potentially nephrotoxic treatment for childhood cancer. Cochrane Database Syst Rev. 2013;8:10
- 35. Kern AJ, Inouve B, Ko JS, Gorin MA, Allaf ME, Goldstein S, et al. Impact of nephrectomy on long-term renal function in non-syndromic children treated for unifocal Wilms tumor. J Pediatr Urol. 2014;10:662–6.

http://dx.doi.org/10.1016/j.jpurol.2014.06.017

- 36. Wilde JCH, Aronson DC, Sznajder B, Van Tinteren H, Powis M, Okoye B, et al. Nephron sparing surgery (NSS) for unilateral Wilms tumor (UWT): the SIOP 2001 experience. Pediatr Blood Cancer. 2014;61:2175–9. http://dx.doi.org/10.1002/pbc.25185
- 37. Zerin JM, Haliloglu M, Cohen MD. Growth of the solitary kidney after nephrectomy in children with unilateral Wilms' tumor. Pediatr Radiol. 1996;26(8):547–52.
- Stefanowicz J, Kosiak M, Kosiak W, Korbus-Kosiak A, Sierota D, Owczuk R. Cystatin C and parenchymal thickness/kidney length ratio in Wilms tumor survivors. Pediatr Blood Cancer. 2010;54:41–6. http://dx.doi.org/10.1002/pbc.22249
- Krawczuk-Rybak M, Kuźmicz M, Kitszel A, Wysocka J, Zalewska B, Olszewska M. Renal function and size after complex treatment of Wilms' tumour. Med Wieku Rozwoj. 2003;7:359–67.
- Di Tullio MT, Casale F, Indolfi P, Polito C, Giuliano M, Martini A, et al.Compensatory hypertrophy and progressive renal damage in children nephrectomized for Wilms' Tumor. Med Pediatr Oncol. 1996;26:325–8. http://dx.doi.org/10.1002/(SICI)1096-911X(199605)26:5<325::AID-MPO4>3.0.CO;2-K

144

Chronic kidney disease in WT survivors

41. Kazama JJ, Kutsuwada K, Ataka K, Maruyama H, Gejyo F. Serum cystatin C reliably detects renal dysfunction in patients with various renal diseases. Nephron. 2002;91:13–20.

http://dx.doi.org/57599

42. Schiavetti A, Altavista P, De Luca L, Andreoli G, Megaro G, Versacci P. Long-term renal function in unilateral non-syndromic renal tumor survivors treated according to International Society of Pediatric Oncology Protocols. Pediatr Blood Cancer. 2015;62:1637–44.

http://dx.doi.org/10.1002/pbc.25558

- 43. Mavinkurve-Groothuis AMC, van de Kract F, Westland R, van Wijk JAE, Loonen JJ, Schreuder MF. Long-term follow-up of blood pressure and glomerular filtration rate in patients with solitary functioning kidney: a comparison between Wilms tumor survivors and nephrectomy for other reasons. Pediatr Nephrol. 2015;Oct 19 [Epub ahead of print].
- 44. Spreafico F, Terenziani M, Testa S, Perotti D, Collini P, Piva L, et al. Long-term renal outcome in adolescent and young adult patients nephroctomized for unilateral Wilms tumor. Letter to the Editor. Pediatr Blood Cancer. 2014;61:1136–37. http://dx.doi.org/10.1002/pbc.24876
- 45. Elli M, Sungur M, Genc G, Ayyildiz P, Dagdemir A, Pinarli FG, et al. The late effects of anticancer therapy after childhood Wilm's tumor: the role of diastolic function and ambulatory blood pressure monitoring. Jpn J Clin Oncol. 2013;43:1004–11. http://dx.doi.org/10.1093/jjco/hyt105
- 46. Sanpakit K, Triwatanawong J, Sumboonnanonda A. Long-term outcome in pediatric renal tumor survivors: experience of a single center. J Pediatr Hematol Oncol. 2013;35:610–13.

http://dx.doi.org/10.1097/MPH.0b013e3182a06265

 Dekkers IA, Blijdorp K, Cransberg K, Pluijm SM, Pieters R, Neggers SJ, et al. Longterm nephrotoxicity in adult survivors of childhood cancer. Clin J Am Soc Nephrol. 2013;8:922–9.

http://dx.doi.org/10.2215/CJN.09980912

Section II

Biology

Chapter 10

Gene Expression in Wilms Tumor: Disturbance of the Wnt Signaling Pathway and MicroRNA Biogenesis

Dirce Maria Carraro,¹ Rodrigo F. Ramalho,¹ Mariana Maschietto²

¹Laboratory of Genomics and Molecular Biology, CIPE, AC Camargo Cancer Center, Sao Paulo, SP, Brazil; ²Brazilian Biosciences National Laboratory (LNBio), Center for Research in Energy and Materials (CNPEM), Campinas, SP, Brazil

Author for correspondence: Dirce Maria Carraro, PhD, Head, Laboratory of Genomics and Molecular Biology, CIPE/AC Camargo Cancer Center, Rua Taguá, 440, 1.andar, São Paulo - SP - 01508-010, Brazil. Email: dirce.carraro@cipe.accamargo.org.br

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch10

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Wilms tumor (WT) originates from the metanephric blastemal cells that are unable to complete the mesenchymal-epithelial transition, resulting in a tumor with triphasic histology, including blastemal, epithelial, and stromal components. WT shows morphological and molecular characteristics that resemble the fetal kidney. Thus, the study of molecular pathways relevant to normal kidney differentiation provides insight into the events that drive

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Carraro et al.

Wilms tumorigenesis. The Wnt signaling pathway has been shown to be crucial for correct kidney differentiation. This pathway is activated by WNT proteins and consists of two highly connected main branches: the canonical (or β -catenin dependent) and the noncanonical (or β -catenin independent). Both branches are essential for controlling embryonic development and adult cell homeostasis. The activation of the canonical Wnt pathway leads to the nuclear accumulation of β -catenin, which acts as a coactivator for transcription factors. In the absence of WNT ligands, this pathway is inactivated by a destruction complex that phosphorylates β -catenin, leading to ubiquitination, proteasomal degradation, and the prevention of β -catenin accumulation in the nucleus. In this context, the expression and mutation analyses of genes involved in Wnt signaling pathways constitute an important approach for understanding WT etiology. Although the activation of the Wnt pathway is well understood in WT samples relative to normal kidney tissue or differentiated kidney cells, there is a remarkable variation among subgroups of WTs. Recently, five WT subgroups were identified, mainly through the use of gene expression data, and only two of them showed clear evidence of Wnt pathway activation, as measured by the presence of β -catenin in the nucleus. Interestingly, some of these subgroups exhibited recurrent germline or somatic mutations in genes involved in microRNA biogenesis, such as DROSHA and DICER. Here, we will review relevant findings regarding Wilms tumorigenesis as revealed by gene expression and mutation analyses, mainly in genes belonging to the Wnt signaling and microRNA biogenesis pathways.

Key words: β-catenin; microRNA biogenesis; Nephrogenesis; Wilms tumor; Wnt signaling pathway

Introduction

Wilms tumor (WT) is an embryonic tumor that is initiated from primitive renal cells that are incapable of completing kidney differentiation. The result is a tumor that recapitulates the earliest step of nephrogenesis and is morphologically and molecularly similar to the fetal kidney. As a consequence, WTs are composed of varying proportions of three morphologically distinct cell types: undifferentiated blastemal cells, epithelial cells ordered into primitive structures, and stromal cells (1). Accordingly, the blastemal component displays an expression profile similar to the earlier stages of kidney development (2).

Signal transduction pathways control signaling from the outside to the inside of a cell through interactions between proteins and cell surface receptors, triggering specific cellular processes, mainly via changes in gene expression. The precise control of gene activation or inactivation is crucial for correct kidney differentiation and function. Disturbances in this process through the mutation of genes that directly or indirectly control gene

Gene expression in Wilms tumor

expression result in the failure of precise kidney development and may in turn lead to renal disease in children, including renal agenesis, dysplasia, hypoplasia, and WT (3). In WT, mutations have been identified in tumor suppressor genes (TSG) and oncogenes (4) (mainly from the WNT signaling pathway) or in genes involved in microRNA (miRNA) biogenesis (5, 6). Thus, as carcinogenesis requires gene mutations, the morphological aspects of WT seem to be dependent on where and when mutations occur during the process of kidney differentiation. In this chapter, we present the current molecular and morphogenetic knowledge about nephrogenesis and WT, focusing on the Wnt signaling and miRNA biogenesis pathways.

Morphogenetic process of kidney development

Kidney development, also known as nephrogenesis, refers to the embryologic origins of this organ. Kidney morphogenesis begins at gastrulation, in the third week of gestation in humans, when the embryo exhibits the three germ layers: ectoderm, mesoderm, and endoderm. The intermediate mesoderm gradually forms the urogenital system, including the pronephros, mesonephros, and metanephros; the first two proceed to develop into transitory kidneys and the third differentiates into the mature and functional kidney (7). The metanephros originates through interactive signals in bidirectional communication between epithelial and mesenchymal cells that ultimately form the nephrons, the functional unit of the kidney. Thus, the entire process of the differentiation of the kidney, with its multifaceted functional structures, involves close interaction between epithelial and mesenchymal cells, in which the signal transduction pathway is imperative. Mesenchymal-epithelial transition (MET) is a crucial process operating during kidney differentiation (7), which comprises the transition from a multipolar or a spindle-shaped mesenchymal cell to a planar assembly of polarized cells known as epithelia. Epithelial cells are stationary and are characterized by apical-basal polarity, tight junctions, and the expression of cell-cell adhesion markers, such as E-cadherin (8), whereas mesenchymal cells do not form cell-cell contacts. Mesenchymal cells can invade through the extracellular matrix and express markers, such as vimentin, fibronectin, N-cadherin, basic helix-loop-helix transcription factor (TWIST), and zinc finger protein SNAI1 (SNAIL) (9).

The exact mechanism that triggers MET in kidney progenitor cells is not entirely known although it has been shown to depend on the silencing of specific genes (e.g., *Osr1* and *Six2*) (7). The morphological result of this process is the formation of a vesicle composed of the metanephric blastema, which further forms the comma-shaped body, followed by the S-shaped body, and then Bowman's capsule, finally culminating in the functional nephron (10). Next, we present some important aspects of WNT signaling, which is a key pathway in MET.

Carraro et al.

Wnt signaling pathway

The Wnt signaling pathway encompasses a variety of signaling cascades activated by the secreted WNT proteins with major involvement in nephrogenesis. The Wnt signaling pathway has been divided into two main branches: canonical and noncanonical. The canonical Wnt pathway (or the Wnt/ β -catenin pathway) operates with the involvement of β -catenin, encoded by *CTNNB1*, whereas the noncanonical (or β -catenin independent) pathway does not involve β -catenin (11, 12). The noncanonical Wnt signaling pathway is mainly divided into the Wnt/calcium (Wnt/Ca²⁺) and planar cell polarity (PCP) pathways. Despite several differences between the two branches, both are activated by the binding of a Wnt ligand to a Frizzled (FZD) family receptor (13, 14).

The canonical Wnt pathway is characterized by its intracellular mediator β -catenin and plays a crucial role in cell fate. In the canonical Wnt pathway, β -catenin can accumulate in the cytoplasm and either be directed to the membrane as a part of the cell-cell adhesion complex or be translocated into the nucleus, acting as a transcriptional coactivator of TCF/LEF family of transcription factors. Thus, β -catenin plays a dual role, either regulating the coordination of cellcell adhesion (in the inactivated Wnt signaling pathway) or acting as a transcriptional cofactor when translocated to the nucleus (in the activated Wnt signaling pathway). Thus, the regulation of cytoplasmic levels of β -catenin by the APC/AXIN1 (adenomatous polyposis coli/) destruction complex (DC) represents a fundamental control step of the canonical Wnt pathway.

The DC is composed of AXIN1, PP2A, GSK3, CK1, Dishevelled (DSH), and APC and marks β -catenin for degradation by the proteasome through ubiquitination. The heterodimer formed by a Wnt ligand and an FDZ (frizzled) receptor (WNT-FDZ heterodimer) interacts with the low-density lipoprotein receptor-related proteins 5 and 6 (LRP5/6), another cell surface protein, recruiting cytoplasmic AXIN1 and preventing the formation of APC/AXIN1. The WNT-FDZ heterodimer recruits and interacts with a series of cytoplasmic proteins to prevent the DC from ubiquitinating β -catenin and targeting it to the proteasome (11, 15). Conversely, in the absence of Wnt proteins, the DC phosphorylates β -catenin, which is further ubiquitinated by an E3 ubiquitin ligase (B-TrCP) and degraded in the proteasome (15). More recently, it was observed that the APC membrane recruitment protein 1 (AMER1) interacts with the APC/AXIN1 DC although its role is not yet completely understood (16). Additionally, controversies about β -catenin ubiquitination and degradation in the context of the AXIN1 complex and about the disassembly of the DC are noted in the literature (17).

In summary, in the canonical Wnt pathway, the binding of WNT proteins to FZD receptors suppresses β -catenin degradation, resulting in its cytoplasmic accumulation, followed by nuclear translocation, which finally releases the expression of certain genes involved in important cellular processes.

Gene expression in Wilms tumor

The noncanonical branch of the Wnt pathway resembles the canonical pathway only in its requirement for Wnt ligands [e.g., silberblick (WNT11) and pipetail (WNT5)], FZD receptors, and the cytoplasmic signal transduction molecule DSH. Upon binding of the noncanonical WNT proteins, complexes belonging to the PCP pathway are asymmetrically distributed in the proximal and distal cell membranes [reviewed in reference (18)]. The other downstream interactions remain unclear, but it is well established that the strabismus (STBM) and prickle (PK) proteins are involved, whereas AXIN1, GSK-3, and β -catenin are not. Despite some similarity between the vertebrate noncanonical Wnt pathway and the *Drosophila* PCP pathway, such as the involvement of DSH, no Wnt ligands are known to be involved in *Drosophila* PCP signaling.

Wnt signaling pathway in nephrogenesis

The Wnt/ β -catenin pathway is one of the multiple signaling pathways that cooperates in the initiation and progression of MET (19). Several members of the Wnt family have been implicated in the induction of epithelial renal vesicles. WNT4 is required and sufficient for the transition of the metanephric mesenchyme to epithelial cells (20). *Wnt4* is also required for tubulogenesis, and it acts through a noncanonical Wnt pathway (21). WNT9B, which is secreted by cells from the ureteric bud, induces the expression of WNT4, and its loss can be rescued with WNT1, a putative canonical Wnt signaling activator (22). Similarly, WNT6 induces tubulogenesis by activating *WNT4* transcription, which leads to the expression of early markers of kidney tubulogenesis PAX2, PAX8, SFRP2, and E-cadherin genes (23).

Another key protein involved in nephrogenesis is *SIX2* (*Sine oculis homeobox 2 gene*), a transcription factor essential for maintaining the self-renewing and multipotent characteristics of nephron progenitor cells (24). *WNT4* is an upstream regulator of *SIX2*, and decreased expression of *SIX2* results in the commitment of the progenitor cells to undergo differentiation via MET (25).

Wnt/ β -catenin signaling also regulates MET through the transcription repressor *SNAIL1*, which is downregulated during embryogenesis, and allows mesenchymal cells to differentiate into epithelia through MET (8, 26, 27).

Wilms tumors and the Wnt signaling pathways

Several studies have confirmed the activation of canonical Wnt signaling pathways in WT. The canonical Wnt signaling pathway, observed by nuclear positivity of β -catenin, is activated in approximately 15–25% of all WTs with a favorable histology (28).

Interestingly, nuclear accumulation of β -catenin in WTs is associated with a mutation in Wilms tumor 1 gene (*WT1*), which in turn is associated with WT cases that show stromal predominance,

Carraro et al.

where the nuclear positivity of the protein is largely confined to the mesenchymal cells (2, 28). However, the genetic and molecular mechanism that underlies these associations is not understood. The *WT1* is known as an inhibitor of Wnt/ β -catenin signaling (29, 30). The antagonistic activities of WT1 and β -catenin probably arises because the two proteins bind to a common transcriptional coactivator, CREB-binding protein. Additionally, the WT1 protein is essential for MET and, hence, for normal embryonic kidney development (31), and genetic deletion and/ or inactivating mutations in WT1 cause severe kidney disorders in mice (32).

APC is a tumor suppressor gene whose protein product is a component of the DC for β -catenin, thus its action negatively regulates the canonical Wnt signaling pathway, and this modulation appears to be essential in nephrogenesis. The APC protein exhibits distinct cellular localization during the differentiation process from the fetal kidney to the mature normal kidney. In earlier stages of kidney development, APC expression is nuclear; in later stages, it is cytoplasmic; and in intermediate stages, it is both nuclear and cytoplasmic. Interestingly, in WT samples, the localization of APC recapitulates that in the earliest stages of the fetal kidney, where APC expression is exclusively nuclear – a pattern that resembles the earliest stage in undifferentiated blastemal cells (2). Thus, the nucleocytoplasmic shuttling of APC may be critical in the context of the activation of the canonical Wnt pathway in WT, as well as in kidney development. Given that APC shuttles into and out of the nucleus (33), it is reasonable to speculate that the nuclear localization of APC might interfere with the export of β -catenin from the nucleus in Wnt-stimulated cells. In this context, nuclear APC positivity could be an indirect indication of Wnt signaling activation in WT (2). Although the involvement of the noncanonical Wnt pathway has been demonstrated in nephrogenesis (25) and cancer (34), few studies directly associate disturbances in this pathway and WT.

The noncanonical Wnt signaling pathway directly affects changes in the cytoskeleton, the PCP pathway, and the regulation of calcium release from the endoplasmic reticulum to control intracellular calcium levels via the Wnt/Ca^{2+} pathway.

PLCG2, a gene in the Wnt/Ca²⁺ pathway, is involved in the control of external calcium entry and in innate immune responses (35), and its expression is modulated during nephrogenesis (2). The mRNA and the protein levels of PLCG2 are reduced in kidney progenitor cells and increased in the mature kidney, where protein expression has been shown to be strongly positive in some cells of the nephron. Accordingly, the *PLCG2* expression pattern appears to be recapitulated in WT at the mRNA and protein levels, showing decreased or predominately negative expression, respectively, in WTs compared with differentiated kidneys (2).

WNT5A and WNT5B, which are members of the noncanonical Wnt signaling pathway, were also recently identified as being altered in WT. WNT5A is likely regulated by PAX2

Gene expression in Wilms tumor

and was found to be expressed at lower levels in WT than in the fetal kidney (36). The expression levels of *WNT5B* mRNA decrease during blastemal cell differentiation. *WNT5B* appears to be associated with the formation of cell polarity, as the WNT5B protein expression has only been observed after the renal vesicle formation, when cells are organizing to form kidney structures, such as glomeruli and tubules, where positivity is observed mainly in the apical cell membrane. Although the mRNA levels of *WNT5B* were observed to be elevated in WT (similar to the undifferentiated blastemal cells that give rise to WT) compared with the differentiated kidneys, the protein expression has been only detected in a minority of cases (37). The mechanism underlying WNT5A and WNT5B signaling remains to be elucidated.

These data provide strong evidence for the involvement of Wnt pathway-related genes in WT, whose pattern in the earliest stages of nephrogenesis is recapitulated in the tumor, marking the disruption of the complete differentiation of the kidney progenitor cells.

Wilms tumor and mutation repertoire

Mutations in *WT1*, a TSG, are present in approximately 20% of WTs. Other TSGs with inactivating mutations include *WTX*, which also occur in approximately 30% of cases (38). The WTX protein has been reported to negatively regulate the canonical WNT pathway, as part of the DC (39). Stabilizing mutations in *CTNNB1* (β -catenin), the major regulator of the canonical WNT pathway, are present in 15% of tumors. The well-known *TP53* gene has been found to be mutated in 5% of WT samples (4). Other genes that have been observed to be mutated in lower frequencies are *DIS3L2* (40), *FBXW7*, and *MYCN* (41). Together, mutations in these genes account for approximately 30% of WT samples.

Despite several lines of evidence supporting the overexpression of a number of downstream genes of the canonical WNT pathway in WT (42, 43), only two of the five currently known WT subtypes show clear evidence of canonical WNT pathway activation (44). These five subgroups (S1–S5) were defined by the hierarchical clustering analysis of expression data from genes of the canonical WT pathway. Only the subgroups, S1 and S2, showed evidence of strong WNT activation. Increased expression of LEF1 and FZD2 and decreased expression of CCND1 and JAG1 characterize the S1 subgroup. Interestingly, some samples from S2 subtype, lacking CTNNB1 or WTX mutations, also showed signals of strong Wnt activation. These findings suggest the presence of other mechanisms for canonical Wnt activation.

The repression of miRNA biogenesis through the inhibition of DROSHA and DICER1 expression impairs accurate kidney differentiation (45–47) and promotes tumorigenesis in several cell lines (48). Accordingly, mutations in genes involved in the miRNA biogenesis were recently identified in a higher proportion of WT samples and appear to be associated

Carraro et al.

with a predominant blastemal histology (5, 6, 49). Mutations in *DROSHA* are found in 12% of WTs, and a recurrent mutation (E1147K) has been shown to affect an RNase IIIb domain of the protein encoded (5, 6). The frequency of the E1147K mutation was further estimated in a validation set of 538 tumors being observed in 11% of the samples (49). If other genes in the miRNA biogenesis pathway are included, such as *DGCR8*, *TARBP2*, *XPO5*, and *DICER*, mutations in this pathway account for approximately 30% of WT samples.

The effect of the E1147K mutation in DROSHA is associated with a predominant reduction in the expression level of mature miRNAs (5, 6). miRNAs are critical regulators of gene expression, and consequently, the defective miRNA biogenesis observed in WT surely makes a crucial contribution to WT development. However, the effects of the deregulation of miRNAs on their target genes, particularly those assumed to be involved in the differentiation of the kidneys, have yet to be established.

Members of the cyclin gene family were recently described as being upregulated in blastemal-type WT samples, especially in samples with recurrent mutations in the SIX1/2 gene (50). This pattern may be an important underlying cause of the continued proliferation of the metanephric mesenchyme.

Interestingly, tumor samples with ectopic mesenchymal elements show upregulation of WNT-related genes, whereas tumors with epithelial elements do not. Moreover, samples from this "WNT-independent" subgroup often show perilobar nephrogenic rests (PLNRs), instead of intralobar nephrogenic rests, which may reflect the likely origin of these tumors in errors that occur later during kidney development (51). More recently, Walz et al. (49) showed a significant statistical association between the presence of PLNR and the mutations in miRNA processing genes. Further analysis may be necessary to clarify the consequences of this probable association.

Conclusion

The origins of WTs are closely related to the processes of kidney development. This is supported by the fact that genes involved in nephrogenesis are altered in WT (Figure 1). The involvement of WNT signaling pathway in kidney embryogenesis was demonstrated by alterations in several genes directly or indirectly. These alterations include mutations (*WT1, WTX, CTNNB1, TP53, DIS3L2, FBXW7,* and *MYCN*) or altered expression (*WNT5, APC,* and *PLCG2*) in WT. Additionally, mutations in genes from the miRNA biogenesis pathway (*DROSHA, DGCR8, TARBP2, XPO5,* and *DICER*) are found in a relatively high frequency in WT cases. Recently, it was demonstrated that one of these mutations affects the function of *DROSHA*, impairing the process of miRNA biogenesis. Currently, there is an urgent need to understand the possible interplay between the miRNA downregulation and the Wnt
Gene expression in Wilms tumor



Figure 1. Disruption of differentiation of the precursor cells of the kidney may result in Wilms tumor formation. Several molecules are associated with nephrogenesis and tumorigenesis. The available evidence suggests that the delicate balance in the expression pattern of these molecules during embryogenesis is the determinant of nephrogenesis and that disturbance in the expression leads to tumorigenesis. The differential expression of selected molecules is depicted beside each triangle (red: higher, green: lower). n: nuclear positivity of protein.

signaling pathway. This knowledge may lead to new perspectives in the design of more effective anticancer therapies for WT.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

The authors thank FAPESP for financial support (2013/23277-8). MM is funded by FAPESP 2015/06281-7.

References

- 1. Beckwith JB, Kiviat NB, Bonadio JF. Nephrogenic rests, nephroblastomatosis, and the pathogenesis of Wilms' tumor. Pediatr Pathol. 1990;10(1–2):1–36.
- Maschietto M, de Camargo B, Brentani H, Grundy P, Sredni ST, Torres C, et al. Molecular profiling of isolated histological components of Wilms tumor implicates a common role for the Wnt signaling pathway in kidney and tumor development. Oncology. 2008;75(1-2):81-91. http://dx.doi.org/10.1159/000155210

.1109/000100210

Carraro et al.

 Iglesias DM, Hueber PA, Chu L, Campbell R, Patenaude AM, Dziarmaga AJ, et al. Canonical WNT signaling during kidney development. Am J Physiol Renal Physiol. 2007;293(2):F494–500.

http://dx.doi.org/10.1152/ajprenal.00416.2006

- Huff V. Wilms' tumours: about tumour suppressor genes, an oncogene and a chameleon gene. Nat Rev Cancer. 2011;11(2):111–21. http://dx.doi.org/10.1038/nrc3002
- Rakheja D, Chen KS, Liu Y, Shukla AA, Schmid V, Chang TC, et al. Somatic mutations in DROSHA and DICER1 impair microRNA biogenesis through distinct mechanisms in Wilms tumours. Nat Commun. 2014;2:4802. http://dx.doi.org/10.1038/ncomms5802
- Torrezan GT, Ferreira EN, Nakahata AM, Barros BD, Castro MT, Correa BR, et al. Recurrent somatic mutation in DROSHA induces microRNA profile changes in Wilms tumour. Nat Commun. 2014;5:4039. http://dx.doi.org/10.1038/ncomms5039
- Costantini F, Kopan R. Patterning a complex organ: branching morphogenesis and nephron segmentation in kidney development. Dev Cell. 2010;18(5):698–712. http://dx.doi.org/10.1016/j.devcel.2010.04.008
- Zeisberg M, Neilson EG. Biomarkers for epithelial-mesenchymal transitions. J Clin Invest. 2009;119(6):1429–37. http://dx.doi.org/10.1172/JCI36183
- Kim YS, Yi BR, Kim NH, Choi KC. Role of the epithelial-mesenchymal transition and its effects on embryonic stem cells. Exp Mol Med. 2014;46:e108. http://dx.doi.org/10.1038/emm.2014.44
- Halt K, Vainio S. Coordination of kidney organogenesis by Wnt signaling. Pediatr Nephrol. 2014;29(4):737–44. http://dx.doi.org/10.1007/s00467-013-2733-z
- Chien AJ, Conrad WH, Moon RT. A Wnt survival guide: from flies to human disease. J Invest Dermatol. 2009;129(7):1614–27. http://dx.doi.org/10.1038/jid.2008.445
- van Amerongen R, Nusse R. Towards an integrated view of Wnt signaling in development. Development. 2009;136(19):3205–14. http://dx.doi.org/10.1242/dev.033910
- Sugimura R, Li L. Noncanonical Wnt signaling in vertebrate development, stem cells, and diseases. Birth Defects Res C Embryo Today. 2010;90(4):243–56. http://dx.doi.org/10.1002/bdrc.20195
- Veeman MT, Axelrod JD, Moon RT. A second canon. Functions and mechanisms of beta-catenin-independent Wnt signaling. Dev Cell. 2003;5(3):367–77.

- Kawakami T, Ren S, Duffield JS. Wnt signalling in kidney diseases: dual roles in renal injury and repair. J Pathol. 2013;229(2):221–31. http://dx.doi.org/10.1002/path.4121
- Major MB, Camp ND, Berndt JD, Yi X, Goldenberg SJ, Hubbert C, et al. Wilms tumor suppressor WTX negatively regulates WNT/beta-catenin signaling. Science. 2007;316(5827):1043–6.

http://dx.doi.org/10.1126/science/1141515

 Niehrs C. The complex world of WNT receptor signalling. Nat Rev Mol Cell Biol. 2012;13(12):767–79.

http://dx.doi.org/10.1038/nrm3470

 Schnell U, Carroll TJ. Planar cell polarity of the kidney. Exp Cell Res. 2014;S0014-4827(14):00492-3.

http://dx.doi.org/10.1016/j.yexcr.2014.11.003

- Lamouille S, Xu J, Derynck R. Molecular mechanisms of epithelial-mesenchymal transition. Nat Rev Mol Cell Biol. 2014;15(3):178–96. http://dx.doi.org/10.1038/nrm3758
- 20. Kispert A, Vainio S, McMahon AP. Wnt-4 is a mesenchymal signal for epithelial transformation of metanephric mesenchyme in the developing kidney. Development. 1998;125(21):4225–34.
- 21. Tanigawa S, Wang H, Yang Y, Sharma N, Tarasova N, Ajima R, et al. Wnt4 induces nephronic tubules in metanephric mesenchyme by a non-canonical mechanism. Dev Biol. 2011;352(1):58–69.

http://dx.doi.org/10.1016/j.ydbio.2011.01.012

- 22. Carroll TJ, Park JS, Hayashi S, Majumdar A, McMahon AP. Wnt9b plays a central role in the regulation of mesenchymal to epithelial transitions underlying organogenesis of the mammalian urogenital system. Dev Cell. 2005;9(2):283–92. http://dx.doi.org/10.1016/j.devcel.2005.05.016
- 23. Itaranta P, Lin Y, Perasaari J, Roel G, Destree O, Vainio S. Wnt-6 is expressed in the ureter bud and induces kidney tubule development in vitro. Genesis. 2002;32(4):259–68.
- 24. Kobayashi A, Valerius MT, Mugford JW, Carroll TJ, Self M, Oliver G, et al. Six2 defines and regulates a multipotent self-renewing nephron progenitor population throughout mammalian kidney development. Cell Stem Cell. 2008;3(2):169–81. http://dx.doi.org/10.1016/j.stem.2008.05.020
- 25. Park JS, Ma W, O'Brien LL, Chung E, Guo JJ, Cheng JG, et al. Six2 and Wnt regulate self-renewal and commitment of nephron progenitors through shared gene regulatory networks. Dev Cell. 2012;23(3):637–51. http://dx.doi.org/10.1016/j.devcel.2012.07.008

Carraro et al.

- 26. Liu Y. New insights into epithelial-mesenchymal transition in kidney fibrosis. J Am Soc Nephrol. 2010;21(2):212–22. http://dx.doi.org/10.1681/ASN.2008121226
- 27. Garcia de Herreros A, Baulida J. Cooperation, amplification, and feedback in epithelial-mesenchymal transition. Biochimica et Biophysica Acta. 2012;1825(2):223-8.
- 28. Perotti D, Hohenstein P, Bongarzone I, Maschietto M, Weeks M, Radice P, et al. Is Wilms tumor a candidate neoplasia for treatment with WNT/beta-catenin pathway modulators? – A report from the renal tumors biology-driven drug development workshop. Mol Cancer Ther. 2013;12(12):2619–27. http://dx.doi.org/10.1158/1535-7163.MCT-13-0335
- 29. Kim MS, Yoon SK, Bollig F, Kitagaki J, Hur W, Whye NJ, et al. A novel Wilms tumor 1 (WT1) target gene negatively regulates the WNT signaling pathway. J Biol Chem. 2010;285(19):14585–93.

http://dx.doi.org/10.1074/jbc.M109.094334

 Chang H, Gao F, Guillou F, Taketo MM, Huff V, Behringer RR. Wt1 negatively regulates beta-catenin signaling during testis development. Development. 2008;135(10):1875–85.

http://dx.doi.org/10.1242/dev.018572

- 31. Martinez-Estrada OM, Lettice LA, Essafi A, Guadix JA, Slight J, Velecela V, et al. Wt1 is required for cardiovascular progenitor cell formation through transcriptional control of Snail and E-cadherin. Nat Genet. 2010;42(1):89–93. http://dx.doi.org/10.1038/ng.494
- 32. Chau YY, Brownstein D, Mjoseng H, Lee WC, Buza-Vidas N, Nerlov C, et al. Acute multiple organ failure in adult mice deleted for the developmental regulator Wt1. PLoS Genet. 2011;7(12):e1002404.

http://dx.doi.org/10.1371/journal.pgen.1002404

33. Henderson BR, Fagotto F. The ins and outs of APC and beta-catenin nuclear transport. EMBO Rep. 2002;3(9):834–9.

http://dx.doi.org/10.1093/embo-reports/kvf181

 Gujral TS, Chan M, Peshkin L, Sorger PK, Kirschner MW, MacBeath G. A noncanonical Frizzled2 pathway regulates epithelial-mesenchymal transition and metastasis. Cell. 2014;159(4):844–56.

http://dx.doi.org/10.1016/j.cell.2014.10.032

35. Yu P, Constien R, Dear N, Katan M, Hanke P, Bunney TD, et al. Autoimmunity and inflammation due to a gain-of-function mutation in phospholipase C gamma 2 that specifically increases external Ca2+ entry. Immunity. 2005;22(4):451–65. http://dx.doi.org/10.1016/j.immuni.2005.01.018

- 36. Tamimi Y, Ekuere U, Laughton N, Grundy P. WNT5A is regulated by PAX2 and may be involved in blastemal predominant Wilms tumorigenesis. Neoplasia. 2008;10(12):1470–80.
- 37. Maschietto M, Trape AP, Piccoli FS, Ricca TI, Dias AA, Coudry RA, et al. Temporal blastemal cell gene expression analysis in the kidney reveals new Wnt and related signaling pathway genes to be essential for Wilms' tumor onset. Cell Death Dis. 2011;2:e224.

http://dx.doi.org/10.1038/cddis.2011.105

- Scott RH, Murray A, Baskcomb L, Turnbull C, Loveday C, Al-Saadi R, et al. Stratification of Wilms tumor by genetic and epigenetic analysis. Oncotarget. 2012;3(3):327–35. http://dx.doi.org/10.18632/oncotarget.468
- Ruteshouser EC, Robinson SM, Huff V. Wilms tumor genetics: mutations in WT1, WTX, and CTNNB1 account for only about one-third of tumors. Genes Chromosomes Cancer. 2008;47(6):461–70.

http://dx.doi.org/10.1002/gcc.20553

- Astuti D, Morris MR, Cooper WN, Staals RH, Wake NC, Fews GA, et al. Germline mutations in DIS3L2 cause the Perlman syndrome of overgrowth and Wilms tumor susceptibility. Nat Genet. 2012;44(3):277–84. http://dx.doi.org/10.1038/ng.1071
- 41. Williams RD, Al-Saadi R, Chagtai T, Popov S, Messahel B, Sebire N, et al. Subtypespecific FBXW7 mutation and MYCN copy number gain in Wilms' tumor. Clin Cancer Res. 2010;16(7):2036–45.

```
http://dx.doi.org/10.1158/1078-0432.CCR-09-2890
```

- 42. Corbin M, de Reynies A, Rickman DS, Berrebi D, Boccon-Gibod L, Cohen-Gogo S, et al. WNT/beta-catenin pathway activation in Wilms tumors: a unifying mechanism with multiple entries? Genes Chromosomes Cancer. 2009;48(9):816–27. http://dx.doi.org/10.1002/gcc.20686
- Zirn B, Samans B, Wittmann S, Pietsch T, Leuschner I, Graf N, et al. Target genes of the WNT/beta-catenin pathway in Wilms tumors. Genes Chromosomes Cancer. 2006;45(6):565–74.

http://dx.doi.org/10.1002/gcc.20319

- 44. Gadd S, Huff V, Huang CC, Ruteshouser EC, Dome JS, Grundy PE, et al. Clinically relevant subsets identified by gene expression patterns support a revised ontogenic model of Wilms tumor: a Children's Oncology Group Study. Neoplasia. 2012;14(8):742–56.
- 45. Harvey SJ, Jarad G, Cunningham J, Goldberg S, Schermer B, Harfe BD, et al. Podocyte-specific deletion of dicer alters cytoskeletal dynamics and causes glomerular disease. J Am Soc Nephrol. 2008;19(11):2150–8. http://dx.doi.org/10.1681/ASN.2008020233

Carraro et al.

 Shi S, Yu L, Chiu C, Sun Y, Chen J, Khitrov G, et al. Podocyte-selective deletion of dicer induces proteinuria and glomerulosclerosis. J Am Soc Nephrol. 2008;19(11):2159–69.

http://dx.doi.org/10.1681/ASN.2008030312

- 47. Zhdanova O, Srivastava S, Di L, Li Z, Tchelebi L, Dworkin S, et al. The inducible deletion of Drosha and microRNAs in mature podocytes results in a collapsing glomerulopathy. Kidney Int. 2011;80(7):719–30. http://dx.doi.org/10.1038/ki.2011.122
- Kumar MS, Lu J, Mercer KL, Golub TR, Jacks T. Impaired microRNA processing enhances cellular transformation and tumorigenesis. Nat Genet. 2007;39(5):673–7. http://dx.doi.org/10.1038/ng2003
- Walz AL, Ooms A, Gadd S, Gerhard DS, Smith MA, Guidry Auvil JM, et al. Recurrent DGCR8, DROSHA, and SIX homeodomain mutations in favorable histology Wilms tumors. Cancer Cell. 2015;27(2):286–97. http://dx.doi.org/10.1016/j.ccell.2015.01.003
- 50. Wegert J, Ishaque N, Vardapour R, Georg C, Gu Z, Bieg M, et al. Mutations in the SIX1/2 pathway and the DROSHA/DGCR8 miRNA microprocessor complex underlie high-risk blastemal type Wilms tumors. Cancer Cell. 2015;27(2):298–311. http://dx.doi.org/10.1016/j.ccell.2015.01.002
- 51. Fukuzawa R, Anaka MR, Weeks RJ, Morison IM, Reeve AE. Canonical WNT signalling determines lineage specificity in Wilms tumour. Oncogene. 2009;28(8):1063–75. http://dx.doi.org/10.1038/onc.2008.455

Chapter 11

Transcriptional Regulation of the Human Thromboxane A₂ Receptor Gene by Wilms' Tumour (WT)1

B. Therese Kinsella

UCD School of Biomolecular and Biomedical Sciences, UCD Conway Institute of Biomolecular and Biomedical Research, University College Dublin, Belfield, Dublin 4, Ireland

Author for correspondence: B. Therese Kinsella, PhD, UCD School of Biomolecular and Biomedical Sciences, UCD Conway Institute of Biomolecular and Biomedical Research, University College Dublin, Belfield, Dublin 4, Ireland. Email: Therese.Kinsella@UCD.IE

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch11

Copyright: The Author.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the author and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

The prostanoid thromboxane $(TX)A_2$ plays a fundamental role in vascular haemostasis and, more recently, is increasingly implicated in various neoplasms including in prostate, breast and bladder cancers, among others. In humans, TXA_2 signals through the TP α and TP β isoforms of the T prostanoid receptor (TP), two structurally related receptors that display both common, over-lapping but also distinct, isoform-specific physiologic roles. Consistent with this, while TP α and TP β are encoded by the same gene, the *TBXA2R*, they are

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

differentially expressed due to their transcriptional regulation by distinct promoters where promoter (Prm) 1 regulates TPa expression and Prm3 regulates TP β . While the clinical evidence for the role of the TXA₂-TP axis in neoplastic progression is increasing, few studies to date have investigated the role of the individual TPa/TP β isoforms in human cancer or indeed in most other diseases in which the TXA₂-TP axis is implicated. Focusing on TPa, this review details the current understanding of the factors regulating its expression and transcriptional regulation through Prm1, including in prostate and breast cancers. Emphasis is placed on the *trans*-acting transcriptional regulators that bind to *cis*-elements within the core and upstream regulatory regions of Prm1 under basal conditions and in response to cellular differentiation. A particular focus is placed on the role of the tumour suppressor Wilms' tumour 1 in the regulation of TPa expression through Prm1 in megakaryoblastic cells of vascular origin and in prostate and breast carcinoma cells. Collectively, this review details current knowledge of the factors determining regulation of the TXA₂-TPa axis and thereby provides a genetic basis for understanding the role of TXA₂ in the progression of certain human cancers.

Key words: Cancer; Gene; Thromboxane receptor; Transcription; Wilms' tumour 1

Introduction

Thromboxane $(TX)A_{\gamma}$ a prostanoid synthesized from arachidonic acid by the sequential enzymatic actions of cyclooxygenase (COX)-1/-2 and TXA synthase (TXS) predominantly in platelets and in monocytes/activated macrophages, plays an essential role in haemostasis regulating platelet aggregation and vascular tone (1). It also induces the constriction of various other types of smooth muscle (SM), including pulmonary, renal and prostate SM, and promotes vascular remodelling in response to endothelial injury contributing to neointimal hyperplasia and restenosis post-stenting (2-4). Accordingly, imbalances in the levels of TXA₂ or of its synthase (TXS) or of its receptor, the T prostanoid receptor (or, in short, the TP), are widely implicated in several cardiovascular, pulmonary, renal and prostate pathologies (1–3, 5). In humans and in other primates, the TP exists as two structurally related isoforms referred to as TP α and TP β , which are identical for their N-terminal 328 amino acid residues but which differ exclusively in their intracellular C-terminal domains (6). While TP α and TP β are encoded by the same TP gene, the *TBXA2R* (Figure 1), they are differentially expressed in several cell/tissue types being transcriptionally regulated by two different promoters referred to as Prm1 and Prm3, respectively, within the TBXA2R (7-10). Functionally, as members of the G-protein-coupled receptor (GPCR) superfamily, TPa and TP β both primarily couple to Gaq-mediated phospholipase C β activation, raising intracellular calcium levels in response to inositol phosphate turnover, but also readily couple to Gα12-mediated RhoA and to extracellular signal-regulated protein kinase (ERK) activation (Figure 2) (6, 11–14). In contrast, TP α and TP β undergo distinct mechanisms of agonist-induced



Figure 1. Structural organization of the TBXA2R gene. Panel A: In humans, a single TBXA2R gene, located on Chr19p13.3, encodes both the TP α and the TP β isoforms of the T prostanoid receptor (the TP), where promoter (Prm) 1 exclusively regulates TPa and Prm3 regulates TPB expression, respectively. The functional role of Prm2 is currently unclear but may contribute to the transcriptional regulation of TPa expression. Panel B: Structurally, TPa and TP β are identical for their N-terminal 328 aa residues but differ exclusively in their intracellular C-terminal domains, where TP α (343 aa) and TP β (407 aa) have 15 and 79 unique aa residues, respectively. Panel C: Schematic representation of Prm1 and the main cis-acting elements and trans-acting factors that regulate TPa expression through Prm1. Prm1 is located at -8500 to -5895 relative to the ATG translational initiation codon within the TBXA2R. Under basal or resting cell conditions, the Core Prm1 region, located between -6294 and -5895, is under the transcriptional regulation of Sp1, Egr1 and NF-E2 but also contains a repressor region (RR3) that is regulated by Wilms' tumour (WT) 1. Prm1 also contains several upstream repressor regions (URR) and upstream activator regions (UARs). The haematopoietic-specific factors Gata-1 and Ets-1 are the main trans-acting factors that bind and regulate UAR1 in megakaryoblastic HEL92.1.7 cells and Ets-1/2 and Oct-1 regulate UAR2 (not shown). In contrast, WT1 is the main *trans*-acting factor that binds to multiple GC-enriched *cis*-elements within URR1, URR2 and RR3 to repress Prm1-directed TPa expression under basal/resting cellular conditions (e.g., in non-differentiated HEL cells).



Figure 2. Summary of the main signalling cascades regulated by TPa and TPβ. 1 (Black): Both TPa and TP β primarily couple to Gaq-mediated phospholipase (PL)C β activation, leading to the generation of IP_3 and mobilization of intracellular calcium (Ca²⁺). Elevation in intracellular Ca²⁺ is the main signalling event that triggers TXA₂-induced platelet activation, including thrombosis, and constriction of various types of smooth muscle (SM), including vascular (V), renal, pulmonary and prostate SM. 2 (Grey): TP α and TP β also couple to G α_{12} -mediated activation of RhoGEF (Rho guanine nucleotide exchange factor), leading to activation of RhoA. Activated (GTP-bound) RhoA, in turn, interacts with a range of effector proteins including Rho kinase (K) 1/2, leading to Ca²⁺-independent (V)SM contraction, to general reorganizations of the cellular cytoskeleton and to a host of events that promotes tumour cell migration and metastasis. In addition, activated RhoA can also interact and activate the effector protein kinase C-related kinase (PRK) 1. PRK1 can also interact with the activated (DHT-bound) androgen receptor (AR) which, in turn, can enhance AR-dependent transcriptional activation by promoting phosphorylation of histone H3 at Thr11 and, hence, androgen-induced chromatin remodelling. 3 (Purple): It has been recently discovered that TPa and TPB can directly interact with and activate PRK1 which, in turn, can also phosphorylate histone H3 at Thr11 to augment androgen-induced chromatin remodelling in response to $TP\alpha/TP\beta$ signalling. 4 (Pink): TPa and TP β also lead to (i) activation of the extracellular signal-regulated protein kinase (ERK) 1/2 cascades (intermediate steps not shown), as well as to (ii) transactivation of the epidermal growth factor receptor (EGFR), both of which (i and ii) account for the ability of TXA, to promote cell proliferation and mitogenesis, including tumour progression. Note: The inflammatory or immune-modulatory roles of the TXA₂-TPa/TP β axis are not shown but may also contribute to their role in certain cancers.

homologous (15, 16) and heterologous (17–19) desensitization to differentially regulate their intracellular signalling. Most notably, signalling by TPa, but not by TP β , is completely desensitized/inhibited by the counter-regulatory *anti*-platelet and vasodilatory agents prostacyclin/prostaglandin I₂ and nitric oxide, which is mediated by direct protein kinase (PK) A and PKG phosphorylation of TPa at Ser³²⁹ and Ser³³¹, respectively, the very first residues within its unique carboxyl-terminal tail domain of TPa and divergent from those of TP β (13, 18). The conclusion from those studies is that TPa is the TP isoform essential for haemostasis/ thrombosis, while the role of TP β in this pathophysiologic process remains unclear (13, 18). Hence, TPa and TP β have both shared and unique patterns of expression and function to mediate the (patho)physiologic actions of the potent autocrine/paracrine mediator TXA₂ in human health and disease.

The role of thromboxane in cancer

In addition to its prominent role within the vasculature, there is growing evidence highlighting a central role for TXA, in human cancers (20, 21). In recent years, evidence supporting this hypothesis has been strengthened by several longitudinal studies showing the prophylactic benefits of long-term daily use of Aspirin in reducing the risk of many prevalent cancers, predominantly gastrointestinal but also breast, lung and prostate cancers (PCa), with numerous clinical trials completed or underway testing the benefits of Aspirin and other COX-1/-2 inhibitors in chemoprevention (22-29). While those longitudinal studies do not specify which COX-1/-2-derived prostanoid metabolite(s) is actually lowered by Aspirin to account for its prophylactic benefits in cancer risk reduction, recent reports strongly suggest that some/many of its anti-cancer effects may be due to its ability to inhibit TXA, generation, as stated a prostanoid more typically associated with thrombosis and cardiovascular disease (20, 21). Indeed, it has long been known that platelets, the main source of TXA, and key target of Aspirin, play a key role in cancer progression promoting cancer cell metastasis, immune evasion and extravasation (30). Furthermore, increased levels of TXA_2 and expression of its synthase and its T prostanoid receptor, the TP, occur in a number of prevalent cancers including, for example, strongly correlating with bladder (31), prostate (32, 33), colorectal (34, 35) and non-small-cell lung cancer (36). Mechanistically, the role of TXA₂ in neoplastic progression is at least partly explained by the ability of the TXA₂-TP axis to regulate key mitogenic/ERK- and RhoA-mediated signalling cascades that contribute to tumour development and metastasis (12, 14) and also by its ability to regulate local inflammation and immunity (37-42), including within the tumour (Figure 2; summary of TXA₂-TP signalling). Hence, aside from its regulation of ERK- and RhoA-mediated processes (Figure 2) (12, 14), TXA, is a potent proinflammatory and immune-modulatory agent being abundantly produced in monocytes/activated macrophages and promotes monocyte chemoattractant protein-1 expression in tumours, recruiting tumour-associated macrophages, and negatively

regulates the interaction between T-cells and dendritic cells, a process essential for adaptive/acquired immunity (38, 39, 43, 44). Moreover, TXA_2 is critical for early B-cell development, also with implications for its role in tumour-infiltrating B-cells (42, 45). Hence, due to its role in tumour growth and metastasis combined with its ability to regulate local inflammation and immunity, the TXA_2 -TP axis can impact at multiple levels within the tumour environment.

Genome-wide association studies also reveal that certain single nucleotide polymorphisms within the TXS gene (the TBXAS1) may predispose individuals to breast cancer (46), while inhibition of TXS activity enhances apoptosis of lung carcinoma A549 cells in vitro, implicating a role for TXA, also in tumour cell survival (36). In the prostate, an increased expression of TXS and the $TP\alpha/TP\beta$ isoforms directly correlate with the tumour Gleason score and pathologic stage (32, 33, 47), where expression of both TXS and the TP is mainly found in areas of perineural invasion, a recognized mechanism by which PCa cells invade the prostatic capsule and metastasize to other tissues (20, 32). Significantly in the context of PCa, through detailed mechanistic studies, we recently discovered that both the TP α and the TP β isoforms directly interact with and regulate signalling by protein kinase C-related kinase/ protein kinase novel (PRK/PKN) (48), a family of 3 AGC kinases that act immediately downstream of phosphatidylinositol 3'kinases, and are strongly, yet differentially, implicated in several cancers (49–51) and in B-cell development (52). Indeed, in addition to acting as Rho GTPase effectors, activation of the PRKs (e.g., PRK1) in response to androgen receptor (AR) signalling within the prostate catalyses phosphorylation of histone (H)3 at Thr11 (H3pThr11) which, in turn, serves as a specific epigenetic marker, and gatekeeper, of androgen-induced chromatin remodelling and transcriptional activation (48, 53-55). Hence, owing to their ability to regulate RhoA-/C-mediated responses, including metastatic processes, combined with their epigenetic priming of tumour cells, members of the PRK family are key chemotherapeutic targets particularly in castrate-resistant prostate cancer, the metastatic lethal form of PCa that occurs following androgen deprivation therapy (53, 55, 56).

Indeed, our research shows that TPa-/TP β -mediated PRK1 activation not only leads to histone H3 threonine 11 phosphorylation in response to TXA₂ but can also cooperate with the AR to enhance the androgen-induced chromatin remodelling (H3pThr11) and transcriptional activation (48). Collectively, these studies raise the exciting possibility that TXA₂, through its ability to directly regulate PRK-induced H3pThr11, may be a strong epigenetic regulator, thereby adding to the range of possible mechanisms, whereby the Aspirin-target TXA₂ may influence the neoplastic growth. Added to this complexity, we recently established that the TPa and TP β isoforms differentially associate with and regulate signalling by the other individual members of the PRKs (PRK1/PKNa, PRK2/PKN γ , PRK3/PKN β) (57). Furthermore, consistent with our previous studies involving PRK1 (48), *si*RNA disruption

of PRK1 and PRK2, but not PRK3, expression eliminates TP-mediated cancer cell responses (proliferation, anchorage-independent growth, migration) and H3pThr11 phosphorylation in the prostate carcinoma PC-3 cell line (57). Identification of a direct, functional interaction of both TP α and TP β with the PRKs provides yet another molecular link accounting for the role of TXA₂ in tumour progression, particularly in prostate and other cancers in which the TXA₂-TP and PRKs are increasingly implicated. Critically, as stated, it suggests that the TXA₂-TP axis may serve as an epigenetic regulator, adding to the range of possible mechanisms whereby the Aspirin-target TXA₂ may influence neoplastic growth.

Factors determining transcriptional regulation of TPa in platelet progenitor megakaryoblastic cells

Collectively, these and numerous other studies provide significant mechanistic insights into the role of TXA₂ and of the TPs (TP α /TP β) in cancer progression. However, with only limited exceptions (20, 31), few of those studies investigated the roles of the individual TP α or TP β isoforms or examined their transcriptional regulation in cancer. To address this and focussing on TP α , the predominant isoform expressed in most cell/tissue types (10), we recently examined its expression in prostate and breast cancer and identified a key role for the tumour suppressor gene product Wilms' tumour (WT)1 in its transcriptional regulation (58). Prior to presenting and discussing these findings, it is first relevant to review knowledge on the transcriptional regulation of TP α in the haematopoietic system where most data and insight is available.

As stated, while TP α and TP β are encoded by the same *TBXA2R* gene (Figure 1), they are differentially expressed being regulated by distinct promoters, whereby promoter (Prm)1 exclusively regulates TP α expression and Prm3 regulates TP β (7–9, 59, 60). Through initial studies carried out in the platelet progenitor megakaryoblastic human erythroleukaemia (HEL) 92.1.7 and K562 cell lineages (9, 60, 61), the transcription factors Sp1 (stimulating protein 1), early growth response 1 (Egr1) and NF-E2 were identified as the key trans-acting factors that bind to the 'core promoter region' of Prm1 to drive basal expression of TPa mRNA (9). In addition, several functional upstream activator regions (UARs; UAR1 and UAR2) and upstream repressor regions (URRs; URR1, URR2 and RR3, where repressor region 3 specifically lies within the core promoter; Figure 1) were identified within Prm1 (9). While GATA-1, Ets-1, Ets-2 and Oct-1 were identified as the main *trans*-acting transcription factors that regulate the UARs in the megakaryoblastic lineages, the tumour suppressor gene product WT1 was found to bind to several GC-enriched consensus cis-elements within the repressor regions (URR1, URR2 and RR3) to repress Prm1, maintaining TPa expression at relatively low levels when cells were cultured under basal conditions (60). However, following differentiation of the pluripotent megakaryoblastic cell lineages to the platelet phenotype, it was established that TPa expression was strongly upregulated, and this occurred

through a complex transcriptional mechanism involving coordinated: (i) *alleviation* of TPa/ Prm1 repression by WT1 by displacement of its binding to its consensus GC-enriched *cis*acting elements within the URRs of Prm1, (ii) *induction* of TPa expression by binding of the transcriptional activator Egr1 to the same *cis*-acting GC elements within Prm1, followed by (iii) *sustained upregulated expression* of TPa through binding of Sp1 also to the same *cis*acting GC elements within Prm1 (Figure 3) (61). Hence, WT1 plays a central role in repressing TPa expression by binding to multiple *cis*-acting elements within the repressor regions (URR1, URR2 and RR3) of Prm1, maintaining TPa expression at low levels under basal/ non-differentiated conditions. However, in response to cellular differentiation, WT1 repression is lifted in favour of sequential high-affinity binding of the transcriptional activators Egr1 followed by Sp1 to the same *cis*-acting GC elements within Prm1 to induce (by Egr1) and maintain (by Sp1) high levels of TPa expression following differentiation, (Figure 3) (61).

In addition to its recognized role in normal and aberrant haematopoiesis (62), WT1 was initially described as a tumour suppressor in Wilms' tumour (WT), a rare form of renal cancer (63–65), but can also play an oncogenic role in certain cancers (66–69). Considering the recognized role of WT1 in WT of the kidney and in other cancers while also acting as a key transcriptional repressor/regulator of TPa expression in megakaryoblastic HEL and K562 cell lineages combined with the increasing awareness of the role of the TXA,-TP axis in neoplastic progression, we recently investigated the possible regulation of TPa/Prm1 by WT1 in prostate and breast cancer, including in the model prostate PC3 and breast MCF-7 [a model oestrogen receptor/oestrogen receptor (ER)-positive breast cancer cell line] and MDA-MB-231 (a model oestrogen receptor/ER, progesterone receptor/PR and Her2/neu triplenegative breast cancer cell line) carcinoma cell lines, respectively. In brief and consistent with the findings in the megakaryoblastic lineages, it was established that WT1 can repress Prm1-directed TPa expression in both the prostate and the breast cancer lineages. Overall, as elaborated upon in detail later in this chapter, the study provided a comprehensive molecular analyses of the factors regulating the TPa expression through Prm1 in the prostate and breast and suggested that aberrant regulation by/or dysfunction of the tumour suppressors WT1, along with hypermethylated in cancer (HIC) 1, may account at least, in part, for the increased association of TXA₂/TP signalling with certain prostate and breast cancers and, potentially, in other cancers in which TXA,, WT1 and/or HIC1 are implicated (58). The reader is referred to the original study for full details on the role of HIC1 in its regulation of TPa/Prm1 expression (58), while this communication will mainly focus on the role of WT1.

Role of Wilms' tumour 1 in regulating TPa expression in prostate and breast cancer

The WT1 gene encodes a zinc finger transcription factor critical for development of the genitourinary, haematopoietic and central nervous systems (62, 70). The finding that mutations



Figure 3. Proposed model for PMA-mediated increases in Prm1 activity. Panels A-E: Proposed model for PMA induction of Prm1/TPa mRNA transcriptional regulation in HEL92.1.7 and K562 megakaryoblastic cells. In resting cells, WT1 binds in a cooperative manner to multiple adjacent GC elements (at -8345, -8281, -8146 and -7831) within Prm1 to impair transcription initiation by the basal transcription apparatus (BTA) and thereby repressing TPa mRNA expression (Panels A and B; Repression). In response to exposure to PMA for ~5 h, the up-regulated expression of Egr1 results in increased high-affinity binding of Egr1 to Prm1, thereby activating Prm1-directed TBXA2R transcription to up-regulate TPa mRNA expression (Panel C; Induction). After exposure to PMA for ~8 h, a further, enhanced increase in Egr1 binding coincides with nuclear export and a resulting reduction in WT1 binding (Panel D; Induction). The decrease in WT1 binding results in de-repression of Prm1 and in a further increase in Egr1 binding, leading to a more pronounced transcriptional activation of Prm1 by the BTA (Panel D; Induction). Following the prolonged exposure to PMA for ~16 h, decreased Egr1 binding coincides with its rapid protein turnover (Panel E). PMA-mediated differentiation of cells can also lead to phosphorylation of Sp1 and/or its increased expression, enhancing its DNA-binding activity. Therefore, the increased affinity of Sp1 for Prm1, coinciding with the decreased Egr1 expression, facilitates binding of Sp1 to Prm1, thereby resulting in a sustained increase in Prm1 activity and TPa expression as the differentiation of HEL and K562 cells progresses toward the platelet phenotype (Panel E; Maintenance). Panel F: Representative chromatin immunoprecipitation (ChIP) analysis of Sp1, Egr1 and WT1 binding in vivo to Prm1 of the TBXA2R in HEL cells as a function of PMA-induced cellular differentiation. Note: Data presented in this figure were reproduced from our previous study (62), and the reader is referred to the original manuscript for the experimental details that led to the proposed model (panels B-E).

within the WT1 gene are a leading cause of the childhood renal cancer WT first led to the suggestion that it might serve as a tumour suppressor (64). However, the fact that non-mutated/wild-type WT1 is also abundantly expressed in a variety of cancers, including cancers of the breast (71), oesophagus (72) and pancreas (73), indicated that WT1 might also play an oncogenic role.

In addition to acting as a regulator of transcription, WT1 can also play a role in post-transcriptional regulation, including in RNA splicing (74, 75), and also in translation (76). Such diverse functions are likely due to fact that WT1 protein exists as multiple isoforms that arise owing to differential splicing and/or the use of multiple translational initiation sites within the WT1 gene (77). Of the most prevalent WT1 isoforms, the best characterized are the variants that differ due to the presence or absence (+/-) of exon 5 and +/-KTS (Lys-Thr-Ser) sequences (Figure 4). More specifically, differential splicing at these two sites yields four different isoforms, namely (+/+), (+/-), (-/+) and (-/-), each of which differ in respect of exon 5 and KTS sequences, respectively. While the -KTS isoforms can act as transcriptional repressors or activators, the +KTS isoforms do not readily bind DNA and, therefore, are less active in the process of transcription [reviewed in reference (62)]. WT1 has four Kruppel-like C₂H₂ fingers within its C-terminal region that share similarity with those of the aforementioned Egr1, another prominent member of the zinc finger family of transcription factors. While the zinc finger domain of WT1 can facilitate its DNA at the consensus Egr1 DNA-binding site (consensus sequence 5'GCG(G/T)GGGCG3'), the binding affinity of WT1 for the Egr1 consensus site is significantly less than that of Egr1 itself (78). In addition to binding to WT1 and/or Egr1 cis-acting elements, WT1 can also bind to another motif termed the Wilms' tumour element (consensus sequence, 5'GCGTGGGAGT3') (79). Hence, depending on the cellular context and/or on the particular promoter, WT1 can therefore serve as a transcriptional repressor or activator. This is exemplified in the case of c-Myc where over-expression of WT1 in K562, a HEL cell line, and in breast cancer cells activates the c-Myc promoter (80) but WT1 represses the c-Myc promoter in HeLa cells (81). WT1 can also complex with other DNA-binding trans-acting co-factors, such as p53 (82), and with certain co-activators or co-repressors, such as CBP (83) or BASP1 (84), respectively, to regulate transcription. It is the identity of these co-factor-binding partners that determines whether WT1 serves as an activator or repressor during transcription. In the case of the TBXA2R, the overwhelming evidence is that WT1 predominantly acts as a transcriptional repressor by binding to multiple GC-enriched cis-acting elements within Prm1 to suppress TPa expression in pluripotent megakaryoblastic cells but that in response to cellular differentiation, this repression is lifted coinciding with the increased expression and binding of Egr1 to the same GC elements to induce TPa expression (Figure 3) (9, 60, 61). In turn, following Egr1-mediated induction, subsequent binding of the constitutive Sp1 to the same *cis*-elements within Prm1 maintains the expression of TPa at high levels in the fully



Figure 4. Schematic representation of Wilms' tumour (WT) 1 isoforms. WT1, encoded by the tumour suppressor gene WT1, can exist as multiple isoforms depending on the translational start site and the inclusion/exclusion of amino acids encoded by exon 5 (encodes a 17 aa sequence) or exon 9 [encodes the 3 aa sequence KTS (Lys-Thr-Ser) located between Zinc Finger (ZF) domains 3 and 4]. The best characterized isoforms are the \pm exon 5 and the \pm exon 9/KTS isoform variants, representing the four most abundantly expressed forms of WT1 with a molecular mass of 52-54 kDa. In addition, the initiation of translation can occur at an upstream CUG codon (-73), resulting in the generation of larger WT1 isoforms with a mass of 62–64 kDa or alternatively can occur at a downstream AUG codon (+127) giving rise to smaller WT1 isoforms of 32-34 kDa. WT1 has four Kruppel-like C₂H₂ zinc fingers, ZF1-ZF4, within its C-terminal DNA-binding domain that share significant identity with those of the early growth response (Egr)1, another member of the zinc finger family of transcription factors. While the -KTS isoforms of WT1 can either repress or activate transcription, the +KTS isoforms have a reduced ability to bind DNA and therefore are less transcriptionally active (62). The increased affinity of Egr1 relative to that of WT1 for binding to the same cis-acting elements accounts for why Egr1 can displace WT1 binding when progressing from transcriptional repression by WT1 to induction by Egr1 as exemplified by the coordinated regulation of Prm1/TPa expression during cell differentiation (Figure 3).

differentiated state (61). The known increased affinity of Egr1 relative to that of WT1 for binding to the same *cis*-acting elements accounts for why Egr1 can displace WT1 binding when progressing from transcriptional repression by WT1 to induction by Egr1 as exemplified by the coordinated regulation of TP α /Prm1 during cell differentiation (Figure 3) (61, 78).

In our recent studies investigating the expression and the transcriptional regulation of TPa through Prm1 in prostate and breast cancer, immunohistochemical analysis confirmed that the expression of TPa correlated with increasing prostate and breast tissue tumour grade (Figure 5), while stimulation of the prostate (PC3) and breast (MCF-7 and MDA-MB-231) carcinoma cell lines with the TXA₂ mimetic U46619 increased both cell proliferation and migration (58). Collectively, these data provided further evidence of a role for the TXA₂-TP signalling axis in prostate and breast cancer progression. In order to identify the factors



Figure 5. Analysis of TP α expression in the prostate and breast. Panels A and B: Immunohistochemical analysis of prostate and breast tissue (benign and increasing tumour grade) screened with an affinity-purified *anti*-TP α antibody (200× magnification; counterstained with haematoxylin). The increased TP α expression coincides with an increased prostate and breast cancer tumour grade. Panel C: Immunohistochemical analysis of full-face benign prostate tissue screened either (i) in the absence of a primary antibody or with the (ii) affinity-purified *anti*-TP α antibody. The specificity of the *anti*-TP α antibody was confirmed, whereby the (iii) immunogenic TP α peptide, (iv) but not a TP β specific-peptide, competed out the *anti*-TP α immune-staining. The arrows in (ii) and (iv) indicate specific detection of TP α expression in the prostate tissue. All sections were counterstained with haematoxylin, and images shown were captured at 200× magnification. *Note*: Data presented in this figure were reproduced from our previous study (58).

regulating TPa in the prostate and breast through Prm1, genetic-based reporter analyses confirmed that the repressor regions, designated URR1, URR2 and RR3 and previously identified within Prm1, are functional in the prostate and breast carcinoma lineages. Furthermore, in each of the prostate and breast carcinoma lineages studied, over-expression of WT1 repressed TPa mRNA and Prm1-directed reporter gene expression, while chromatin immunoprecipitation analysis confirmed that WT1 binds *in vivo* to each of the consensus GC-enriched *cis*-elements within the repressor regions of Prm1. Furthermore, in the prostate and breast cellular systems, it was established that the tumour suppressor HIC1 represses TPa mRNA expression through its binding to a functional *cis*-element, referred to as the HIC1(b) element, within Prm1 in PC3 and MCF-7 cells, while a second HIC1 element,

referred to as HIC1(a), was identified within Prm1 through bioinformatics analyses but was not found to be functional (58).

Among the many cell-specific differences we observed in the transcriptional activity of Prm1 in the prostate and breast carcinoma-derived cell types, it was noteworthy that the UAR1/UAR2 repressor regions within Prm1, previously identified in the megakaryoblastic HEL92.1.7 and K562 cell lines (9) where they are regulated by GATA-1, Ets-1, Ets-2 and Oct-1, were not found to be functionally active in the prostate PC3 or breast MCF-7 lineages (Table 1). These observations pointed to clear cell-/tissue-specific differences in the regulation of TPa expression through Prm1 in the haematopoietic system versus prostate and/or breast tissues. Moreover, as summarized in Table 1, our analysis of Prm1 in the prostate and breast carcinoma lineages revealed additional cell-specific differences in the upstream regulatory regions. Included in this is a prostate-specific Novel URR identified in PC3 cells, a novel UAR1 in PC3 and MCF-7 cells and an additional Novel UAR2 in PC3, MCF-7 and MDA-MB-231 cells (58). Detailed bioinformatic analysis by us revealed a putative oestrogen response element (ERE) within the Novel URR (-7962 to -7859; Table 1), suggesting that this region might function as a binding site for the oestrogen and/or ARs (85). The finding that oestrogen production increases in men with age, mainly due to aromatase conversion of androgens to oestrogens, indicates that like androgens (86), oestrogens may also play a role in PC progression (87, 88). Whether the ERE within the Novel URR of Prm1 regulates the TPa expression in the prostate remains to be investigated.

With regard to the Novel UAR1 (-7504 to -6848; Table 1) identified in both the prostate PC3 and the ER-positive breast MCF-7 lines, but not in the triple-negative breast cancer MDA-MB-231 cell line, several putative *cis*-acting binding elements were identified through bioinformatics, including multiple EREs that lie in close proximity to each other. While remaining speculative, binding of the ER to some or all of these adjacent EREs may explain why the Novel UAR1 repressor region is functional in both the ER-positive PC3 and the MCF-7 lines but not active in the ER-negative MDA-MB-231 cell line.

In terms of the novel activator regions identified within Prm1 and confirmed to be transcriptionally active in all prostate and breast cell lineages examined, the Novel UAR2 located between -6648 and -6492 of Prm1 was of particular note (Table 1). Specifically, similar to that found in all other repressor regions (UAR1, UAR2 and RR3) within Prm1, putative binding elements for Egr1 and stimulating protein (Sp) 1 were also identified within this novel prostate-/breast-specific Novel UAR2 region of Prm1. As stated, our previous studies in differentiated megakaryoblastic HEL and K562 lineages established that both the inducible Egr1 and the constitutive Sp1 factors bind to multiple GC-consensus WT1 *cis*-elements within the UAR1, UAR2 and RR3 repressor regions of Prm1 to strongly up-regulate the TPa mRNA expression (9, 60, 61). Hence, it is indeed possible, if not likely, that the expression of TPa through Prm1 in the prostate and

		Cell type ¹			
Regulatory regions initially identified in megakaryoblastic lineages ²					
Regulatory region ²	Position within Prm1 ³	HEL 92.1.7	PC3	MCF-7	MDA-MB-231
URR1	-8500 to -7962	Yes	Yes	Yes	Yes
URR2	-6848 to -6648	Yes	Yes	Yes	Yes
RR3	-6258 to -6123	Yes	Yes	Yes	Yes
UAR1	-7962 to -7859	Yes	No	No	No
UAR2	-7859 to -7504	Yes	No	No	Yes
Core	-6320 to -5895	Yes	Yes	Yes	Yes
Novel regulatory regions identified in prostate or breast cancer cell lineages ¹					
Regulatory region ¹	Position within Prm1	HEL 92.1.7	PC3	MCF-7	MDA-MB-231
Novel URR	-7962 to -7859	No	Yes	No	No
Novel UAR1	-7504 to -6848	No	Yes	Yes	No
Novel UAR2	-6648 to -6492	No	Yes	Yes	Yes

Table 1. Activator and repressor regions within Prm1

¹This table has been adapted from our previous study (58).

²From references (9, 60, 61).

³The nucleotide numbers given are relative to the translational initiation codon at +1 within the *TBXA2R*.

breast may be subject to a similar type of complex transcriptional regulation involving occupancy of the common/shared and Novel prostate-/breast-specific GC-enriched *cis*-acting elements that can act as consensus binding sites for WT1, Egr1 and/or Sp1, where temporal occupancy may possibly be determined by the (patho)physiologic setting. Clarity on this matter remains to be experimentally determined and is necessary to shed further light on the transcriptional regulation of TP α in both the normal and the malignant prostate/breast tissue and potentially in other tissues in which the TXA₂-TP axis is implicated. Critically, given that Egr1 serves as a master regulator in several key aspects of prostate and breast cancer progression (89, 90), investigation of the interplay between WT1 and Egr1 in the regulation of TP α expression through Prm1 within the *TBXA2R* as a function of tumour grade merits detailed investigation.

Conclusions and future perspectives

The prostanoid TXA_2 plays a central role in haemostasis and is widely implicated in a range of cardiovascular, renal, pulmonary and prostate diseases (1-3, 5). In humans, TXA_2 signals

through TP α and TP β , two structurally related TP isoforms that display both common, overlapping and isoform-specific physiologic roles (6). TP α and TP β are encoded by the same gene, the TBXA2R, but are differentially expressed being regulated by distinct promoters where Prm1 regulates TPa expression, mainly involving NF-E2, Sp1, GATA-1, Ets-1, WT-1/Egr1, and Prm3 regulates TP β , involving cFos/cJun and Oct-1/-2 (7–9, 59, 60). While the clinical evidence for the role of the TXA₂-TP axis in neoplastic progression is increasing, few studies to date have investigated the role of the individual TP α /TP β isoforms in human cancer or indeed in most other diseases in which TXA₂ is implicated (20, 31). Focussing on TPa, we investigated its expression and transcriptional regulation through Prm1 in prostate and breast cancer and established that the tumour suppressor protein WT1 plays a key role in regulating its expression in both tissue types. Critically, it was established that WT1 can repress TPa expression through binding to multiple GC-consensus *cis*-elements within previously recognized regulatory regions within Prm1, while several other prostate- and/or breast-specific novel regulatory regions were identified accounting for cell-/tissue-specific regulation of TPa. Taken together with previous studies in platelet progenitor megakaryoblastic lineages (9, 60, 61), the findings provide a strong genetic basis for understanding the diverse physiological roles played by TXA₂-TP axis within the vasculature and other systems, including potentially in the progression of certain human cancers. These studies may also provide a mechanistic basis accounting, at least in part, for the prophylactic benefits of Aspirin in reducing certain cancer risks by lowering the overall TXA, levels.

WT1 has recently emerged as an important target in immune-therapy approaches to treat certain cancers, and ongoing clinical trials involving the WT peptide 1 vaccination are proving positive in reducing tumour growth in breast and lung cancers, in leukaemia and, more recently, in glioblastoma (91–93). As the TXA_2 -TP axis has been implicated in the development of prostate and breast cancer (20, 33, 47, 94, 95), the findings herein suggest that aberrant WT1 regulation of TP α expression may contribute to such cancers and potentially to WT itself. In addition, bearing in mind that the TP α and TP β isoforms display a number of important functional similarities but also differences in terms of their signalling (6) and regulation including in certain cancers (96, 97), coupled with the fact that they are regulated by distinct promoters within the *TBXA2R* gene, it will be of considerable interest to investigate the expression and transcriptional regulation of TP β in human cancers, in particular in prostate and breast cancer.

Abbreviations

ADT, androgen deprivation therapy; AR, androgen receptor; ChIP, chromatin immunoprecipitation; COX, cyclooxygenase/prostaglandin G2/H2 synthase; CRPC, castrate-resistant prostate cancer; C-tail, carboxyl-terminal tail; Egr1, early growth response 1; ER, oestrogen receptor; ERE, oestrogen response element; ERK, extracellular signal-regulated protein kinase; GPCR, G-protein-coupled receptor; HEL, human erythroleukaemia; H3Thr11, histone H3 threonine 11; IHC,

immunohistochemistry; PCa, prostate cancer; PDK-1, 3-phosphoinositide-dependent protein kinase-1; PG, prostaglandin; PI3'K, phosphatidylinositol 3'kinase; PK, protein kinase; PRK, protein kinase C-related kinase; RR, repressor region; SDM, site-directed mutagenesis; Sp1, stimulating protein 1; TP, thromboxane prostanoid receptor; TX, thromboxane; TXS, TXA synthase; UAR, upstream activator region; URR, upstream repressor region; WT1, Wilms' tumour 1.

Conflict of Interest

The author declares no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgements

This work was funded by an Irish Cancer Society Research Award, co-funded by the Health Research Board Ireland and the Movember Foundation (Grant no. PCI12Kin/MCRG2012/3) and by the Programme for Research in Third Level Institutions (PRTLI; MolCellBiol) and the European Regional Development Fund. The author (BTK) wishes to acknowledge the many colleagues who contributed to the research cited in this communication, in particular Drs AnneMarie M. Gannon, Garret L. Keating, Eamon P. Mulvaney and Helen M. Reid.

References

- Woodward DF, Jones RL, Narumiya S. International Union of Basic and Clinical Pharmacology. LXXXIII: classification of prostanoid receptors, updating 15 years of progress. Pharmacol Rev. 2011;63(3):471–538. http://dx.doi.org/10.1124/pr.110.003517
- 2. Davi G, Santilli F, Vazzana N. Thromboxane receptors antagonists and/or synthase inhibitors. Handb Exp Pharmacol. 2012;(210):261–86.
- Nakahata N. Thromboxane A2: physiology/pathophysiology, cellular signal transduction and pharmacology. Pharmacol Ther. 2008;118(1):18–35. http://dx.doi.org/10.1016/j.pharmthera.2008.01.001
- Strittmatter F, Gratzke C, Weinhold P, Steib CJ, Hartmann AC, Schlenker B, et al. Thromboxane A2 induces contraction of human prostate smooth muscle by Rho kinase- and calmodulin-dependent mechanisms. Eur J Pharmacol. 2011;650(2–3):650–5. http://dx.doi.org/10.1016/j.ejphar.2010.10.052
- Hennenberg M, Miljak M, Herrmann D, Strittmatter F, Walther S, Rutz B, et al. The receptor antagonist picotamide inhibits adrenergic and thromboxane-induced contraction of hyperplastic human prostate smooth muscle. Am J Physiol Renal Physiol. 2013;305(10):F1383-90.
 http://dx/doi.org/10.1152/cinrepal.00280.2012

http://dx.doi.org/10.1152/ajprenal.00380.2013

- 6. Kinsella BT. Thromboxane A2 signalling in humans: a 'Tail' of two receptors. Biochem Soc Trans. 2001;29(Pt 6):641–54.
- Coyle AT, Kinsella BT. Characterization of promoter 3 of the human thromboxane A receptor gene. A functional AP-1 and octamer motif are required for basal promoter activity. FEBS J. 2005;272(4):1036–53.

http://dx.doi.org/10.1111/j.1742-4658.2004.04538.x

- 8. Coyle AT, Miggin SM, Kinsella BT. Characterization of the 5' untranslated region of alpha and beta isoforms of the human thromboxane A2 receptor (TP). Differential promoter utilization by the TP isoforms. Eur J Biochem. 2002;269(16):4058–73.
- Gannon AM, Kinsella BT. Regulation of the human thromboxane A2 receptor gene by Sp1, Egr1, NF-E2, GATA-1, and Ets-1 in megakaryocytes. J Lipid Res. 2008;49(12):2590–604.

http://dx.doi.org/10.1194/jlr.M800256-JLR200

- 10. Miggin SM, Kinsella BT. Expression and tissue distribution of the mRNAs encoding the human thromboxane A2 receptor (TP) alpha and beta isoforms. Biochim Biophys Acta. 1998;1425(3):543–59.
- Miggin SM, Kinsella BT. Thromboxane A(2) receptor mediated activation of the mitogen activated protein kinase cascades in human uterine smooth muscle cells. Biochim Biophys Acta. 2001;1539(1–2):147–62.
- 12. Miggin SM, Kinsella BT. Regulation of extracellular signal-regulated kinase cascades by alpha- and beta-isoforms of the human thromboxane A(2) receptor. Mol Pharmacol. 2002;61(4):817–31.
- 13. Walsh M, Foley JF, Kinsella BT. Investigation of the role of the carboxyl-terminal tails of the alpha and beta isoforms of the human thromboxane A(2) receptor (TP) in mediating receptor: effector coupling. Biochim Biophys Acta. 2000;1496(2–3):164–82.
- 14. Wikstrom K, Kavanagh DJ, Reid HM, Kinsella BT. Differential regulation of RhoAmediated signaling by the TPalpha and TPbeta isoforms of the human thromboxane A2 receptor: independent modulation of TPalpha signaling by prostacyclin and nitric oxide. Cell Signal. 2008;20(8):1497–512.

http://dx.doi.org/10.1016/j.cellsig.2008.04.006

 Kelley-Hickie LP, Kinsella BT. Homologous desensitization of signalling by the beta (beta) isoform of the human thromboxane A2 receptor. Biochim Biophys Acta. 2006;1761(9):1114–31.

http://dx.doi.org/10.1016/j.bbalip.2006.07.012.

 Kelley-Hickie LP, O'Keeffe MB, Reid HM, Kinsella BT. Homologous desensitization of signalling by the alpha (alpha) isoform of the human thromboxane A2 receptor: a specific role for nitric oxide signalling. Biochim Biophys Acta. 2007;1773(6):970–89. http://dx.doi.org/10.1016/j.bbamcr.2007.03.012

- 17. Foley JF, Kelley LP, Kinsella BT. Prostaglandin D(2) receptor-mediated desensitization of the alpha isoform of the human thromboxane A(2) receptor. Biochem Pharmacol. 2001;62(2):229–39.
- Reid HM, Kinsella BT. The alpha, but not the beta, isoform of the human thromboxane A2 receptor is a target for nitric oxide-mediated desensitization. Independent modulation of Tp alpha signaling by nitric oxide and prostacyclin. J Biol Chem. 2003;278(51):51190–202.

http://dx.doi.org/10.1074/jbc.M309314200

 Walsh MT, Foley JF, Kinsella BT. The alpha, but not the beta, isoform of the human thromboxane A2 receptor is a target for prostacyclin-mediated desensitization. J Biol Chem. 2000;275(27):20412–23.

http://dx.doi.org/10.1074/jbc.M907881199

- 20. Ekambaram P, Lambiv W, Cazzolli R, Ashton AW, Honn KV. The thromboxane synthase and receptor signaling pathway in cancer: an emerging paradigm in cancer progression and metastasis. Cancer Metastasis Rev. 2011;30(3–4):397–408. http://dx.doi.org/10.1007/s10555-011-9297-9
- Yang Y, Tang LQ, Wei W. Prostanoids receptors signaling in different diseases/ cancers progression. J Recept Signal Transduct Res. 2013;33(1):14–27. http://dx.doi.org/10.3109/10799893.2012.752003
- Langley RE, Burdett S, Tierney JF, Cafferty F, Parmar MK, Venning G. Aspirin and cancer: has aspirin been overlooked as an adjuvant therapy? Br J Cancer. 2011;105(8):1107–13.

http://dx.doi.org/10.1038/bjc.2011.289

- Langley RE, Rothwell PM. Potential biomarker for aspirin use in colorectal cancer therapy. Nat Rev Clin Oncol. 2013;10(1):8–10. http://dx.doi.org/10.1038/nrclinonc.2012.216
- Rothwell PM, Algra A, Amarenco P. Medical treatment in acute and long-term secondary prevention after transient ischaemic attack and ischaemic stroke. Lancet. 2011;377(9778):1681–92.

http://dx.doi.org/10.1016/S0140-6736(11)60516-3

- 25. Rothwell PM, Fowkes FG, Belch JF, Ogawa H, Warlow CP, Meade TW. Effect of daily aspirin on long-term risk of death due to cancer: analysis of individual patient data from randomised trials. Lancet. 2011;377(9759):31–41. http://dx.doi.org/10.1016/S0140-6736(10)62110-1
- 26. Rothwell PM, Price JF, Fowkes FG, Zanchetti A, Roncaglioni MC, Tognoni G, et al. Short-term effects of daily aspirin on cancer incidence, mortality, and non-vascular death: analysis of the time course of risks and benefits in 51 randomised controlled trials. Lancet. 2012;379(9826):1602–12.

http://dx.doi.org/10.1016/S0140-6736(11)61720-0

- 27. Rothwell PM, Wilson M, Elwin CE, Norrving B, Algra A, Warlow CP, et al. Longterm effect of aspirin on colorectal cancer incidence and mortality: 20-year followup of five randomised trials. Lancet. 2010;376(9754):1741–50. http://dx.doi.org/10.1016/S0140-6736(10)61543-7
- 28. Rothwell PM, Wilson M, Price JF, Belch JF, Meade TW, Mehta Z. Effect of daily aspirin on risk of cancer metastasis: a study of incident cancers during randomised controlled trials. Lancet. 2012;379(9826):1591–601. http://dx.doi.org/10.1016/S0140-6736(12)60209-8
- 29. Cuzick J, Thorat MA, Bosetti C, Brown PH, Cook NR, Ford LG, et al. Estimates of benefits and harms of prophylactic use of aspirin in the general population. Ann Oncol. 2015;26(1):47–57.

http://dx.doi.org/10.1093/annonc/mdu225

- Ho-Tin-Noe B, Goerge T, Wagner DD. Platelets: guardians of tumor vasculature. Cancer Res. 2009;69(14):5623–6. http://dx.doi.org/10.1158/0008-5472.CAN-09-1370
- 31. Moussa O, Ciupek A, Watson DK, Halushka PV. Urinary thromboxane B(2) and thromboxane receptors in bladder cancer: Opportunity for detection and monitoring. Prostaglandins Other Lipid Mediat. 2011;96(1–4):41–4. http://dx.doi.org/10.1016/j.prostaglandins.2011.09.002
- 32. Dassesse T, de Leval X, de Leval L, Pirotte B, Castronovo V, Waltregny D. Activation of the thromboxane A2 pathway in human prostate cancer correlates with tumor Gleason score and pathologic stage. Eur Urol. 2006;50(5):1021–31; discussion 31. http://dx.doi.org/10.1016/j.eururo.2006.01.036
- Nie D, Che M, Zacharek A, Qiao Y, Li L, Li X, et al. Differential expression of thromboxane synthase in prostate carcinoma: role in tumor cell motility. Am J Pathol. 2004;164(2):429–39.

http://dx.doi.org/10.1016/S0002-9440(10)63133-1

- 34. Gustafsson A, Hansson E, Kressner U, Nordgren S, Andersson M, Lonnroth C, et al. Prostanoid receptor expression in colorectal cancer related to tumor stage, differentiation and progression. Acta Oncol. 2007;46(8):1107–12. http://dx.doi.org/10.1080/02841860701403061
- 35. Sakai H, Suzuki T, Takahashi Y, Ukai M, Tauchi K, Fujii T, et al. Upregulation of thromboxane synthase in human colorectal carcinoma and the cancer cell proliferation by thromboxane A2. FEBS Lett. 2006;580(14):3368–74. http://dx.doi.org/10.1016/j.febslet.2006.05.007
- 36. Cathcart MC, Gately K, Cummins R, Kay E, O'Byrne KJ, Pidgeon GP. Examination of thromboxane synthase as a prognostic factor and therapeutic target in non-small cell lung cancer. Mol Cancer. 2011;10:25. http://dx.doi.org/10.1186/1476-4598-10-25

- Hirata T, Narumiya S. Prostanoids as regulators of innate and adaptive immunity. Adv Immunol. 2012;116(2012):143–74. http://dx.doi.org/10.1016/B978-0-12-394300-2.00005-3
- 38. Kabashima K, Murata T, Tanaka H, Matsuoka T, Sakata D, Yoshida N, et al. Thromboxane A2 modulates interaction of dendritic cells and T cells and regulates acquired immunity. Nat Immunol. 2003;4(7):694–701. http://dx.doi.org/10.1038/ni943
- 39. Moalli F, Cupovic J, Thelen F, Halbherr P, Fukui Y, Narumiya S, et al. Thromboxane A2 acts as tonic immunoregulator by preferential disruption of low-avidity CD4+ T cell-dendritic cell interactions. J Exp Med. 2014;211(13):2507–17. http://dx.doi.org/10.1084/jem.20140137
- 40. Thomas DW, Rocha PN, Nataraj C, Robinson LA, Spurney RF, Koller BH, et al. Proinflammatory actions of thromboxane receptors to enhance cellular immune responses. J Immunol. 2003;171(12):6389–95.
- 41. Yang CW, Unanue ER. Neutrophils control the magnitude and spread of the immune response in a thromboxane A2-mediated process. J Exp Med. 2013;210(2):375–87. http://dx.doi.org/10.1084/jem.20122183
- Yang Q, Shi M, Shen Y, Cao Y, Zuo S, Zuo C, et al. COX-1-derived thromboxane A2 plays an essential role in early B-cell development via regulation of JAK/STAT5 signaling in mouse. Blood. 2014;124(10):1610–21. http://dx.doi.org/10.1084/jem.20122183
- 43. Li X, Tai HH. Thromboxane A2 receptor-mediated release of matrix metalloproteinase-1 (MMP-1) induces expression of monocyte chemoattractant protein-1 (MCP-1) by activation of protease-activated receptor 2 (PAR2) in A549 human lung adenocarcinoma cells. Mol Carcinog. 2013;53(8):659–66. http://dx.doi.org/10.1002/mc.22020
- Li X, Tai HH. Activation of thromboxane A2 receptor (TP) increases the expression of monocyte chemoattractant protein -1 (MCP-1)/chemokine (C-C motif) ligand 2 (CCL2) and recruits macrophages to promote invasion of lung cancer cells. PLoS One. 2013;8(1):e54073.

http://dx.doi.org/10.1371/journal.pone.0054073

- Linnebacher M, Maletzki C. Tumor-infiltrating B cells: the ignored players in tumor immunology. Oncoimmunology. 2012;1(7):1186–8. http://dx.doi.org/10.4161/onci.20641
- 46. Abraham JE, Harrington P, Driver KE, Tyrer J, Easton DF, Dunning AM, et al. Common polymorphisms in the prostaglandin pathway genes and their association with breast cancer susceptibility and survival. Clin Cancer Res. 2009;15(6):2181–91. http://dx.doi.org/10.1158/1078-0432.CCR-08-0716

- 47. Nie D, Guo Y, Yang D, Tang Y, Chen Y, Wang MT, et al. Thromboxane A2 receptors in prostate carcinoma: expression and its role in regulating cell motility via small GTPase Rho. Cancer Res. 2008;68(1):115–21. http://dx.doi.org/10.1158/0008-5472.CAN-07-1018
- 48. Turner EC, Kavanagh DJ, Mulvaney EP, McLean C, Wikstrom K, Reid HM, et al. Identification of an interaction between the TPα and TP{beta} isoforms of the human thromboxane A2 receptor with protein kinase C-related kinase (PRK) 1. Implications for prostate cancer. J Biol Chem. 2011;286(17):15440–57. http://dx.doi.org/10.1074/jbc.M110.181180
- Hutchinson CL, Lowe PN, McLaughlin SH, Mott HR, Owen D. Differential binding of RhoA, RhoB, and RhoC to protein kinase C-related kinase (PRK) isoforms PRK1, PRK2, and PRK3: PRKs have the highest affinity for RhoB. Biochemistry. 2013;52(45):7999–8011.

http://dx.doi.org/10.1021/bi401216w

Lachmann S, Jevons A, De Rycker M, Casamassima A, Radtke S, Collazos A, et al. Regulatory domain selectivity in the cell-type specific PKN-dependence of cell migration. PLoS One. 2011;6(7):e21732.

http://dx.doi.org/10.1371/journal.pone.0021732

- 51. Leenders F, Mopert K, Schmiedeknecht A, Santel A, Czauderna F, Aleku M, et al. PKN3 is required for malignant prostate cell growth downstream of activated PI 3-kinase. EMBO J. 2004;23(16):3303–13. http://dx.doi.org/10.1038/sj.emboj.7600345
- 52. Yasui T, Sakakibara-Yada K, Nishimura T, Morita K, Tada S, Mosialos G, et al. Protein kinase N1, a cell inhibitor of Akt kinase, has a central role in quality control of germinal center formation. Proc Natl Acad Sci U S A. 2012;109(51):21022–7. http://dx.doi.org/10.1073/pnas.1218925110
- 53. Metzger E, Imhof A, Patel D, Kahl P, Hoffmeyer K, Friedrichs N, et al. Phosphorylation of histone H3T6 by PKCbeta(I) controls demethylation at histone H3K4. Nature. 2010;464(7289):792–6.

http://dx.doi.org/10.1038/nature08839

 Metzger E, Muller JM, Ferrari S, Buettner R, Schule R. A novel inducible transactivation domain in the androgen receptor: implications for PRK in prostate cancer. EMBO J. 2003;22(2):270–80.

http://dx.doi.org/10.1093/emboj/cdg023

55. Metzger E, Yin N, Wissmann M, Kunowska N, Fischer K, Friedrichs N, et al. Phosphorylation of histone H3 at threonine 11 establishes a novel chromatin mark for transcriptional regulation. Nat Cell Biol. 2008;10(1):53–60. http://dx.doi.org/10.1038/ncb1668

- 56. Karantanos T, Corn PG, Thompson TC. Prostate cancer progression after androgen deprivation therapy: mechanisms of castrate resistance and novel therapeutic approaches. Oncogene. 2013;32(49):5501–11. http://dx.doi.org/10.1038/onc.2013.206
- 57. O'Sullivan AG, Mulvaney EP, Hyland PB, Kinsella BT. Protein kinase C-related kinase 1 and 2 play an essential role in thromboxane-mediated neoplastic responses in prostate cancer. Oncotarget. 2015;6(28):26437–56.
- 58. Keating GL, Reid HM, Eivers SB, Mulvaney EP, Kinsella BT. Transcriptional regulation of the human thromboxane A2 receptor gene by Wilms' tumor (WT)1 and hypermethylated in cancer (HIC) 1 in prostate and breast cancers. Biochim Biophys Acta. 2014;1839(6):476–92.

http://dx.doi.org/10.1016/j.bbagrm.2014.04.010

59. Coyle AT, Kinsella BT. Synthetic peroxisome proliferator-activated receptor gamma agonists rosiglitazone and troglitazone suppress transcription by promoter 3 of the human thromboxane A2 receptor gene in human erythroleukemia cells. Biochem Pharmacol. 2006;71(9):1308–23.

http://dx.doi.org/10.1016/j.bcp.2006.01.011

- 60. Gannon AM, Kinsella BT. The Wilms' tumour suppressor protein WT1 acts as a key transcriptional repressor of the human thromboxane A2 receptor gene in megakaryocytes. J Cell Mol Med. 2009;13(11–12):4571–86. http://dx.doi.org/10.1111/j.1582-4934.2008.00599.x
- 61. Gannon AM, Turner EC, Reid HM, Kinsella BT. Regulated expression of the alpha isoform of the human thromboxane A2 receptor during megakaryocyte differentiation: a coordinated role for WT1, Egr1, and Sp1. J Mol Biol. 2009;394(1):29–45. http://dx.doi.org/10.1016/j.jmb.2009.09.007
- 62. Ariyaratana S, Loeb DM. The role of the Wilms tumour gene (WT1) in normal and malignant haematopoiesis. Expert Rev Mol Med. 2007;9(14):1–17. http://dx.doi.org/10.1017/S1462399407000336
- Gessler M, Poustka A, Cavenee W, Neve RL, Orkin SH, Bruns GA. Homozygous deletion in Wilms tumours of a zinc-finger gene identified by chromosome jumping. Nature. 1990;343(6260):774–8. http://dx.doi.org/10.1038/343774a0
- 64. Haber DA, Buckler AJ, Glaser T, Call KM, Pelletier J, Sohn RL, et al. An internal deletion within an 11p13 zinc finger gene contributes to the development of Wilms' tumor. Cell. 1990;61(7):1257–69.
- 65. Call KM, Glaser T, Ito CY, Buckler AJ, Pelletier J, Haber DA, et al. Isolation and characterization of a zinc finger polypeptide gene at the human chromosome 11 Wilms' tumor locus. Cell. 1990;60(3):509–20.

- 66. Koesters R, Linnebacher M, Coy JF, Germann A, Schwitalle Y, Findeisen P, et al. WT1 is a tumor-associated antigen in colon cancer that can be recognized by in vitro stimulated cytotoxic T cells. Int J Cancer. 2004;109(3):385–92. http://dx.doi.org/10.1002/ijc.11721
- 67. Miwa H, Beran M, Saunders GF. Expression of the Wilms' tumor gene (WT1) in human leukemias. Leukemia. 1992;6(5):405–9.
- 68. Loeb DM, Evron E, Patel CB, Sharma PM, Niranjan B, Buluwela L, et al. Wilms' tumor suppressor gene (WT1) is expressed in primary breast tumors despite tumor-specific promoter methylation. Cancer Res. 2001;61(3):921–5.
- 69. Rodeck U, Bossler A, Kari C, Humphreys CW, Gyorfi T, Maurer J, et al. Expression of the wt1 Wilms' tumor gene by normal and malignant human melanocytes. Int J Cancer. 1994;59(1):78–82.
- 70. Roberts SG. Transcriptional regulation by WT1 in development. Curr Opin Genet Dev. 2005;15(5):542–7.

http://dx.doi.org/10.1016/j.gde.2005.08.004

- 71. Silberstein GB, Van Horn K, Strickland P, Roberts CT, Jr., Daniel CW. Altered expression of the WT1 wilms tumor suppressor gene in human breast cancer. Proc Natl Acad Sci U S A. 1997;94(15):8132–7.
- 72. Oji Y, Yano M, Nakano Y, Abeno S, Nakatsuka S, Ikeba A, et al. Overexpression of the Wilms' tumor gene WT1 in esophageal cancer. Anticancer Res. 2004;24(5B):3103–8.
- 73. Oji Y, Nakamori S, Fujikawa M, Nakatsuka S, Yokota A, Tatsumi N, et al. Overexpression of the Wilms' tumor gene WT1 in pancreatic ductal adenocarcinoma. Cancer Sci. 2004;95(7):583–7.
- 74. Davies RC, Calvio C, Bratt E, Larsson SH, Lamond AI, Hastie ND. WT1 interacts with the splicing factor U2AF65 in an isoform-dependent manner and can be incorporated into spliceosomes. Genes Dev. 1998;12(20):3217–25.
- 75. Ladomery M, Sommerville J, Woolner S, Slight J, Hastie N. Expression in Xenopus oocytes shows that WT1 binds transcripts in vivo, with a central role for zinc finger one. J Cell Sci. 2003;116(Pt 8):1539–49.
- 76. Bor YC, Swartz J, Morrison A, Rekosh D, Ladomery M, Hammarskjold ML. The Wilms' tumor 1 (WT1) gene (+KTS isoform) functions with a CTE to enhance translation from an unspliced RNA with a retained intron. Genes Dev. 2006;20(12):1597–608.

http://dx.doi.org/10.1101/gad.1402306

 Hohenstein P, Hastie ND. The many facets of the Wilms' tumour gene, WT1. Hum Mol Genet. 2006;15 Spec No 2:R196–201. http://dx.doi.org/10.1093/hmg/ddl196

- 78 Rauscher FJ, 3rd, Morris JF, Tournay OE, Cook DM, Curran T. Binding of the Wilms' tumor locus zinc finger protein to the EGR-1 consensus sequence. Science. 1990;250(4985):1259–62.
- 79. Nakagama H, Heinrich G, Pelletier J, Housman DE. Sequence and structural requirements for high-affinity DNA binding by the WT1 gene product. Mol Cell Biol. 1995;15(3):1489–98.
- Han Y, San-Marina S, Liu J, Minden MD. Transcriptional activation of c-myc protooncogene by WT1 protein. Oncogene. 2004;23(41):6933–41. http://dx.doi.org/10.1038/sj.onc.1207609
- 81. Hewitt SM, Hamada S, McDonnell TJ, Rauscher FJ, 3rd, Saunders GF. Regulation of the proto-oncogenes bcl-2 and c-myc by the Wilms' tumor suppressor gene WT1. Cancer Res. 1995;55(22):5386–9.
- Maheswaran S, Park S, Bernard A, Morris JF, Rauscher FJ, 3rd, Hill DE, et al. Physical and functional interaction between WT1 and p53 proteins. Proc Natl Acad Sci U S A. 1993;90(11):5100–4.
- Wang W, Lee SB, Palmer R, Ellisen LW, Haber DA. A functional interaction with CBP contributes to transcriptional activation by the Wilms tumor suppressor WT1. J Biol Chem. 2001;276(20):16810–6.

http://dx.doi.org/10.1074/jbc.M009687200

- 84. Carpenter B, Hill KJ, Charalambous M, Wagner KJ, Lahiri D, James DI, et al. BASP1 is a transcriptional cosuppressor for the Wilms' tumor suppressor protein WT1. Mol Cell Biol. 2004;24(2):537–49.
- 85. Tostes RC, Nigro D, Fortes ZB, Carvalho MH. Effects of estrogen on the vascular system. Braz J Med Biol Res. 2003;36(9):1143–58.
- Ricke WA, Wang Y, Cunha GR. Steroid hormones and carcinogenesis of the prostate: the role of estrogens. Differentiation. 2007;75(9):871–82. http://dx.doi.org/10.1111/j.1432-0436.2007.00224.x
- 87. Carruba G. Estrogen and prostate cancer: an eclipsed truth in an androgen-dominated scenario. J Cell Biochem. 2007;102(4):899–911. http://dx.doi.org/10.1002/jcb.21529
- Ellem SJ, Risbridger GP. The dual, opposing roles of estrogen in the prostate. Ann N Y Acad Sci. 2009;1155:174–86.

http://dx.doi.org/10.1111/j.1749-6632.2009.04360.x

- 89. Adamson E, de Belle I, Mittal S, Wang Y, Hayakawa J, Korkmaz K, et al. Egr1 signaling in prostate cancer. Cancer Biol Ther. 2003;2(6):617–22.
- 90. Krones-Herzig A, Adamson E, Mercola D. Early growth response 1 protein, an upstream gatekeeper of the p53 tumor suppressor, controls replicative senescence. Proc Natl Acad Sci U S A. 2003;100(6):3233–8. http://dx.doi.org/10.1073/pnas.2628034100

- 91. Oka Y, Tsuboi A, Oji Y, Kawase I, Sugiyama H. WT1 peptide vaccine for the treatment of cancer. Curr Opin Immunol. 2008;20(2):211–20. http://dx.doi.org/10.1016/j.coi.2008.04.009
- 92. Oka Y, Tsuboi A, Fujiki F, Li Z, Nakajima H, Hosen N, et al. WT1 peptide vaccine as a paradigm for "cancer antigen-derived peptide"-based immunotherapy for malignancies: successful induction of anti-cancer effect by vaccination with a single kind of WT1 peptide. Anticancer Agents Med Chem. 2009;9(7):787–97.
- 93. Hashimoto N, Tsuboi A, Kagawa N, Chiba Y, Izumoto S, Kinoshita M, et al. Wilms tumor 1 peptide vaccination combined with temozolomide against newly diagnosed glioblastoma: safety and impact on immunological response. Cancer Immunol Immunother. 2015;64(6):707–16. http://dx.doi.org/10.1007/s00262-015-1674-8
- 94. Nie D, Lamberti M, Zacharek A, Li L, Szekeres K, Tang K, et al. Thromboxane A(2) regulation of endothelial cell migration, angiogenesis, and tumor metastasis. Biochem Biophys Res Commun. 2000;267(1):245–51. http://dx.doi.org/10.1006/bbrc.1999.1840
- 95. Kajita S, Ruebel KH, Casey MB, Nakamura N, Lloyd RV. Role of COX-2, thromboxane A2 synthase, and prostaglandin I2 synthase in papillary thyroid carcinoma growth. Mod Pathol. 2005;18(2):221–7. http://dx.doi.org/10.1038/modpathol.3800285
- 96. Moussa O, Ashton AW, Fraig M, Garrett-Mayer E, Ghoneim MA, Halushka PV, et al. Novel role of thromboxane receptors beta isoform in bladder cancer pathogenesis. Cancer Res. 2008;68(11):4097–104.

http://dx.doi.org/10.1158/0008-5472.CAN-07-6560

97. Moussa O, Ciupek A, Watson DK, Halushka PV. Urinary thromboxane B2 and thromboxane receptors in bladder cancer: opportunity for detection and monitoring. Prostaglandins Other Lipid Mediat. 2011;96(1–4):41–4. http://dx.doi.org/10.1016/j.prostaglandins.2011.09.002

Chapter 12

The Inflammatory Microenvironment in Wilms Tumors

Paramahamsa Maturu^{1,2}

¹Department of Genetics, The University of Texas MD Anderson Cancer Center, 1515 Holcombe Blvd, Unit 1010, Houston, TX 77030, USA; ²Department of Pediatrics, Section of Neonatology, Texas Children's Hospital, Baylor College of Medicine, 1102 Bates Avenue, Houston, TX 77030, USA

 Author for correspondence: Paramahamsa Maturu, PhD, Department of Pediatrics, Section of Neonatology, Texas Children's Hospital, Baylor College of Medicine,
1102 Bates Avenue, MC: FC520, Houston, TX 77030, USA. Email: maturu@bcm.edu

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch12

Copyright: The Author.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the author and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

For the past several decades, the role of inflammation in different types of tumors has been well defined. The significance of inflammation including the presence of various immune cells and inflammatory marker analysis of tumors helped the clinicians to use new treatment methods, which lead to high cure rates but failed to do so in some tumors due to lack of information about the tumor microenvironment. Although the importance of inflammation in various adult malignancies has been well defined, by contrast, Wilms tumor (WT), the most common childhood kidney cancer, which represents 6% of all pediatric tumors, has

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Maturu

not been well studied. Nearly 75% of the WT cases have been noticed in children less than 5 years of age with a higher incidence at 2 to 3 years. Thus, very little is known about the inflammatory microenvironment in the development of WT. This inflammatory microenvironment may initiate oncogenic transformation, and in some instances, genetic and epigenetic modifications in tumor cells can also generate an inflammatory microenvironment that further supports tumor progression. Thus, the tumor microenvironment is highly dynamic, and linking the modulating factors and various inflammatory cells with tumor progression is of considerable interest. Although to some extent the currently used WT treatment methods such as surgical removal, chemotherapy, and radiation therapy are successful, the youngest children are at high risk for the irreversible adverse side effects. Thus, there is a need for alternative therapy/therapies exposing the child to the minimum possible adverse effects. This chapter gives a special focus on the inflammatory microenvironment of human WT with a comprehensive picture of various immune cells and other inflammatory markers. This may aid in the use of new therapeutic targets for the efficacious treatment of WT with the combination of currently adapted therapies or alone.

Key words: Cancer; Immune cells; Inflammation; Microenvironment; Wilms tumor

Introduction

Following the transformation of a normal cell to malignant or tumor cell, the inflammatory mediators promote the tumor growth, by inducing the proliferation and the evading immune surveillance. The unregulated inflammatory microenvironment plays a central role in the initiation and progression of tumor. In general, inflammation is initiated by the recruitment of a wide range of immune cells that affect malignant cells through the production of cytokines, chemokines, growth factors, prostaglandins, reactive oxygen and nitrogen species, proteases, and other bioactive molecules, which can act in an autocrine and/or paracrine manner (1). Altogether, this environment with various factors is known inflammatory tumor microenvironment. These inflammatory markers are very critical components to establish a link between inflammation and cancer although the activation of these inflammatory markers is influenced by various factors. This inflammatory microenvironment progresses the tumor cells with endowed immunosuppressive properties. Hence, the immune destruction property has now been proposed as one of the "hallmarks of cancer." Thus, the role of inflammation and inflammatory microenvironment in cancer is generally accepted and is an essential component of many tumors even though its relationship with inflammation has not been demonstrated (1-4). So far, the molecular mechanisms involved in establishing this inflammatory tumor microenvironment were not clearly understood and established. This may be due to multifaceted role of inflammatory markers/mediators, such as cytokines, chemokines, oncogenes, enzymes, transcription factors, and immune cells, in the tumor

Inflammation in Wilms tumors

microenvironment. Till date, studies are still going on to elucidate the complete link between the cancer and the inflammation. For the past one decade, studies using knockout animals have unraveled to some extent the molecular mechanisms that link inflammation and cancer in adult-onset cancers but not in pediatric cancers (5). These studies show that the inflammatory microenvironment is very important in tumor development. The inflammatory conditions may initiate or promote oncogenic transformation, or genetic and epigenetic changes in malignant cells can also generate an inflammatory microenvironment that further supports tumor progression (2). It is important to note that the acute inflammation regresses the tumor growth, whereas the chronic inflammation progresses the tumor. Thus, there is a need to be a balance between antitumor immunity and tumor-promoting immune activity within a tumor microenvironment that consists of tumor cells, stroma (including fibroblasts and endothelial cells), innate immune cells, and adaptive immune cells.

What is Wilms tumor and what are the various components of Wilms tumor?

Wilms tumor (WT) is the most common pediatric kidney cancer, which represents 6% of all pediatric tumors, and 9 out of 10 kidney cancers in children are WTs. Nearly 75% of the WT cases have been noticed in children less than 5 years of age with a higher incidence at 2 to 3 years. It is the most common cause of a renal mass in a child and more prevalent in the people of African descent (6, 7). WT is an undifferentiated mesodermal tumor, which consists of variable amount of embryonic renal elements, such as blastema, epithelium, and stroma (8, 9). The etiology of this childhood tumor is largely due to genetic alterations or mutations in the *WT1*, *CTNNB1*, and/or *WTX1* genes.

Most of the WTs are unilateral and most often involve only one tumor, but it has been observed that around 5% to 10% of children with WTs have more than one tumor in the same kidney. Only about 5% of children with WTs have bilateral disease. Most often, the size of the WT is much larger than that of the kidney before they were diagnosed and metastasized to other organs (10).

The mechanism/mechanisms of this pediatric cancer development at present is less clear, and the whole etiology of these diseases is also not completely understood. In general, pediatric cancers will not arise from epithelial tissues and will have different causative mechanisms than adult tumors. It is assumed that most of the childhood cancers arise as a result of inherited and/or acquired genetic events during embryogenesis (11).

Although in general the currently used WT treatment methods such as surgical removal, chemotherapy, and radiation therapy are successful, young children are still at high risk for the irreversible adverse side effects. In addition, a considerable number of patients relapse (20– 25%), and part of these tumors resist to current therapies and progress (12, 13). Thus, the main

Maturu

challenge is better stratification and development of novel therapeutic targets/approaches to eliminate or minimize these side effects and deficiencies. Such novel approaches critically depend on the in-depth understanding of the tumor microenvironment and on the mediating factors responsible for WT progression. Hence, this chapter focuses on the inflammatory microenvironment of human WT with a comprehensive picture of various immune cells and other inflammatory markers. This may aid in the advent of new therapeutic targets for the efficacious treatment of WT with the combination of currently adapted therapies or alone.

Types of Wilms tumors

Based on the histology, WTs are categorized into two major groups.

Unfavorable histology (anaplastic WTs)

In these tumors, the tumor cells vary widely, and the nuclei is very large and distorted. This is called anaplasia. The anaplastic tumors are very hard to cure. In preoperative chemo-therapy, such as in the International Society of Paediatric Oncology (SIOP) settings, also cases with chemo-resistant blastemal subtype, are considered at high risk of relapse.

Favorable histology

These are nonanaplastic tumors. Interestingly, more than 9 out of 10 WTs have a favorable histology. This type of tumors can easily be cured (10).

What is known about kidney cancer and inflammation?

There are not many studies available to relate WT and inflammation with the complete analysis of WT inflammatory microenvironment. In a comparative analysis of adult tumors, Vakkila et al. (14) reported that human WTs were infiltrated with macrophages and to a very less extent with T lymphocytes. This study was incomplete because it was confined to one or two immune cell markers. The other two different groups independently observed cyclooxygenase-2 (COX-2) expression in human WT ubiquitously in all cases, independent of the type (15) and stage (16) of neoplasm. However, these studies were again restricted to only one inflammatory marker, COX-2. The coexpression of the hypoxia-inducible factor (HIF-1α and its one of the target genes, vascular endothelial growth factor (VEGF), was reported in human WTs. This finding suggested the possible role of hypoxic cascade driving the tumor angiogenesis, growth, and progression (17). In addition, very early studies on isolation and culturing of tumor-infiltrating leukocytes (TIL) with different doses of cytokines in human WT comparing with other pediatric tumors were also reported (18). But none of these studies was able to give a comprehensive view of tumor microenvironment in human WT. It is therefore critical and relevant to know the whole picture of tumor microenvironment, whereas its role in
Inflammation in Wilms tumors

Wilms tumorigenesis has not been widely explored. Because there was not much information available about the complete analysis of the inflammatory microenvironment, we recently reported a comprehensive overview of various inflammatory markers and immune cells (qualitative and quantitative) in human WTs by immunohistochemistry (19).

Molecular links between WT and inflammation

Although there are a plethora of publications to link inflammation and adult tumor development, only few studies are available to relate the molecular links between WT and inflammation. Some of the recent findings are summarized below.

Immune cell infiltration

Our qualitative and quantitative immunohistochemical examination of immune cells in WTs (19) revealed infiltration of both adaptive and innate immune cells in tumors, similar to that previously reported in five WT samples (14) in a comparative study with adult tumors. However, our examination of a larger panel of tumors revealed that the extent of infiltration varied among tumors and among different histologically distinct regions within the same tumor, and also there was a difference in the in the quantity and infiltration pattern of adaptive and innate immune cells. Interestingly, while adaptive immune cells (T cells and B cells) were mostly localized to tumor stroma, innate immune cells [e.g., tumor-associated macrophages (TAMs), tumor-infiltrating neutrophils (TINs), and mast cells (MCs)], were not only predominantly localized to tumor stroma but also present in all other regions of the tumor. This different spatial localization suggested that a similar spatial pattern of chemical mediators, including chemokines and cytokines and other inflammatory proteins, might exist, either as a cause or as an effect of the presence of immune cells, which have been demonstrated to be recruited by, and also, in some cases, produce such mediators. To assess this possibility, we analyzed the expression and the intratumor localization of COX-2, HIF-1a, p-Stat3, p-ERK1/2, and the angiogenic marker, VEGF, in human WTs.

The following section describes the role of adaptive immune cells.

T lymphocytes

Human WTs were highly infiltrated with CD3+ T cells when compared with control kidney tissues in our earlier study (19). These T lymphocytes were almost absent in control kidney sections. Strikingly, although tumor stroma has many of the T lymphocytes when compared with other regions such as epithelium and blastema, the peritumoral area adjacent to tumor islands also has a huge number of infiltrating T lymphocytes. The peritumoral infiltration of this mononuclear T lymphocytes was greater than intratumoral (in blastemal, epithelial, and stromal regions) area of the tumor. Thus, this mononuclear T-cell infiltration was detected intensely in peritumoral region of the tumor in most of the cases we analyzed.

B Lymphocytes

B lymphocytes (CD20+) were also scattered in intra- and peritumoral region of WT with complete absence in control kidney sections, suggesting that these mononuclear lymphocytes (both T and B lymphocytes) followed the same kind of inflammatory cell infiltration pattern (19). Again, the density of CD20+ B-cell infiltration tended to be higher in peritumoral area than in intratumoral stromal region of the WT. In some of the tumors, we found only very few or absent in the tumor stroma, with aggregated infiltration of B lymphocytes found in most of the tumor-adjacent regions.

The role of innate immune cells is described below.

Macrophages

Although the intratumoral regions have infiltrating CD68+ macrophages (CD68MØ) in human WT, the majority of these CD68MØ were mostly dispersed extensively in tumor stroma in our earlier reported study (19). In contrary to the T and B lymphocytes, the CD68MØ within the tumor islands were also present in blastemal and epithelial regions although they were sparse when compared with tumor stroma. These CD68MØ were mostly in direct contact with the adjacent tumor cells in the invasive front. Surprisingly, although there were peritumoral CD68MØ, they were not comparable with the very highly infiltrated intratumoral regions. This observation is absolutely opposite to the lymphocyte (both T and B) infiltration, which we observed earlier. Very clear staining either in the membrane or in the cytoplasm was observed, with no staining in the nucleus. The spatial uniformity of the macrophage infiltration and density in the intratumoral region was maintained. But some tumors showed considerably less CD68MØ infiltration in some areas.

Neutrophils

TINs were identified in the intratumoral region of human WT (19). These TINs were mostly concentrated in the blastemal or epithelial regions to a lesser extent in the tumor stroma. There is a huge remarkable difference in the density of these cells in these different regions of the tumor. Most of these TINs were either intraepithelial or intrablastemal or, to some extent, were in the stroma, which is adjacent to the differentiated epithelial tissue. Overall, these TINs followed the tumorocentric distribution, concentrating mostly in neoplastic area as a massive infiltrate and diminishing its number or density distant from the neoplasm in almost all the WT cases in the current study. This is also true with anaplastic histology tumors, but the size of the neutrophils was slightly bigger in these tumors. TINs were not detected in the normal kidney.

Mast cells

MCs have been identified in the tumor microenvironment of various human neoplasias; we first confirmed that the MCs also infiltrate human WT (19). The infiltrating MCs were

Inflammation in Wilms tumors

distributed mainly in the invasive area of most of the human WTs. MCs were found in very small groups around neoplastic cells in tumor stroma and also in the peritumor areas but were almost absent in other intratumoral areas such as blastema and epithelium.

Together with these, various immune cell infiltration clearly demonstrates that the tumor inflammatory microenvironment is also present in human WT.

Inflammatory mediators

The inflammatory mediators can induce genetic and epigenetic changes that result in aberrations in critical biochemical pathways responsible for maintaining the cellular homeostasis, which leads to progression of cancer (1, 3, 4, 20). These inflammatory mediators may be of many types, such as cytokines, chemokines, free radicals, prostaglandins, growth factors, and enzymes such as COX.

COX-2

Positive immunoreactivity for COX-2 protein was observed in the entire tumor sections stained with diffuse moderate-to-strong cytoplasmic expression in the blastemal and the epithelial components and with very intense staining in tumor stroma. The infiltrating immune cells and other cells such as fibroblasts in the stroma were immune reactive for COX-2 protein. However, some of the tumors with anaplastic histology showed strong nuclear localization COX-2. The staining pattern and the intensity varied from tumor to tumor. Normal kidney samples showed weak to moderate staining in the cytoplasm of tubular epithelial cells. However, very weak or no staining was observed in the renal interstitial cells or glomeruli. We also investigated the correlation of COX-2 expression with the other inflammatory markers such as HIF-1, Stat-3, and VEGF. In addition, two different groups independently observed COX-2 expression in human WT ubiquitously in all cases, independent of the type (15) and stage (16) of neoplasm. COX-2 expression has been reported in other kidney cancers (renal cell carcinoma) (21), but not in pediatric tumors. In addition, Lee et al. (22) reported that the inhibition of COX-2 by SC-236 disrupted the tumor vascular mural cell recruitment and survival signaling in an orthotopic xenograft model of human WT. And another group (22) reported that the use of the same COX-2 inhibitor reduced tumor metastasis and inflammatory signaling during the blockade of VEGF in orthotopic SKNEP1 model of pediatric cancer.

HIF-1a

Very prominent nuclear localization of the HIF-1a protein expression was noticed in most of the cases evaluated in blastema, stroma, and epithelium along with negative HIF-1a expression in matched control kidney slides as reported earlier (19). In addition, some tumor specimens showed cytoplasmic granular staining in the cell cytosol and membranous (only in blastema) expression in blastemal and stromal compartments. The immune cell infiltrate of

tumor stroma was immunoreactive for HIF-1 α protein as observed for COX-2 expression. Thus, the stromal expression of HIF-1 α resembles the COX-2 expression. HIF-1 α overexpression was reported in a significant proportion of WTs (23). In their study, they found no significant association between the expression of HIF-1 α and clinicopathological variables in WTs resected following chemotherapy. In addition, the coexpression of HIF-1 α has been reported with the angiogenic marker VEGF (17).

VEGF

VEGF expression was observed in most of the specimens (19). Although majority of the tumors showed VEGF expression in the infiltrating immune cells, connective tissue, or fibroblasts in tumor stroma similar to COX-2 and HIF-1a expression, but the blastemal and epithelial cell components were also immunoreactive, to some extent, in some tumor specimens. In the normal kidney samples, VEGF expression was observed in the proximal and distal convoluted tubules. Rowe et al. (24) reported that the anti-VEGF antibody suppressed primary tumor growth and metastasis in experimental models of WT. And the combination of low-dose topotecan and anti-VEGF antibody therapy suppressed the tumor growth and metastasis in experimental WT mice more durably than either agent alone (25). The immunohistochemical expression of VEGF-C and VEGFR-2 in the stromal and epithelial components of WT was reported (26) and indicated a potent unfavorable risk factor and directed the use of antiangiogenic treatment strategies to control the tumor growth.

Phosphorylated-Stat3 (p-Stat3)

The p-Stat3 expression was predominantly confined to the nucleus with almost undetectable cytoplasmic staining in all WT cases evaluated (19). Immunoreactivity of p-STAT3 was not detected in the control kidney tissue. Majority of the tumors showed the expression of p-Stat3 in the infiltrating immune cells in the tumor stroma, as well as in blastemal region, and these were very little or absent in epithelial cells. In addition, p-STAT3-expressing cells were found in the peritumoral area adjacent to the tumor islands. Moreover, the positive cells in this peritumoral area were found to be with stronger expression of p-STAT3 in the nucleus. Significantly higher nuclear immunoreactivity for p-Stat3 was also found in tumors compared with normal kidney sections. Furthermore, the expression of p-STAT3 was positively correlated with the TAM, CD3+ T cells, B cells, and inflammatory markers such as COX-2, HIF-1 α , and VEGF. Zhang et al. (27) reported that the p-STAT3 expression in WT may correlate with progression and predict unfavorable prognosis and a new therapeutic target for metastatic WTs.

Phosphorylated pERK1/2

The expression of pERK1/2 protein was detected with very diffuse in the cytoplasm and more prominent staining in the nucleus in most of the WT cases (19), but to a small

Inflammation in Wilms tumors

extent, we were also able to see the cytoplasmic expression in normal kidney. However, the expression of phospho-mitogen-activated protein kinases (MAPK-/ERK1/2)-positive nuclei was observed in both peritumoral and tumoral islands. In most of the tumor cases, the expression was localized in tumor stroma with some extent in blastemal cells. Epithelial cell component of the tumors was almost absent with either cytoplasmic or nuclear pERK1/2 expression. The stromal expression was similar to COX2, HIF-1 α , and VEGF expressions. The correlation between the p-ERK expression and other immune cell markers was also assessed. Significantly higher expression of pERK1/2 was observed in tumors than in control kidneys. It has been observed that the Wt1 ablation and insulin growth factor-2 (IGF2) upregulation resulted in WTs with elevated ERK1/2 phosphorylation in mice (28).

Inducible nitric oxide synthase (iNOS)

Although there are not many studies available on the expression of iNOS in WTs, we observed the iNOS expression (19) in tumor stroma with intense nuclear or cytoplasmic staining in most of the cases and diffuse cytoplasmic staining in blastemal cells of the tumor. The inflammatory immune cells within the tumor stroma were highly immunoreactive for iNOS in some of the WT specimens. Surprisingly, immunoreactivity for iNOS was also detected in the peritumoral area of some of the tumor sections. However, none of the epithelial cells expressed iNOS. In addition, areas around the tumor with neovascularization showed positive staining for iNOS. However, no significant immunoreactivity for iNOS was detected in the control kidney sections.

Nitrotyrosine (NT)

In our observation (19), WTs showed NT expression in the cytoplasm of the inflammatory immune cells of tumor stroma, as well as with much diffused cytoplasmic staining in the blastemal and epithelial regions of the tumors. The NT expression was observed very rarely in the peritumoral region. The expression was mostly localized within the tumor.

Chemokines and cytokines

Chemokines play an important role in tumor development and metastasis. The expression or secretion of these chemokines in the tumor microenvironment of various cancers, including breast, ovarian, pancreatic, melanoma, lung cancers, etc, have been reported. The expression of chemokines and their receptors is altered in many malignancies, and it leads to aberrant chemokine receptor signaling. Although the chemokine expression has been reported in various cancers, there is not much information available in human WTs. However, the role of ELR-CXC chemokine family members CXCL2 and CXCL7 and their receptor CXCR2 was expressed at the earliest stages of metanephric development in the rat, and signaling through this receptor was required for the survival and maintenance of the undifferentiated metanephric mesenchyme (MM) (29).

Other markers expressed in WTs

CITED1

In general, CITED1 is expressed at high levels in the condensed metanephric mesenchyme (MM) surrounding ureteric bud (UB)tips, is downregulated temporally as these cells begin to differentiate into early epithelial structures, and is not expressed in differentiated elements of the adult kidney (30). WTs arise from the undifferentiated renal progenitor cells. CITED1, which is a transcriptional regulator, blocks the metanephric mesenchymal-to-epithelial transition and is expressed in the blastema of both the developing kidney and WTs. The overexpression of CITED1 in a human WT cell line significantly increased proliferation in vitro, and mutation of its functionally critical transactivation domain (DCR2) significantly reduced proliferation (31). CITED1 expression was observed in blastemal cell populations of both experimental rat nephroblastomas and human WTs, and that primary human WTs presenting with disseminated disease show the highest level of CITED1 expression (32). Rivera and Haber (33) reported the Cited1 expression in the undifferentiated MM cells of WTs, and its expression was primarily confined to the nucleus. These studies suggest CITED1 as a marker of primitive blastema in WTs, and its persistent expression and altered subcellular localization in the condensed MM might have a role in WT initiation and progression. And another possibility is that persistent expression of CITED1 in metanephric blastema may have adverse developmental role in the pathogenesis of WTs.

B7-H1

A membrane glycoprotein, B7-H1, has been reported to act as an important coregulator of antigen-specific T-cell-mediated immunity (34, 35). This is normally expressed by the macrophage lineage cells and is aberrantly expressed by multiple human malignancies (34–36). Interestingly, tumor B7-H1 has been observed to induce T-cell apoptosis or anergy, thereby downregulating the host antitumoral immunity (34). B7-H1 expression has been observed in WTs, and its expression correlated with tumor biology and is associated with an increased risk of recurrence in patients with favorable-histology tumors (36). Because B7-H1 is involved in T-cell apoptosis, its expression in WTs may be related to inflammation. Thus, B7-H1 expression may be used as a prognostic marker, which indicates the aggressive behavior for favorable-histology WT. In addition, B7-H1 may be used to distinguish patients with favorable-histology WT, who require aggressive treatment and unfavorable-histology tumors, and who are at very low risk for disease recurrence and death and to avoid unnecessary overtreatment (36).

CD44

CD44 is also a membrane glycoprotein like B7-H1 expressed in a variety of cells, including those of epithelial, mesenchymal, and hematopoietic origin (37). CD44 expression has been observed in three different isoforms, such as CD44s, CD44v5, and CD44v10. The expression

Inflammation in Wilms tumors

of CD44s has been observed in all three components of the WTs. Among the isoforms, overexpression of CD44v5 in blastemal cells of WT correlated with tumor stage, clinical progression, and tumor-related death (37). Therefore, CD44v5 expression may be used as a good prognostic marker in identifying WT patients with a high tendency for distant metastases. Several studies have indicated that the increased expression of the CD44 gene is associated with metastatic disease. Studies by various other groups also indicated that the expression of CD44 isoforms may be a good prognostic marker for various cancers (38–40).

Carbonic anhydrase IX (CAIX)

CAIX is a membrane glycoprotein, which plays an important role in the growth and survival of tumor cells under normoxic, as well as hypoxic, conditions (23, 41, 42). CAIX (mRNA and protein) expression has been reported to be upregulated in the untreated and treated WTs when compared with normal kidneys and WT precursor lesions (nephrogenic rests) (23). There was no correlation between CA expression and clinicopathological variables, including metastatic status in postchemotherapy-treated WTs. Cellular localization studies in untreated WTs suggest that CAIX and HIF-1a are regulated by hypoxic and nonhypoxic mechanisms.

Role of immune cell infiltration with COX-2 pathway components

The expression of the inflammatory markers such as COX-2, HIF-1, iNOS, p-ERK1/2, and VEGF was predominantly localized to tumor stroma similar to the expression of TAMs. The codistribution of major inflammatory marker COX-2 with TAM infiltration in the tumor stroma was observed in our study (19). This study suggests that the infiltration of inflammatory immune cells and the expression of inflammatory markers in the tumor stroma are related. This observation suggests a correlation between the infiltrating immune cells and the activated cytokines and chemokines. This TAM infiltration was further confirmed (F4/80 expression) in the mouse model of WT (19). TAM infiltration is known to be induced by COX-2 in the tumor microenvironment (43), especially in the tumor stroma, and TAMs can also induce the expression of COX-2 (44). We have reported earlier that the colocalization of COX-2 and TAMs in the tumor stroma (19) may activate each other in the tumor microenvironment. The mechanisms responsible for the abundant COX-2 expression in WTs are that the infiltrating immune cells themselves could be overexpressing COX-2, or tumor fibroblasts may be generating COX-2 in response to macrophage infiltration, or fetal mitogen IGF2 may induce COX2 by MEK/ERK pathway (45).

Cross talk between immune cells and other markers in WT microenvironment

As indicated earlier, TAMs are also involved in the production of proangiogenic factors, such as transforming growth factor β and VEGF (46, 47), and of immunosuppressive chemokines and cytokines, such as interleukin 10 and prostaglandin E2, which contribute to

tumor angiogenesis (46, 48, 49). Thus, the TAM infiltration might play a significant role in the increased VEGF expression and also in the vascularization of the tumors. The correlation and localization of TAMs in the tumor stroma with the expression of various inflammatory protein markers, such as COX-2, HIF-1, p-ERK1/2, iNOS, and NT, suggest a functional association of TAM infiltration with the overexpression of these markers and vice versa in WTs (19) and demonstrate the existence of a highly inflammatory microenvironment in this disease.

Possible mechanism/mechanisms responsible for COX-2 pathway activation

Reports from our laboratory indicated that p-ERK1/2 was induced in mice, which are engineered to overexpress IGF2 and along with ablation of Wt1 gene, and also in human WTs (28), suggesting a role for the ERK signaling in WT development. The robust expression of COX-2 and p-ERK1/2 in tumors may be a consequence of IGF2 overexpression in WTs because IGF2mediated COX2 expression has been reported in other tumors (45). Thus, the upregulated COX-2 expression creates an inflammatory microenvironment in WTs, which may be mediated by the enhanced p-ERK signaling is depicted in Figure 1. In the tumor microenvironment, COX-2 can also activate the expression of HIF-1 through its enzymatic product prostaglandin E2 (45, 50). Furthermore, this upregulated expression of p-ERK1/2 stabilizes the HIF-1 α protein by preventing its degradation via the blockage of prolyl hydroxylase activity, which regulates HIF-1 or activates HIF-1a protein (Figure 1). Spatially similar expression of COX-2 and HIF-1 was observed in WTs (19), suggesting the role of COX-2 in HIF-1 activation. COX-2 activation of HIF-1 can also occur through hypoxia (17, 51) or hypoxia-independent mechanisms (52), with the involvement of p-ERK1/2 (53). In addition, it has been reported that PGE2, the end product of COX-2 pathway, can also enhance HIF-1 transcriptional activity (51). HIF-1 can also directly upregulate the expression of COX-2 during hypoxia (54) and thus form a feedback loop to continually activate the COX-2 pathway (Figure 1). Hence, it may be assumed that IGF2 affects the inflammation, hyperproliferation, and angiogenesis in WTs by IGF2-induced Cox-2-mediated p-ERK1/2 pathway. Therefore, we speculate that COX-2 in this WT microenvironment may drive the inflammation and upregulate the aforementioned downstream targets.

Possible therapeutic targets in Wilms tumors

On the basis of the above evidences, it was found that WTs have highly inflammatory microenvironment, which further provides a link for the inflammatory etiology of cancer. The overexpression of different inflammatory markers provides a rationale for their use in the prevention and treatment of cancer. More specifically, our observation (19) strongly supports the therapeutic value of blocking COX-2 in WTs. The overexpression of inflammatory markers in tumors, in particular COX-2, has provided a rationale for their targeting in prevention and treatment of many cancers (55–59) by COX-2-specific inhibitors alone (60, 61) or in combination

Inflammation in Wilms tumors



Figure 1. Possible mechanism/mechanisms responsible for COX-2 pathway activation: COX-2 activation of HIF-1 can also occur through hypoxia-dependent by PGE2 or hypoxia-independent mechanisms with the involvement of p-ERK1/2. HIF-1 can also directly upregulate the expression of COX-2 during hypoxia and thus form a feedback loop to continually activate the COX-2 pathway leading to activation. Hence, IGF2 affects the inflammation, hyperproliferation, and angiogenesis in WTs by IGF2-induced COX-2-mediated p-ERK1/2 pathway.

with other inhibitors (62). COX-2 inhibition may serve as an appropriate target for therapeutic intervention because all the downstream targets of COX-2 pathway components may be controlled or inhibited too. Also, COX-2 blockade may be effective in WT therapy owing to the inhibitory effect of COX-2 inhibitors in controlling the immune cell infiltration and tumor-promoting angiogenesis, thereby controlling tumor growth. Elucidating the molecular basis for the accumulation of the different inflammatory protein markers in tumors requires further in-depth study and warrants further investigation of this COX-2-mediated pathway.

Conclusions

The colocalization of TAMs in the tumor stroma along with COX-2 and its pathway components, such as HIF-1 and p-ERK1/2, suggests a functional association of TAM

infiltration with the overexpression of these markers and vice versa in WTs and demonstrate the existence of a highly inflammatory microenvironment in this cancer. The overexpression of inflammatory marker COX-2 has provided a rationale for their targeting COX-2 pathway using COX-2-specific inhibitors alone or in combination with other inhibitors, which may be effective in treating this childhood cancer.

Conflict of Interests

The author declares no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

The expert scientific help of Vicki Huff, Professor, Department Genetics, MD Anderson Cancer Center, Houston, TX, USA, is greatly appreciated.

References

1. Grivennikov SI, Greten FR, Karin M. Immunity, inflammation, and cancer. Cell. 2010;140(6):883–99.

http://dx.doi.org/10.1016/j.cell.2010.01.025

 Mantovani A, Allavena P, Sica A, Balkwill F. Cancer-related inflammation. Nature. 2008;454(7203):436–44.

http://dx.doi.org/10.1038/nature07205

Colotta F, Allavena P, Sica A, Garlanda C, Mantovani A. Cancer-related inflammation, the seventh hallmark of cancer: links to genetic instability. Carcinogenesis. 2009;30(7):1073–81.

http://dx.doi.org/10.1093/carcin/bgp127

- Nikitovic D, Tzardi M, Berdiaki A, Tsatsakis A, Tzanakakis GN. Cancer microenvironment and inflammation: role of hyaluronan. Front Immunol. 2015;6:169. http://dx.doi.org/10.3389/fimmu.2015.00169
- 5. Mantovani A. Cancer: inflaming metastasis. Nature. 2009;457(7225):36–7. http://dx.doi.org/10.1038/457036b
- Dumba M, Jawad N, McHugh K. Neuroblastoma and nephroblastoma: a radiological review. Cancer Imaging. 2015;15 (1):5. http://dx.doi.org/10.1186/s40644-015-0040-6
- Stones DK, Hadley GP, Wainwright RD, Stefan DC. The impact of ethnicity on Wilms tumor: characteristics and outcome of a South African cohort. Int J Pediatr. 2015;2015 706058.

http://dx.doi.org/10.1155/2015/706058

Inflammation in Wilms tumors

- Gessler M, Poustka A, Cavenee W, Neve RL, Orkin SH, Bruns GA. Homozygous deletion in Wilms tumours of a zinc-finger gene identified by chromosome jumping. Nature. 1990;343 (6260):774–8. http://dx.doi.org/10.1038/343774a0
- 9. Call KM, Glaser T, Ito CY, Buckler AJ, Pelletier J, Haber DA, et al. Isolation and characterization of a zinc finger polypeptide gene at the human chromosome 11 Wilms' tumor locus. Cell. 1990;60 (3):509–20.
- 10. Society AC. Cancer Facts & Figures 2015. Atlanta, Ga: American Cancer Society, 2015.
- 11. Morgenstern DA, Anderson J. Inflammation: what role in pediatric cancer? Pediatr Blood Cancer. 2012;58 (5):659–64.
 - http://dx.doi.org/10.1002/pbc.24008
- Green DM, Grigoriev YA, Nan B, Takashima JR, Norkool PA, D'Angio GJ, et al. Congestive heart failure after treatment for Wilms' tumor: a report from the National Wilms' Tumor Study group. J Clin Oncol. 2001;19 (7):1926–34.
- Yi Y, Polosukhina D, Love HD, Hembd A, Pickup M, Moses HL, et al. A murine model of K-ras and beta-catenin induced renal tumors express high levels of E2F1 and resemble human Wilms Tumor. J Urol. 2015;194:1762–1770. http://dx.doi.org/10.1016/j.juro.2015.04.090
- Vakkila J, Jaffe R, Michelow M, Lotze MT. Pediatric cancers are infiltrated predominantly by macrophages and contain a paucity of dendritic cells: a major nosologic difference with adult tumors. Clin Cancer Res. 2006;12 (7 Pt 1):2049–54. http://dx.doi.org/10.1158/1078-0432.CCR-05-1824
- Giordano G, Campanini N, Donofrio V, Bertolini P, Falleti J, Grassani C, et al. Analysis of Cox-2 expression in Wilms' tumor. Pathol Res Pract. 2008;204 (12):875–82.

http://dx.doi.org/10.1016/j.prp.2008.06.008

- Fridman E, Pinthus JH, Kopolovic J, Ramon J, Mor O, Mor Y. Expression of cyclooxygenase-2 in Wilms tumor: immunohistochemical study using tissue microarray methodology. J Urol. 2006;176 (4 Pt 2):1747–50. http://dx.doi.org/10.1016/j.juro.2006.03.118
- 17. Karth J, Ferrer FA, Perlman E, Hanrahan C, Simons JW, Gearhart JP, et al. Coexpression of hypoxia-inducible factor 1-alpha and vascular endothelial growth factor in Wilms' tumor. J Pediatr Surg. 2000;35 (12):1749–53. http://dx.doi.org/10.1053/jpsu.2000.19241
- Rivoltini L, Arienti F, Orazi A, Cefalo G, Gasparini M, Gambacorti-Passerini C, et al. Phenotypic and functional analysis of lymphocytes infiltrating paediatric tumours, with a characterization of the tumour phenotype. Cancer Immunol Immunother. 1992;34 (4):241–51.

- Maturu P, Overwijk WW, Hicks J, Ekmekcioglu S, Grimm EA, Huff V. Characterization of the inflammatory microenvironment and identification of potential therapeutic targets in Wilms tumors. Transl Oncol. 2014;7 (4):484–92. http://dx.doi.org/10.1016/j.tranon.2014.05.008
- Fernandes JV, Cobucci RN, Jatoba CA, Fernandes TA, de Azevedo JW, de Araujo JM. The role of the mediators of inflammation in cancer development. Pathol Oncol Res. 2015;21 (3):527–34.

http://dx.doi.org/10.1007/s12253-015-9913-z

- 21. Mungan MU, Gurel D, Canda AE, Tuna B, Yorukoglu K, Kirkali Z. Expression of COX-2 in normal and pyelonephritic kidney, renal intraepithelial neoplasia, and renal cell carcinoma. Eur Urol. 2006;50 (1):92–7; discussion 97. http://dx.doi.org/10.1016/j.eururo.2005.12.039
- 22. Lee A, Frischer J, Serur A, Huang J, Bae JO, Kornfield ZN, et al. Inhibition of cyclooxygenase-2 disrupts tumor vascular mural cell recruitment and survival signaling. Cancer Res. 2006;66 (8):4378–84.

http://dx.doi.org/10.1158/0008-5472.CAN-05-3810

- Dungwa JV, Hunt LP, Ramani P. Overexpression of carbonic anhydrase and HIF-1alpha in Wilms tumours. BMC Cancer. 2011;11:390. http://dx.doi.org/10.1186/1471-2407-11-390
- 24. Rowe DH, Huang J, Kayton ML, Thompson R, Troxel A, O'Toole KM, et al. Anti-VEGF antibody suppresses primary tumor growth and metastasis in an experimental model of Wilms' tumor. J Pediatr Surg. 2000;35 (1):30–2; discussion 32–3.
- Soffer SZ, Moore JT, Kim E, Huang J, Yokoi A, Manley C, et al. Combination antiangiogenic therapy: increased efficacy in a murine model of Wilms tumor. J Pediatr Surg. 2001;36 (8):1177–81.

http://dx.doi.org/10.1053/jpsu.2001.25747

- 26. Nowicki M, Ostalska-Nowicka D, Kaczmarek M, Miskowiak B, Witt M. The significance of VEGF-C/VEGFR-2 interaction in the neovascularization and prognosis of nephroblastoma (Wilms' tumour). Histopathology. 2007;50 (3):358–64. http://dx.doi.org/10.1111/j.1365-2559.2007.02613.x
- 27. Zhang LJ, Liu W, Gao YM, Qin YJ, Wu RD. The expression of IL-6 and STAT3 might predict progression and unfavorable prognosis in Wilms' tumor. Biochem Biophys Res Commun. 2013;435 (3):408–13.

http://dx.doi.org/10.1016/j.bbrc.2013.04.102

28. Hu Q, Gao F, Tian W, Ruteshouser EC, Wang Y, Lazar A, et al. Wt1 ablation and Igf2 upregulation in mice result in Wilms tumors with elevated ERK1/2 phosphorylation. J Clin Invest. 2011;121 (1):174–83. http://dx.doi.org/10.1172/JCI43772 Levashova ZB, Sharma N, Timofeeva OA, Dome JS, Perantoni AO. ELR+-CXC chemokines and their receptors in early metanephric development. J Am Soc Nephrol. 2007;18 (8):2359–70.

http://dx.doi.org/10.1681/ASN.2006040380

 Boyle S, Shioda T, Perantoni AO, de Caestecker M. Cited1 and Cited2 are differentially expressed in the developing kidney but are not required for nephrogenesis. Dev Dyn. 2007;236 (8):2321–30. http://dx.doi.org/10.1002/dvdy.21242

http://dx.doi.org/10.1002/dvdy.21242

- 31. Lovvorn HN, 3rd, Boyle S, Shi G, Shyr Y, Wills ML, Perantoni AO, et al. Wilms' tumorigenesis is altered by misexpression of the transcriptional co-activator, CITED1. J Pediatr Surg. 2007;42 (3):474–81. http://dx.doi.org/10.1016/j.jpedsurg.2006.10.054
- 32. Lovvorn HN, Westrup J, Opperman S, Boyle S, Shi G, Anderson J, et al. CITED1 expression in Wilms' tumor and embryonic kidney. Neoplasia. 2007;9 (7):589–600.
- 33. Rivera MN, Haber DA. Wilms' tumour: connecting tumorigenesis and organ development in the kidney. Nat Rev Cancer. 2005;5 (9):699–712. http://dx.doi.org/10.1038/nrc1696
- Dong H, Strome SE, Salomao DR, Tamura H, Hirano F, Flies DB, et al. Tumor-associated B7-H1 promotes T-cell apoptosis: a potential mechanism of immune evasion. Nat Med. 2002;8 (8):793–800.

http://dx.doi.org/10.1038/nm730

35. Dong H, Zhu G, Tamada K, Chen L. B7-H1, a third member of the B7 family, co-stimulates T-cell proliferation and interleukin-10 secretion. Nat Med. 1999;5 (12):1365–9.

http://dx.doi.org/10.1038/70932

- 36. Routh JC, Ashley RA, Sebo TJ, Lohse CM, Husmann DA, Kramer SA, et al. B7-H1 expression in Wilms tumor: correlation with tumor biology and disease recurrence. J Urol. 2008;179 (5):1954–9; discussion 1959–60. http://dx.doi.org/10.1016/j.juro.2008.01.056
- Ghanem MA, Van Steenbrugge GJ, Van Der Kwast TH, Sudaryo MK, Noordzij MA, Nijman RJ. Expression and prognostic value of CD44 isoforms in nephroblastoma (Wilms tumor). J Urol. 2002;168 (2):681–6.
- 38. Stauder R, Eisterer W, Thaler J, Gunthert U. CD44 variant isoforms in non-Hodgkin's lymphoma: a new independent prognostic factor. Blood. 1995;85 (10):2885–99.
- Favrot MC, Combaret V, Lasset C. CD44 a new prognostic marker for neuroblastoma. N Engl J Med. 1993;329 (26):1965. http://dx.doi.org/10.1056/NEJM199312233292615

- 40. Speiser P, Wanner C, Tempfer C, Mittelbock M, Hanzal E, Bancher-Todesca D, et al. CD44 is an independent prognostic factor in early-stage cervical cancer. Int J Cancer. 1997;74 (2):185–8.
- Chiche J, Ilc K, Laferriere J, Trottier E, Dayan F, Mazure NM, et al. Hypoxia-inducible carbonic anhydrase IX and XII promote tumor cell growth by counteracting acidosis through the regulation of the intracellular pH. Cancer Res. 2009;69 (1):358–68. http://dx.doi.org/10.1158/0008-5472.CAN-08-2470
- 42. Robertson N, Potter C, Harris AL. Role of carbonic anhydrase IX in human tumor cell growth, survival, and invasion. Cancer Res. 2004;64 (17):6160–5. http://dx.doi.org/10.1158/0008-5472.CAN-03-2224
- 43. Tjiu JW, Chen JS, Shun CT, Lin SJ, Liao YH, Chu CY, et al. Tumor-associated macrophage-induced invasion and angiogenesis of human basal cell carcinoma cells by cyclooxygenase-2 induction. J Invest Dermatol. 2009;129 (4):1016–25. http://dx.doi.org/10.1038/jid.2008.310
- Hou Z, Falcone DJ, Subbaramaiah K, Dannenberg AJ. Macrophages induce COX-2 expression in breast cancer cells: role of IL-1beta autoamplification. Carcinogenesis. 2011;32 (5):695–702.

http://dx.doi.org/10.1093/carcin/bgr027

 45. Kim HJ, Kim TY. IGF-II-mediated COX-2 gene expression in human keratinocytes through extracellular signal-regulated kinase pathway. J Invest Dermatol. 2004;123 (3):547–55.

```
http://dx.doi.org/10.1111/j.0022-202X.2004.23317.x
```

- 46. Mantovani A, Sozzani S, Locati M, Allavena P, Sica A. Macrophage polarization: tumor-associated macrophages as a paradigm for polarized M2 mononuclear phagocytes. Trends Immunol. 2002;23 (11):549–55.
- 47. Bingle L, Brown NJ, Lewis CE. The role of tumour-associated macrophages in tumour progression: implications for new anticancer therapies. J Pathol. 2002;196 (3):254–65. http://dx.doi.org/10.1002/path.1027
- Balkwill F, Charles KA, Mantovani A. Smoldering and polarized inflammation in the initiation and promotion of malignant disease. Cancer Cell. 2005;7 (3):211–7. http://dx.doi.org/10.1016/j.ccr.2005.02.013
- Pollard JW. Tumour-educated macrophages promote tumour progression and metastasis. Nat Rev Cancer. 2004;4 (1):71–8. http://dx.doi.org/10.1038/nrc1256
- 50. Kirkpatrick K, Ogunkolade W, Elkak A, Bustin S, Jenkins P, Ghilchik M, et al. The mRNA expression of cyclo-oxygenase-2 (COX-2) and vascular endothelial growth factor (VEGF) in human breast cancer. Curr Med Res Opin. 2002;18 (4):237–41. http://dx.doi.org/10.1185/030079902125000633

51. Kaidi A, Qualtrough D, Williams AC, Paraskeva C. Direct transcriptional up-regulation of cyclooxygenase-2 by hypoxia-inducible factor (HIF)-1 promotes colorectal tumor cell survival and enhances HIF-1 transcriptional activity during hypoxia. Cancer Res. 2006;66 (13):6683–91.

http://dx.doi.org/10.1158/0008-5472.CAN-06-0425

- 52. Zhong H, Agani F, Baccala AA, Laughner E, Rioseco-Camacho N, Isaacs WB, et al. Increased expression of hypoxia inducible factor-1alpha in rat and human prostate cancer. Cancer Res. 1998;58 (23):5280–4.
- 53. Bilton RL, Booker GW. The subtle side to hypoxia inducible factor (HIFalpha) regulation. Eur J Biochem. 2003;270 (5):791–8.
- 54. Greenhough A, Smartt HJ, Moore AE, Roberts HR, Williams AC, Paraskeva C, Kaidi A. The COX-2/PGE2 pathway: key roles in the hallmarks of cancer and adaptation to the tumour microenvironment. Carcinogenesis. 2009;30 (3):377–86. http://dx.doi.org/10.1093/carcin/bgp014
- 55. Rahman M, Selvarajan K, Hasan MR, Chan AP, Jin C, Kim J, et al. Inhibition of COX-2 in colon cancer modulates tumor growth and MDR-1 expression to enhance tumor regression in therapy-refractory cancers in vivo. Neoplasia. 2012;14 (7):624–33.
- 56. Sarkar FH, Adsule S, Li Y, Padhye S. Back to the future: COX-2 inhibitors for chemoprevention and cancer therapy. Mini Rev Med Chem. 2007;7 (6):599–608.
- 57. Arun B, Goss P. The role of COX-2 inhibition in breast cancer treatment and prevention. Semin Oncol. 2004;31 (2 Suppl 7):22–9.
- 58. Brandao RD, Veeck J, Van de Vijver KK, Lindsey P, de Vries B, van Elssen CH, et al. A randomised controlled phase II trial of pre-operative celecoxib treatment reveals anti-tumour transcriptional response in primary breast cancer. Breast Cancer Res. 2013;15 (2):R29. http://dx.doi.org/10.1186/bcr3409
- 59. Yang YH, Yang YH, Cheng CL, Ho PS, Ko YC. The role of chemoprevention by selective cyclooxygenase-2 inhibitors in colorectal cancer patients a population-based study. BMC Cancer. 2012;12:582.

http://dx.doi.org/10.1186/1471-2407-12-582

- 60. Bae SH, Jung ES, Park YM, Kim BS, Kim BK, Kim DG, et al. Expression of cyclooxygenase-2 (COX-2) in hepatocellular carcinoma and growth inhibition of hepatoma cell lines by a COX-2 inhibitor, NS-398. Clin Cancer Res. 2001;7 (5):1410–8.
- 61. Parashar B, Latha Shankar S, O'Guin K, Butler J, Vikram B, Shafit-Zagardo B. Inhibition of human neuroblastoma cell growth by CAY10404, a highly selective Cox-2 inhibitor. J Neurooncol. 2005;71 (2):141–8.

http://dx.doi.org/10.1007/s11060-004-1721-3

62. Dickens DS, Cripe TP. Effect of combined cyclooxygenase-2 and matrix metalloproteinase inhibition on human sarcoma xenografts. J Pediatr Hematol Oncol. 2003;25 (9):709–14.

Section III

WT1 Gene Aberrations in Other Malignancies

Chapter 13

WT1 in Cardiac Development and Disease

Sjoerd N. Duim, Marie-José Goumans, Boudewijn P. T. Kruithof

Department of Molecular Cell Biology, Leiden University Medical Center, Leiden, The Netherlands

Authors for correspondence: Boudewijn P. T. Kruithof, PhD and Marie-José Goumans, PhD, Department of Molecular Cell Biology, Leiden University Medical Center, Einthovenweg 20, 2333 ZC Leiden, The Netherlands. Email: B.P.T.Kruithof@lumc.nl; M.J.T.H.Goumans@lumc.nl

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch13

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

The heart is essential for realizing the distribution of oxygen and nutrients throughout the body. Therefore, the heart is the first organ to develop and is already functional in its most primitive structure during embryogenesis. Recent studies indicate that the transcription factor Wilms' tumor-1 (WT1) is important for many aspects of cardiac development. WT1 expression is first observed in the proepicardium, a group of progenitor cells that give rise to a mesothelial sheet covering the heart, the epicardium. WT1 expression in epicardial cells is required for their epithelial-to-mesenchymal transformation forming epicardium-derived cells that will contribute to the formation of coronary vessels and interstitial fibroblasts. Endothelial cells within the heart also express WT1, whereas the endothelial cells in other

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

parts of the embryo do not. The endothelial expression of WT1 during cardiac development is likely to be important for vascular formation. After cardiac injury, WT1 is temporally upregulated in the epicardium and in the endothelial cells in the infarcted area and border zone, which points to a potential important role for WT1 in cardiac repair and regeneration. In this chapter, we describe the many faces of WT1 within the heart.

Key words: Cardiac development; Cardiac regeneration; Endothelial cells; Epicardium; Wilms' tumor-1

Introduction

The pleiotropic molecule Wilms' tumor-1 (WT1) is an transcription factor that was first discovered in renal tumors (1, 2). It contains four zinc-finger motifs at the C-terminus which are important for the binding of DNA to activate gene expression. Besides its function to activate the transcription of genes, WT1 is also involved in posttranscriptional processes (3). The expression of WT1 is essential during development of multiple organs, including kidneys, gonads, spleen, and the heart (4, 5). In the developing heart, WT1 is strongly expressed in the outer layer, i.e. the epicardium, and in the cardiac endothelial cells. After myocardial infarction this WT1 expression reemerges in both lineages. In this chapter, we describe the dynamic expression of WT1 during development and after cardiac injury. We focus on the multiple roles of WT1 including epithelial-to-mesenchymal transformation and angiogenesis in cardiac development, repair and regeneration.

WT1 in cardiac development

The expression of WT1 in the epicardium during early cardiac development

Knockout of *Wt1* in mice revealed that this transcription factor plays an essential role in cardiac development. In the absence of WT1, the vasculature of the heart is not formed, which disturbs proper formation of the heart and therefore results in prenatal death (3, 4). Already in the developing embryo, the heart is essential for the supply of oxygen and nutrients. Therefore, it is the first organ to develop and function during embryogenesis. The heart has a mesodermal origin and is formed through gastrulation. The earliest recognizable structure is the primitive heart tube, which is formed at embryonic day (E) 8 in mouse, corresponding with day 21 postfertilization in human. This hollow structure consists of two cardiac cell populations, namely cardiomyocytes on the outside and endothelial cells on the inside, which are separated by cardiac jelly (5) (Figure 1). The primitive tube elongates and undergoes rightward looping between E8.5 and E10.0 in mouse (days 23 and 28 in human) (Figure 1). Subsequent remodeling of the heart involves formation and expansion of the chambers, and formation of valves and septa, resulting in a septated four-chambered heart (6, 7)



Figure 1. Early cardiac development and the expression of WT1. E8.0: The primitive tube consists of cardiomyocytes (brown) on the outside and endothelial cells (red) on the inside, which are separated by cardiac jelly. E8.5–E10.5: The primitive tube elongates and undergoes rightward looping. At the inflow tract of the developing heart, the proepicardium arises and the epicardium formation has initiated. E11.5: The four chambers of the heart can be recognized, and septa and valves are being formed. The expression of WT1 is indicated in green. AVC, atrioventricular cushion; EPI, epicardium; LA, left atrium; LV, left ventricle; OFT, outflow tract; PEO, proepicardium; RA, right atrium; RV, right ventricle.

(Figure 1). Even though the heart is already functional at an early stage of fetal development, the growth and maturation of the heart are not completed until after birth (8).

A third population of cardiac cells envelopes the heart during development, the epicardium. The cells forming the epicardium are derived from the proepicardium, a heterogeneous transient cluster of cells (9), located at the base of the inflow tract of the developing heart (Figure 1). The earliest expression of WT1 in the heart is found in cells of the proepicardium, at E9.5 (10–12) (Figure 1 and 2a). Proepicardial cells reach the bare heart tube by formation of a tissue bridge or by free-floating vesicles and spread over the myocardium covering the complete heart, forming the epicardium. Recently, we have shown that the covering of the myocardium with WT1-positive cells occurs in a dorsal to ventral pattern between E9.75 and E10.5. In addition, the epicardial covering of the right ventricle is delayed and less dense compared to the left ventricle. Complete covering of the myocardium with WT1-positive cells is established in the mouse by E12.5 (11) (Figure 1). During human cardiac development, complete covering of the myocardium is observed at week 5 (13). After enveloping the entire heart, the epicardium remains positive for WT1 both in mouse and in human during embryonic development (3, 10–18) (Figure 1 and Figure 2a).

After the formation of the epicardium, a subset of epicardial cells undergoes epithelialto-mesenchymal transition (EMT), resulting in epicardium-derived cells (EPDCs) (19, 20)



Figure 2. Expression of WT1 and the fate of epicardial cells. (a) The epicardium arises from the proepicardium. A subset of cells undergo EMT and migrate into the subepicardium and myocardium. WT1 (green) is expressed by cells in the proepicardium, epicardium, EPDC, endothelial cells, and endocardial cells. (b) EPDCs contribute to the formation of fibroblasts, smooth muscle cells and endothelial cells; the contribution to cardiomyocytes and endocardial cells is controversial. Cardiac fibroblasts can be subdivided into interstitial fibroblasts, perivascular fibroblast, and valvular interstitial cells.

(Figure 2a). Lineage-tracing studies have shown that EPDCs migrate into the myocardium and contribute to cardiac fibroblasts, endothelial and smooth muscle cells of the cardiac blood vessels, and cardiomyocytes (9, 12, 19, 21–23), although the contribution to the latter is still under debate (11, 14, 16, 24) (Figure 2b). The contribution of WT1-positive cells to endothelial cells however is minimal (12).

The role of WT1 in embryonic epithelial-to-mesenchymal transition

In addition to the essential role of WT1 in the formation of the epicardium (3, 4), several studies have shown that WT1 serves as a regulator of epicardial EMT. *Wt1*-knockout mice show a reduction of subepicardial mesenchyme (3). In addition, epicardial cells are unable to detach from the epicardium, and EPDCs do not migrate into the subepicardium (18, 25).

Epicardial EMT is regulated by WT1 via multiple genes and pathways. Knockdown of *Wt1* in epicardial cells reduced the expression of SNAIL and SLUG, whereas the

downstream target E-cadherin was upregulated. SNAIL and SLUG are key regulators of the EMT process and have an inhibitory effect on the expression of the epithelial marker E-cadherin (26). WT1 directly promotes EMT by enhancing the expression of SNAIL and inhibiting the expression of E-cadherin (25). Furthermore, WT1 was shown to be a positive upstream regulator of the Wnt pathway, which influences diverse aspects of cardiogenesis and is important for epicardial EMT (27). Knockdown of *Wt1* resulted in a decrease of the downstream effectors of the Wnt pathway, *Ctnnb1* and *Lef1* (18). WT1 directly regulates RALDH2, an enzyme involved in retinoic acid (RA) synthesis, and is expressed in the epicardium (18, 28, 29). RA signaling is essential during embryonic development, and RA deficiency has been shown to result in cardiac abnormalities, similar to the phenotype of *Wt1*-knockout mice (20, 30). In chicken, induction of RA signaling in EPDCs results in upregulation of WT1 (31), indicating a positive feedback loop between WT1 and RA.

Although most *in vivo* studies suggest an inducing role for WT1 in epicardial EMT, *in vitro* repression of WT1 induced the transformation of both human and mice cobblestone-like EPDCs into spindle-shaped cells, indicating a context-dependent and possibly, concentration-dependent function of WT1 (32, 33).

Expression of WT1 in cardiac endothelial cells

The classical consensus is that EPDCs lose their expression of WT1 during their migration from subepicardium into the myocardial layer (12, 34). We have recently shown that the expression of WT1 is not restricted to the epicardium and subepicardium but is also present in the myocardial layer during development from E12.5 onward in mice and from week 5 after fertilization in humans (11, 13, 14) (Figure 2a and Figure 3). This expression of WT1 starts at the epicardial side and expands toward the luminal site of the heart (13, 14). The myocardial wall of the heart consists of an outer condensed part, the compact layer, and an inner loosely arranged part, the trabecular layer. Before birth, the expression of WT1 within the myocardium in mice is restricted to the compact layer and expands into the trabeculae in the neonatal heart (14). In human, the trabecular expression of WT1 is already observed before birth (13). The cellular composition of the heart comprises cardiomyocytes, cardiac fibroblasts, smooth muscle cells, and endothelial cells (35). Interestingly, in the myocardial layer of the heart, we found the expression of WT1 in endothelial cells of both small capillaries and the larger coronary vessels in mice and human (13, 14) (Figure 2a and Figure 3). In mice, the expression of WT1 in endothelial cells is still present at neonatal stages but gradually decreases before adulthood (14). In human, the expression of WT1 in endothelial cells of at least the arteries decreases before birth (13). Another difference between mice and human is



Figure 3. WT1 is expressed by cardiac endothelial cells during development. (a) Overview of the mouse heart at E17.5. WT1 is expressed in the epicardium and endothelial cells [coexpression of WT1 (green) and PECAM-1 (red)]. (b, c) Magnification of the connection between aorta and coronary vessel shows the specific cardiac expression of WT1 in endothelial cells. (d) WT1 is expressed in the vasculature of the heart during murine development. (e) WT1 is expressed in the vasculature of the heart during murine development. (e) WT1 is expressed in the vasculature of the heart during human development. A, aorta; CA, coronary artery; LA, left atrium; LV, left ventricle; RA, right atrium; RV, right ventricle.

the widespread expression of WT1 in endocardial cells at early stages during human cardiogenesis (Figure 2a). The differences in WT1 expression between mouse and human can be explained by the difference in maturation time during pregnancy, as well as the differences in dimensions (36, 37).

WT1 in the infarcted heart

The epicardial expression of WT1 decreases after birth and remains at low levels during normal homeostasis (Figure 4a). The endothelial expression of WT1 in the adult heart is low and mostly observed in some capillaries and cardiac veins (14). In contrast to the quiescent appearance during adult physiological conditions, after myocardial infarction (MI) the epicardial and endothelial cells re-express WT1 (Figure 4d).

MI is the most common type of ischemic heart disease and the leading cause of death in the Western world (38, 39). During MI, coronary occlusion leads to a reduced supply of oxygen to the cardiac muscle, resulting in massive cell death of cardiomyocytes. The extracellular matrix (ECM) is degraded ensuring infiltration of inflammatory cells, which remove the cellular debris generated during acute cardiac injury (40, 41). The expression of WT1 in the epicardium is already upregulated 1 day after infarction and is induced throughout the entire

epicardial layer of the heart (42). The subepicardium thickens by resurgence of the EMT process and is most pronounced at the infarcted area (43) (Figure 4b). The re-expression of WT1 and the revival of epicardial EMT suggest that WT1 regains its fetal role after MI. In the subepicardium, WT1 is expressed in the aSMA-positive cells but not in the endothelial cells (44) (data not shown) (Figure 4d). Lineage tracing of epicardial cells indicated that the WT1-positive cells do not migrate into the infarcted area (44, 45). Remarkably, priming the mouse heart before MI with Thymosin beta-4 resulted in the migration of epicardial cells into the myocardium and functional differentiation into cardiomyocytes after MI (22).



Figure 4. Expression of WT1 in the adult and injured heart. (a) The epicardial expression of WT1 is almost absent in the adult heart. (b) Myocardial infarction results in the loss of cardiomyocytes and the replacement by fibrotic scar tissue. After myocardial infarction, the epicardial expression of WT1 is reactivated in the entire heart. The thickening of the epicardium is most pronounced at the infarcted area. (c) Cardiac remodeling following myocardial infarction results in ventricular dilatation and impairment of cardiac pumping. (d) Myocardial infarction results in re-expression of WT1 in the epicardium, epicardial EMT, migration of EPDCs into the subepicardium, expression of WT1 in myofibroblasts of the subepicardium and in endothelial cells of the infarcted area and border zone. The expression of WT1 throughout adulthood and after injury is indicated in green. LV, left ventricle; RV, right ventricle.

In the infarcted area, an increase in granulation tissue is observed approximately 3 days after MI, which is characterized by the presence of interstitial fibroblasts, myofibroblasts, and forming blood vessels (12, 14, 43, 46, 47). As opposed to the expression in the subepicardium, upregulation of WT1 expression is present in endothelial cells and not in (myo) fibroblasts in the myocardial layer (14) (data not shown; Figure 4). Initially, upregulation of WT1 is observed in endothelial cells of the border zone and subsequently in the infarcted area, with a peak endothelial expression of WT1 at day 7 after MI (47). As time progresses, vessels in the infarcted area become more mature, fibrotic scar forms, and the expression of WT1 disappears (14). Interestingly, in the border zone, the expression of WT1 in endothelial cells is still detectable 4 weeks after MI, indicating that this region still undergoes active remodeling. The expression of WT1 in the epicardium gradual decreases after the first week to return to quiescent levels at 3 months after infarction (42) (Figure 4c). The remodeling and maturation of the fibrotic scar result in ventricular dilatation and impairment of cardiac functioning (48, 49) (Figure 4c).

The molecular mechanism causing WT1 reactivation in epicardial and endothelial cells is unclear. Inflammation might be a potential trigger for the activation of WT1 expression after MI. Proinflammatory cytokines, such as TNF- α , IL-1 β , and IL-6, are upregulated within the first few hours after injury (50). These cytokines are able to activate the transcription of NF- κ B (51), which is highly present after MI (52–54). NF- κ B upregulates WT1 expression (55), thereby potentially activating the epicardium after MI (Figure 5). The peak of NF- κ B induction is found at day 3 after MI (56), the same day the first signs of angiogenesis, including the endothelial WT1 expression, are visible in the border zone (47). Interestingly, WT1 has an anti-inflammatory role because it inhibits the expression of inflammatory cytokines by stimulation of IL-10 (57) (Figure 5).

The upregulation of epicardial WT1 after injury might also be caused by soluble factors released by the myocardium within the pericardial fluid (PF). Injection of PF from MI patients into the pericardial cavity of mice induces the expression of WT1 in epicardial cells in the absence of infarction (58, 59). In addition, PF of patients affects gene expression in epicardial cells that are involved in EMT; among others, the expression of SNAIL and TWIST is stimulated (59). After MI, PF contains an increased number of exosomes, which are small extracellular microvesicles. These vesicles contain bioactive molecules and are important for intracellular communication and activation (60). Recent proteomic analysis by Foglio et al. (59) showed that clusterin is highly enriched in exosomes of PF of patients after MI. Clusterin is involved in EMT in prostate cancer (61), and administration of clusterin in the pericardial cavity induced EMT in epicardial cells (59).

The upregulation of endothelial WT1 after injury might be caused by hypoxia (Figure 5). Hypoxia is a well-known condition that induces vascular formation during development



Figure 5. Working model of the regulation of WT1 and its target genes. The expression of WT1 is activated during embryonic development and in the adult heart after injury. Upregulation of WT1 is caused by hypoxia, growth factors, and inflammation. WT1 in turn is able to activate multiple genes that are important for the regulation of different processes, including EMT, angiogenesis, remodeling of the ECM.

and after MI via the hypoxia-inducible factor-1-alpha (HIF1 α) (62). The expression of WT1 can be directly upregulated by hypoxia through the HIF1 α -responsive elements in the WT1 promoter (63) (Figure 5). *In vitro* exposure of human endothelial cells to hypoxia increased the expression of WT1 (14, 64). Important for the response to hypoxia is that one of the downstream targets of WT1 is VEGF (65), one of the most potent angiogenic factors, both in embryonic vascular formation and in the growth of blood vessels after injury (66) (Figure 5).

The role of WT1 in endothelial cells

The expression of WT1 in endothelial cells is only found in the heart and not in other organs of the developing embryo (14) (Figure 3a–3c). The cardiac-specific expression of WT1 is supported by a recent study that identified the unique gene expression profiles of endothelial cells, isolated form different organs (67). In both human and mouse, the expression of WT1

in cardiac endothelial cells is significantly higher compared to noncardiac endothelial cells, confirming cardiac-specific expression of WT1 in endothelial cells (67). In fact, overexpression of WT1 was sufficient to differentiate endothelial cells into a more cardiac specialized population. The importance of WT1 for the development of blood vessels is highlighted by the re-expression in the cardiac vasculature after MI in mouse and after exposure of rats to hypoxia (14, 64). A study by Coosemans et al. (68) claimed the expression of WT1 in cardiac endothelial cells of patients that died after MI. Although the expression of WT1 in cardiac endothelial cells is unique during normal conditions, the expression is also present in endothelial cells in other organs in a pathological condition. WT1 is found in endothelial cells of the skin in patients with chronic dermatitis (69), and WT1 has been observed in endothelial cells in a wide variety of tumors (68–72).

It is unclear why under physiological conditions WT1 expression is found only in cardiac endothelial cells. In contrast to other organs, the heart has the unique feature that it is exposed to cyclic strain (73). It is known that mechanical forces during early development play an important role in cardiac morphology (74). In addition, cyclic strain is able to regulate the process of EMT (75). It is therefore tempting to speculate that stimulation of EMT by cyclic strain is regulated by an upregulation of the expression of WT1. Alternatively, WT1 might be induced by TGF β , which is known to be upregulated by cyclic strain. TGF β is able to upregulate WT1 expression via Par-4 (76–78). On the contrary, WT1 works as a negative feedback loop on TGF β , by repressing its expression (79, 80).

At the very early stages of development, the fetal heart is predominantly dependent on glucose metabolism and shortly after birth the heart energy metabolism switches to fatty acid oxidation (81). Facilitating the uptake of fatty acids is a unique feature of cardiac endothelial cells (67, 82). WT1 expression is known to be essential for the cardiac endothelial fingerprint; therefore, WT1 might be important for the regulation of cardiac endothelial cell metabolism.

Patients with Denys–Drash syndrome (DDS), carrying partial-loss-of-function mutations in the *WT1* gene, develop glomerulosclerosis. In addition, the capillaries of the glomeruli show abnormal development. A cause of these malformations is found in a strong decrease in the expression of platelet endothelial cell adhesion molecule-1 (PECAM-1) in endothelial cell of the glomeruli (83). PECAM-1 is part of intercellular junctions and is present in mature vascular structures; additionally, its expression is upregulated during formation and remodeling of vascular networks (84, 85). WT1 is a positive regulator of PECAM-1 (71); this may explain the poor organization of capillaries in patients with DDS. Knockdown of *WT1* in human endothelial cells confirms the importance of WT1 in the formation of vascular networks, as these cells are unable to form proper networks (14, 64). The angiogenesis role of WT1 is supported by a reduced sprouting capacity in an aortic ring angiogenesis

assay from mice lacking WT1 expression in endothelial cells. In addition, the same study revealed that the vessel density in matrigel plugs after subcutaneously injection, in mice lacking WT1 expression in endothelial cells, is significantly reduced compared to wild-type animals (71). Furthermore, deletion of WT1 in endothelial cells resulted in major reduction in cardiac vessel formation during mouse cardiac development supporting the presence of WT1 and the essential role of WT in cardiac endothelial cells (86).

Endothelial cells are anchored to a basement membrane that ensures structural and organizational stability. During vascular formation and remodeling, reorganization of the ECM is essential (87, 88). Remodeling of the ECM is the net result between synthesis and decomposition of the ECM (89). Degradation of the ECM is facilitated by an increase and activation of latent matrix metalloproteinases (MMPs). WT1 is able to directly upregulate the expression of MMP9 (70), thereby facilitating the degradation of ECM. The basement membrane is mostly made up of collagen IV, which is degraded by MMP9 (90). Interestingly, the study of Johnson et al. (91) showed that in the absence of MMP9 revascularization of infarcted tissue is strongly impaired, confirming that remodeling is essential for angiogenesis. The role of WT1 in remodeling is further strengthened by proteomic analysis in patients with DDS. Glomerular podocytes with WT1 mutations have a disturbed production of proteins forming the cytoskeleton (92). Furthermore, the expression of intermediate filament Nestin is regulated by WT1 (93). The expression of Nestin is increased in regenerating tissue and is believed to participate in cellular remodeling and angiogenesis (94-96). Coexpression of WT1 and Nestin was found in the epicardium and endothelial cells of the embryonic heart in mice (93) and in vascular endothelium of patients who died after MI (97, 98).

Changes in the cytoskeleton are also required for cells to adapt to a less differentiated phenotype, allowing them to proliferate and migrate. Nestin is present in proliferating progenitor cells and positively regulates proliferation and migration (96). Within the epicardium, a positive correlation was found between WT1 and proliferation (99). In addition, WT1 plays a role in regulating the cell cycle. *In vitro* studies knocking down WT1 in human endothelial cells show reduced proliferation and migration (14, 72). Proliferation of endothelial cells is directly regulated by WT1 via Cyclin D1 (14, 100), one of the many regulators of the cell cycle and present in the G1 phase (101). Interestingly, the expression of WT1 is upregulated in embryonic stem cells during embryonic body differentiation, a proliferative period for mesenchymal cells. Upon cellular differentiation, the expression of WT1 was reduced (25). The positive role of WT1 on migration might be the result of direct repression of the promoter of Cxcl10, an inhibitory chemokine preventing angiogenesis (17).

Finally, WT1 is known to play a role in apoptosis. This was already noticed in 1993 in *Wt1*-knockout mice; embryonic tissue of the kidney showed more cell death compared to wild-type littermates (4). Over the last years, it has become clear that WT1 can directly

regulate genes involved in apoptosis; however, it depends on the cellular context if WT1 has a pro- or anti-apoptotic effect (102). Future research is required to investigate the role of WT1 in apoptosis of endothelial cells; potentially WT1 protects the forming and maturing blood vessels against cell death.

Together these observations suggest a role for WT1 in the remodeling, proliferation, and migration of cardiac endothelial cells and the formation of a proper vascular network (Figure 5). Stress factors such as hypoxia and inflammation are likely to play a role in the activation of WT1 both during cardiac development and in the response after injury (Figure 5). Future research, focusing on the molecular mechanisms, can hopefully reveal all pathways by which the angiogenic function of WT1 can be explained.

Clinical perspective

Restoring the cardiac blood flow is the most important treatment of ischemic cardiomyopathy at this moment. To improve cardiac output, the cardiac wall consisting of cardiomyocytes, fibroblasts, endothelial and smooth muscle cells, has to be rebuild. Transplantation of cardiac stem cells after infarction improved the function (103-105); however, difficulties in acquisition of human tissue and *in vitro* expansion of cells limit the clinical applicability. An interesting approach would be to take advantage of the properties of WT1-expressing cells. The differentiation potential of WT1-expressing epicardial cells during development into the vasculature, fibroblasts and cardiomyocytes has positioned the epicardium as a promising target (20, 106, 107). WT1-expressing stem cell-like cells are residing in the epicardium (58, 108). In addition, Chong and colleagues (109) showed that cardiac colony-forming units originate from WT1-positive cells within the epicardium. Finally, activation of WT1 expression within the epicardium after injury revives the fetal differentiation potential in the epicardial cells. Facilitating the differentiation of these WT1-expressing cells towards cardiomyocytes, fibroblasts, endothelial and smooth muscles cells could provide a great tool to improve cardiac regeneration after injury. In addition, the indicated role of WT1 in the endothelial cells during development and injury has positioned the WT1-expressing endothelial cells as a potential target for improving angiogenesis in the diseased area.

A recent study indicates a role for the epicardium in autonomic modulation during early development. Within the initial stages of epicardial formation, WT1 in epicardial cells is coexpressed with the neuronal markers TUBB3 and NCAM (15). Interestingly, the expression of WT1 was also found in the ventral region of the neural tube, as well as the roof of the 4th ventricle of the brain, supporting the neuronal phenotype of the epicardium. Dysfunctioning of the cardiac autonomic nervous system plays a role in the pathogenesis of arrhythmias (110) and hypertension (111) and is involved in disease progression in heart failure (112). Understanding the mechanistic role of WT1 in the formation of the cANS

might help to unravel processes that govern normal cANS development and opens possibilities for treatment after cardiac injury.

WT1 is assoaciated with major generation processes during cardiac development like formation of epicardium, cardiac vasculature, valves, cANS, and also myocardial wall maturation, but also with major regeneration processes during cardiac repair like scar formation and angiogenesis. More knowledge on the upregulation of WT1 in cardiac cells and their subsequent response can contribute to the development and improvement of therapeutic strategies for cardiac repair, and thereby restoring a functional contractile cardiac wall.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

Not relevant.

References

- Call KM, Glaser T, Ito CY, Buckler AJ, Pelletier J, Haber DA, et al. Isolation and characterization of a zinc finger polypeptide gene at the human chromosome 11 Wilms' tumor locus. Cell. 1990;60:509–20.
- Rivera MN, Haber DA. Wilms' tumour: connecting tumorigenesis and organ development in the kidney. Nat Rev Cancer. 2005;5:699–712. http://dx.doi.org/10.1038/nrc1696
- Moore AW, McInnes L, Kreidberg J, Hastie ND, Schedl A. YAC complementation shows a requirement for Wt1 in the development of epicardium, adrenal gland and throughout nephrogenesis. Development. 1999;126:1845–57.
- 4. Kreidberg JA, Sariola H, Loring JM, Maeda M, Pelletier J, Housman D, et al. WT-1 is required for early kidney development. Cell. 1993;74:679–91.
- Manner J, Wessel A, Yelbuz TM. How does the tubular embryonic heart work? Looking for the physical mechanism generating unidirectional blood flow in the valveless embryonic heart tube. Dev Dyn. 2010;239:1035–46. http://dx.doi.org/10.1002/dvdy.22265
- Moorman AF, Christoffels VM. Cardiac chamber formation: development, genes, and evolution. Physiol Rev. 2003;83:1223–67. http://dx.doi.org/10.1152/physrev.00006.2003

- 7. Anderson RH, Webb S, Brown NA, Lamers W, Moorman A. Development of the heart: (2) Septation of the atriums and ventricles. Heart. 2003;89:949-58.
- 8. Kruithof BP, Krawitz SA, Gaussin V. Atrioventricular valve development during late embryonic and postnatal stages involves condensation and extracellular matrix remodeling. Dev Biol. 2007;302:208-17.

http://dx.doi.org/10.1016/j.ydbio.2006.09.024

9. Katz TC, Singh MK, Degenhardt K, Rivera-Feliciano J, Johnson RL, Epstein JA, et al. Distinct compartments of the proepicardial organ give rise to coronary vascular endothelial cells. Dev Cell. 2012;22:639-50.

http://dx.doi.org/10.1016/j.devcel.2012.01.012

- 10. Armstrong JF, Pritchard-Jones K, Bickmore WA, Hastie ND, Bard JB. The expression of the Wilms' tumour gene, WT1, in the developing mammalian embryo. Mech Dev. 1993;40:85-97.
- 11. Vicente-Steijn R, Scherptong RW, Kruithof BP, Duim SN, Goumans MJ, Wisse LJ, et al. Regional differences in WT-1 and Tcf21 expression during ventricular development: implications for myocardial compaction. PLoS One. 2015;10:e0136025. http://dx.doi.org/10.1371/journal.pone.0136025
- Zhou B, Ma Q, Rajagopal S, Wu SM, Domian I, Rivera-Feliciano J, et al. Epicar-12. dial progenitors contribute to the cardiomyocyte lineage in the developing heart. Nature. 2008;454:109-13.

http://dx.doi.org/10.1038/nature07060

- 13. Duim SN, Smits AM, Kruithof BP, Goumans MJ. The roadmap of WT1 protein expression in the human fetal heart. J Mol Cell Cardiol. January 2016;90:139-45. http://dx.doi.org/10.1016/j.yjmcc.2015.12.008
- Duim SN, Kurakula K, Goumans MJ, Kruithof BP. Cardiac endothelial cells express 14. Wilms' tumor-1: Wt1 expression in the developing, adult and infarcted heart. J Mol Cell Cardiol. 2015;81:127-35.

http://dx.doi.org/10.1016/j.yjmcc.2015.02.007

- Kelder TP, Duim SN, Vicente-Steijn R, Végh AMD, Kruithof BPT, Smits AM, et 15. al. The epicardium as modulator of the cardiac autonomic response during early development. J Mol Cell Cardiol. 2015;89(Pt B):251-9. http://dx.doi.org/10.1016/j.yjmcc.2015.10.025
- 16. Rudat C, Kispert A. Wt1 and epicardial fate mapping. Circ Res. 2012;111:165-9. http://dx.doi.org/10.1161/CIRCRESAHA.112.273946
- 17. Velecela V, Lettice LA, Chau YY, Slight J, Berry RL, Thornburn A, et al. WT1 regulates the expression of inhibitory chemokines during heart development. Hum Mol Genet. 2013;22:5083-95. http://dx.doi.org/10.1093/hmg/ddt358

- von Gise A, Zhou B, Honor LB, Ma Q, Petryk A, Pu WT. WT1 regulates epicardial epithelial to mesenchymal transition through beta-catenin and retinoic acid signaling pathways. Dev Biol. 2011;356:421–31. http://dx.doi.org/10.1093/hmg/ddt358
- Gittenberger-de Groot AC, Vrancken Peeters MP, Mentink MM, Gourdie RG, Poelmann RE. Epicardium-derived cells contribute a novel population to the myocardial wall and the atrioventricular cushions. Circ Res. 1998;82:1043–52.
- 20. Wessels A, Perez-Pomares JM. The epicardium and epicardially derived cells (EPDCs) as cardiac stem cells. Anat Rec A Discov Mol Cell Evol Biol. 2004;276:43–57. http://dx.doi.org/10.1002/ar.a.10129
- Cai CL, Martin JC, Sun Y, Cui L, Wang L, Ouyang K, et al. A myocardial lineage derives from Tbx18 epicardial cells. Nature. 2008;454:104–8. http://dx.doi.org/10.1038/nature06969
- 22. Smart N, Bollini S, Dube KN, Vieira JM, Zhou B, Davidson S, et al. De novo cardiomyocytes from within the activated adult heart after injury. Nature. 2011;474:640-4.

http://dx.doi.org/10.1038/nature10188

 Smart N, Risebro CA, Melville AA, Moses K, Schwartz RJ, Chien KR, et al. Thymosin beta4 induces adult epicardial progenitor mobilization and neovascularization. Nature. 2007;445:177–82.

http://dx.doi.org/10.1038/nature05383

- 24. Christoffels VM, Grieskamp T, Norden J, Mommersteeg MT, Rudat C, Kispert A. Tbx18 and the fate of epicardial progenitors. Nature. 2009;458:E8–9; discussion E10. http://dx.doi.org/10.1038/nature07916
- 25. Martinez-Estrada OM, Lettice LA, Essafi A, Guadix JA, Slight J, Velecela V, et al. Wt1 is required for cardiovascular progenitor cell formation through transcriptional control of Snail and E-cadherin. Nat Genet. 2010;42:89–93. http://dx.doi.org/10.1038/ng.494
- 26. Peinado H, Portillo F, Cano A. Transcriptional regulation of cadherins during development and carcinogenesis. Int J Dev Biol. 2004;48:365–75. http://dx.doi.org/10.1387/ijdb.041794hp
- 27. Zamora M, Manner J, Ruiz-Lozano P. Epicardium-derived progenitor cells require betacatenin for coronary artery formation. Proc Natl Acad Sci U S A. 2007;104:18109–14. http://dx.doi.org/10.1073/pnas.0702415104
- Guadix JA, Ruiz-Villalba A, Lettice L, Velecela V, Munoz-Chapuli R, Hastie ND, et al. Wt1 controls retinoic acid signalling in embryonic epicardium through transcriptional activation of Raldh2. Development. 2011;138:1093–7. http://dx.doi.org/10.1242/dev.044594

- 29. Moss JB, Xavier-Neto J, Shapiro MD, Nayeem SM, McCaffery P, Drager UC, et al. Dynamic patterns of retinoic acid synthesis and response in the developing mammalian heart. Dev Biol. 1998;199:55–71. http://dx.doi.org/10.1006/dbio.1998.8911
- Hoover LL, Burton EG, Brooks BA, Kubalak SW. The expanding role for retinoid signaling in heart development. ScientificWorldJournal. 2008;8:194–211. http://dx.doi.org/10.1100/tsw.2008.39
- 31. Braitsch CM, Combs MD, Quaggin SE, Yutzey KE. Pod1/Tcf21 is regulated by retinoic acid signaling and inhibits differentiation of epicardium-derived cells into smooth muscle in the developing heart. Dev Biol. 2012;368:345–57. http://dx.doi.org/10.1016/j.ydbio.2012.06.002
- 32. Bax NA, van Oorschot AA, Maas S, Braun J, van Tuyn J, de Vries AA, et al. In vitro epithelial-to-mesenchymal transformation in human adult epicardial cells is regulated by TGFbeta-signaling and WT1. Basic Res Cardiol. 2011;106:829–47. http://dx.doi.org/10.1007/s00395-011-0181-0
- 33. Takeichi M, Nimura K, Mori M, Nakagami H, Kaneda Y. The transcription factors Tbx18 and Wt1 control the epicardial epithelial-mesenchymal transition through bi-directional regulation of Slug in murine primary epicardial cells. PloS One. 2013;8:e57829. http://dx.doi.org/10.1371/journal.pone.0057829
- 34. Perez-Pomares JM, Phelps A, Sedmerova M, Carmona R, Gonzalez-Iriarte M, Munoz-Chapuli R, et al. Experimental studies on the spatiotemporal expression of WT1 and RALDH2 in the embryonic avian heart: a model for the regulation of myocardial and valvuloseptal development by epicardially derived cells (EPDCs). Dev Biol. 2002;247:307–26.
- 35. Banerjee I, Fuseler JW, Price RL, Borg TK, Baudino TA. Determination of cell types and numbers during cardiac development in the neonatal and adult rat and mouse. Am J Physiol Heart Circul Physiol. 2007;293:H1883–91. http://dx.doi.org/10.1152/ajpheart.00514.2007
- Krishnan A, Samtani R, Dhanantwari P, Lee E, Yamada S, Shiota K, et al. A detailed comparison of mouse and human cardiac development. Pediatr Res. 2014;76:500–7.

http://dx.doi.org/10.1038/pr.2014.128

37. Wessels A, Sedmera D. Developmental anatomy of the heart: a tale of mice and man. Physiol Genomics. 2003;15:165–76.

http://dx.doi.org/10.1152/physiolgenomics.00033.2003

Go AS, Mozaffarian D, Roger VL, Benjamin EJ, Berry JD, Borden WB, et al. Heart disease and stroke statistics – 2013 update: a report from the American Heart Association. Circulation. 2013;127:e6–e245.
 http://dx.doi.org/10.1161/CIP.0b012o21828124od

http://dx.doi.org/10.1161/CIR.0b013e31828124ad

- Mathers CD, Loncar D. Projections of global mortality and burden of disease from 2002 to 2030. PLoS Med. 2006;3:e442. http://dx.doi.org/10.1371/journal.pmed.0030442
- 40. Moore L, Fan D, Basu R, Kandalam V, Kassiri Z. Tissue inhibitor of metalloproteinases (TIMPs) in heart failure. Heart Fail Rev. 2012;17:693–706. http://dx.doi.org/10.1007/s10741-011-9266-y
- Dingenouts CK, Goumans MJ, Bakker W. Mononuclear cells and vascular repair in HHT. Front Genet. 2015;6:114. http://dx.doi.org/10.3389/fgene.2015.00114
- 42. van Wijk B, Gunst QD, Moorman AF, van den Hoff MJ. Cardiac regeneration from activated epicardium. PloS One. 2012;7:e44692. http://dx.doi.org/10.1371/journal.pone.0044692
- 43. Desmouliere A, Geinoz A, Gabbiani F, Gabbiani G. Transforming growth factor-beta 1 induces alpha-smooth muscle actin expression in granulation tissue myofibroblasts and in quiescent and growing cultured fibroblasts. J Cell Biol. 1993;122:103–11.
- 44. Zhou B, Honor LB, He H, Ma Q, Oh JH, Butterfield C, et al. Adult mouse epicardium modulates myocardial injury by secreting paracrine factors. J Clin Invest. 2011;121:1894–904.
- 45. Zhou B, Honor LB, Ma Q, Oh JH, Lin RZ, Melero-Martin JM, et al. Thymosin beta 4 treatment after myocardial infarction does not reprogram epicardial cells into cardiomyocytes. J Mol Cell Cardiol. 2012;52:43–7. http://dx.doi.org/10.1016/j.yjmcc.2011.08.020
- 46. Fan D, Takawale A, Lee J, Kassiri Z. Cardiac fibroblasts, fibrosis and extracellular matrix remodeling in heart disease. Fibrogenesis Tissue Repair. 2012;5:15. http://dx.doi.org/10.1186/1755-1536-5-15
- 47. Zhao T, Zhao W, Chen Y, Ahokas RA, Sun Y. Vascular endothelial growth factor (VEGF)-A: role on cardiac angiogenesis following myocardial infarction. Microvasc Res. 2010;80:188–94.

http://dx.doi.org/10.1016/j.mvr.2010.03.014

- Cleutjens JP, Blankesteijn WM, Daemen MJ, Smits JF. The infarcted myocardium: simply dead tissue, or a lively target for therapeutic interventions. Cardiovasc Res. 1999;44:232–41.
- 49. Sun Y, Weber KT. Infarct scar: a dynamic tissue. Cardiovasc Res. 2000;46:250–6.
- 50. Nian M, Lee P, Khaper N, Liu P. Inflammatory cytokines and postmyocardial infarction remodeling. Circ Res. 2004;94:1543–53. http://dx.doi.org/10.1161/01.RES.0000130526.20854.fa
- 51. Lawrence T. The nuclear factor NF-kappaB pathway in inflammation. Cold Spring Harb Perspect Biol. 2009;1:a001651. http://dx.doi.org/10.1101/cshperspect.a001651

- 52. Chandrasekar B, Freeman GL. Induction of nuclear factor kappaB and activation protein 1 in postischemic myocardium. FEBS Lett. 1997;401:30–4.
- 53. Hall G, Hasday JD, Rogers TB. Regulating the regulator: NF-kappaB signaling in heart. J Mol Cell Cardiol. 2006;41:580–91. http://dx.doi.org/10.1016/j.yjmcc.2006.07.006
- 54. Morgan EN, Boyle EM, Jr., Yun W, Griscavage-Ennis JM, Farr AL, Canty TG, Jr., et al. An essential role for NF-kappaB in the cardioadaptive response to ischemia. Ann Thorac Surg. 1999;68:377–82.
- 55. Dehbi M, Hiscott J, Pelletier J. Activation of the wt1 Wilms' tumor suppressor gene by NF-kappaB. Oncogene. 1998;16:2033–9. http://dx.doi.org/10.1038/sj.onc.1201747
- 56. Tillmanns J, Carlsen H, Blomhoff R, Valen G, Calvillo L, Ertl G, et al. Caught in the act: in vivo molecular imaging of the transcription factor NF-kappaB after myocar-dial infarction. Biochem Biophys Res Commun. 2006;342:773–4. http://dx.doi.org/10.1016/j.bbrc.2006.02.024
- 57. Sciesielski LK, Kirschner KM, Scholz H, Persson AB. Wilms' tumor protein Wt1 regulates the Interleukin-10 (IL-10) gene. FEBS Lett. 2010;584:4665–71. http://dx.doi.org/10.1016/j.febslet.2010.10.045
- 58. Limana F, Bertolami C, Mangoni A, Di Carlo A, Avitabile D, Mocini D, et al. Myocardial infarction induces embryonic reprogramming of epicardial c-kit(+) cells: role of the pericardial fluid. J Mol Cell Cardiol. 2010;48:609–18. http://dx.doi.org/10.1016/j.yjmcc.2009.11.008
- 59. Foglio E, Puddighinu G, Fasanaro P, D'Arcangelo D, Perrone GA, Mocini D, et al. Exosomal clusterin, identified in the pericardial fluid, improves myocardial performance following MI through epicardial activation, enhanced arteriogenesis and reduced apoptosis. Int J Cardiol. 2015;197:333–47. http://dx.doi.org/10.1016/j.ijcard.2015.06.008
- Sluijter JP, Verhage V, Deddens JC, van den Akker F, Doevendans PA. Microvesicles and exosomes for intracardiac communication. Cardiovasc Res. 2014;102:302–11.

http://dx.doi.org/10.1093/cvr/cvu022

- 61. Shiota M, Zardan A, Takeuchi A, Kumano M, Beraldi E, Naito S, et al. Clusterin mediates TGF-beta-induced epithelial-mesenchymal transition and metastasis via Twist1 in prostate cancer cells. Cancer Res. 2012;72:5261–72. http://dx.doi.org/10.1158/0008-5472.CAN-12-0254
- Carmeliet P. Mechanisms of angiogenesis and arteriogenesis. Nat Med. 2000;6:389–95.
 http://du/doi.org/10.1028/74/51

http://dx.doi.org/10.1038/74651
WT1 in the heart

- 63. Wagner KD, Wagner N, Wellmann S, Schley G, Bondke A, Theres H, et al. Oxygenregulated expression of the Wilms' tumor suppressor Wt1 involves hypoxia-inducible factor-1 (HIF-1). FASEB J. 2003;17:1364–6. http://dx.doi.org/10.1096/fj.02-1065fje
- 64. Wagner KD, Wagner N, Bondke A, Nafz B, Flemming B, Theres H, et al. The Wilms' tumor suppressor Wt1 is expressed in the coronary vasculature after myocardial infarction. FASEB J. 2002;16:1117–9. http://dx.doi.org/10.1096/fj.01-0986fje
- 65. McCarty G, Awad O, Loeb DM. WT1 protein directly regulates expression of vascular endothelial growth factor and is a mediator of tumor response to hypoxia. J Biol Chem. 2011;286:43634-43.

http://dx.doi.org/10.1074/jbc.M111.310128

66. Carmeliet P, Ferreira V, Breier G, Pollefeyt S, Kieckens L, Gertsenstein M, et al. Abnormal blood vessel development and lethality in embryos lacking a single VEGF allele. Nature. 1996;380:435–9.
 http://dx.doi.org/10.1038/380:435–0.

http://dx.doi.org/10.1038/380435a0

67. Coppiello G, Collantes M, Sirerol-Piquer MS, Vandenwijngaert S, Schoors S, Swinnen M, et al. Meox2/Tcf15 heterodimers program the heart capillary endothelium for cardiac fatty acid uptake. Circulation. 2015;131:815–26. http://dx.doi.org/10.1161/CIRCULATIONAHA.114.013721

68. Coosemans A, Van Hove T, Verbist G, Moons L, Neven P, Moerman P, et al. Wilms' tumour gene 1 (WT1) positivity in endothelial cells surrounding epithelial uterine tumours. Histopathology. 2009;54:384–7.

http://dx.doi.org/10.1111/j.1365-2559.2008.03213.x

- 69. Timar J, Meszaros L, Orosz Z, Albini A, Raso E. WT1 expression in angiogenic tumours of the skin. Histopathology. 2005;47:67–73. http://dx.doi.org/10.1111/j.1365-2559.2005.02169.x
- Katuri V, Gerber S, Qiu X, McCarty G, Goldstein SD, Hammers H, et al. WT1 regulates angiogenesis in Ewing Sarcoma. Oncotarget. 2014;5:2436–49. http://dx.doi.org/10.18632/oncotarget.1610
- 71. Wagner KD, Cherfils-Vicini J, Hosen N, Hohenstein P, Gilson E, Hastie ND, et al. The Wilms' tumour suppressor Wt1 is a major regulator of tumour angiogenesis and progression. Nat Commun. 2014;5:5852. http://dx.doi.org/10.1038/ncomms6852
- Wagner N, Michiels JF, Schedl A, Wagner KD. The Wilms' tumour suppressor WT1 is involved in endothelial cell proliferation and migration: expression in tumour vessels in vivo. Oncogene. 2008;27:3662–72. http://dx.doi.org/10.1038/sj.onc.1211044

Duim et al.

- 73. Gupta V, Grande-Allen KJ. Effects of static and cyclic loading in regulating extracellular matrix synthesis by cardiovascular cells. Cardiovasc Res. 2006;72:375–83. http://dx.doi.org/10.1016/j.cardiores.2006.08.017
- Miller CE, Donlon KJ, Toia L, Wong CL, Chess PR. Cyclic strain induces proliferation of cultured embryonic heart cells. In Vitro Cell Dev Biol Anim. 2000;36:633–9.

http://dx.doi.org/10.1290/1071-2690(2000)036<0633:CSIPOC>2.0.CO;2

- 75. Balachandran K, Alford PW, Wylie-Sears J, Goss JA, Grosberg A, Bischoff J, et al. Cyclic strain induces dual-mode endothelial-mesenchymal transformation of the cardiac valve. Proc Natl Acad Sci U S A. 2011;108:19943–8. http://dx.doi.org/10.1073/pnas.1106954108
- 76. Berasain C, Herrero JI, Garcia-Trevijano ER, Avila MA, Esteban JI, Mato JM, et al. Expression of Wilms' tumor suppressor in the liver with cirrhosis: relation to hepatocyte nuclear factor 4 and hepatocellular function. Hepatology. 2003;38:148–57.

http://dx.doi.org/10.1053/jhep.2003.50269

77. Chaudhry P, Fabi F, Singh M, Parent S, Leblanc V, Asselin E. Prostate apoptosis response-4 mediates TGF-beta-induced epithelial-to-mesenchymal transition. Cell Death Dis. 2014;5:e1044.

http://dx.doi.org/10.1038/cddis.2014.7

 Richard DJ, Schumacher V, Royer-Pokora B, Roberts SG. Par4 is a coactivator for a splice isoform-specific transcriptional activation domain in WT1. Genes Dev. 2001;15:328–39.

http://dx.doi.org/10.1101/gad.185901

- 79. Sakairi T, Abe Y, Kopp JB. TGF-beta1 reduces Wilms' tumor suppressor gene expression in podocytes. Nephrol Dial Transplant. 2011;26:2746–52. http://dx.doi.org/10.1093/ndt/gfr061
- Dey BR, Sukhatme VP, Roberts AB, Sporn MB, Rauscher FJ, 3rd, Kim SJ. Repression of the transforming growth factor-beta 1 gene by the Wilms' tumor suppressor WT1 gene product. Mol Endocrinol. 1994;8:595–602. http://dx.doi.org/10.1210/mend.8.5.8058069
- Kong B, Liu YL, Lu XD. Microarray-bioinformatics analysis of altered genomic expression profiles between human fetal and infant myocardium. Chin Med J. 2008;121:1257–64.
- 82. Iruretagoyena JI, Davis W, Bird C, Olsen J, Radue R, Teo Broman A, et al. Metabolic gene profile in early human fetal heart development. Mol Hum Reprod. 2014;20:690–700.

http://dx.doi.org/10.1093/molehr/gau026

WT1 in the heart

- Natoli TA, Liu J, Eremina V, Hodgens K, Li C, Hamano Y, et al. A mutant form of the Wilms' tumor suppressor gene WT1 observed in Denys-Drash syndrome interferes with glomerular capillary development. J Am Soc Nephrol. 2002;13:2058–67.
- 84. Chen Z, Tzima E. PECAM-1 is necessary for flow-induced vascular remodeling. Arterioscler Thromb Vasc Biol. 2009;29:1067–73. http://dx.doi.org/10.1161/ATVBAHA.109.186692
- 85. Yang S, Graham J, Kahn JW, Schwartz EA, Gerritsen ME. Functional roles for PECAM-1 (CD31) and VE-cadherin (CD144) in tube assembly and lumen formation in three-dimensional collagen gels. Am J Pathol. 1999;155:887–95. http://dx.doi.org/10.1016/S0002-9440(10)65188-7
- 86. Cano E, Carmona R, Ruiz-Villalba A, Rojas A, Chau YY, Wagner KD, et al. Extracardiac septum transversum/proepicardial endothelial cells pattern embryonic coronary arterio-venous connections. Proc Natl Acad Sci U S A. 2016;113(3):656–61. http://dx.doi.org/10.1073/pnas.1509834113
- Stupack DG, Cheresh DA. ECM remodeling regulates angiogenesis: endothelial integrins look for new ligands. Sci STKE. 2002;2002:pe7. http://dx.doi.org/10.1126/stke.2002.119.pe7
- Heissig B, Hattori K, Friedrich M, Rafii S, Werb Z. Angiogenesis: vascular remodeling of the extracellular matrix involves metalloproteinases. Curr Opin Hematol. 2003;10:136-41.
- Davis GE, Senger DR. Endothelial extracellular matrix: biosynthesis, remodeling, and functions during vascular morphogenesis and neovessel stabilization. Circ Res. 2005;97:1093–107.

http://dx.doi.org/10.1161/01.RES.0000191547.64391.e3

- 90. Zeng ZS, Cohen AM, Guillem JG. Loss of basement membrane type IV collagen is associated with increased expression of metalloproteinases 2 and 9 (MMP-2 and MMP-9) during human colorectal tumorigenesis. Carcinogenesis. 1999;20:749–55.
- 91. Johnson C, Sung HJ, Lessner SM, Fini ME, Galis ZS. Matrix metalloproteinase-9 is required for adequate angiogenic revascularization of ischemic tissues: potential role in capillary branching. Circ Res. 2004;94:262–8. http://dx.doi.org/10.1161/01.RES.0000111527.42357.62
- 92. Viney RL, Morrison AA, van den Heuvel LP, Ni L, Mathieson PW, Saleem MA, et al. A proteomic investigation of glomerular podocytes from a Denys-Drash syndrome patient with a mutation in the Wilms tumour suppressor gene WT1. Proteomics. 2007;7:804–15.

http://dx.doi.org/10.1002/pmic.200600666

93. Wagner N, Wagner KD, Scholz H, Kirschner KM, Schedl A. Intermediate filament protein nestin is expressed in developing kidney and heart and might be regulated

Duim et al.

by the Wilms' tumor suppressor Wt1. Am J Physiol Regul Integr Comp Physiol. 2006;291:R779-87.

http://dx.doi.org/10.1152/ajpregu.00219.2006

 Ishiwata T, Kudo M, Onda M, Fujii T, Teduka K, Suzuki T, et al. Defined localization of nestin-expressing cells in L-arginine-induced acute pancreatitis. Pancreas. 2006;32:360–8.

http://dx.doi.org/10.1097/01.mpa.0000220860.01120.21

- 95. Teranishi N, Naito Z, Ishiwata T, Tanaka N, Furukawa K, Seya T, et al. Identification of neovasculature using nestin in colorectal cancer. Int J Oncol. 2007;30:593–603.
- 96. Suzuki S, Namiki J, Shibata S, Mastuzaki Y, Okano H. The neural stem/progenitor cell marker nestin is expressed in proliferative endothelial cells, but not in mature vasculature. J Histochem Cytochem. 2010;58:721–30. http://dx.doi.org/10.1369/jhc.2010.955609
- 97. Mokry J, Cizkova D, Filip S, Ehrmann J, Osterreicher J, Kolar Z, et al. Nestin expression by newly formed human blood vessels. Stem Cells Dev. 2004;13:658–64. http://dx.doi.org/10.1089/scd.2004.13.658
- 98. Mokry J, Pudil R, Ehrmann J, Cizkova D, Osterreicher J, Filip S, et al. Re-expression of nestin in the myocardium of postinfarcted patients. Virchows Arch. 2008;453:33–41. http://dx.doi.org/10.1007/s00428-008-0631-8
- 99. Wu M, Smith CL, Hall JA, Lee I, Luby-Phelps K, Tallquist MD. Epicardial spindle orientation controls cell entry into the myocardium. Dev Cell. 2010;19:114–25. http://dx.doi.org/10.1016/j.devcel.2010.06.011
- 100. Xu C, Wu C, Xia Y, Zhong Z, Liu X, Xu J, et al. WT1 promotes cell proliferation in non-small cell lung cancer cell lines through up-regulating cyclin D1 and p-pRb in vitro and in vivo. PLoS One. 2013;8:e68837.
 - http://dx.doi.org/10.1371/journal.pone.0068837
- 101. Baldin V, Lukas J, Marcote MJ, Pagano M, Draetta G. Cyclin D1 is a nuclear protein required for cell cycle progression in G1. Genes Dev. 1993;7:812–21.
- 102. Loeb DM. WT1 influences apoptosis through transcriptional regulation of Bcl-2 family members. Cell Cycle. 2006;5:1249–53.
- 103. Yi BA, Wernet O, Chien KR. Pregenerative medicine: developmental paradigms in the biology of cardiovascular regeneration. J Clin Invest. 2010;120:20–8. http://dx.doi.org/10.1172/JCI40820
- 104. Kajstura J, Urbanek K, Rota M, Bearzi C, Hosoda T, Bolli R, et al. Cardiac stem cells and myocardial disease. J Mol Cell Cardiol. 2008;45:505–13. http://dx.doi.org/10.1016/j.yjmcc.2008.05.025
- Moerkamp AT, Goumans MJ. Cardiac regeneration: stem cells and beyond. Curr Med Chem. 2012;19:5993–6002.

WT1 in the heart

 Aguiar C, Brunt KR. Wilms' tumor 1 (re)activation in evidence for both epicardial progenitor and endothelial cells for cardiovascular regeneration. J Mol Cell Cardiol. 2015;84:112–5.

http://dx.doi.org/10.1016/j.yjmcc.2015.04.001

- 107. Smits A, Riley P. Epicardium-derived heart repair. J Dev Biol. 2014;2:84.
- 108. Limana F, Zacheo A, Mocini D, Mangoni A, Borsellino G, Diamantini A, et al. Identification of myocardial and vascular precursor cells in human and mouse epicardium. Circ Res. 2007;101:1255–65.

http://dx.doi.org/10.1161/CIRCRESAHA.107.150755

 Chong JJ, Chandrakanthan V, Xaymardan M, Asli NS, Li J, Ahmed I, et al. Adult cardiac-resident MSC-like stem cells with a proepicardial origin. Cell Stem Cell. 2011;9:527–40.

http://dx.doi.org/10.1016/j.stem.2011.10.002

110. Shen MJ, Zipes DP. Role of the autonomic nervous system in modulating cardiac arrhythmias. Circ Res. 2014;114:1004–21.

http://dx.doi.org/10.1161/CIRCRESAHA.113.302549

111. Mancia G, Grassi G. The autonomic nervous system and hypertension. Circ Res. 2014;114:1804–14.

http://dx.doi.org/10.1161/CIRCRESAHA.114.302524

Florea VG, Cohn JN. The autonomic nervous system and heart failure. Circ Res. 2014;114:1815–26.
http://dx.doi.org/10.1161/CIPCPECALIA_114.202580

http://dx.doi.org/10.1161/CIRCRESAHA.114.302589

Chapter 14

Functional Role of WT1 in Prostate Cancer

Gail C. Fraizer,¹ Kurtis Eisermann,^{1,2} Sony Pandey,¹ Adina Brett-Morris,³ Anton Bazarov,¹ Sarah Nock,¹ Nirmala Ghimirey,¹ Steven J. Kuerbitz⁴

¹Kent State University, Kent, OH, USA; ²University of Pittsburgh, Pittsburgh, PA, USA; ³Case Western Reserve University, Cleveland, OH, USA; ⁴Akron Children's Hospital, Akron, OH

Author for correspondence: Gail C. Fraizer, MPH, PhD, Kent State University, Kent, OH, USA. Email: gfraizer@kent.edu

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch14

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

Although initial discoveries of Wilms tumor 1 (WT1) expression in extrarenal disease generated controversy, we and others have examined WT1 expression in non-Wilms cancers and have demonstrated that the WT1(A) isoform, lacking the lysine-threonine-serine tripeptide (KTS) insertion, transcriptionally regulates the expression of growth control genes in other cancer types. Here, we review our evidence that WT1 is expressed in prostate cancer (PC) epithelial cells and regulates PC critical genes. That WT1 may promote metastatic disease is consistent with previous findings that WT1 suppressed E-cadherin and enhanced motility of PC cells with low migratory and metastatic potential. Recent findings led us to ask

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor)

ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

whether WT1 acts as an angiogenic switch in PC. Although vascular endothelial growth factor (VEGF) is regulated at several levels and by a number of different factors, a mechanistic understanding of WT1-mediated transcriptional regulation in PC cells was previously lacking. Here, we discuss the evidence of WT1- and androgen receptor (AR)-binding sites in the VEGF promoter and show the potential for cooperation between hormone and WT1. These findings revealed that in AR-intact PC cells, WT1 was sufficient to upregulate VEGF transcription, and WT1 expression enhanced the hormone activation of VEGF expression. This notion that WT1 can activate an angiogenic switch in PC cells, to enhance tumor growth and progression to metastatic disease, is consistent with our understanding of the oncogenic nature of WT1 overexpression in inappropriate tissues or at inappropriate times. The potential for WT1 to promote both tumor angiogenesis and PC cell migration suggests that WT1 regulates genes that promote PC progression to lethal metastatic disease. Therapies targeting WT1 in PC may reduce metastatic spread and increase overall survival.

Key words: AR; E-cadherin; Prostate cancer; Transcription; VEGF; WT1

Introduction

The WT1 gene is a member of the early growth response gene I (EGR-1) family of transcription factors containing four Kruppel-like zinc fingers in the carboxyl terminus that bind nucleic acids (both DNA and RNA). The functions of the Wilms tumor 1 (WT1) protein are isoformspecific and reflect its structural domains (1). The four major isoforms of WT1 are formed by alternative splicing at two sites resulting in the inclusion or exclusion of (1) exon V and/ or (2) a lysine-threonine-serine tripeptide (KTS) in exon 9 that alters the relative orientation of the 3rd and 4th zinc fingers and affects the DNA-binding structure. The isoform WT1(A), which lack both exon V and the KTS tripeptide, binds DNA and functions as a transcription factor, while isoform WT1(D) contains both elements and can function as a post-transcriptional regulator in certain contexts. Additional, less common isoforms initiate from internal or upstream start sites. Three of the four Cys₂-His₂ zinc fingers (2, 3) of the (-)KTS isoforms are involved in binding a common G-rich DNA consensus sequence, GNGNGGGNG, as well as the related Egr-1 recognition elements (4). The importance of the zinc finger domain for DNA binding is underscored by congenital syndromes associated with naturally occurring WT1 mutations, such as the Denys-Drash syndrome and Frasier syndrome, characterized by urogenital anomalies and elevated risk of Wilms tumor or gonadoblastoma, respectively (5, 6). While controversy exists over the ability of mutant forms of WT1 to bind DNA, it is possible that protein interaction sites remaining within the mutant WT1 protein could play a direct role in these anomalies (7). Indeed, controversy exists over the role of normal cytoplasmic WT1 protein, with some evidence supporting a shuttle function, as WT1 contains cytoplasmic and nuclear localization signals, as well as a nuclear export signal (8). The activity of

the cytoplasmic form may be related to phosphorylation status, as phospho-WT1 is thought to be retained in the cytoplasm (9, 10). Alternatively, as both +/- KTS isoforms have been identified in polysomes and bound to polyA RNPs (11), a post-transcriptional function has been suggested. Interestingly, one example of post-transcriptional regulation of vascular endothelial growth factor (VEGF) by WT1 involved transcriptional regulation of a splicing factor kinase that, in turn, altered VEGF splicing in podocytes (12). Recent evidence indicating the association of WT1 protein with histone and chromatin modifying enzymes also suggests an epigenetic function for WT1 [reviewed in reference (1)], mediated, in part, by WT1 recruitment of DNA methyltransferase DNMT1 and polycomb group protein enhancer of zeste homolog 2 (EZH2) (13) and CREB-binding protein (CBP), a histone acetyltransferase (14). Additionally, the evidence of epigenetic regulation of WT1 expression by lncRNA in acute leukemia (15) suggests that WT1 is intimately involved in both direct transcriptional and indirect epigenetic regulation. Thus, study of WT1 as a regulator of gene expression in key developmental processes, such as hematopoiesis, continues to be relevant.

Developmental expression of WT1

WT1 expression in the developmental processes was initially viewed as growth suppressive and necessary for cell differentiation, consistent with its earliest descriptions as a tumor suppressor gene (TSG). Within the developing kidney and genitourinary system, the timing of WT1 expression is exquisitely controlled, and once kidney development occurs, WT1 expression is tightly restricted to podocytes (16). WT1 is an essential regulator of nephrogenesis (17-19) and is expressed in both normal podocytes and in some Wilms tumors (18, 20). In addition to the kidney, WT1 is normally expressed in many other organs (6), including hematopoietic tissues such as the spleen, fetal liver, bone marrow, and lymph nodes, gonads, and peripheral nervous system (3, 21–25). However, its role is ambiguous depending on the organ involved and whether epithelial or mesenchymal differentiation occurs. For instance, in the normal development of the kidneys and the urogenital system, WT1 is needed to induce mesenchymal-epithelial transition (MET) leading to the formation of nephrons (26)and kidneys (16). In MET, the mesenchymal cells undergo multiple morphological changes associated with differentiation into epithelial cells and condensation into structures forming the organ. WT1 expression accompanies the opposite developmental role, epithelial to mesenchymal transition (EMT), in the developing heart where epithelial cells transform into motile mesenchymal cells that contribute to the organ's cellular structure and generate important signals (27). Furthermore, it has been demonstrated that WT1 is required for cardiovascular progenitor cell formation through the upregulation of Snail and downregulation of E-cadherin, two of the major molecules involved in EMT (28). Although WT1 has been proposed to regulate EMT by repressing E-cadherin; more recently, WT1 has been linked to the regulation of epicardial EMT through the β -catenin and retinoic acid signaling pathways (29). Interestingly, it has

been found that WT1 transcriptionally activates Snail with partial maintenance of E-cadherin, and WT1 is associated with epithelial characteristics in kidney cells and in clear cell renal cell carcinoma (31). Thus, in these examples, WT1 induces an epithelial-mesenchymal hybrid transition defined by Snail upregulation with E-cadherin maintenance, a tumor cell differentiation state in which cancer cells retain both mesenchymal and epithelial features that may contribute to tumor cell plasticity and tumor progression (30). Similarly in prostate cancer (PC), a partial EMT with features of both epithelial and mesenchymal cells has been observed (31). The transformation of metanephric mesenchyme to epithelial cells within the condensing glomeruli also is similar to the metastatic process of cancer cells, whereby motile cancer cells, after extravasation, must revert back to their epithelial state to survive at the metastatic site (32). Because WT1 is required for normal MET within the developing kidney, it seems plausible that it may also play a role in the metastatic MET process. Yet, little is known about the requirement for WT1 expression during the metastatic process.

WT1 expression in non-Wilms cancer

CD34+ hematopoietic progenitor cells express WT1, but like metanephric mesenchyme, once hematopoietic progenitors become lineage-committed then expression of WT1 is highly restricted within a small subpopulation of cells [reviewed in reference (33)]. Increased expression appears to persist in cancer cells, and WT1 expression in tumor tissue exceeds that of the normal cell counterparts. This dysregulated expression was regarded as an indicator of a potential growth-promoting effect of WT1 and led to the controversy over whether WT1 was truly a TSG as originally identified in Wilms tumor or whether WT1 was actually an oncogenedriving cancer cell proliferation and blocking differentiation, as observed in leukemia cell lines (25). As evidence accumulates on different tumor types that overexpress WT1 relative to their normal counterparts, it is clear that WT1 has a dichotomous role in cancer, and indeed, WT1 has been referred to as a chameleon [reviewed in references (33) and (25)]. Within the hematopoietic system, it is clear that WT1 can behave as a survival gene, enhancing cell viability, but also can induce quiescence, depending on the differentiation state of the leukemia cells involved [reviewed in reference (33)]. Many studies have shown elevated WT1 expression in diverse cancer types, including leukemia (34-37), breast (38-40), Ewing sarcoma (41), ovarian (42), mesothelioma, and pulmonary adenocarcinomas (43). Additionally, WT1 is being investigated as a potential prognostic marker for both leukemia and breast cancer (39, 44).

Expression and potential role of WT1 in PC

WT1 is expressed mainly during development, and it plays an important role in adrenogonadal development and sex determination [reviewed in reference (45)] via its regulation of SRY (46), so its expression in hormone-responsive tumors such as breast, ovarian, and

prostate was not unexpected. We and others initially identified WT1 mRNA in cultured PC cells (47–51)and then WT1 mRNA and protein in PC tissues (49). Because the prostate is a complex tissue and PC is a heterogeneous disease, we used laser capture microdissection (LCM) to isolate distinct cell-type populations from epithelial and stromal tissues in PC and identified *WT1* among the nearly 500 genes whose expression was significantly different between epithelial and stromal PC cells (49). Results of microarray analysis are posted at NCBI (Geo #GSE 20758). This differential expression of *WT1* in PC epithelial cells was validated by quantitative real-time polymerase chain reaction(PCR) and relevance confirmed by analysis of additional frozen tumor tissue biopsies and tissue microarray (TMA) sections (49). This cell-specific expression suggests a potential role for *WT1* in PC, likely involving the acquisition of characteristics necessary for metastatic growth of PC.

Metastatic disease is associated with a marked increase in the risk of mortality among PC patients. Ninety-nine percent of patients who develop primary PC are expected to live at least 5 years after diagnosis (52). Ninety-eight percent are alive after 10 years, and 94% live for at least 15 years if the disease remains localized. By contrast, patients with metastatic disease at diagnosis have a 5-year survival rate of only 28% (52). The process of metastasis requires that cancer cells acquire characteristics of enhanced motility and invasiveness. That WT1 may be involved in PC metastasis was suggested by immunohistochemical analysis of PC TMAs, demonstrating that WT1 protein was more often expressed in high Gleason grade PC epithelial cells than that in low grade, and it was not observed in nonneoplastic prostate tissue (49). Others have also suggested that WT1 could serve as a marker for PC progression (53). While Devilard et al. (53) demonstrated the expression of WT1 by microarray analysis in a hormone-refractory LuCaP xenograft PC progression model, our results provide the most complete evidence of elevated WT1 mRNA and protein in prostate tumors, and our study was the first to identify WT1 expression in LCM human prostate epithelial tissue (49). We confirmed the relevance of the microarray analysis of LCM-captured tissue RNA by real-time PCR quantifying WT1 expression in 20 additional sets of paired tumor and non-neoplastic tissues. WT1 mRNA levels were elevated in 70% of invasivestage T3 tumors examined when compared to the adjacent non-neoplastic tissue. Similarly, in three of four established PC cell lines, WT1 expression was also significantly higher than the nontumorigenic, immortalized prostate epithelial cell line RWPE-1 (49). Further analysis of WT1 protein in formalin-fixed, paraffin-embedded TMAs identified WT1 expression in 65% of tumor samples (of Gleason grade 6–10) and, importantly, the absence of expression in non-neoplastic and benign prostatic hyperplasia (BPH) samples. WT1 expression in high-grade PC may indicate that WT1-responsive pathways promote the slow progression of latent PC to aggressive, hormone-refractory PC. Two possible mechanisms whereby WT1 expression in prostate could enhance metastatic tumor growth warrant discussion.

WT1 target genes relevant in PC

The transcriptionally active isoform of the Wilms tumor gene, WT1(A), regulates a large family of genes involved in growth control, sex determination, and genitourinary development [for reviews see references (6, 16, 54)]. We and others have demonstrated that WT1 regulates important PC pathways – both growth-promoting pathways, e.g., insulin-like growth factor axis (55, 56)and androgen signaling via androgen receptor (AR) (46, 50), and growth suppressing/apoptotic pathways via Bcl-2 (57–61). Recently, WT1 has been shown to control differentiation of epicardial cells by repressing E-cadherin expression, thereby inducing mesenchymal transformation (EMT) resulting in vascular endothelial cells, smooth muscle cells, and cardiomyocytes in the heart (28). WT1 could similarly facilitate the metastatic progression of PC cells by inducing EMT, which is marked by loss of epithelial markers such as E-cadherin and gain of mesenchymal markers such as N-cadherin. WT1 could also enhance metastatic tumor growth by inducing expression of the angiogenic regulatory gene, VEGF. Together, these gene regulatory functions could promote acquisition of the lethal metastatic phenotype of PC.

WT1 suppression of E-cadherin promotes cell motility

Initial studies in NIH-3T3 cells, in which it was demonstrated that E-cadherin is a WT1 target gene (62), and studies in cardiac epithelial cells have established the role of WT1 in E-cadherin regulation (28). E-cadherin is a transmembrane protein that mediates epithelial cell-cell interactions in the adherent junctions of the plasma membrane (63) through homophilic proteinprotein interactions (64). Downregulation of E-cadherin results in increased invasiveness of distinct types of cancer, such as gastric (65, 66), breast (67), ovary (68, 69), endometrial (70), thyroid (71), hepatocellular carcinoma (72), oral (73), and pancreatic (74), and has been well documented in prostate adenocarcinoma (75-77). In PC, E-cadherin expression has been shown to be reduced by activation of AKT signaling (78), by high expression of transcription factors such as Snail (79, 80), Slug (81), Twist (82) and WT1 (48), and by hypermethylation of the E-cadherin promoter (83). The loss of this important cell adhesion molecule is a critical early event in invasion and metastasis that leads to the conversion from a stationary to a migratory cell phenotype (84). When cancer cells acquire motility and invasiveness, they exhibit marked morphological changes, lose epithelial features, and acquire a more mesenchymal phenotype (EMT) (85, 86). Interestingly, androgen exposure has been reported to increase levels of Snail, decrease levels of E-cadherin and β -catenin, and induce expression of the mesenchymal marker N-cadherin in PC cells (87). TGF- β also has been implicated in induction of EMT in PC through activation of SMAD3 (88) and promotion of PARP4 nuclear localization with the subsequent increase of Snail, vimentin and N-cadherin, and decrease of E-cadherin (89).

While initial experiments associated growth suppression and characteristics of epithelial differentiation, including upregulation of E-cadherin, with stable expression of WT1 in NIH 3T3 cells

(62), more recent studies in cardiac epithelial cells showed that WT1 transcriptionally repressed E-cadherin expression both directly and indirectly by the upregulation of Snail (28). Furthermore, it has been demonstrated that WT1 expression promotes metastasis and invasion in nonsmall-cell lung carcinoma patients through the suppression of E-cadherin (90). In the context of PC, we observed that WT1 expression was inversely related to E-cadherin expression in several PC cell lines, and, importantly, WT1 expression correlated with migratory potential (48). Mechanistic studies showed that WT1 could bind to the E-cadherin promoter *in vivo* and decrease the E-cadherin promoter activity through a novel-binding site located at -146 bp upstream from the transcription start site. Additionally, overexpression of WT1 in LNCaP cells decreased E-cadherin mRNA expression (2-fold, $p \le 0.05$). Although LNCaP cells have low migratory potential as measured in migration chamber assays, forced expression of WT1 not only suppressed E-cadherin but also enhanced LNCaP cell migration 3-fold compared to control vector-transfected cells ($p \le 0.001$). Moreover, silencing WT1 in PC3 cells, which exhibit higher WT1 expression and greater migratory potential, reduced their motility in migration chambers by 50% compared to scrambled control-transfected cells ($p \le 0.01$). This strong inhibition of motility was confirmed in wound-healing assays showing a 4.4× reduction in the motility of siWT1 RNA-transfected PC3 cells compared to controls ($p \le 0.001$) (48). Our study, the first to undertake a complete analysis of the effect of WT1 on E-cadherin expression and motility in PC cells, thus demonstrated that WT1 binding decreased activity of the E-cadherin promoter in the presence of WT1 and that repression of E-cadherin expression led to an increase in cell migration (Figure 1). Suppression of E-cadherin expression and enhancement of motility are both associated with EMT.

WT1 may contribute to tumor angiogenesis via regulation of VEGF

We have demonstrated that, in addition to enhancing PC migration by suppressing E-cadherin expression, WT1 also upregulates VEGF, thereby potentially promoting tumor angiogenesis and metastasis. VEGF is a mitogen secreted by tumor cells that is essential for tumor angiogenesis and is necessary for tumor growth beyond 1–3 mm³ in volume (91). VEGF regulation is complex and occurs at both transcriptional and post-transcriptional levels (92, 93). While the VEGF promoter lacks a TATA-binding site, it contains a GC-rich core promoter region and additional distal enhancer sites including hypoxia response elements that bind hypoxia-inducible factor (HIF1)-alpha.

Coexpression of WT1 and VEGF

WT1 was previously shown to play a role in neovascularization in the proliferative response of coronary vasculature to regional ischemia (94). In vascular cells, WT1 expression was associated with an increase in proangiogenic molecules such as VEGF (95). Similarly, both VEGF and WT1 are elevated in some PC cells (96), consistent with its ability to regulate growth control pathways important in PC (46, 50, 55–61). Additionally, WT1



Figure 1. Proposed mechanism for WT1 regulation of motility of prostate cancer cells. WT1 transcriptionally represses E-cadherin, which would lead to loss of cell adhesion and promote prostate cancer cell migration, potentially enhancing the epithelial to mesenchymal transition (EMT) of PC cells expressing WT1.

and VEGF are coexpressed in both normal podocytes and some Wilms tumors (18, 20, 97). These findings of coordinate expression led to suggestions that WT1 plays an important regulatory role in developmental and tumor angiogenesis (20, 51, 98). For all these reasons, it seemed likely that VEGF was a physiologically relevant target of WT1 regulation in the prostate. In Ewing sarcoma cell lines, knockdown of WT1 expression using WT1specific shRNA downregulated VEGF mRNA expression and decreased angiogenic activity (99). Conversely, overexpression of WT1 upregulated VEGF mRNA and increased angiogenic activity (99). Additionally, WT1 bound to the promoter of VEGF and increased promoter activity in response to hypoxia in Ewing sarcoma cells (100). Together, these results demonstrated that WT1 could directly regulate VEGF expression in Ewing sarcoma cells.

Regulation of VEGF by WT1 in prostate cancer

We assessed the WT1-mediated regulation of VEGF in PC cells. WT1-binding sites predicted by *in silico* analysis of the VEGF proximal promoter (101, 102) were demonstrated functional by reporter assays and protein binding *in vitro* and *in vivo* using electrophoretic mobility shift assay (EMSA) and chromatin immunoprecipitation (ChIP) assays, respectively. The latter result indicated the ability of WT1 protein to bind to native chromatin in LNCaP PC cells (101) and is consistent with results of luciferase reporter assays, showing that WT1 upregulates the VEGF promoter (102) in LNCaP cells. One of the functional binding sites identified initially as an Egr1-binding site was verified to bind WT1 by ChIP analysis in LNCaP cells (101). Site-directed mutagenesis of the proximal VEGF promoter construct V411 (Figure 2A)

was used to determine if this site was necessary for WT1-mediated transcriptional activation of the VEGF promoter. Cotransfection of a green fluorescent protein (GFP)-tagged WT1 expression construct (103) and mutant reporter into LNCaP cells revealed that disruption of this site significantly decreased the ability of WT1 to upregulate the proximal VEGF promoter (Figure 2B). We then asked whether this same pattern of regulation was occurring in other hormone-responsive PC cell lines. Testing two other hormone-responsive PC cell lines, CWR22Rv1 and C4-2, we found that WT1 regulated the VEGF proximal promoter (Figure 2C and 2D) similarly in all three cell lines. Thus, the data showed that WT1 bound and activated the VEGF proximal promoter in several PC cell lines.

The enhanced expression of VEGF mRNA in WT1-transfected LNCaP cells confirmed the *in vitro* promoter activation studies. Although overexpression of WT1 increased VEGF mRNA levels, the converse was not true (data not shown). Knockdown of WT1 expression in LNCaP cells using siRNA did not significantly affect VEGF mRNA levels. Together, these results indicate that WT1 is sufficient to upregulate VEGF expression, but not necessary, suggesting that other transcription factors (possibly SP1) play a role in the androgen activation of VEGF (104). Additionally, WT1-mediated regulation of VEGF appears to be cell specific as transfection of hormone-insensitive PC3 cells did not enhance VEGF promoter activity, and WT1 appears to repress the VEGF promoter in embryonic kidney HEK293 cells (102).

Combined and rogen and WT1 activation of VEGF expression in hormone-responsive PC

Although hormone responsive (105–108), the VEGF promoter lacks canonical AR or estrogen receptor (ER)-binding sites. VEGF regulation by estrogen in endometrial and breast cancer cells involves interactions of ER- α and Sp1 (or Sp3) with GC boxes in the core promoter region of VEGF (-66 to -47 bases from start site) (108, 109). VEGF mRNA levels were significantly induced in ZR-75 breast cancer cells treated with estradiol, and the intact GC-rich core VEGF promoter region (-66 to -47) was required for such activation. The relevance of Sp1 and Sp3 in estradiol regulation of VEGF in breast cancer was suggested by binding assays *in vitro* (by EMSA) and *in vivo* (by ChIP). Similarly, multiple groups have shown that androgen treatment of human fetal fibroblasts and LNCaP cells significantly increases VEGF mRNA expression levels (110–112, 102). Additionally, VEGF protein levels have been demonstrated to be upregulated after the treatment of LNCaP cells with hormone (106), and the androgen antagonist flutamide blocked this upregulation (113). The mechanism of androgen-mediated regulation of VEGF expression, however, is less well understood.

In examining the mechanism of androgen-mediated regulation of VEGF expression, we identified AR/GC sites within the VEGF GC-rich core. Based on our earlier *in silico* analyses of the VEGF promoter (101) and the discovery that site-directed mutation of three AR half-sites did not eliminate hormone activation of the VEGF promoter (104), we hypothesized that



Figure 2. WT1 activates the VEGF promoter in prostate cancer cells via a WT1-binding site within the proximal VEGF promoter. (A) Site-directed mutagenesis of a predicted WT1-binding site (red box) was performed on the VEGF proximal promoter construct V411. (B) LNCaP cells were cotransfected with GFP-WT1 or the empty CMV expression construct, pcDNA3.1, along with 250 ng of either the wild-type V411(left) or the mutant (right) reporter constructs. (C) Cotransfection of CWR22Rv1 and (D) C42 prostate cancer cells with V411 and CMV or WT1 expression construct as described above. Luciferase activity was normalized to protein concentration, and values represent average normalized luciferase activity (+SEM) relative to empty vector control. Experiments were repeated three times in triplicate, and statistical significance was determined by Student's t-test (**p< 0.01, ***p< 0.001).

WT1 might regulate the hormone-responsive VEGF promoter. Thus, we asked whether AR might bind at other sites via interaction with other zinc finger transcription factors (ZFTFs), such as SP1, EGR-1, or WT1. We hypothesized that if AR–ZFTF interactions were important mediators of androgen response, then cognate-binding sites should be located within the

promoter regions of hormone-responsive genes expressed in PC (101, 114). As expected, nonclassical AR half-sites were identified adjacent to WT1/EGR1/Sp1 sites in 8 of 11 promoters analyzed including VEGF (114). Binding at one of the three predicted nonclassical androgen receptor element half-sites (ARE-I) in the VEGF promoter region was tested by ChIP analysis of hormone-treated, WT1-transfected LNCaP cells (114). Endogenous AR and Sp1 proteins, along with exogenous WT1, were immunoprecipitated from native chromatin of these hormone-treated cells, indicating that the predicted WT1, Sp1, and AR sites in the VEGF proximal promoter region were functional and suggesting that the three factors may bind individually or as a complex. Based on these in silico predictions, we proposed three alternative models for AR-mediated regulation of VEGF promoter activity. The models differ primarily in the manner that AR binds the VEGF promoter (Figure 3). The first model proposes that AR binds to AREs as a dimer (Figure 3, model i), in the classical way that AR binds to many androgen-responsive genes, such as prostate-specific antigen (PSA). However, there have also been reports that noncanonical monomeric ARE half-sites are important (115-117). Thus, the second proposed model (Figure 3, model ii) shows AR monomers binding to an ARE half-site and bridged to WT1 (or other ZFTF, such as Sp1 or Egr1) binding sites by cofactors (marked as?), such as CBP or SRC-1; alternatively, AR dimers may bind to half-site ARE and bridge to WT1-binding site. Because AR is known to interact with Sp1, Egr1, and potentially WT1, the third and final model (Figure 3, model iii) proposes that AR is not bound to an ARE-binding site but is tethered via a ZFTF, which is bound to the G-rich VEGF promoter at either Egr1-/WT1-binding sites or GC boxes (SP1-/Sp3-binding sites).

To test the model for WT1 AR interaction, we examined the WT1 site within 200 bp of the ARE site to determine whether WT1 would modulate the hormone response of the proximal VEGF promoter. Cells were serum-starved to deplete androgens, cotransfected with the VEGF proximal promoter and either WT1 or empty vector control, then treated with 5 nM R1881, an androgen analog, or vehicle control dimethyl sulfoxide (DMSO) (Figure 4A). Luciferase assays confirmed that either hormone or WT1 alone increased VEGF transcription 3- to 4-fold compared to cytomegalovirus (CMV) empty vector, vehicle control. However, the combination of WT1 and 5nM R1881 activated this reporter construct more than 12-fold (Figure 4A), suggesting that their interaction strongly enhanced hormone response. This strong upregulation suggested that WT1 and AR may form a complex in the nucleus and bind the G-rich and the AR half-site (similar to Figure 3, model ii). Nuclear lysates from WT1-transfected LNCaP cells grown in full serum (containing endogenous hormone) were co-immunoprecipitated with WT1 and AR antibodies. Immunoblot analysis revealed that complexes precipitated by antibodies specific for WT1 also contained AR protein (Figure 4B). Conversely, AR-immunoprecipitated complexes contained low levels of WT1 protein (data not shown). Together, these results indicate that WT1 may interact with AR to enhance androgen induction VEGF expression in PC cells.

Surprisingly, the GC-rich VEGF core promoter (-88 to +51), which lacks AR half-sites, but contains multiple EGR-1/WT1/Sp1 overlapping sites, also demonstrated a hormone response. Consequently, the third model we tested (Figure 3, model iii) proposes that AR is not bound to an ARE but is tethered via SP1, which is bound to GC boxes in the VEGF core promoter. Because estrogen regulation of the VEGF core promoter has been shown to require Sp1 sites in breast cancer cells (109), we asked whether androgen might regulate VEGF in a similar fashion in PC cells. Sp1-associated binding of AR to novel-binding sites in the VEGF promoter was demonstrated in vivo by ChIP analysis in LNCaP cells (101, 104, 114). AR and Sp1 formed a nuclear complex and were shown to bind to the VEGF core promoter in hormone-treated CWR22Rv1 PC cells (104). Suppression of Sp1 binding in the VEGF core promoter by mutation of a specific Sp1-binding site abrogated VEGF promoter activation by androgen. Additionally, treatment with mithramycin A, which blocks access of proteins to GC-rich DNA, significantly reduced Sp1 binding and VEGF expression. Together, these results indicated that another mechanism of androgen-mediated induction of VEGF expression in PC cells involved interaction of AR with a specific, critical Sp1-binding site in the VEGF core promoter region (104) similar to that described here for WT1 interaction at the proximal promoter region. Overall hormone activation of the VEGF promoter region is enhanced by interaction of AR with transcription factor-binding partners in PC cells.



Figure 3. Proposed models of androgen regulation of VEGF in prostate cancer. Three potential ways that androgen is proposed to bind AR and regulate VEGF in prostate cancer: (i) AR binding to androgen response elements (AREs) as a dimer, (ii) monomeric AR binding to half-site ARE and bridged by unknown factor (?) to WT1 at its binding site, or (iii) AR tethering to WT1 at WT1-binding site, but not bound to ARE.



Figure 4. (A) WT1 and AR interact to activate the VEGF proximal promoter. (A) Serum-starved LNCaP cells were cotransfected as described above with V411 reporter and either GFP-WT1 or pcDNA3.1 vector control DNA. Transfected cells were treated with 0 or 5 nM R1881 in media containing 10% charcoal-stripped FCS. Values shown represent mean normalized luciferase activity (and SEM) relative to pCDNA3.1 empty vector control in the absence of hormone treatment (white). Each experiment was performed in triplicate and repeated twice. (B) The interaction of AR and WT1 in transfected LNCaP cells was demonstrated by coimmunoprecipitation of nuclear protein lysates by either WT1 Ab (Epitomics) or serum IgG. Immunoprecipitated (IP) proteins were electrophoresed and immunoblotted (IB) with either AR (top) or WT1 (bottom) antibodies. Left lanes show nuclear lysates, middle and right lanes show proteins immunoprecipitated by IgG or WT1 Ab, respectively.

Conclusion

Here, we review evidence that WT1 is expressed in PC epithelial cells and transcriptionally regulates PC critical genes. The relevance of WT1 to PC has been shown by finding that WT1 mRNA and protein are more often expressed in high-grade, invasive PC than low-grade localized tumors and that WT1 is not expressed in BPH or non-neoplastic prostate tissue (49). The identification of potential WT1-binding sites in the regulatory sequences of cancer-critical genes expressed in PC epithelial cells, together with the demonstration of WT1 protein bound to these gene promoters in native chromatin of transfected LNCaP cells, supported the notion that elevated WT1 expression in prostate epithelial cells affects transcriptional modulation of homeostatic genes important for PC (101). That WT1 may

promote metastatic disease is consistent with previous findings that WT1 suppressed E-cadherin, thereby increasing motility and metastatic potential of PC cells (48). The fact that WT1 transcriptionally upregulated VEGF expression and enhanced hormone induction of VEGF (102) suggested that WT1 could activate an angiogenic switch in PC cells. Taken together, the potential for WT1 to promote tumor angiogenesis and PC cell migration would suggest that WT1 regulates genes that enhance tumor growth and promote progression to lethal metastatic disease. These functions of WT1 are consistent with an oncogenic, not a tumor-suppressive, role and suggest that WT1 expression might serve as a marker for PC progression (53). Furthermore, therapies targeting WT1 in PC may block metastatic spread and increase the overall survival.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgments

We gratefully acknowledge the receipt of the VEGF luciferase constructs from Dr. K. Xie and the GFP-WT1 construct from Dr. A Ward. Funding was provided by NIH-1CA33160 (GF), the Kent State University Research Council (GF), Sigma Xi G2009101485 (KE), and the Akron Children's Hospital Foundation. This review is dedicated to the memory of Dr. Grady F. Saunders.

References

- Toska E, Roberts SG. Mechanisms of transcriptional regulation by WT1 (Wilms' tumour 1). Biochem J. 2014 Jul 1;461(1):15–32. http://dx.doi.org/10.1042/BJ20131587
- Madden SL, Cook DM, Morris JF, Gashler A, Sukhatme VP, Rauscher FJ. Transcriptional repression mediated by the WT1 Wilms tumor gene product. Science. 1991;253(5027):1550–3.
- Renshaw J, King-Underwood L, Pritchard-Jones K. Differential splicing of exon 5 of the Wilms tumour (WT1) gene. Genes Chromosomes Cancer. 1997;19(4):256–66.
- Rauscher FJ, Morris JF, Tournay OE, Cook DM, Curran T. Binding of the Wilms' tumor locus zinc finger protein to the EGR-1 consensus sequence. Science. 1990;250(4985):1259–62.
- Pelletier J. Germline mutations in the Wilms' tumor suppressor gene are associated with abnormal urogenital development in Denys-Drash syndrome. Cell. 1991;67(2):437-47.

- 6. Mrowka C. Wilms' tumor suppressor gene WT1: from structure to renal pathophysiologic features. J Am Soc Nephrol. 2000;11:S106–15.
- Tajinda K, Carroll J, Roberts CT Jr. Regulation of insulin-like growth factor I receptor promoter activity by wild-type and mutant versions of the WT1 tumor suppressor 1. Endocrinology. 1999;140(10):4713–24.

http://dx.doi.org/10.1210/endo.140.10.7065

- Vajjhala P, Macmillan E, Gonda T, Little M. The Wilms' tumour suppressor protein, WT1, undergoes CRM1-independent nucleocytoplasmic shuttling. FEBS Lett. 2003;554(1-2):143–8.
- 9. Ye Y, Raychaudhuri B, Gurney A, Campbell C, Williams B. Regulation of WT1 by phosphorylation: inhibition of DNA binding, alteration of transcriptional activity and cellular translocation. EMBO J. 1996;15(20):5606.
- Sakamoto Y, Yoshida M, Semba K, Hunter T. Inhibition of the DNA-binding and transcriptional repression activity of the Wilms' tumor gene product, WT1, by cAMP-dependent protein kinase-mediated phosphorylation of ser-365 and ser-393 in the zinc finger domain. Oncogene. 1997;15(17):2001–12. http://dx.doi.org/10.1038/sj.onc.1201391
- Niksic M, Slight J, Sanford JR, Caceres JF, Hastie ND. The Wilms' tumour protein (WT1) shuttles between nucleus and cytoplasm and is present in functional polysomes. Hum Mol Genet. 2004;13(4):463–71.

http://dx.doi.org/10.1093/hmg/ddh040

- Amin EM, Oltean S, Hua J, Gammons MV, Hamdollah-Zadeh M, Welsh GI, et al. WT1 mutants reveal SRPK1 to be a downstream angiogenesis target by altering VEGF splicing. Cancer cell. 2011;20(6):768–80. http://dx.doi.org/10.1016/j.ccr.2011.10.016
- Xu B, Zeng DQ, Wu Y, Zheng R, Gu L, Lin X, et al. Tumor suppressor menin represses paired box gene 2 expression via Wilms tumor suppressor protein-polycomb group complex. J Biol Chem. 2011 Apr 22;286(16):13937–44. http://dx.doi.org/10.1074/jbc.M110.197830
- Wang W, Lee SB, Palmer R, Ellisen LW, Haber DA. A functional interaction with CBP contributes to transcriptional activation by the Wilms tumor suppressor WT1. J Biol Chem. 2001 May 18;276(20):16810–6. http://dx.doi.org/10.1074/jbc.M009687200
- McCarty G, Loeb DM. Hypoxia-sensitive epigenetic regulation of an antisense-oriented lncRNA controls WT1 expression in myeloid leukemia cells. PLoS One. 2015 Mar 20;10(3):e0119837.
 http://dx.doi.org/10.1271/journal.pope.0110827

http://dx.doi.org/10.1371/journal.pone.0119837

- Rivera MN, Haber DA. Wilms' tumour: connecting tumorigenesis and organ development in the kidney. Nat Rev Cancer. 2005 Sep;5(9):699–712. http://dx.doi.org/10.1038/nrc1696
- 17. Kreidberg JA, Sariola H, Loring JM, Maeda M, Pelletier J, Housman D, et al. WT-1 is required for early kidney development. Cell. 1993;74(4):679–691.
- Armstrong JF, Pritchard-Jones K, Bickmore WA, Hastie ND, Bard JB. The expression of the Wilms' tumour gene, WT1, in the developing mammalian embryo. Mech Dev. 1993;40(1):85–97.
- 19. Hammes A, Guo J, Lutsch G, Leheste J, Landrock D, Ziegler U, et al. Two splice variants of the Wilms' tumor 1 gene have distinct functions during sex determination and nephron formation. Cell. 2001;106(3):319–29.
- Baudry D, Faussillon M, Cabanis MO, Rigolet M, Zucker JM, Patte C, et al. Changes in WT1 splicing are associated with a specific gene expression profile in Wilms' tumour. Oncogene. 2002 Aug 15;21(36):5566–73. http://dx.doi.org/10.1038/sj.onc.1205752
- 21. Fraizer GC, Patmasiriwat P, Zhang X, Saunders GF. Expression of the tumor suppressor gene WT1 in both human and mouse bone marrow. Blood. 1995 Dec 15;86(12):4704–6.
- 22. Algar E. A review of the Wilms' tumor 1 gene (WT1) and its role in hematopoiesis and leukemia. J Hematother Stem Cell Res. 2002;11(4):589–99. http://dx.doi.org/10.1089/15258160260194749
- 23. Parenti R, Puzzo L, Vecchio GM, Gravina L, Salvatorelli L, Musumeci G, et al. Immunolocalization of Wilms' tumor protein (WT1) in developing human peripheral sympathetic and gastroenteric nervous system. Acta Histochem. 2014;116(1):48–54. http://dx.doi.org/10.1016/j.acthis.2013.05.003
- 24. Mundlos S, Pelletier J, Darveau A, Bachmann M, Winterpacht A, Zabel B. Nuclear localization of the protein encoded by the Wilms' tumor gene WT1 in embryonic and adult tissues. Development. 1993;119(4):1329–41.
- Yang L, Han Y, Suarez Saiz F, Minden MD. A tumor suppressor and oncogene: the WT1 story. Leukemia. 2007 May;21(5):868–76.
- 26. Davies JA, Ladomery M, Hohenstein P, Michael L, Shafe A, Spraggon L, et al. Development of an siRNA-based method for repressing specific genes in renal organ culture and its use to show that the Wt1 tumour suppressor is required for nephron differentiation. Hum Mol Genet. 2004;13(2):235–46. http://dx.doi.org/10.1093/hmg/ddh015
- Hohenstein P, Hastie ND. The many facets of the Wilms' tumour gene, WT1. Hum Mol Genet. 2006;15(suppl 2):R196–201. http://dx.doi.org/10.1093/hmg/ddl196

- 28. Martínez-Estrada OM, Lettice LA, Essafi A, Guadix JA, Slight J, Velecela V, et al. Wt1 is required for cardiovascular progenitor cell formation through transcriptional control of snail and E-cadherin. Nat Genet. 2009;42(1):89–93. http://dx.doi.org/10.1038/ng.494
- 29. von Gise A, Zhou B, Honor LB, Ma Q, Petryk A, Pu WT. WT1 regulates epicardial epithelial to mesenchymal transition through β-catenin and retinoic acid signaling pathways. Dev Biol. 2011:356(2):421–31.
 http://dx.doi.org/10.1016/j.wdbio.2011.05.668

http://dx.doi.org/10.1016/j.ydbio.2011.05.668

- 30. Sampson VB, David JM, Puig I, Patil PU, de Herreros AG, Thomas GV, et al. Wilms' tumor protein induces an epithelial-mesenchymal hybrid differentiation state in clear cell renal cell carcinoma. 2014, PLoS One 9(7):e102041. doi:10.1371/journal.pone.0102041.
- 31. Putzke AP, Ventura AP, Bailey AM, Akture C, Opoku-Ansah J, Çeliktaş M, et al. Metastatic progression of prostate cancer and e-cadherin: regulation by Zeb1 and src family kinases. Am J Pathol. 2011;179(1):400–10. http://dx.doi.org/10.1016/j.ajpath.2011.03.028
- 32. Hanahan D, Weinberg RA. Hallmarks of cancer: the next generation. Cell. 2011;144(5):646–74.

http://dx.doi.org/10.1016/j.cell.2011.02.013

- 33. Huff V. Wilms' tumours: about tumour suppressor genes, an oncogene and a chameleon gene. Nat Rev Cancer. 2011;11(2):111–21. http://dx.doi.org/10.1038/nrc3002
- 34. Ostergaard M, Olesen LH, Hasle H, Kjeldsen E, Hokland P. WT1 gene expression: an excellent tool for monitoring minimal residual disease in 70% of acute myeloid leukaemia patients results from a single-centre study. Br J Haematol. 2004 Jun;125(5):590–600.

http://dx.doi.org/10.1111/j.1365-2141.2004.04952.x

- 35. Inoue K, Sugiyama H, Ogawa H, Nakagawa M, Yamagami T, Miwa H, et al. WT1 as a new prognostic factor and a new marker for the detection of minimal residual disease in acute leukemia. Blood. 1994 November 1, 1994;84(9):3071–9.
- 36. Patmasiriwat P, Fraizer G, Kantarjian H, Saunders GF. WT1 and GATA1 expression in myelodysplastic syndrome and acute leukemia. Leukemia. 1999 Jun;13(6):891–900.
- 37. Ito K, Oji Y, Tatsumi N, Shimizu S, Kanai Y, Nakazawa T, et al. Antiapoptotic function of 17AA(+)WT1 (Wilms' tumor gene) isoforms on the intrinsic apoptosis pathway. Oncogene. 2006;25(3):4217–29.

http://dx.doi.org/10.1038/sj.onc.1209455

Silberstein GB, Van Horn K, Strickland P, Roberts CT, Daniel CW. Altered expression of the WT1 Wilms tumor suppressor gene in human breast cancer. Proc Natl Acad Sci. 1997;94(15):8132–7.

- 39. Miyoshi Y, Ando A, Egawa C, Taguchi T, Tamaki Y, Tamaki H, et al. High expression of Wilms' tumor suppressor gene predicts poor prognosis in breast cancer patients. Clin Cancer Res. 2002;8(5):1167–1171.
- 40. Nakatsuka S, Oji Y, Horiuchi T, Kanda T, Kitagawa M, Takeuchi T, et al. Immunohistochemical detection of WT1 protein in a variety of cancer cells. Mod Pathol. 2006;19(6):804–14.

http://dx.doi.org/10.1038/modpathol.3800588

 Werner H, Idelman G, Rubinstein M, Pattee P, Nagalla SR, Roberts CT. A novel EWS-WT1 gene fusion product in desmoplastic small round cell tumor is a potent transactivator of the insulin-like growth factor-I receptor (IGF-IR) gene. Cancer Lett. 2007;247:84–90.

http://dx.doi.org/10.1016/j.canlet.2006.03.027

- 42. Netinatsunthorn W, Hanprasertpong J, Dechsukhum C, Leetanaporn R, Geater A. WT1 gene expression as a prognostic marker in advanced serous epithelial ovarian carcinoma: an immunohistochemical study. BMC Cancer. 2006;6:90. http://dx.doi.org/10.1186/1471-2407-6-90
- Foster MR, Johnson JE, Olson SJ, Allred DC. Immunohistochemical analysis of nuclear versus cytoplasmic staining of WT1 in malignant mesotheliomas and primary pulmonary adenocarcinomas. Arch Pathol Lab Med. 2001 Oct;125(10):1316–20.

http://dx.doi.org/10.1043/0003-9985(2001)125<1316:IAONVC>2.0.CO;2

- Karakas T, Miething C, Maurer U, Weidmann E, Ackermann H, Hoelzer D, et al. The coexpression of the apoptosis-related genes bcl-2 and wt1 in predicting survival in adult acute myeloid leukemia. Leukemia. 2002;16:846–54. http://dx.doi.org/10.1038/sj.leu.2402434
- Bandiera R, Sacco S, Vidal VP, Chaboissier M, Schedl A. Steroidogenic organ development and homeostasis: a WT1-centric view. Mol Cell Endocrinol. 2015;408:145–55.

http://dx.doi.org/10.1016/j.mce.2015.01.009

- 46. Shimamura R, Fraizer GC, Trapman J, Lau Y, Saunders GF. The Wilms' tumor gene WT1 can regulate genes involved in sex determination and differentiation: SRY, Mullerian-inhibiting substance, and the androgen receptor. Clin Cancer Res. 1997 Dec;3(12 Pt 2):2571–80.
- 47. Dong G, Rajah R, Vu T, Hoffman A, Rosenfeld R, Robert CT Jr, et al. Decreased expression of the Wilm's tumor suppressor gene WT1 and increased IGF-II and IGF-1R in the prostatic stroma of patients with benign prostatic hyperplasia. J Clin Endocrinol Metab. 1997;82:2198–203.

http://dx.doi.org/10.1210/jcem.82.7.4067

- Brett A, Pandey S, Fraizer G. The Wilms' tumor gene (WT1) regulates E-cadherin expression and migration of prostate cancer cells. Mol Cancer. 2013;12(3). http://dx.doi.org/10.1186/1476-4598-12-3
- Gregg J, Brown K, Mintz E, Piontkivska H, Fraizer G. Analysis of gene expression in prostate cancer epithelial and interstitial stromal cells using laser capture microdissection. BMC Cancer. 2010;10(1):165. http://dx.doi.org/10.1186/1471-2407-10-165
- 50. Zaia A, Fraizer GC, Piantanelli L, Saunders GF. Transcriptional regulation of the androgen signaling pathway by the Wilms' tumor suppressor gene WT1. Anticancer Res. 2001 Jan-Feb;21(1A):1–10.
- 51. Graham K, Li W, Williams BR, Fraizer G. Vascular endothelial growth factor (VEGF) is suppressed in WT1-transfected LNCaP cells. Gene Expr. 2006;13(1):1–14.
- Skolarus TA, Wolf A, Erb NL, Brooks DD, Rivers BM, Underwood W, et al. American cancer society prostate cancer survivorship care guidelines. CA Cancer J Clin. 2014;64(4):225–49.

http://dx.doi.org/10.3322/caac.21234

- 53. Devilard E, Bladou F, Ramuz O, Karsenty G, Dalès JP, Gravis G, et al. FGFR1 and WT1 are markers of human prostate cancer progression. BMC Cancer. 2006;6(1):272. http://dx.doi.org/10.1186/1471-2407-6-272
- 54. Roberts SGE. Transcriptional regulation by WT1 in development. Curr Opin Genet Dev. 2005;15(5):542–7.

http://dx.doi.org/10.1016/j.gde.2005.08.004

- 55. Werner H, Rauscher FJ, 3rd, Sukhatme VP, Drummond IA, Roberts CT, Jr, LeRoith D. Transcriptional repression of the insulin-like growth factor I receptor (IGF-I-R) gene by the tumor suppressor WT1 involves binding to sequences both upstream and downstream of the IGF-I-R gene transcription start site. J Biol Chem. 1994 Apr 29;269(17):12577–82.
- Drummond I, Madden SL, Rohwer-Nutter P, Bell GI, Sukhatme VP, Rauscher FJ. Repression of the insulin-like growth factor II gene by the Wilms tumor suppressor WT1. Science. 1992;257(5070):674–8.
- 57. Hewitt SM, Hamada S, McDonnell TJ, Rauscher FJ, 3rd, Saunders GF. Regulation of the proto-oncogenes bcl-2 and c-myc by the Wilms' tumor suppressor gene WT1. Cancer Res. 1995 Nov 15;55(22):5386–9.
- 58. Heckman C, Mochon E, Arcinas M, Boxer LM. The WT1 protein is a negative regulator of the normal bcl-2 allele in t (14; 18) lymphomas. J Biol Chem. 1997;272(31):19609–14.
- Morrison DJ, English MA, Licht JD. WT1 induces apoptosis through transcriptional regulation of the proapoptotic Bcl-2 family member Bak. Cancer Res. 2005;65(18):8174–82.

http://dx.doi.org/10.1158/0008-5472.CAN-04-3657

- 60. Johnstone RW, See RH, Sells SF, Wang J, Muthukkumar S, Englert C, et al. A novel repressor, par-4, modulates transcription and growth suppression functions of the Wilms' tumor suppressor WT1. Mol Cell Biol. 1996 Dec;16(12):6945–56. PMCID:PMC231698.
- 61. Cheema SK, Mishra SK, Rangnekar VM, Tari AM, Kumar R, Lopez-Berestein G. Par-4 transcriptionally regulates bcl-2 through a WT1-binding site on the bcl-2 promoter. J Biol Chem. 2003 May 30;278(22):19995–20005. http://dx.doi.org/10.1074/jbc.M205865200
- 62. Hosono S, Gross I, English MA, Hajra KM, Fearon ER, Licht JD. E-cadherin is a WT1 target gene. J Biol Chem. 2000;275(15):10943–53.
- 63. Takeichi M. Cadherin cell adhesion receptors as a morphogenetic regulator. Science. 1991;251(5000):1451–5.
- 64. Blaschuk OW, Sullivan R, David S, Pouliot Y. Identification of a cadherin cell adhesion recognition sequence. Dev Biol. 1990;139(1):227–9.
- 65. Machado JC, Soares P, Carneiro F, Rocha A, Beck S, Blin N, et al. E-cadherin gene mutations provide a genetic basis for the phenotypic divergence of mixed gastric carcinomas. Lab Invest.1999;79(4):459–65.
- Caldeira J, Figueiredo J, Bras-Pereira C, Carneiro P, Moreira AM, Pinto MT, et al. E-cadherin-defective gastric cancer cells depend on laminin to survive and invade. Hum Mol Genet. 2015 Aug 5, ddv312.
- 67. Berx G, Cleton-Jansen A, Nollet F, De Leeuw W, Van de Vijver M, Cornelisse C, et al. E-cadherin is a tumour/invasion suppressor gene mutated in human lobular breast cancers. EMBO J. 1995;14(24):6107.
- 68. Wang Y, Zhao X, Shuai Z, Li C, Bai Q, Yu X, et al. Snail promotes epithelial-mesenchymal transition and invasiveness in human ovarian cancer cells. Int J Clin Exp Med. 2015;8(5):7388.
- 69. Sawada K, Mitra AK, Radjabi AR, Bhaskar V, Kistner EO, Tretiakova M, et al. Loss of E-cadherin promotes ovarian cancer metastasis via α5-integrin, which is a therapeutic target. Cancer Res. 2008;68(7):2329–39.

http://dx.doi.org/10.1158/0008-5472.CAN-07-5167

- 70. Mell LK, Meyer JJ, Tretiakova M, Khramtsov A, Gong C, Yamada SD, et al. Prognostic significance of E-cadherin protein expression in pathological stage I-III endometrial cancer. Clin Cancer Res. 2004;10(16):5546–53. http://dx.doi.org/10.1158/1078-0432.CCR-0943-03
- Brabant G, Hoang-Vu C, Cetin Y, Dralle H, Scheumann G, Mölne J, et al. E-cadherin: a differentiation marker in thyroid malignancies. Cancer Res. 1993;53(20):4987–93.
- 72. Lu S, Zhou W, Wei H, He L, Li L. MTBP promotes the invasion and metastasis of hepatocellular carcinoma by enhancing the MDM2-mediated degradation of E-cadherin. Dig Dis Sci. 2015:1–10.

- 73. Santoro A, Bufo P, Russo G, Cagiano S, Papagerakis S, Bucci P, et al. Expression and clinical implication of cyclooxygenase-2 and e-cadherin in oral squamous cell carcinomas. Cancer Biol Ther. 2015 [Epub ahead of print].
- 74. Hwang IK, Kim H, Lee YS, Kim J, Cho JY, Yoon Y, et al. The presence of pancreatic intraepithelial neoplasia-3 in a background of chronic pancreatitis in pancreatic cancer patients. Cancer Sci. 2015;106:1408–13.
- 75. Annicotte JS, Iankova I, Miard S, Fritz V, Sarruf D, Abella A, et al. Peroxisome proliferator-activated receptor {gamma} regulates E-cadherin expression and inhibits growth and invasion of prostate cancer. Mol Cell Biol. 2006;26(20):7561–74. http://dx.doi.org/10.1128/MCB.00605-06
- 76. Cheng L, Nagabhushan M, Pretlow TP, Amini SB, Pretlow TG. Expression of E-cadherin in primary and metastatic prostate cancer. Am J Pathol. 1996;148(5):1375.
- 77. Umbas R, Isaacs WB, Bringuier PP, Xue Y, Debruyne FMJ, Schalken JA. Relation between aberrant α-catenin expression and loss of e-cadherin function in prostate cancer. Int J Cancer. 1997;74(4):374–7.
- 78. Barber AG, Castillo-Martin M, Bonal DM, Jia AJ, Rybicki BA, Christiano AM, et al. PI3K/AKT pathway regulates e-cadherin and desmoglein 2 in aggressive prostate cancer. Cancer Med. 2015;4(8):1258–71. http://dx.doi.org/10.1002/cam4.463
- 79. Deep G, Jain AK, Ramteke A, Ting H, Vijendra KC, Gangar SC, et al. SNAI1 is critical for the aggressiveness of prostate cancer cells with low E-cadherin. Mol Cancer. 2014 Feb 24;13:37.

http://dx.doi.org/10.1186/1476-4598-13-37

- 80. Poblete CE, Fulla J, Gallardo M, Muñoz V, Castellón EA, Gallegos I, et al. Increased SNAIL expression and low syndecan levels are associated with high gleason grade in prostate cancer. Int J Oncol. 2014;44(3):647–54. http://dx.doi.org/10.3892/ijo.2014.2254
- Xie Y, Liu S, Lu W, Yang Q, Williams KD, Binhazim AA, et al. Slug regulates E-cadherin repression via p19Arf in prostate tumorigenesis. Mol Oncol. 2014;8(7):1355–64.

http://dx.doi.org/10.1016/j.molonc.2014.05.006

 Liu G, Yang H, Liu T, Lin Y. Expression and significance of E-cadherin, N-cadherin, transforming growth factor-β1 and twist in prostate cancer. Asian Pac J Trop Med. 2014;7(1):76–82.

http://dx.doi.org/10.1016/S1995-7645(13)60196-0

 Mostafavi-Pour Z, Kianpour S, Dehghani M, Mokarram P, Torabinejad S, Monabati A. Methylation of integrin a4 and E-cadherin genes in human prostate cancer. Pathol Oncol Res. 2015;21(4):1–7.

- Arya M, Bott SR, Shergill IS, Ahmed HU, Williamson M, Patel HR. The metastatic cascade in prostate cancer. Surg Oncol. 2006;15(3):117–28. http://dx.doi.org/10.1016/j.suronc.2006.10.002.
- Christiansen JJ, Rajasekaran AK. Reassessing epithelial to mesenchymal transition as a prerequisite for carcinoma invasion and metastasis. Cancer Res. 2006;66(17):8319–26. http://dx.doi.org/10.1158/0008-5472.CAN-06-0410
- 86. Guarino M, Rubino B, Ballabio G. The role of epithelial-mesenchymal transition in cancer pathology. Pathology. 2007;39(3):305–18. http://dx.doi.org/10.1080/00313020701329914
- 87. Zhu ML, Kyprianou N. Role of androgens and the androgen receptor in epithelial-mesenchymal transition and invasion of prostate cancer cells. FASEB J. 2010 Mar;24(3):769–77.

http://dx.doi.org/10.1096/fj.09-136994

 Yao B, Zhao J, Li Y, Li H, Hu Z, Pan P, et al. Elf5 inhibits TGF-β-driven epithelialmesenchymal transition in prostate cancer by repressing SMAD3 activation. Prostate. 2015;75(8):872–82.

http://dx.doi.org/10.1002/pros.22970

 Chaudhry P, Fabi F, Singh M, Parent S, Leblanc V, Asselin E. Prostate apoptosis response-4 mediates TGF-β-induced epithelial-to-mesenchymal transition. Cell Death Dis. 2014;5(2):e1044.

http://dx.doi.org/10.1038/cddis.2014.7

- 90. Wu C, Zhu W, Qian J, He S, Wu C, Chen Y, et al. WT1 promotes invasion of NSCLC via suppression of CDH1. J Thorac Oncol. 2013 Sep;8(9):1163–9. http://dx.doi.org/10.1097/JTO.0b013e31829f6a5f
- 91. Fox WD, Higgins B, Maiese KM, Drobnjak M, Cordon-Cardo C, Scher HI, et al. Antibody to vascular endothelial growth factor slows growth of an androgen-independent xenograft model of prostate cancer. Clin Cancer Res. 2002 Oct;8(10):3226–31.
- 92. Loureiro RM, D'Amore PA. Transcriptional regulation of vascular endothelial growth factor in cancer. Cytokine Growth Factor Rev. 2005;16(1):77–89. http://dx.doi.org/10.1016/j.cytogfr.2005.01.005
- 93. de Brot S, Ntekim A, Cardenas R, James V, Allegrucci C, Heery DM, et al. Regulation of vascular endothelial growth factor in prostate cancer. Endocr Relat Cancer. 2015 Jun;22(3):R107–23.

http://dx.doi.org/10.1530/ERC-15-0123

Wagner KD, Wagner N, Vidal VPI, Schley G, Wilhelm D, Schedl A, et al. The Wilms' tumor gene Wt1 is required for normal development of the retina. EMBO J. 2002;21(6):1398–405.

http://dx.doi.org/10.1093/emboj/21.6.1398

 Scholz H, Kirschner KM. Oxygen-dependent gene expression in development and cancer: lessons learned from the Wilms' tumor gene, WT1. Front Mol Neurosci. 2011;4:4.

http://dx.doi.org/10.3389/fnmol.2011.00004

- 96. Mazzucchelli R, Montironi R, Santinelli A, Lucarini G, Pugnaloni A, Biagini G. Vascular endothelial growth factor expression and capillary architecture in high-grade PIN and prostate cancer in untreated and androgen-ablated patients. Prostate. 2000;45(1):72–9.
- 97. Karth J, Ferrer FA, Perlman E, Hanrahan C, Simons JW, Gearhart JP, et al. Coexpression of hypoxia-inducible factor 1-alpha and vascular endothelial growth factor in Wilms' tumor. J Pediatr Surg. 2000;35(12):1749–53. http://dx.doi.org/10.1053/jpsu.2000.19241
- 98. Gao X, Chen X, Taglienti M, Rumballe B, Little MH, Kreidberg JA. Angioblastmesenchyme induction of early kidney development is mediated by Wt1 and vegfa. Development. 2005 Dec;132(24):5437–49. http://dx.doi.org/10.1242/dev.02095
- 99. Katuri V, Gerber S, Qiu X, McCarty G, Goldstein SD, Hammers H, et al. WT1 regulates angiogenesis in Ewing sarcoma. Oncotarget. 2014 May 15;5(9):2436–49. http://dx.doi.org/10.18632/oncotarget.1610
- McCarty G, Awad O, Loeb DM. WT1 directly regulates expression of vascular endothelial growth factor and is a mediator of tumor response to hypoxia. J Biol Chem. 2011 286(51):43634–43.

http://dx.doi.org/10.1074/jbc.M111.310128

- 101. Eisermann K, Tandon S, Bazarov A, Brett A, Fraizer G, Piontkivska H. Evolutionary conservation of zinc finger transcription factor binding sites in promoters of genes co-expressed with WT1 in prostate cancer. BMC Genomics. 2008 Jul 16;9:337. http://dx.doi.org/10.1186/1471-2164-9-337
- 102. Hanson J, Gorman J, Reese J, Fraizer G. Regulation of vascular endothelial growth factor, VEGF, gene promoter by the tumor suppressor, WT1. Front Biosci. 2007;12:2279–90.
- Dutton J, Lahiri D, Ward A. Different isoforms of the Wilms' tumour protein WT1 have distinct patterns of distribution and trafficking within the nucleus. Cell Prolif. 2006;39(6):519–35.

http://dx.doi.org/10.1111/j.1365-2184.2006.00409.x

104. Eisermann K, Broderick CJ, Bazarov A, Moazam MM, Fraizer GC. Androgen upregulates vascular endothelial growth factor expression in prostate cancer cells via an Sp1 binding site. Mol Cancer. 2013;12:7. http://dx.doi.org/10.1186/1476-4598-12-7

105. Sordello S, Bertrand N, Plouet J. Vascular endothelial growth factor is up-regulated in vitro and in vivo by androgens. Biochem Biophys Res Commun. 1998 Oct 9;251(1):287–90.

http://dx.doi.org/10.1006/bbrc.1998.9328

- 106. Joseph IB, Nelson JB, Denmeade SR, Isaacs JT. Androgens regulate vascular endothelial growth factor content in normal and malignant prostatic tissue. Clin Cancer Res. 1997 Dec;3(12 Pt 1):2507–11.
- 107. Aslan G, Cimen S, Yorukoglu K, Tuna B, Sonmez D, Mungan U, et al. Vascular endothelial growth factor expression in untreated and androgen-deprived patients with prostate cancer. Pathol Res Pract. 2005;201(8):593–8. http://dx.doi.org/10.1016/j.prp.2005.07.003
- 108. Mueller MD, Vigne JL, Minchenko A, Lebovic DI, Leitman DC, Taylor RN. Regulation of vascular endothelial growth factor (VEGF) gene transcription by estrogen receptors alpha and beta. Proc Natl Acad Sci U S A. 2000 Sep 26;97(20):10972–7. http://dx.doi.org/10.1073/pnas.200377097
- 109. Stoner M, Wormke M, Saville B, Samudio I, Qin C, Abdelrahim M, et al. Estrogen regulation of vascular endothelial growth factor gene expression in ZR-75 breast cancer cells through interaction of estrogen receptor a and SP proteins. Oncogene. 2004;23:1052–63. http://dx.doi.org/10.1038/sj.onc.1207201
- 110. Levine AC, Liu XH, Greenberg PD, Eliashvili M, Schiff JD, Aaronson SA, et al. Androgens induce the expression of vascular endothelial growth factor in human fetal prostatic fibroblasts. Endocrinology. 1998 Nov;139(11):4672–8. http://dx.doi.org/10.1210/endo.139.11.6303
- 111. Li J, Wang E, Rinaldo F, Datta K. Upregulation of VEGF-C by androgen depletion: the involvement of IGF-IR-FOXO pathway. Oncogene. 2005;24(35):5510–20. http://dx.doi.org/10.1038/sj.onc.1208693
- 112. Stewart RJ, Panigrahy D, Flynn E, Folkman J. Vascular endothelial growth factor expression and tumor angiogenesis are regulated by androgens in hormone responsive human prostate carcinoma: evidence for androgen dependent destabilization of vascular endothelial growth factor transcripts. J Urol. 2001 Feb;165(2):688–93. http://dx.doi.org/10.1097/00005392-200102000-00095
- 113. Mabjeesh NJ, Willard MT, Frederickson CE, Zhong H, Simons JW. Androgens stimulate hypoxia-inducible factor 1 activation via autocrine loop of tyrosine kinase receptor/phosphatidylinositol 3'-kinase/protein kinase B in prostate cancer cells. Clin Cancer Res. 2003;9(7):2416–25.
- Eisermann K, Bazarov A, Brett A, Knapp E, Piontkivska H, Fraizer G. Uncovering androgen responsive regulatory networks in prostate cancer. In: Bioinformatics, 2009. OCCBIO'09. Ohio Collaborative Conference on;IEEE;2009: p. 99–103.

- 115. Wang Q, Li W, Liu XS, Carroll JS, Jänne OA, Keeton EK, et al. A hierarchical network of transcription factors governs androgen receptor-dependent prostate cancer growth. Mol Cell. 2007;27(3):380–92. http://dx.doi.org/10.1016/j.molcel.2007.05.041
- Bolton EC, So AY, Chaivorapol C, Haqq CM, Li H, Yamamoto KR. Cell- and gene-specific regulation of primary target genes by the androgen receptor. Genes Dev. 2007 August 15;21(16):2005–17. http://dx.doi.org/10.1101/gad.1564207
- 117. Massie CE, Adryan B, Barbosa-Morais NL, Lynch AG, Tran MG, Neal DE, et al. New androgen receptor genomic targets show an interaction with the ETS1 transcription factor. EMBO Rep. 2007 Sep;8(9):871–8. http://dx.doi.org/10.1038/sj.embor.7401046

Chapter 15

Functional Roles of Wilms' Tumor 1 (WT1) in Malignant Brain Tumors

Noriyuki Kijima,¹ Naoya Hashimoto,² Yasuyoshi Chiba,³ Yasunori Fujimoto,⁴ Haruo Sugiyama,⁵ Toshiki Yoshimine⁴

¹Department of Neurosurgery, Osaka National Hospital, Osaka, Japan; ²Department of Neurosurgery, Kyoto Prefectural University of Medicine, Kyoto, Japan; ³Department of Neurosurgery, Osaka Medical Center and Research Institute for Maternal and Child Health, Osaka, Japan; ⁴Department of Neurosurgery, Osaka University, Graduate School of Medicine, Osaka, Japan; ⁵Department of Functional Diagnostic Science, Osaka University Graduate School of Medicine, Osaka, Japan

Author for correspondence: Noriyuki Kijima, MD, PhD, Department of Neurosurgery, Osaka National Hospital, Osaka, Japan. Email: norikiji@onh.go.jp

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch15

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/

Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

The pleiotropic transcription factor, Wilms' tumor 1 (WT1), is expressed in the majority of glioblastomas, the most common malignant brain tumors of adulthood. Despite intensive treatment, including surgery and chemoradiotherapy, the prognosis for patients with glioblastoma remains very poor. Encouragingly, immunotherapy targeting WT1 has proven to be effective in recurrent glioblastoma, suggesting that

In: Wilms Tumor. Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Kijima et al.

this approach may be an important new treatment modality for the disease. However, WT1 appears to function as a context-dependent tumor suppressor or oncogene, and the functional roles of WT1 in the pathogenesis of glioblastoma, and other types of brain tumors, have not been extensively studied. With this in mind, we briefly review WT1 expression data for a range of different brain tumor classes and address the role of WT1 in the regulation of proliferation and apoptosis in glioblastoma. We generated WT1 knockdown glioblastoma cells by using shRNA-expressing lentivirus. Proliferation was reduced and apoptosis increased in WT1 knockdown glioblastoma cells compared with control cells *in vitro*. Consistent with these data, when WT1 knockdown glioblastoma cells or control glioblastoma cells were intracranially injected into the immunodeficient mice, tumor growth was significantly reduced in WT1 knockdown cells compared with that in control cells. Thus, WT1 is an oncogene that regulates cell proliferation and apoptosis in glioblastoma.

Key words: Ependymoma; Glioblastoma; Medulloblastoma; Meningioma; Oligodendroglioma

Introduction

Wilms' tumor 1 (*WT1*) is a pleiotropic transcription factor expressed in various types of hematological malignancies and solid tumors (1–9). *WT1* was first defined as a tumor suppressor gene (10–15). However, accumulating evidence suggests that the *WT1* can act as an oncogene in some contexts. For example, the growth of a range of different WT1-expressing cancer cell types is inhibited by *WT1* antisense oligomer (16, 17) and *WT1*-specific shRNA (18). Furthermore, overexpression of WT1 promotes cell growth (19–21), migration, and invasion (22). Overexpression of WT1 also inhibits apoptosis (23) and induces tumorigenicity in leukemia (24). However, the functional roles of WT1 in the pathogenesis of malignant brain tumors have not been extensively studied.

Glioblastoma is one of the most common malignant brain tumors. Despite intensive treatment, including surgery, radiation, and chemotherapy, the prognosis is still very poor, and the median survival is only 12–15 months (25). Improved treatments are urgently required to improve the prognosis of glioblastoma patients, and various therapies have been tested or are in development. In this regard, WT1 peptide vaccine immunotherapy has proven to be effective in recurrent glioblastoma (26), suggesting that WT1 is a valid therapeutic target in glioblastoma.

We briefly review the available WT1 expression data for a range of different brain tumor classes and address the involvement of WT1 in the regulation of proliferation and apoptosis in glioblastoma cells. In addition, we also investigated whether WT1 is involved in glioblastoma tumorigenicity in an intracranial xenograft model.

WT1 in malignant brain tumors

WT1 expression in malignant brain tumors

Glioma

Glioma is one of the most common types of malignant brain tumor. According to the WHO classification of central nervous system tumors, glioma is divided into four different grades depending on the malignant potential. Immunohistochemical analyses demonstrated that WT1 is expressed in many gliomas (5, 26–29) (Figure 1) and expression is variable depending on the tumor grade. In less-aggressive grade II glioma and diffuse astrocytoma, WT1 expression was lower than that in grade IV glioblastoma. Furthermore, Rauscher et al. (29) reported that WT1 expression is a prognostic marker of WHO grade II and III tumors, and WT1 expression is reduced in recurrent tumors.

Oligodendroglioma

Oligodendroglioma is a type of glioma that is thought to originate from brain oligodendrocytes. It occurs primarily in adults (9.4% of all primary brain and central nervous system



Figure 1. (A)-(D) Immunohistochemical analysis of WT1 expression in glioblastomas. Many glioblastomas express WT1 by immunohistochemistry.

Kijima et al.

tumors) but also affect children (4% of all primary brain tumors). WT1 is also expressed in oligodendrogliomas, and its expression is elevated in higher grade tumors (29).

Medulloblastoma

Medulloblastoma is the most common type of malignant pediatric brain tumor. A previous report showed that WT1 transcripts were detectable in five of nine primary medulloblastoma tumors (30), but in a separate study, WT1 expression was not detectable by immunohistochemistry (27). It is unclear whether this discrepancy is related to the heterogeneous WT1 expression levels in the four distinct molecular subtypes of medulloblastoma that have been described (31–33). Additional detailed studies are required to address the levels and the significance of WT1 expression in the pathogenesis of the various medulloblastoma subtypes.

Ependymoma

Ependymoma is a neuroepithelial malignancy of the central nervous system, which occurs in both children and adults. Although Idowu et al. (34) and Yeung et al. (35) reported that WT1 is expressed in ependymoma by immunohistochemistry, the significance of WT1 expression in ependymoma pathogenesis remains to be determined.

Meningioma

Meningioma is a common and predominantly benign intracranial tumor, which is classified as grade I according to the WHO classification of central nervous system tumors. Less than 10 percent of meningioma cases are classified as malignant WHO grade II or III tumors. The expression of WT1 in meningioma is controversial. Singh et al. (36) reported that WT1 is not expressed in WHO grade I meningiomas by immunohistochemistry, while Iwami et al. (37) showed that at the mRNA level, WT1 is expressed in many meningioma samples irrespective of WHO grade. Furthermore, Iwami et al. (37) reported that WT1 could be a therapeutic target for skull base meningioma.

In summary, WT1 is expressed in many different classes of intracranial tumors, including gliomas, oligodendrogliomas, ependymomas, and meningiomas. However, in most cases, the significance of WT1 expression in the pathogenesis of brain tumor remains unclear. In part, this is related to the fact that WT1 expression will need to be assessed in many more representatives of the various histological and/or molecular brain tumor subtypes to generate the statistically robust conclusions. At present, there are more than 100 histological subtypes of brain tumors according to the WHO classification, many of which are rare, and of which only a specific subset predominates in children, and WT1 expression data are limited or not available. In addition, consensus will be required to determine the WT1 detection method that is most appropriate for the comparison of data across laboratories.

The available data suggest that the expression of WT1 in some major brain tumor classes, most notably glioma, is likely to play an important role in tumor initiation and progression.
WT1 in malignant brain tumors

Based on this, the impetus is provided to assess WT1 expression in all brain tumor types to determine the validity of WT1 as a therapeutic target across the brain tumor spectrum.

Functional roles of WT1 in glioblastoma pathogenesis

Most studies investigating the functional roles of WT1 in the pathogenesis of malignant brain tumors have focused on glioblastoma. In other types of brain tumors, the functional roles of WT1 have not been assessed. Previous reports suggest that WT1 is involved in driving cell proliferation (38) and inhibiting apoptosis (18, 38) in glioblastoma. In an earlier study conducted in our laboratory (38), we transduced two glioblastoma cell lines, U87MG and U251, with lentivirus carrying WT1 shRNA to address the effect of WT1 knockdown on cell proliferation. We found that cell proliferation was significantly reduced (Figure 2A and 2B), suggesting that WT1 is involved in proliferation of glioblastoma cells. We also examined the effect of WT1 on glioblastoma progression in vivo by transducing U87MG cells with WT1 shRNA or control shRNA followed by intracranial injection into the immunodeficient $Rag2^{-/-}$ gamma $c^{-/-}$ mice. There was a significant difference in survival between the mice injected with U87MG cells transduced with WT1 shRNA and those injected with control U87MG cells. We also found that all the mice inoculated with U87MG cells transduced with control shRNA died of glioblastoma within 40 days, whereas none of the mice injected with WT1-shRNA-treated U87MG cells died of glioblastoma by the same time point (Figure 2C). These data demonstrated that WT1 knockdown significantly inhibited glioblastoma growth in vivo.

Immunohistochemical analysis of formalin-fixed, paraffin-embedded tumor sections from mice inoculated with U87MG cells transduced with WT1 shRNA or control shRNA revealed that the Ki67 proliferation index was higher in control tumors compared with those transduced with WT1 shRNA (Figure 2D). Consistent with the *in vitro* data, these findings also suggest that WT1 drives cell proliferation and tumor formation *in vivo*.

We also investigated the differences in mRNA expression of selected apoptosis-related genes in U87MG and U251 cells transduced with WT1 shRNA or control shRNA by realtime PCR. We found that apoptosis-related genes such as *MAP3K5*, *PIK3CA*, and *p53* were upregulated in both U87MG and U251 WT1 knockdown cells compared with those in control cells. Extending these findings, we examined the differences in apoptosis between glioblastoma cells transduced with WT1 shRNA and control shRNA using an Annexin-V-Fluos kit. We found that the number of apoptotic cells was higher in U87MG and U251 WT1 knockdown cells compared with that in U87MG cells transduced with control shRNA (Figure 3A and 3B). These results showed that apoptosis was promoted in both U87MG and U251 WT1 knockdown cells, suggesting that WT1 drives glioblastoma tumorigenicity by regulating apoptosis.

Kijima et al.



Figure 2. WT1 is involved in glioblastoma tumorigenicity *in vivo*. (A) Cell proliferation rate of U87MG cells transduced with control shRNA and WT1 shRNA. (B) Cell proliferation rate of U251 cells transduced with control shRNA and WT1 shRNA. (C) All the mice injected with U87MG cells transduced with control shRNA died of glioblastoma within 40 days after tumor cell xenograft, whereas none of the mice injected with U87MG cells transduced with WT1 shRNA succumbed by 40 days' post-transplant. (D) Immunohistochemical staining for the Ki67 proliferation marker in tumor samples from mice injected with U87MG cells transduced with WT1 shRNA or control shRNA.

In addition to our report, roles of WT1 in glioblastoma pathogenesis have been described in several studies (5, 39–41). Oji et al. (5) reported that the growth of U87MG cells was inhibited by a WT1 antisense oligomer consistent with WT1 being involved in the regulation of cell proliferation in glioblastoma. Tatsumi et al. (18) found that WT1 inhibited apoptosis in the A172 glioblastoma cell line. Clark et al. (39) reported that WT1 regulated cell proliferation in U251 cells and that U251 WT1 knockdown cells showed significantly lower tumorigenicity in a subcutaneous nude mouse model. However, contrasting data were presented in several other studies using different glioblastoma cell lines. Chidambaram et al. (40) reported that silencing of WT1 reduced the invasiveness of U1242 glioblastoma cells but had no effect on the U1242 cell proliferation *in vitro*. Clark et al. (39) found that transduction of T98G cells with WT1 shRNA had no effect on apoptosis compared with those transduced with control shRNA *in vitro*. These results suggest that the regulation of cell proliferation and apoptosis

WT1 in malignant brain tumors



Figure 3. WT1 regulates apoptosis in glioblastoma cells. (A) The number of apoptotic cells per field and representative pictures of Annexin V-positive U87MG cells following transduction with control shRNA or WT1 shRNA. (B) The number of apoptotic cells per field and representative pictures of Annexin V-positive U251 cells following transduction with control shRNA or WT1 shRNA.

by WT1 *in vitro* is dependent on the specific glioblastoma cell line being studied, potentially reflecting subtype-specific molecular characteristics.

In summary, the functional roles of WT1 in glioblastoma pathogenesis remain controversial although the weight of evidence is consistent, with WT1 being involved in regulating cell proliferation and apoptosis in at least some glioblastoma. Clearly, more work is required to comprehensively address the functional roles of WT1 in the initiation and progression of glioblastoma and the many other types of malignant brain tumor that have not yet been adequately studied.

Immunotherapy targeting WT1 peptide for malignant glioma

Cancer vaccination is one of the immunotherapeutic strategies that have been developed to target many solid tumors. Recently, a large number of tumor-associated antigens, including WT1, were identified and assessed as potential candidates as cancer vaccines. Tumor antigen epitopes associated with human leukocyte antigen (HLA) class I molecules were

Kijima et al.

recognized by cytotoxic T lymphocytes, providing a potential mechanism for direct tumor cell killing. Furthermore, recent studies reported that systemic immunotherapy can induce an antitumor response within the immunologically privileged brain. These findings suggest that the peptide-based cancer immunotherapy could be a potent therapeutic strategy for the treatment of malignant brain tumors.

Izumoto et al. (26) carried out a phase II clinical trial of WT1 peptide vaccine immunotherapy for recurrent glioblastomas and found that the approach was effective in this context. To improve the efficacy of the treatment, we have been assessing a possible combination of WT1 peptide vaccine immunotherapy with temozolomide, a standard chemotherapeutic agent for newly diagnosed malignant glioma patients (42). We found that the combination therapy was tolerable without serious side effects. We are now moving on to the phase II clinical trial of combination WT1 peptide vaccine immunotherapy and temozolomide chemotherapy for newly diagnosed glioblastoma. This represents one example of combined immunotherapy with chemotherapeutic or other immunotherapeutic modalities, such as antiangiogenic agents or checkpoint inhibitors. Although WT1 peptide vaccine immunotherapy for malignant brain tumors has promise, more extensive studies are needed to determine the clinical efficacy of this approach.

Conclusions

WT1 expression in malignant brain tumors varies depending on the tumor type. WT1 functions as an oncogene in at least some glioblastomas, in part, by regulating cell proliferation and apoptosis. Overall, these findings suggest that WT1 is a valid molecular target for the treatment of glioblastoma and potentially a range of other malignant brain tumors.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

Not relevant.

References

 Oji Y, Inohara H, Nakazawa M, Nakano Y, Akahani S, Nakatsuka S, et al. Overexpression of the Wilms' tumor gene WT1 in head and neck squamous cell carcinoma. Cancer Sci. 2003;94(6):523–9. Oji Y, Miyoshi S, Maeda H, Hayashi S, Tamaki H, Nakatsuka S, et al. Overexpression of the Wilms' tumor gene WT1 in de novo lung cancers. Int J Cancer. 2002;100(3):297–303.

http://dx.doi.org/10.1002/ijc.10476

- 3. Oji Y, Miyoshi Y, Koga S, Nakano Y, Ando A, Nakatsuka S, et al. Overexpression of the Wilms' tumor gene WT1 in primary thyroid cancer. Cancer Sci. 2003;94(7):606–11.
- 4. Oji Y, Nakamori S, Fujikawa M, Nakatsuka S, Yokota A, Tatsumi N, et al. Overexpression of the Wilms' tumor gene WT1 in pancreatic ductal adenocarcinoma. Cancer Sci. 2004;95(7):583-7.
- 5. Oji Y, Suzuki T, Nakano Y, Maruno M, Nakatsuka S, Jomgeow T, et al. Overexpression of the Wilms' tumor gene W T1 in primary astrocytic tumors. Cancer Sci. 2004;95(10):822–7.
- Oji Y, Yamamoto H, Nomura M, Nakano Y, Ikeba A, Nakatsuka S, et al. Overexpression of the Wilms' tumor gene WT1 in colorectal adenocarcinoma. Cancer Sci. 2003;94(8):712–7.
- Oji Y, Yano M, Nakano Y, Abeno S, Nakatsuka S, Ikeba A, et al. Overexpression of the Wilms' tumor gene WT1 in esophageal cancer. Anticancer Res. 2004;24(5B):3103–8.
- Ueda T, Oji Y, Naka N, Nakano Y, Takahashi E, Koga S, et al. Overexpression of the Wilms' tumor gene WT1 in human bone and soft-tissue sarcomas. Cancer Sci. 2003;94(3):271–6.
- Clark AJ, Dos Santos WG, McCready J, Chen MY, Van Meter TE, Ware JL, et al. Wilms tumor 1 expression in malignant gliomas and correlation of +KTS isoforms with p53 status. J Neurosurg. 2007;107(3):586–92. http://dx.doi.org/10.3171/JNS-07/09/0586
- Coppes MJ, Campbell CE, Williams BR. The role of WT1 in Wilms tumorigenesis. FASEB J. 1993;7(10):886–95.
- 11. Rauscher. The WT1 Wilms tumor gene product: a developmentally regulated transcription factor in the kidney that functions as a tumor suppressor. FASEB J. 1993;7(10):896–903.
- 12. Haber DA, Park S, Maheswaran S, Englert C, Re GG, Hazen-Martin DJ, et al. WT1mediated growth suppression of Wilms tumor cells expressing a WT1 splicing variant. Science. 1993;262(5142):2057–9.
- Algar EM, Kenney MT, Simms LA, Smith SI, Kida Y, Smith PJ. Homozygous intragenic deletion in the WT1 gene in a sporadic Wilms' tumour associated with high levels of expression of a truncated transcript. Hum Mutat. 1995;5(3):221–7. http://dx.doi.org/10.1002/humu.1380050306

Kijima et al.

 14. Little M, Wells C. A clinical overview of WT1 gene mutations. Hum Mutat. 1997;9(3):209-25. http://dx.doi.org/10.1002/(SICI)1098-1004(1997)9:3<209::AID-

HUMU2>3.0.CO;2-2

- 15. Loeb DM, Sukumar S. The role of WT1 in oncogenesis: tumor suppressor or oncogene? Int J Hematol. 2002;76(2):117–26.
- Algar EM, Khromykh T, Smith SI, Blackburn DM, Bryson GJ, Smith PJ. A WT1 antisense oligonucleotide inhibits proliferation and induces apoptosis in myeloid leukaemia cell lines. Oncogene. 1996;12(5):1005–14.
- Yamagami T, Sugiyama H, Inoue K, Ogawa H, Tatekawa T, Hirata M, et al. Growth inhibition of human leukemic cells by WT1 (Wilms tumor gene) antisense oligodeoxynucleotides: implications for the involvement of WT1 in leukemogenesis. Blood. 1996;87(7):2878–84.
- Tatsumi N, Oji Y, Tsuji N, Tsuda A, Higashio M, Aoyagi S, et al. Wilms' tumor gene WT1-shRNA as a potent apoptosis-inducing agent for solid tumors. Int J Oncol. 2008;32(3):701–11.
- Inoue K, Tamaki H, Ogawa H, Oka Y, Soma T, Tatekawa T, et al. Wilms' tumor gene (WT1) competes with differentiation-inducing signal in hematopoietic progenitor cells. Blood. 1998;91(8):2969–76.
- 20. Tsuboi A, Oka Y, Ogawa H, Elisseeva OA, Tamaki H, Oji Y, et al. Constitutive expression of the Wilms' tumor gene WT1 inhibits the differentiation of myeloid progenitor cells but promotes their proliferation in response to granulocyte-colony stimulating factor (G-CSF). Leuk Res. 1999;23(5):499–505.
- 21. Li H, Oka Y, Tsuboi A, Yamagami T, Miyazaki T, Yusa S, et al. The lck promoterdriven expression of the Wilms tumor gene WT1 blocks intrathymic differentiation of T-lineage cells. Int J Hematol. 2003;77(5):463–70.
- 22. Jomgeow T, Oji Y, Tsuji N, Ikeda Y, Ito K, Tsuda A, et al. Wilms' tumor gene WT1 17AA(-)/KTS(-) isoform induces morphological changes and promotes cell migration and invasion in vitro. Cancer Sci. 2006;97(4):259–70. http://dx.doi.org/10.1111/j.1349-7006.2006.00169.x
- Ito K, Oji Y, Tatsumi N, Shimizu S, Kanai Y, Nakazawa T, et al. Antiapoptotic function of 17AA(+)WT1 (Wilms' tumor gene) isoforms on the intrinsic apoptosis pathway. Oncogene. 2006;25(30):4217–29.

http://dx.doi.org/10.1038/sj.onc.1209455

24. Nishida S, Hosen N, Shirakata T, Kanato K, Yanagihara M, Nakatsuka S, et al. AML1-ETO rapidly induces acute myeloblastic leukemia in cooperation with the Wilms tumor gene, WT1. Blood. 2006;107(8):3303–12. http://dx.doi.org/10.1182/blood-2005-04-1656

- 25. Stupp R, Mason WP, van den Bent MJ, Weller M, Fisher B, Taphoorn MJ, et al. Radiotherapy plus concomitant and adjuvant temozolomide for glioblastoma. N Engl J Med. 2005;352(10):987–96. http://dx.doi.org/10.1056/NEJMoa043330
- 26. Izumoto S, Tsuboi A, Oka Y, Suzuki T, Hashiba T, Kagawa N, et al. Phase II clinical trial of Wilms tumor 1 peptide vaccination for patients with recurrent glioblastoma multiforme. J Neurosurg. 2008;108(5):963–71.
 http://dui.org/10/2171/JNS/2008/108/5/00(2)

http://dx.doi.org/10.3171/JNS/2008/108/5/0963

- Nakahara Y, Okamoto H, Mineta T, Tabuchi K. Expression of the Wilms' tumor gene product WT1 in glioblastomas and medulloblastomas. Brain Tumor Pathol. 2004;21(3):113–6.
- 28. Chiba Y, Hashimoto N, Tsuboi A, Rabo C, Oka Y, Kinoshita M, et al. Prognostic value of WT1 protein expression level and MIB-1 staining index as predictor of response to WT1 immunotherapy in glioblastoma patients. Brain Tumor Pathol. 2010;27(1):29–34. http://dx.doi.org/10.1007/s10014-010-0265-9
- Rauscher J, Beschorner R, Gierke M, Bisdas S, Braun C, Ebner FH, et al. WT1 expression increases with malignancy and indicates unfavourable outcome in astrocytoma. J Clin Pathol. 2014;67(7):556–61. http://dx.doi.org/10.1136/jclinpath-2013-202114
- 30. Dennis SL, Manji SS, Carrington DP, Scarcella DL, Ashley DM, Smith PJ, et al. Expression and mutation analysis of the Wilms' tumor 1 gene in human neural tumors. Int J Cancer. 2002;97(5):713–5.
- Northcott PA, Korshunov A, Witt H, Hielscher T, Eberhart CG, Mack S, et al. Medulloblastoma comprises four distinct molecular variants. J Clin Oncol. 2011;29(11):1408–14. http://dx.doi.org/10.1200/JCO.2009.27.4324
- 32. Northcott PA, Korshunov A, Pfister SM, Taylor MD. The clinical implications of medulloblastoma subgroups. Nat Rev Neurol. 2012;8(6):340–51. http://dx.doi.org/10.1038/nrneurol.2012.78
- 33. Taylor MD, Northcott PA, Korshunov A, Remke M, Cho YJ, Clifford SC, et al. Molecular subgroups of medulloblastoma: the current consensus. Acta Neuropathol. 2012;123(4):465–72.

http://dx.doi.org/10.1007/s00401-011-0922-z

34. Idowu MO, Rosenblum MK, Wei XJ, Edgar MA, Soslow RA. Ependymomas of the central nervous system and adult extra-axial ependymomas are morphologically and immunohistochemically distinct – a comparative study with assessment of ovarian carcinomas for expression of glial fibrillary acidic protein. Am J Surg Pathol. 2008;32(5):710–8.

http://dx.doi.org/10.1097/PAS.0b013e318159a2b4

Kijima et al.

- 35. Yeung JT, Hamilton RL, Okada H, Jakacki RI, Pollack IF. Increased expression of tumor-associated antigens in pediatric and adult ependymomas: implication for vaccine therapy. J Neurooncol. 2013;111(2):103–11. http://dx.doi.org/10.1007/s11060-012-0998-x
- 36. Singh A, Mishra AK, Ylaya K, Hewitt SM, Sharma KC, Saxena S. Wilms tumor-1, claudin-1 and ezrin are useful immunohistochemical markers that help to distinguish schwannoma from fibroblastic meningioma. Pathol Oncol Res. 2012;18(2):383–9. http://dx.doi.org/10.1007/s12253-011-9456-x
- 37. Iwami K, Natsume A, Ohno M, Ikeda H, Mineno J, Nukaya I, et al. Adoptive transfer of genetically modified Wilms' tumor 1-specific T cells in a novel malignant skull base meningioma model. Neuro Oncol. 2013;15(6):747–58. http://dx.doi.org/10.1093/neuonc/not007
- 38. Kijima N, Hosen N, Kagawa N, Hashimoto N, Kinoshita M, Oji Y, et al. Wilms' tumor 1 is involved in tumorigenicity of glioblastoma by regulating cell proliferation and apoptosis. Anticancer Res. 2014;34(1):61–7.
- Clark AJ, Ware JL, Chen MY, Graf MR, Van Meter TE, Dos Santos WG, et al. Effect of WT1 gene silencing on the tumorigenicity of human glioblastoma multiforme cells. J Neurosurg. 2010;12(1):18–25.

http://dx.doi.org/10.3171/2008.11.JNS08368

- 40. Chidambaram A, Fillmore HL, Van Meter TE, Dumur CI, Broaddus WC. Novel report of expression and function of CD97 in malignant gliomas: correlation with Wilms tumor 1 expression and glioma cell invasiveness. J Neurosurg. 2012;116(4):843–53. http://dx.doi.org/10.3171/2011.11.JNS111455
- Somasundaram A, Ardanowski N, Opalak CF, Fillmore HL, Chidambaram A, Broaddus WC. Wilms tumor 1 gene, CD97, and the emerging biogenetic profile of glioblastoma. Neurosurg Focus. 2014;37(6):E14. http://dx.doi.org/10.3171/2014.9.FOCUS14506
- 42. Hashimoto N, Tsuboi A, Kagawa N, Chiba Y, Izumoto S, Kinoshita M, et al. Wilms tumor 1 peptide vaccination combined with temozolomide against newly diagnosed glioblastoma: safety and impact on immunological response. Cancer Immunol Immunother. 2015;64(6):707–16.

http://dx.doi.org/10.1007/s00262-015-1674-8

Chapter 16

Wilms' Tumor Gene (WT1) Expression and Minimal Residual Disease in Acute Myeloid Leukemia

Giovanni Rossi, Maria Marta Minervini, Angelo Michele Carella, Lorella Melillo, Nicola Cascavilla

Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, San Giovanni Rotondo, Italy

Author for correspondence: Giovanni Rossi, PhD, MD, Department of Hematology and Stem Cell Transplantation Unit, IRCCS "Casa Sollievo della Sofferenza" Hospital, v.le Cappuccini 1, 71013, San Giovanni Rotondo, Italy. Email: giovannirossi.fr@gmail.com

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ch16

Copyright: The Authors.

Licence: This open access article is licensed under Creative Commons Attribution 4.0 International (CC BY 4.0). http://creativecommons.org/licenses/by/4.0/ Users are allowed to share (copy and redistribute the material in any medium or format) and adapt (remix, transform, and build upon the material for any purpose, even commercially), as long as the authors and the publisher are explicitly identified and properly acknowledged as the original source.



Abstract

The identification of minimal residual disease (MRD) has led to substantial improvements in early recognition of the recurrence of acute myeloid leukemia (AML). Flow cytometry (FC), real-time quantitative polymerase chain reaction (RQ-PCR) and fluorescence in situ hybridization are useful methods for the detection of MRD in AML patients although molecular monitoring of leukemia-specific rearranged (RUNX1-RUNX1T1 and CBFB-MYH11) or mutated genetic (NPM1, CEBPA) sequences represents the most sensitive methodology.

In: *Wilms Tumor*. Marry M. van den Heuvel-Eibrink (Editor) ISBN: 978-0-9944381-1-9; Doi: http://dx.doi.org/10.15586/codon.wt.2016 Codon Publications, Brisbane, Australia

Besides, more than 50% of all AML patients lack one of these specific sequences, so it is crucial to identify molecular targets applicable for the majority of patients. WT1 is overexpressed at the mRNA level in 80-90% of AML cases at diagnosis in both peripheral blood and bone marrow, and is detectable in a consistent low range in normal donors. These features have led to its adoption for MRD detection using RQ-PCR. A European LeukemiaNet Study found the magnitude of WT1 log reduction after induction chemotherapy to be an independent predictor of relapse. Other studies showed a poorer outcome in patients having WT1 levels above reference thresholds at specific time points. WT1 expression was compared with other modalities of MRD assessment, such as RQ-PCR of specific fusion genes and FC, but no differences in terms of predictive value emerged. Finally, some authors translated the use of WT1 in the clinic giving donor lymphocytes infusions to patients with increasing WT1-mRNA levels after allogeneic stem cell transplantation and obtaining an improvement of survival in this subset. Data collected on WT1 expression over the past years provided evidence for the use of this molecular marker to stratify high-risk AML patients. It can also be used as a marker for early interventional therapy, but further studies are needed to demonstrate it.

Key words: Acute myeloid leukemia; Allogeneic stem cell transplantation; Minimal residual disease; Multiparameter flow cytometry; WT1 expression

Introduction

WT1 is an important regulatory molecule involved in cell growth and development. The presence of zinc fingers in the C-terminal half of the protein confers WT1 the role of a potent transcriptional factor, including important genes for cellular growth and metabolism among the targets (1). It has been found that WT1 can either enhance or repress the expression of specific target genes, depending on the levels of WT1 expression, the isoforms, the location of the transcriptional start site, and the cell type in which the experiment was performed (2, 3). In human hematopoietic cells, WT1 appears to behave as a tumor suppressor gene as the overexpression of WT1 in early human bone marrow (BM) cells leads to growth arrest and reduced colony formation. Indeed, in normal human BM, WT1 is expressed at extremely low levels and is confined to the primitive CD34+ population of cells (4, 5). Besides, WT1 is highly overexpressed in the BM or peripheral blood (PB) of a variety of leukemias, and these evidences support the role of WT1 as an oncogene in this subset (6, 7). Increased levels of WT1 expression can be found in both acute lymphoblastic and acute myeloid leukemia (AML) although more frequently in AML (frequencies varying from 73% to 91%) (8-10). Following the discovery of overexpression of WT1, there has been growing evidence that the WT1 expression levels may have a prognostic role in AML. In 139 de novo AML, Bergmann et al. (11) observed that the probability of the 3-year overall survival (OS) was 59% in patients with low WT1 levels compared to 21% in patients with high levels. Similarly, Galimberti et al. (12) showed a higher probability of disease progression in AML patients presenting high WT1 levels, and recently, Nomdedeu et al. (13) also confirmed the prognostic role of high WT1 levels at diagnosis in a larger study population. However, these data are in contrast with results reported by others where WT1 levels did not correlate with the outcome (8-10, 14), thus suggesting a controversial role for WT1 expression at presentation. On the contrary, a greater agreement was found among groups that have used WT1 levels as a marker of minimal residual disease (MRD) in AML remission BMs (less than 5% of blast cells). In particular, WT1 expression has been shown to predict disease progression in AML patients treated with conventional chemotherapy (8-10, 15-17) and patients undergone allogeneic stem cell transplantation (allo-SCT) (18-22). Furthermore, when WT1 expression was compared with widely used techniques in monitoring MRD such as multiparameter flow cytometry (MFC) (23) or specific molecular targets such as fusion genes transcripts (PML-RARa, AML-ETO1, and CBFb-MYH11), comparable sensitivities were found in predicting the relapse in AML. Thus, we addressed our review on the main papers that focused on the predictive role of WT1 expression as an MRD marker in AML patients, as well as results from comparison between WT1 and other methodologies in monitoring MRD.

WT1 as a minimal residual disease marker after conventional chemotherapy

Many studies have shown that the assessment of MRD may prove useful to better stratify high-risk patients and address treatment intensity in AML (Table 1). The most sensitive method for this strategy involves the detection of fusion genes derived from chromosome translocations, such as PML-RARa, AML-ETO1, and CBFb-MYH11 (24, 25), and more recently gene mutations such as NPM1 (26, 27). Besides, more than 50% of AML lack known genetic lesions or clonality markers suitable for MRD monitoring. Thus, alternative markers for MRD are highly sought, and WT1 gene has been suggested as a candidate. Nondiseasespecific genes should be abnormally high expressed in malignant cells when compared with normal controls to be used as an MRD marker. Cilloni et al. (8) first showed that the number of WT1 copies in 71 AML BMs and 14PB was 27,669 (ranges: 1,081–121,086) and 10,244 \times 10⁴ (ranges: 758–86,140) copies of Abelson gene (ABL) mRNA, respectively. Conversely, WT1 levels were extremely low in normal samples: median number of WT1 copies was 78 (range: 3-180) and 4 (1-22) × 10⁴ ABL in BM and PB samples, respectively. Second, in order to assess the significance of the WT1 expression for the detection of MRD, the authors monitored WT1 levels in 10 AML patients characterized by the presence of fusion gene transcripts (CBFb-MYH11 and AML1-ETO);a good parallelism between sequential WT1 and fusion transcripts values was found: some patients who remained in complete remission (CR) (28) constantly showed WT1 values within the normal range, while patients who experienced a relapse showed a conversion to WT1 levels above the normal range in concomitance with fusion

Authors	MRD (cutoff)	LOG reduction	Time of assessment	Main results
Weisser et al. (16)*	0.4%	≥2 Log</td <td>16-60 vs 61-120 vs 121-180 days after start of therapy</td> <td>Within 61–120 and 121–180 days, levels ≤0.4%, and ≥2 log reduction were associ- ated with improved EFS and OS</td>	16-60 vs 61-120 vs 121-180 days after start of therapy	Within 61–120 and 121–180 days, levels ≤0.4%, and ≥2 log reduction were associ- ated with improved EFS and OS
Cilloni et al. (9)*	250 × 10 ⁴ ABL copies	≥2 Log</td <td>Postinduction/ postconsolidation</td> <td>WT1 transcript reduction ≥2 log after induction, and WT1 levels more than 250 × 10⁴ ABL copies after consolidation predicted a significantly increased risk of relapse</td>	Postinduction/ postconsolidation	WT1 transcript reduction ≥2 log after induction, and WT1 levels more than 250 × 10 ⁴ ABL copies after consolidation predicted a significantly increased risk of relapse
Nomdedeu et al. (13)*	170 and 100× 10 ⁴ ABL copies	-	Postinduction/ postintensification	WT1 levels greater than 170 copies after induc- tion and 100 copies after intensification identified patients with the highest probability to relapse and die
Lapillone et al. (29)†	50 × 10 ⁴ ABL copies	-	Postinduction	WT1 > 50× 10 ⁴ ABL copies after induction is an inde- pendent prognostic risk factor of relapse and death
Rossi et al. (38)*	77 × 10 ⁴ ABL copies	≥1.96Log</td <td>Postinduction/ postconsolidation</td> <td>Only postinduction MRD ≥ 77× 10⁴ ABL copies and log reduction ≤1.96 predicted a shorter DFS and OS</td>	Postinduction/ postconsolidation	Only postinduction MRD ≥ 77× 10 ⁴ ABL copies and log reduction ≤1.96 predicted a shorter DFS and OS

Table 1. WT1 expression after conventional chemotherapy

*Study performed on adults.

†Study performed on children.

transcript increasing although patients were still in CR. Indeed, the quantitative assessment of WT1 transcript allows to distinguish between normal and pathological samples, as well as increasing WT1 levels above the normal range can be prognostically significant during the follow-up of patients. Weisser et al. (16), some years later, confirmed a significant correlation between WT1 levels and fusion genes (96%, median r = 0.996) in a similar study population. The authors also showed that more than 2 log reduction of WT1 levels within 61 and 180 days from the start of chemotherapy was associated with a significantly improved OS and event-free survival (EFS). Comparable results were published by the European LeukemiaNet (ELN) study group (9). In order to standardize the WT1 assay, Cilloni et al. (9)

WT1 expression in minimal residual disease

first undertook a systematic evaluation of nine published and "in-house" real-time quantitative polymerase chain reaction (PCR) assays in a quality control study involving 11 ELN laboratories. Then, the selected ELN WT1 assay was applied to samples from 129 follow-up patients, and a significantly increased risk of relapse was found in patients achieving less than 2 log reduction in WT1 transcripts after induction therapy (p = 0.004). This study suggests that application of a standardized WT1 assay early during the patients' therapy could potentially be used to refine risk stratification in AML and decisions on the role of allogeneic transplant in first morphological CR. Recently, in a large study population of AML (n = 584), Nomdedeu et al. (13) defined three different prognostic groups after induction and intensification on the basis of WT1 levels. Patients having more than 170 copies after induction and more than 100 copies after intensification showed the highest probability to relapse and the lowest to OS. On pediatric AML also, similar results were obtained when WT1 was investigated in AML. Lapillone et al. (29) observed that WT1 higher than 50×10^4 ABL copies after induction was an independent prognostic risk factor of relapse (p = 0.002) and death (p = 0.002) 0.02) in pediatric AML. Published results conferred to WT1 an important role in monitoring MRD and stratifying patients with AML, similarly to results obtained by MFC (30-37). When the techniques were compared, a different role was addressed to each one on the basis of the timing of assessment and quantification of MRD or log reduction. Generally, our group and others showed that detection of MRD by WT1 expression and MFC had comparable prognostic value and technical performance described in terms of sensitivity (sens), specificity (spec), predictive value (PV), and likelihood ratio (LR) (23). Besides, when we compared log reduction with MRD measured after conventional chemotherapy by both WT1 expression and MFC, important differences between the two methodologies were found (38). Log reduction and MRD well predicted the outcome at both timing of assessment according as both methodologies, but WT1 log reduction after induction (spec 84.2%, sens 46.2%, LR+ 2.92, LR- 0.64) identified the relapse better than the MRD (spec 57.7%, sens 84.2%, LR+ 1.99, LR- 0.27) and opposite results were true after consolidation for MFC (spec 80.8%, sens 57.9%, LR+ 3.01, LR- 0.52 vs spec 73.1%, sens 63.2%, LR+ 2.35, LR- 0.50 for MRD and log reduction, respectively), thus confirming what was previously published about either WT1 or MFC singularly.

WT1 as minimal residual disease marker in allogeneic stem cell transplantation

Allo-SCT represents the only effective therapy for high-risk patients with AML in first or subsequent CR. Nevertheless, relapse remains a crucial issue in this setting, and new methods able to prevent it are needed (39). Cytogenetics and response after induction therapy were uniformly recognized as predictors of relapse, but there is a growing evidence that quantification of MRD is also a powerful, independent predictor of prognosis (Table 2). Ogawa et al. (18) studied the impact of WT1 levels after allo-SCT on the relapse and the capability to prevent it

Authors	MRD (cutoff)	Time of assessment	Intervention MRD-based	Main results
Ogawa et al. (18)*	10-4-10-2	Post-transplant	Immune inter- ventions (discontinuation of immunosup- pressive therapy or DLI)	Probability to relapse within 40 days was significantly associated with WT1 expres- sion levels. Among high- risk patients, a significantly longer doubling time of WT1 levels in patients who under- went preemptive measures
Zhao et al. (19)*	0.60%	Pre- and post-transplant (+120 days)	Immune inter- ventions (DLI, tapering of immunosuppres- sive therapy) when WT1 levels were >0.60%	Greater than 0.60% after transplant has been shown as an independent risk fac- tor for DFS and OS. High- risk patients who received immune interventions displayed a longer OS
Pozzi et al. (20)*	100 × 10 ⁴ ABL copies	Pre- and post-transplant	DLI if MRD > 180 × 10 ⁴ ABL copies	Post-transplant WT1 expres- sion was the strongest predictor of relapse. Patients with increasing WT1 levels received DLI and showed an improved OS
Rossi et al. (43)*	138× 10 ⁴ ABL copies	Pre- and post-transplant (+30 days)	_	A shorter DFS was found in patients having high levels (≥138 copies) of WT1 at day +30 from transplant. The combination of MFC and WT1 may be preferred for preemptive immune inter- ventions

Table 2. WT1 expression in allogeneic stem cell transplantation

*Study performed on adults.

by preemptive therapeutic measures in patients with leukemias [AML, acute lymphoblastic leukemia (ALL), chronic myeloid leukemia (CML)]. First, the authors showed that the probability of relapse that occurred within 40 days significantly increased according to the increase in *WT1* expression levels (100% for 1.0×10^{-2} to 5.0×10^{-2} , 44.4% for 4.0×10^{-3} to 1.0×10^{-2} , 10.2% for 4.0×10^{-4} to 4.0×10^{-3} , and 0.8% for $<4.0 \times 10^{-4}$). Then, among high-risk patients, they found a significantly longer doubling time of WT1 levels in patients who underwent the discontinuation of immunosuppressive therapy or donor lymphocyte infusions (DLIs). In conclusion, they stated that WT1 was a very useful marker to predict and manage the relapse following the allo-SCT. Similar data were reported by Zhao et al. (19), who investigated the

prognostic significance of WT1 expression in a large study population (n = 138) of AL (AML, ALL) patients following allo-SCT. After measuring MRD by WT1 levels at designed time points, the authors showed that WT1 levels ≥0.60% before allo-SCT indicated higher rates of relapse post-transplant. Similarly, WT1 levels ≥0.60% at median time of +120 days from transplant was associated with lower DFS and OS. Besides, 20 patients showing high levels of WT1 expression received modified DLI, and a median of 0.22% of WT1 levels was observed after intervention. Indeed, patients showing a recurrence trend after allo-SCT, did not experience it due to interventions MRD -based. Recently, Pozzi et al. (20) also confirmed that AML patients in CR before transplant and with a median expression of WT1 >100 × 104 ABL had a higher relapse risk (53% vs 26%) and a lower 5-year survival (36% vs 62%) when compared with patients who had less than this cutoff. Similar results were obtained when the threshold of WT1 ≤100 copies was considered at 30 days after allo-SCT. Thirty-eight patients achieving a CR but exceeding 180 × 104 ABL copies post-transplant were eligible for immune intervention by DLI: 17 patients received DLI and 21 did not. The interval between MRD positivity and relapse was significantly longer in patients receiving DLI. These studies clearly defined the predictive effect of WT1 expression on relapse in AML patients who underwent allo-SCT. In particular, post-transplant WT1 expression was the strongest predictor of outcome in multivariate analysis and was found to be a useful marker to select patients for preemptive immune intervention (DLI, tapering of immunosuppressive therapy). Comparable data were reported in a smaller number of patients monitored before and after transplant (21, 22). However, discordant results on prognosis were obtained when MFC and WT1 levels were compared (40-42). In our recent paper, we investigated technical performance of MRD detected by the two techniques at different time points, before and after transplant. At day +30 post-transplant, we recommended to study MRD by either or both methods, as it had a strong predictive role. Although posttransplant WT1 measurement is a valuable and essential marker for MRD monitoring also in our series, the combination of MFC and WT1 may be preferred to a single one when further treatments should be administered to prevent the relapse. In fact, double-positive MRD after allo-SCT correlated with a higher probability to experience a recurrence, based on higher product between specificity and sensitivity (43).

Conclusions

The relapse remains the main cause of treatment failure and death in AML. Although more than 80% of patients achieves a CR after conventional chemotherapy, a significant number of them experiences a recurrence disease (44). Indeed, more stringent criteria of response than CR are needed. The monitoring of leukemia-specific gene mutation by PCR represents the gold standard method to stratify patients on the basis of the risk to relapse. Unfortunately, more than 50% of AML cases lack one of these specific genes, and new genes to detect MRD are desirable. WT1 is a transcriptional factor, which has found an important role in acute

leukemias as MRD marker. To date, all published papers have confirmed the prognostic value of WT1 levels in AML patients achieving a CR after chemotherapy or allo-SCT. Indeed, despite the controversial role of WT1 expression at the presentation of disease, WT1 levels higher than the given thresholds in AML remission BM predicted the risk of relapse and death. The main concerns grown on this technique referred to cutoff that should be used and the influence of regenerating BM on quantification of the number of WT1 copies. Although WT1 assay has been standardized by ELN, methods to determine the positive threshold of MRD differentiate from one to another study group, with values ranging from 50 to 250 × 10⁴ ABL. Further, WT1 transcript values were not univocally normalized with respect to the number of ABL. On the contrary, regenerating CD34+ cells may be WT1 levels, affecting the sensitivity (45). According to the better sensitivity of 2log reduction compared to MRD and the amply demonstrated prognostic value of this cutoff after induction chemotherapy, the log reduction of copy number may overcome these pitfalls. Finally, post-transplant MRD positive by WT1 is a strong predictor of outcome, and it has been found that WT1 levels may be useful for preemptive immune intervention after transplant. Besides, the low product between sensitivity and specificity for WT1 expression suggests using another method such as MFC to detect MRD and decide for further treatments in case of double positivity.

Conflict of Interest

The authors declare no potential conflicts of interest with respect to research, authorship and/or publication of this article.

Acknowledgment

Not relevant.

References

- Haber DA, Sohn RL, Buckler AJ, Pelletier J, Call KM, Housman DE. Alternative splicing and genomic structure of the Wilms tumor gene WT1. Proc Natl Acad Sci U S A. 1991;88:9618–22.
- Reddy JC, Hosono S, Licht JD. The transcriptional effect of WT1 is modulated by choice of expression vector. J Biol Chem. 1995;270:29976–82.
- Wang ZY, Qiu QQ, Huang J, Gurrieri M, Deuel TF. Products of alternatively spliced transcripts of the Wilms' tumor suppressor gene, wt1, have altered DNA binding specificity and regulate transcription in different ways. Oncogene. 1995;10:415–22.
- 4. Hosen N, Sonoda Y, Oji Y, Kimura T, Minamiguchi H, Tamaki H, et al. Very low frequencies of human normal CD34+ haematopoietic progenitor cells express the

Wilms' tumour gene WT1 at levels similar to those in leukaemia cells. Br J Haematol. 2002;116:409–20.

- 5. Baird PN, Simmons PJ. Expression of the Wilms' tumor gene (WT1) in normal hemopoiesis. Exp Hematol. 1997;25:312–20.
- Miyagi T, Ahuja H, Kubota T, Kubonishi I, Koeffler HP, Miyoshi I. Expression of the candidate Wilm's tumor gene (WT1) in human leukemias. Leukemia. 1992;6:405-9.
- 7. Miwa H, Beran M, Saunders GF. Expression of the Wilm's tumor gene (WT1) in human leukemia. Leukemia. 1992;6:405–9.
- Cilloni D, Gottardi E, De Micheli D, Serra A, Volpe G, Messa F, et al. Quantitative assessment of WT1 expression by real time quantitative PCR may be useful tool for monitoring minimal residual disease in acute leukemia patients. Leukemia. 2002;16:2115–21.

http://dx.doi.org/10.1038/sj.leu.2402675

- 9. Cilloni D, Renneville A, Hermitte F, Hills RK, Daly S, Jovanovic JV, et al. Real-time quantitative polymerase chain reaction detection of minimal residual disease by standardized WT1 assay to enhance risk stratification in acute myeloid leukemia: a European LeukemiaNet Study. J Clin Oncol. 2009;27:5195–201. http://dx.doi.org/10.1200/JCO.2009.22.4865
- Østergaard M, Olesen LH, Hasle H, Kjeldsen E, Hokland P. WT1 gene expression: an excellent tool for monitoring minimal residual disease in 70% of acute myeloid leukaemia patients results from a single centre study. Br J Haematol. 2004;125:590–600.

http://dx.doi.org/10.1111/j.1365-2141.2004.04952.x

- 11. Bergmann J, Weidmann E, Fenchel K, Mitrou PS, Hoelzer D, Bergmann L. The expression of the Wilms' tumor gene in acute myelocytic leukemia as a possible marker for leukemic blast cells. Leukemia. 1994;8:2138–43.
- Galimberti S, Ghio F, Guerrini F, Ciabatti E, Grassi S, Ferreri MI, et al. WT1 expression levels at diagnosis could predict long-term time-to- progression in adult patients affected by acute myeloid leukaemia and myelodysplastic syndrome. Br J Haematol. 2010;149:451–62.

http://dx.doi.org/10.1111/j.1365-2141.2009.08063.x

- Nomdedeu J, Hoyos M, Carricondo M, Bussaglia E, Estivill C, Esteve J, et al. Bone marrow WT1 levels at diagnosis, post-induction and post-intensification in adult de novo AML. Leukemia. 2013;27:2157–64. http://dx.doi.org/10.1038/leu.2013.111
- 14. Schmid D, Heinze G, Linnerth B. Prognostic significance of WT1 gene expression at diagnosis in adult de novo acute myeloid leukemia. Leukemia. 1997;11:639–43.

- 15. Garg M, Moore H, Tobal K, Liu Yin JA. Prognostic significance of quantitative analysis of WT1 gene transcripts by competitive reverse transcription polymerase chain reaction in acute leukaemia. Br J Haematol. 2003;123:49–59.
- Weisser M, Kern W, Rauhut S, Schoch C, Hiddemann W, Haferlach T, et al. Prognostic impact of RT-PCR-based quantification of WT1 gene expression during MRD monitoring of acute myeloid leukemia. Leukemia. 2005;19:1416-23. http://dx.doi.org/10.1038/sj.leu.2403809
- Andersson C, Li X, Lorenz F, Golovleva I, Wahlin A, Li A. Reduction in WT1 gene expression during early treatment predicts the outcome in patients with acute myeloid leukemia. Diagn Mol Path. 2012;21:225–33. http://dx.doi.org/10.1097/PDM.0b013e318257ddb9
- Ogawa H, Tamaki H, Ikegame K, Soma T, Kawakami M, Tsuboi A, et al. The usefulness of monitoring WT1 gene transcripts for the prediction and management of relapse following allogeneic stem cell transplantation in acute type leukemia. Blood. 2003;101:1698–704. http://dx.doi.org/10.1182/blood-2002-06-1831
- Zhao XS, Jin S, Zhu HH, Xu LP, Liu DH, Chen H, et al. Wilms' tumor gene 1 expression: an independent acute leukemia prognostic indicator following allogeneic hematopoietic SCT. Bone Marrow Transplant. 2012;47(4):499–507. http://dx.doi.org/10.1038/bmt.2011.121
- Pozzi S, Geroldi S, Tedone E, Luchetti S, Grasso R, Colombo N, et al. Leukaemia relapse after allogeneic transplants for acute myeloid leukaemia: predictive role of WT1 expression. Br J Haematol. 2013;160:503–9. http://dx.doi.org/10.1111/bjh.12181
- 21. Candoni A, Tiribelli M, Toffoletti E, Cilloni D, Chiarvesio A, Michelutti A, et al. Quantitative assessment of WT1 gene expression after allogeneic stem cell transplantation is a useful tool for monitoring minimal residual disease in acute myeloid leukemia. Eur J Haematol. 2009;82 (1):61–8.

http://dx.doi.org/10.1111/j.1600-0609.2008.01158.x

- 22. Valkova V, Polák J, Marková M, Vítek A, Hájková H, Sálek C, et al. Minimal residual disease detectable by quantitative assessment of WT1 gene before allogeneic stem cell transplantation in patients in first remission of acute myeloid leukemia has an impact on their future prognosis. Clin Transplant. 2013;27:21–9. http://dx.doi.org/10.1111/ctr.12046
- Rossi G, Minervini MM, Carella AM, de Waure C, di Nardo F, Melillo L, et al. Comparison between flow cytometry and WT1 RNA quantification in monitoring minimal residual disease in acute myeloid leukemia without specific molecular targets. Leuk Res. 2012;36(4):401–6.
 http://dx.doi.org/10.1016/j.leukros.2011.11.020

http://dx.doi.org/10.1016/j.leukres.2011.11.020

24. Schnittger S, Weisser M, Schoch C, Hiddemann W, Haferlach T, Kern W. New score predicting for prognosis in PML-RARA+, AML-ETO, or CBFB-MYH 11+ acute myeloid leukemia based on quantification of fusion transcripts. Blood. 2003;102:2746–55.

http://dx.doi.org/10.1182/blood-2003-03-0880

- 25. Ommen HB, Schnittger S, Jovanovic JV, Ommen IB, Hasle H, Østergaard M, et al. Strikingly different molecular relapse kinetics in NPM1c, PML-RARA, RUNX1-RUNX1T1, and CBFB-MYH11 acute myeloid leukemias. Blood. 2010;115:198–205. http://dx.doi.org/10.1182/blood-2009-04-212530
- 26. Papadaki C, Dufour A, Seibl M, Schneider S, Bohlander SK, Zellmeier E, et al. Monitoring minimal residual disease in acute myeloid leukaemia with NPM1 mutations by quantitative PCR: clonal evolution is a limiting factor. Br J Haematol. 2009;144:517–23.

http://dx.doi.org/10.1111/j.1365-2141.2008.07488.x

27. Krönke J, Schlenk RF, Jensen KO, Tschürtz F, Corbacioglu A, Gaidzik VI, et al. Monitoring of minimal residual disease in NPM1-mutated acute myeloid leukemia: a study from the German-Austrian acute myeloid leukemia study group. J Clin Oncol. 2011;29:2709–16.

http://dx.doi.org/10.1200/JCO.2011.35.0371

- 28. Cheson BD, Bennett JM, Kopecky KJ, Büchner T, Willman CL, Estey EH, et al. International Working Group for Diagnosis. Revised recommendations of response Criteria. Treatments outcomes and reporting standards for therapeutic trials in acute myeloid leukemia. J Clin Oncol. 2003;21:4642–9. http://dx.doi.org/10.1200/JCO.2003.04.036
- 29. Lapillone H, Renneville A, Auvrignon A, Flamant C, Blaise A, Perot C, et al. High WT1 expression after induction therapy predicts high risk of relapse and death in pediatric acute myeloid leukemia. J Clin Oncol. 2006;24:1507–15.
- Venditti A, Buccisano F, Del Poeta G, Maurillo L, Tamburini A, Cox C, et al. Level of minimal residual disease after consolidation therapy predicts outcome in acute myeloid leukemia. Blood. 2000;96:3948–52.
- 31. San Miguel JF, Vidriales MB, López-Berges C, Díaz-Mediavilla J, Gutiérrez N, Cañizo C, et al. Early immunophenotypical evaluation of minimal residual disease in acute myeloid leukemia identifies different patient risk groups and may contribute to postinduction treatment stratification. Blood. 2001;98:1746–51.
- 32. Buccisano F, Maurillo L, Gattei V, Del Poeta G, Del Principe MI, Cox MC, et al. The kinetics of reduction of minimal residual disease impacts on duration of response and survival of patients with acute myeloid leukemia. Leukemia. 2006;20:1783–9. http://dx.doi.org/10.1038/sj.leu.2404313

- 33. Al-Mawali A, Gillis D, Lewis I. The use of receiver operating characteristic Analysis for detection of minimal residual disease using five-color multiparameter flow cytometry in acute myeloid leukemia identifies patients with high risk of relapse. Cytometry Part B (Clinical Cytometry). 2009;76B:91–101. http://dx.doi.org/10.1002/cyto.b.20444
- 34. Kern W, Voskova D, Schoch C, Schnittger S, Hiddemann W, Haferlach T. Prognostic impact of early response to induction therapy as assessed by multiparameter flow cytometry in acute myeloid leukemia. Haematologica. 2004;89:528–40.
- 35. Maurillo L, Buccisano F, Del Principe MI, Del Poeta G, Spagnoli A, Panetta P, et al. Toward optimization of postremission therapy for residual disease –positive patients with acute myeloid leukemia. J Clin Oncol. 2008;26(30):4944–51. http://dx.doi.org/10.1200/JCO.2007.15.9814
- 36. Walter RB, Gooley TA, Wood BL, Milano F, Fang M, Sorror ML, et al. Impact of pretransplantation minimal residual disease, as detected by multiparametric flow cytometry, on outcome of myeloablative hemopoietic cell transplantation for acute myeloid leukemia. J Clin Oncol. 2011;29(9):1190–7. http://dx.doi.org/10.1200/JCO.2010.31.8121
- 37. Zhao XS, Liu YR, Zhu HH, Hu LP, Liu KY, Huang HJ. Monitoring MRD with flow cytometry: an effective method to predict relapse for ALL patients after allogeneic hematopoietic stem cell transplantation. Ann Hematol. 2012;91(2):183–92.
- 38. Rossi G, Minervini MM, Melillo L, di Nardo F, de Waure C, Scalzulli PR, et al. Predictive role of minimal residual disease and log clearance in acute myeloid leukemia: a comparison between multiparameter flow cytometry and Wilm's tumor 1 levels. Ann Hematol. 2014;93:1149–57.

http://dx.doi.org/10.1007/s00277-014-2029-9

- 39. Kröger N, Bacher U, Bader P, Böttcher S, Borowitz MJ, Dreger P, et al. NCI First International Workshop on the biology, prevention, and treatment of relapse after allogeneic hematopoietic stem cell transplantation: report from the committee on disease-specific methods and strategies for monitoring relapse following allogeneic stem cell transplantation. Part I: methods, acute leukemias, and myelodysplastic syndromes. Biol Blood Marrow Transplant. 2010;16:1187–211. http://dx.doi.org/10.1016/j.bbmt.2010.06.008
- 40. Kwon M, Martínez-Laperche C, Infante M, Carretero F, Balsalobre P, Serrano D, et al. Evaluation of minimal residual disease by real-time quantitative PCR of Wilm's tumor I expression in patients with acute myelogenous leukemia after allogeneic stem cell transplantation: correlation with flow cytometry and chimerism. Biol Blood Marrow Transplant. 2012;1:1–8.

41. Miyazaki T, Fujita H, Fujimaki K, Hosoyama T, Watanabe R, Tachibana T, et al. Clinical significance of minimal residual disease detected by multidimensional flow cytometry: serial monitoring after allogeneic stem cell transplantation for acute leukemia. Leuk Res. 2012;36:998–1003.

http://dx.doi.org/10.1016/j.leukres.2012.04.005

42. Rossi G, Carella AM, Minervini MM, Savino L, Fontana A, Pellegrini F, et al. Minimal residual disease after allogeneic stem cell transplantation: a comparison among multiparametric flow cytometry, Wilms tumor 1 expression and chimerism status (complete chimerism versus low level mixed chimerism) in acute leukemia. Leuk Lymphoma. 2013;54:2660–6.

http://dx.doi.org/10.3109/10428194.2013.789508

- 43. Rossi G, Carella AM, Minervini MM, di Nardo F, Waure Cd, Greco MM, et al. Optimal time-points for minimal residual disease monitoring change on the basis of method used in patients with acute myeloid leukemia who underwent allogeneic stem cell transplantation: a comparison between multiparameter flow cytometry and Wilm's tumor 1 expression. Leuk Res. 2015;39:138–43. http://dx.doi.org/10.1016/j.leukres.2014.11.011
- 44. Mayer RJ, Davis RB, Schiffer CA, Berg DT, Powell BL, Schulman P, et al. Intensive postremission chemotherapy in adult with acute myeloid leukemia. Cancer and leukemia Group B. New Engl J Med. 1994;331:896–903. http://dx.doi.org/10.1056/NEJM199410063311402
- 45. Alonso-Dominguez JM, Tenorio M, Velasco D, Abalo L, Lozano S, Villarrubia J, et al. Correlation of WT1 expression with the burden of total and residual leukemic blasts in bone marrow samples of acute myeloid leukemia patients. Cancer Genetics. 2012;205:190–1.

http://dx.doi.org/10.1056/NEJM199410063311402

A

Abdominal Doppler US, 47 Abdominal imaging, 63 Abdominal pain, 45 Abelson gene (ABL) mRNA, 275 Actinomycin D, 87, 96, 98, 105 Adaptive immune cells, 193 Adjuvant chemotherapy, 87, 96, 116 Adults, WT in, 26, 217 AFP (alpha-fetoprotein), 86 Age as a prognostic factor, 26, 27 Allogeneic stem cell transplantation (allo-SCT), 275, 277, 278 Alternative lengthening of telomeres (ALT) mechanism, 36 Anaplasia, 8, 10, 24, 26, 65, 66, 192 Anaplastic histology, 9, 65, 96, 194, 195 Anaplastic Wilms' tumours, 8, 9, 10 Androgen receptor element half-sites (ARE-I), 245 Anthracyclines, 70, 98, 135 Antigen-presenting cells (APCs), 114, 154 APC membrane recruitment protein 1 (AMER1), 152 Appearance, 5, 16 Arterial hypertension, 140 Arterial line, 49 Aspirin, 167, 168, 169, 177

B

B lymphocytes (CD20+), 194 B7-H1 expression, 198 Beckwith-Wiedemann syndrome (BWS), 34, 44, 46, 134 Beta-catenin, 34, 152, 154, 155 β HCG (beta-human chorionic gonadotropin), 86 Bilateral Wilms tumours (BWTs), 51, 61, 62, 132, 134 Blastema, 5, 6, 8, 16, 83 Blastemal-type WTs, 4, 6, 7, 13, 15, 16, 156 Brenner theory, 136

С

Cancer vaccination, 114, 115, 267 Canonical Wnt pathway, 152, 154, 155 Carbonic anhydrase IX (CAIX), 199 Carboplatin, 66, 105 Cardiac development, WT1 in, 212 Cavitron ultrasonic surgical aspirator, 67 CD3+ T cells, 193 CD4⁺ helper T cell, 114, 115 CD8⁺ killer T cell, 114 CD20+ B-cell infiltration, 194 CD34+ cells, 280 CD34+ hematopoietic progenitor cells, 238 CD44, 198 CD68+ macrophages (CD68MØ), 194 Chemokines, 197, 199 Chemotherapy, 6, 24, 36, 51, 64, 65, 66, 78, 87, 98, 103, 105, 106, 119, 136 Chemotherapy before surgery strategy, 25 Chemotherapy-induced change (CIC), 7 Chest X-ray, 45 Childhood ERWTs, 78, 79, 88 Children, WT in, 26 Children's Oncology Group (COG), 24, 46, 62

Chromatin immunoprecipitation (ChIP) assay, 174, 242 Chromosome 11p13, 34 Chronic kidney disease (CKD), 131, 132, 135, 136, 137, 140 CITED1, 198 Clinical presentation, 45, 85 Clinical relevance of age, 25 c-Myc promoter, 172 COG staging system, 13 Color Doppler ultrasonography, 100, 103 Common Terminology Criteria for Adverse Events (CTCAE), 119 Completely necrotic WT, 7 Components of Wilms tumor, 191 Computerized tomography (CT), 45, 46 Conventional chemotherapy, WT1 expression after, 275, 276, 277, 279 CREB-binding protein (CBP), 156, 237 Cut surface of tumour, 5 Cyclooxygenase-2 (COX-2) expression, 192, 195, 196, 199, 200, 201, 202 Cyclophosphamide, 53, 105 Cystic nephroma (CN), 16 Cystic partially differentiated nephroblastoma (CPDN), 16 Cystic WTs, 16 Cytogenetics, 277 Cytokines, 190, 192, 193, 197 Cytotoxic T cells (CTLs), 116, 121

D

Dactinomycin, 66 Dendritic cells (DCs), 113, 114, 115, 117, 122 Denys-Drash syndrome (DDS), 34, 44, 65, 72, 133, 220, 236 Diagnosis, 3, 5, 14, 32, 45, 78, 85 Differential diagnosis, 3, 14, 16, 32, 36, 37, 87 Diffuse hyperplastic perilobar nephroblastomatosis, 13 Dimethyl sulfoxide (DMSO), 245 Donor lymphocyte infusions (DLIs), 278 Doppler, ultrasound with, 47, 63 Doxorubicin, 66, 87, 98, 105 *Drosophila* PCP signaling, 153

Е

Early growth response 1 (Egr1), 169, 170, 171, 172, 173, 175 E-cadherin, 151, 215, 237, 238, 240, 241, 248 18F-FDG-PET-CT imaging, 64 Electrophoretic mobility shift assay (EMSA), 242 Embryonic development of kidney, 32 Embryonic nephrogenic tissue, 83 Endothelial cells, 215, 217, 218, 219, 220, 221 End-stage renal disease (ESRD), 132 Enhancer of zeste homolog 2 (EZH2), 237 Enzyme-linked immunosorbent spot (ELISPOT) assays, 120 Ependymoma, 264 Epicardium-derived cells (EPDCs), 213 Epithelial cells, 151 Epithelial to mesenchymal transition (EMT), 213, 237 Epithelial-type WT, 7, 16 Estrogen receptor (ER)-binding sites, 243 Etoposide, 53, 105, 122 Event-free survival (EFS) time, 100 Ewing sarcoma, 16 Extracellular matrix (ECM), 216 Extracellular signal-regulated protein kinase (ERK) activation, 164 Extrarenal Wilms' tumor (ERWT), 77, 84, 85,86

F

Favorable histology, 27, 33, 84, 88, 192 Features of WT, 15

Fillet, 70 5-F Pigtail catheter, 97 Foley catheter, 49 4-F/5-F Cobra catheter, 98 Foxd1-positive cells, 33 Frasier syndrome, 236 Frizzled (FZD) family receptor, 152

G

Gain-of-function mutations, 34 Gemcitabine, 123 Gene activation/inactivation, 150 Gene expression in WT, 149 Genetic abnormalities in WT, 33 Gerota fascia, 49, 67, 70 Glioblastoma, 262, 263, 265 Glioma, 263 Glomerulosclerosis, 34 G-protein-coupled receptor (GPCR) superfamily, 164 Granulocyte colony-stimulating factor (G-CSF), 123 Granulocyte-macrophage colonystimulating factor (GM-CSF), 115 Green fluorescent protein (GFP)-tagged WT1 expression, 243 Gross appearance, 5

Η

HIF-1α, 195 High telomerase reverse transcriptase (hTERT) staining, 36 Histological components of WT, 36 Histological features, 5–9 Histology, 3 Histopathological characteristics, 31 Histopathological features, 34 HLA-A*24:02 tetramer, 120 Horseshoe kidney and ERWT, 85 Human leukocyte antigen (HLA) molecules, 114, 267
Human thromboxane A₂ receptor gene, 163
Hyperplastic perilobar nephroblastomatosis, 14, 15
Hyperplastic rests, 63
Hypertension, 135
Hypertrophy of a solitary kidney, 140
Hypoxia-inducible factor-1-alpha (HIF1α), 219, 241

Ι

Ice-bath kidney cooling, 67, 68, 69 Ifosfamide, 53, 105 IL-4, 115 Immature DCs, 117 Immune cell infiltration, 193, 199 Immune cells, cross talk between, 199 Immune monitoring with tetramer analysis, 120 Immunohistochemical stains, 37 Immunosuppressive therapy, 278 Inducible nitric oxide synthase (iNOS), 197 Infants, WT in, 25 Infarcted heart, WT1 in, 216 Inflammation, WT and, 193 Inflammatory mediators, 195 Inflammatory microenvironment in WT, 189 Inguinal ERWT, 87 Innate immune cells, 194 Interleukin 10, 199 Intracardiac extension of WT, 51 Intralobar nephrogenic rest (ILNR), 12 Intravenous chemotherapy, 98

Κ

Ki-67 index, 36 Kidney cancer, 192

Kidney damage, 136, 137 Kidney development, 151 Kocher' maneuver, 49

L

Laser capture microdissection (LCM), 239 LNCaP cells, 241 Lymph node sampling, 49

Μ

Macrophages, 194 Magnetic resonance imaging (MRI), 45 Malignant brain tumors, WT1 expression in, 261 Malignant glioma, 267 Malignant renal tumors, 32, 33 Management of WT, 47 Manufacture of DC vaccine, 116 Mast cells (MCs), 193, 194 Matrix metalloproteinases (MMPs), 221 Mature DCs (mDCs), 116 Medulloblastoma, 264 Meningioma, 264 Mesenchymal cells, 151 Mesenchymal-epithelial transition (MET), 151, 237 Mesonephros, 83, 85 Metanephric adenoma, 37 Metanephric kidney, 33 Metanephric mesenchyme (MM), 197, 198 Metastatic disease, 239 Microdeletion 11p13 syndrome, 132 MicroRNA (miRNA) biogenesis, 151 Minimal residual disease (MRD), 275 Minimal residual disease marker, WT1 as, 275, 277 Mithramycin A, 246 Molecular characteristics of WT, 31

Monophasic WTs, 14 Multicentric tumours, 5, 10 Multifocal NRs, 13 Multiparameter flow cytometry (MFC), 275 Mutation repertoire, 155 MYCN gene, 9

Ν

Nasogastric tube, 49 National Wilms Tumor Study (NWTS), 46, 47, 50, 86, 100 National Wilms Tumor Study Group (NWTSG), 46, 51, 52 Necrotic WT, 7 Neoadjuant transcatheter arterial chemoembolization, 95 Neoadjuvant chemotherapy, 66 Nephrectomy specimen, 9 Nephroblastoma, 4, 32, 44, 62 Nephroblastomatosis, 12, 13, 14, 15, 63 Nephrogenesis, 151 Nephrogenic rests (NRs), 5, 12, 63, 83 Nephrons, 33 Nephron-sparing surgery (NSS), 48, 50, 63, 134 Nestin, expression of, 221 Neuroblastoma (case), 119 Neutrophils, 194 NF-E2, 169, 177 Nitrotyrosine (NT), 197 Noncanonical Wnt signaling pathway, 152, 154 Nonmetastatic WT, 25 Non-viable tumour, 11 Non-Wilms cancer, WT1 expression in, 238 Non-WT renal tumors, 36 Non-WT1-syndromic BWTs, 134

0

Oligodendroglioma, 263 Oncogenes, 151 Overall survival (OS) time, 100

P

Paediatric renal tumours, 9 Papillary renal cell carcinoma, 16 Pathogenesis of WT, 33 Perilobar nephrogenic rest (PLNR), 12, 156 Perioperative US, 51 Peritoneal reflection covering right Wilms tumor, 51 PERK1/2 protein, 196 Phosphorylated pERK1/2, 196 Phosphorylated-Stat3 (p-Stat3) expression, 196 Phycoerythrin (PE)-conjugated human immunodeficiency virus, 120 Pirarubicin, 98 Planar cell polarity (PCP) pathway, 152 Platelet endothelial cell adhesion molecule-1 (PECAM-1), 220 Platinum compounds, 53 Platinum-based combination chemotherapy, 106 PLCG2 expression, 154 Polymerase chain reaction (PCR), 239, 277 Post-operative chemotherapy, 66 Preoperative chemotherapy, 5, 7, 8, 9, 13, 16, 17, 65, 104, 105 Prickle (PK) proteins, 153 Primary nephrectomy for WT, 104 Primary renal tumours in childhood, 4 Prm1, 164, 165, 169, 170, 171, 172, 173, 174, 175, 176, 177 Prm3, 164, 165, 169, 177 Prognosis, 24, 44, 71 Proinflammatory cytokines, 218

Prominent cystic appearance, WT with, 16 Pronephros, 83 Prostaglandin E2, 199 Prostate cancer, 235, 242 Protein kinase C-related kinase/protein kinase novel (PRK/PKN), 168

R

Radiotherapy, 70 Rance, Thomas F., 44 Recessive tumor suppressor genes, 34 Recurrent disease, 52 Renal cell carcinomas, 16 Renal function in Wilms tumour survivors, 138 Renal tissue preservation, 64 Renal tumours, 4 Renal vein, ligation of, 54 Residual renal parenchyma, 67, 71 Retroperitoneal ERWT, 87 Rhabdoid tumour, 16

S

Schwartz equation, 137 Seldinger technique, 97 Signal transduction pathways, 150 SIOP WT 2001 trial, 50 Sipuleucel-T (Provenge), 115, 116 SIX2 (Sine oculis homeobox 2 gene), 153 Six2-positive mesenchymal cells, 33 Small round blue cell tumours, 6 Smooth muscle (SM), 164 Snail upregulation, 238 Societe Internationale D'oncologie Pediatrique (SIOP), 9, 11, 44, 46, 48, 52, 62 Solitary kidney, 135, 142 Sp1 binding, 246 Staging, 46

Steinman, Ralph, 114
Stimulating protein (Sp) 1, 178
Stockholm working classification of renal tumors, 44
Strabismus (STBM), 153
Stromal-type WTs, 7, 16
Surgical management, 46
Syndromic tumors, 34
Systemic chemotherapy, 95

Т

T lymphocytes, 193 T prostanoid receptor (TP), 164, 177 T prostanoid receptor (TP) a, 165, 166, 167, 169, 172, 174, 177 T prostanoid receptor (TP) β, 165, 166, 167, 169, 177 Tamm-Horsfall protein, 11 TBXA2R gene, 164, 165, 169, 172 Telomerase activity, 36 Telomere maintenance, 36 Telomere stability, 36 Temozolomide, 268 Tetramer analysis, immune monitoring with, 120 Therapeutic targets in WT, 200 Thoracic ERWT, 87 Thromboxane (TX)A₂, 164, 167, 170 Thymosin beta-4, 217 Titanium clips, 49 TP53 gene, 8, 155 Transcatheter arterial chemoembolization (TACE), 95, 97, 99, 105 Transient catheter, 49 Trendelenburg, Frederick, 62 Tumor angiogenesis, WT1 and, 241 Tumor suppressor gene (TSG), 151, 237 Tumor-associated macrophages (TAMs), 193 Tumor-infiltrating leukocytes (TIL), 192

Tumor-infiltrating neutrophils (TINs), 193, 194 Types of WT, 192

U

U1242 glioblastoma cells, 266 U87MG cells, 266 Ultrasonic scalpel, 51 Ultrasonography (US), 45 Ultrasound examination of WTS, 140 Ultrasound with Doppler, 63 Unfavorable histology, 192 Unilateral nephrectomy for WT, 50 Unilateral WT (UWT), 132, 134, 135, 137

V

Varicocele, 45 Vascular endothelial growth factor (VEGF), 192, 196, 237, 241, 242, 243 Viable tumour, 6 Vincristine, 66, 87, 105 Vindesine, 98 Von Willebrand disease, 46

W

WAGR syndrome, 34, 44
Wilms, Carl Max Wilhelm, 44, 62
Wilms tumor 1 (WT1) protein, 236
Wilms tumor 1 gene (WT1), 153, 154
Wilms tumorigenesis, 33
Wilms' tumor (case), 119
Wilms' tumor 1 (WT1), 34, 116, 121, 133, 134, 169, 211, 212, 216, 219, 222, 235, 236, 237, 238, 240, 241, 261, 262, 266, 267, 274, 275, 277
Wnt signaling pathway, 150, 152, 153
WNT5A signaling, 154, 155

WNT5B signaling, 154, 155 WNT6, 153 WNT9B, 153 Wolffian duct, 33 WT survivors (WTS), 131 WT1 antigen-specific CTLs (WT1-CTLs), 120, 121 WT1-expressing stem cell-like cells, 222 WT2 gene, 34, 44

Doi: http://dx.doi.org/10.15586/codon.wt.2016.ind

Printed in Australia